SPECTRAL EDGE PROPERTIES OF PERIODIC ELLIPTIC OPERATORS

A Dissertation

by

MINH TUAN KHA

Submitted to the Office of Graduate and Professional Studies of Texas A&M University in partial fulfillment of the requirements for the degree of

DOCTOR OF PHILOSOPHY

Chair of Committee,	Peter Kuchment
Committee Members,	Gregory Berkolaiko
	Kenneth Dykema
	Valery Pokrovsky
Head of Department,	Emil Straube

August 2017

Major Subject: Mathematics

Copyright 2017 Minh Tuan Kha

ABSTRACT

In this dissertation, we study some spectral problems for periodic elliptic operators arising in solid state physics, material sciences, and differential geometry. More precisely, we are interested in dealing with various effects near and at spectral edges of such operators. We use the name "threshold effects" for the features that depend only on the infinitesimal structure (e.g., a finite number of Taylor coefficients) of the dispersion relation at a spectral edge.

We begin with an example of a threshold effect by describing explicitly the asymptotics of the Green's function near a spectral edge of an internal gap of the spectrum of a periodic elliptic operator of second-order on Euclidean spaces, as long as the dispersion relation of this operator has a non-degenerate parabolic extremum there. This result confirms the expectation that the asymptotics of such operators resemble the case of the Laplace operator.

Then we generalize these results by establishing Green's function asymptotics near and at gap edges of periodic elliptic operators on abelian coverings of compact Riemannian manifolds. The interesting feature we discover here is that the torsion-free rank of the deck transformation group plays a more important role than the dimension of the covering manifold.

Finally, we provide a combination of the Liouville and the Riemann-Roch theorems for periodic elliptic operators on abelian co-compact coverings. We obtain several results in this direction for a wide class of periodic elliptic operators. As a simple application of our Liouville-Riemann-Roch inequalities, we prove the existence of non-trivial solutions of polynomial growth of certain periodic elliptic operators on noncompact abelian coverings with prescribed zeros, provided that such solutions grow fast enough.

DEDICATION

To my father Xuong Kha, my mother Sang Lam, and my sister Loan Kha.

ACKNOWLEDGMENTS

First, it is my great pleasure to express my deepest gratitude to my advisor Professor Peter Kuchment for his endless support of my research during my Ph.D study. I am indebted to him for shaping me as a mathematician as well as helping and encouraging me through my tough times in life. I also thank Professors Yehuda Pinchover and Andy Raich for their collaboration, discussion, support, and advice.

I would like to thank the rest of my Ph.D committee. Professor Ken Dykema's valuable support began when I just entered the department and has continued all these years. Thank you for the insight gained from our discussions. I also thank Professor Gregory Berkolaiko for many interesting and very helpful mathematical conversations. Professor Valery Pokrovsky was very kind to agree to be in my commitee. Thank you for the time and effort spent looking into my work.

I am also grateful to many people from the department of Mathematics at Texas A&M University. Special thanks go to Professor Peter Howard, whose helpfulness, patience, and encouragement went beyond any expectation during my entire six years here; Ms. Monique Stewart for her tireless work and help; Professor Ciprian Foias for his inspiring lectures and precious advices in research; Professor Dean Baskin for many enthusiastic conversations and his great teaching microlocal analysis.

I also feel thankful to have many great friends who have made my life enjoyable and comfortable in College Station; among them are Yeong Chyuan Chung, Wai-Kit Chan, Guchao Zheng, Adrian Barquero Sanchez, Mahishanka Withanachchi, Fatma Terzioglu, Jimmy Corbin.

Thank you to Hanh Nguyen, for all her love and support.

I don't even know how to begin thanking my family. Thank you, Mom and Dad, for giving birth to me and your unconditional love, endless care, and spiritual support throughout my life. There is no way I could have accomplished this without you.

This dissertation is partly based on the research carried out under the support of the NSF grant DMS-1517938. I express my gratitude to the NSF for this support.

CONTRIBUTORS AND FUNDING SOURCES

Contributors

This work was supported by a thesis (or) dissertation committee consisting of Professor Peter Kuchment [advisor], Professor Gregory Berkolaiko, and Professor Kenneth Dykema of the Department of Mathematics and Professor Valery Pokrovsky of the Department of Physics.

The results in Chapter 3 were obtained from a collaboration of the student with Professor Peter Kuchment (Texas A&M University) and Professor Andrew Raich (University of Arkansas), and this will be published in an article listed in the Biographical Sketch (see [40]). The results in Chapter 5 were obtained together with Professor Peter Kuchment.

All other work conducted for the dissertation was completed by the student independently (see [41,42]).

Funding Sources

Graduate study was supported by a Graduate Teaching Assistantship from the Department of Mathematics at Texas A&M University and also partially supported by the National Science Foundation.

TABLE OF CONTENTS

				Pa	ige
ABS	STRA	ACT		•	ii
DEI	DICA	TION		•	iii
ACF	KNO	WLED	GMENTS	•	iv
CON	NTR]	IBUTO	RS AND FUNDING SOURCES	•	vi
TAE	BLE (OF COI	NTENTS	•	vii
LIST	ГOF	FIGU	RES	•	х
1. I	INTF	RODUC	TION	•	1
2. I	FLO	QUET-I	BLOCH THEORY FOR PERIODIC ELLIPTIC OPERATORS	•	9
	2.1 2.2	Periodi Floque	c elliptic operators		9 12
3. (S	GRE SPEC RIOF	EN'S F CTRA (R	UNCTION ASYMPTOTICS NEAR THE INTERNAL EDGES OF DF PERIODIC ELLIPTIC OPERATORS. SPECTRAL GAP INTE-		21
	3.1 3.2 3.3 3.4 3.5 3.6	Introdu Assum Proof c On loc: A Floq 3.5.1 3.5.2 3.5.3 Asymp	action	• • • •	21 22 30 32 32 33 34 38 39 40
	3.7	3.6.1 3.6.2 The ful 3.7.1 3.7.2	The asymptotics of the leading term of the Green's function \dots Estimates of the integral J \dots		40 53 57 58 62

	3.8	Some results on parameter-dependent toroidal Ψ DOs	71
	3.9	Some auxiliary statements	75
		3.9.1 A lemma on the principle of non-stationary phase	75
		3.9.2 The Weierstrass preparation theorem	76
		3.9.3 Proofs of Proposition 3.4.1 and Lemma 3.5.1	77
		3.9.4 Regularity of eigenfunctions $\phi(z, x)$	84
	3.10	Concluding remarks	89
4.	GRE	EEN'S FUNCTION ASYMPTOTICS OF PERIODIC ELLIPTIC OPERA-	
	TOR	RS ON ABELIAN COVERINGS OF COMPACT MANIFOLDS	90
	4.1	Introduction	90
	4.2	Notions and preliminary results	92
		4.2.1 Group actions and abelian coverings	92
		4.2.2 Additive and multiplicative functions on abelian coverings	96
		4.2.3 Some notions and assumptions	00
	4.3	The main results \ldots	05
	4.4	A Floquet-Bloch reduction of the problem	.09
		4.4.1 The Floquet transforms on abelian coverings and a Floquet reduc-	
		tion of the problem	.09
		4.4.2 Isolating the leading term in $R_{s,\lambda}$ and a reduced Green's function . 1	13
	4.5	Some auxiliary statements	17
	4.6	Proofs of the main results	20
	4.7	Proofs of technical statements	23
		4.7.1 Proof of Proposition 4.2.8	23
		4.7.2 Proof of Proposition 4.2.10	24
		4.7.3 Proof of Theorem 4.4.6	25
	4.8	Concluding remarks	40
5.	A LI	IOUVILLE-RIEMANN-ROCH THEOREM ON ABELIAN COVERINGS. 1	42
	5.1	Introduction	42
	5.2	Some preliminaries for Liouville type results	43
		5.2.1 Some preliminaries for periodic elliptic operators on abelian cov-	
		erings	43
		5.2.2 Floquet-Bloch solutions and Liouville theorem on abelian coverings 1	47
		5.2.3 Explicit formulas for dimensions of spaces $V_N^{\infty}(A)$	50
		5.2.4 A characterization of the spaces $V_N^p(A)$	52
	5.3	The Gromov-Shubin version of the Riemann-Roch theorem for elliptic op-	
		erators on noncompact manifolds	54
		5.3.1 Some notions and preliminaries	54
		5.3.2 Point divisors	57
		5.3.3 Rigged divisors	58

		5.3.4	Gromov-Shubin theorem on noncompact manifolds	160
	5.4	The m	ain results	166
		5.4.1	Non-empty Fermi surface	166
		5.4.2	Empty Fermi surface	174
	5.5	Proofs	of the main results	179
	5.6	Applic	cations of Liouville-Riemann-Roch theorems to specific operators .	200
		5.6.1	Periodic operators with nondegenerate spectral edges	201
		5.6.2	Periodic operators with Dirac points	206
		5.6.3	Non-self-adjoint second order elliptic operators	209
	5.7	Some	auxiliary statements and proofs of technical lemmas	213
		5.7.1	Some properties of Floquet functions on abelian coverings	213
		5.7.2	Some basic facts about the family $\{A(k)\}_{k\in\mathbb{C}^d}$	215
		5.7.3	Some properties of Floquet transforms on abelian coverings	217
		5.7.4	A Schauder type estimate	220
		5.7.5	A variant of Dedekind's lemma	221
		5.7.6	Proofs of technical statements	223
	5.8	Conclu	Jding remarks	234
6.	SUN	/MARY	AND CONCLUSIONS	235
RI	EFER	ENCES		237
A	PPEN	DIX A.	A TOPOLOGICAL APPROACH TO DEFINING ADDITIVE FUNC	2-
	TIO	NS ON	CO-COMPACT RIEMANNIAN NORMAL COVERINGS	246
	A.1	Introd	uction	246
	A.2	Additi	ve functions on co-compact normal coverings	247

LIST OF FIGURES

FIGUR	E Pa	age
2.1	An example of $\sigma(L)$	19
3.1	An illustration of the regions $M_{s,C}$ and $N_{s,C}$ when $k_0 \equiv 0.$	78

1. INTRODUCTION

Periodic media play a crucial role in solid state physics as well as in other areas (e.g., meta- and nano-materials). A remarkable example is the study of crystals: The atoms in a perfect crystal are placed in a periodic order and this order induces many interesting properties of the material. Mathematically speaking, to describe the one-electron model of solid state physics [6], one uses the stationary Schrödinger operator $-\Delta + V$ with a periodic potential V that represents the field created by the lattice of ions in the crystal. In general, one can study elliptic PDEs with periodic coefficients arising naturally from other contexts. For instance, periodic magnetic Schrödinger operators, overdetermined systems like Maxwell equations in a periodic medium, periodic elliptic operators on abelian coverings of compact manifolds or finite graphs are also of interest. It has been known for a long time that in mathematical physics, the Floquet-Bloch theory is a standard technique for studying such operators. Although periodic elliptic PDEs have been studied extensively, there are still many theoretical problems to address as well as new applications to explore such as metamaterials (e.g., graphene and photonic crystals), carbon nanostructures, topological insulators, waveguides and so on.

In Chapter 2, we briefly discuss periodic elliptic operators and then give a quick review of the Floquet transform and the Floquet-Bloch theory. Here we introduce definitions, notations, and preliminary results that will be needed in the subsequent chapters.

In Chapter 3, we describe the behavior at infinity of Green's functions near gap edges for periodic elliptic operators in \mathbb{R}^d ($d \ge 2$), which is an example of a threshold effect. These asymptotics are relevant for many important applications in Anderson localization, impurity spectra, Martin boundaries and random walks. The original motivation comes from the well-known asymptotic behavior of the Green's function of the Laplacian in \mathbb{R}^d outside its spectrum: along with an algebraic decay factor, the Green's function decays exponentially at infinity with the rate of decay controlled by the distance to the spectrum. One may ask how the Green's functions of other classes of elliptic operators behave at infinity: Would it resemble the Laplacian case? A natural candidate is the class of periodic self-adjoint elliptic operators of the second-order.

Exponential decay estimates of Schwartz kernels of resolvents, such as the Combes-Thomas estimates [18] are well-known. However, as most of them are obtained by operatortheoretic techniques, they do not capture precisely the anisotropy of the asymptotics for periodic operators. Moreover, an additional algebraically decaying factor besides the exponential decay rate is lost in this approach.

Since one can compute the asymptotics of the Green's function of the Laplacian by using Fourier transform, it is natural to use Floquet theory in the periodic case. Consider a periodic self-adjoint elliptic operator L on \mathbb{R}^d , where $d \ge 2$. Its spectrum has a band-gap structure, so it is reasonable to look at the Green's function around the spectral edges of a gap. A special case is of the bottom of the spectrum, for which results were established by Babilott [9] and then by Murata-Tsuchida [60].

Here we attack the case of an internal spectral edge, where the corresponding band function $\lambda_j(k)$ has an isolated nondegenerate extremum.¹ By using Floquet transform and a localization argument, we reduce the problem to studying the behavior of a scalar integral that involves a germ of the branch λ_j . Here only the local structure around the edge of the branch λ_j is solely responsible for the main term of the scalar integral, and this is where nondegeneracy of the extremum is used to compute the asymptotics. This also confirms the expectation that the asymptotics resemble the ones for the Laplacian. Indeed, the asymptotics of the Green's function $G_{\lambda}(x, y)$ of L at an energy level λ that is

¹Non-degeneracy means that the Hessian matrix of the corresponding band function $\lambda_j(k)$ at the gap edge is non-degenerate.

sufficiently close to a simple and nondegenerate spectral edge can be described roughly as follows:

$$G_{\lambda}(x,y) \sim \frac{e^{-\gamma_s |x-y|}}{|x-y|^{(d-1)/2}} \times C(\lambda,x,y), \quad \text{where} \quad s := \frac{(x-y)}{|x-y|}.$$
 (1.1)

Here $\gamma_s > 0$ depends on the direction s only and $C(\lambda, x, y)$ is a bounded term depending on the local structure of the dispersion relation of L around the spectral edge. Hence, the rate of exponential decay is usually non-isotropic, unlike in the Laplacian case. The additional algebraic decay is also captured.

In Chapter 4, we extend the above results to periodic elliptic operators on an abelian cover of a compact base. An abelian cover X is a normal Riemannian cover such that its deck transformation group G is abelian and the quotient space M = X/G is a compact Riemannian manifold. Without loss of generality, one can assume $G = \mathbb{Z}^d$. Let $d_X(\cdot, \cdot)$ be the Riemannian distance on X. Note that the dimension n of the manifold X could be different from the rank d of the deck group G. Let A be a self-adjoint elliptic operator of second-order on X such that it is periodic, i.e., A commutes with actions of G. One can define a Floquet transform on X so that the machinery of Floquet theory still works well. The notions of dispersion relations, band-gaps, and spectral edges are defined similarly to the flat case and thus, the question of finding the asymptotics of Green's function $G_{A,\lambda}(x, y)$ of A at a level λ that is near to a non-degenerate gap edge still makes sense. At first glance, one may expect that the leading term of the Green's function of A can be represented in terms of the distance $d_X(x, y)$ like the formula (1.1) in the flat case. However, this is just 'almost' true as we show that the asymptotics should look like

$$G_{A,\lambda}(x,y) \sim \frac{e^{-\gamma_s |h(x) - h(y)|}}{|h(x) - h(y)|^{(d-1)/2}} \times \tilde{C}(\lambda, x, y), \quad \text{where} \quad s = \frac{h(x) - h(y)}{|h(x) - h(y)|}.$$
 (1.2)

Here, the mapping h from X to \mathbb{R}^d is an analog on an abelian covering of the coordinate function x in the flat case. ² It satisfies $|h(x) - h(y)| \sim d_X(x,y)$ when x and y are sufficiently far apart. The positive constant γ_s depends on s, and the term $\tilde{C}(\lambda, x, y)$ is again bounded and does not contribute much to the asymptotics at infinity. An interesting feature is that the rank d of G plays a more important role than the dimension n of X, since n does not enter explicitly into (1.2). This feature is in line with Gromov's idea that the large scale geometry of X is captured mostly by its deck group G. Furthermore, the result of [52] concerning the behavior at exactly a spectral edge is also extended to co-compact abelian coverings.

In Chapter 5, we combine the Riemann-Roch and the Liouville theorems for periodic elliptic operators on abelian coverings. The classical Riemann-Roch theorem for compact Riemann surfaces has been extended in various ways to higher dimensional settings. Instead of taking the viewpoint from algebraic geometry, Gromov and Shubin took a motivation from classical analysis of solutions of general elliptic equations with point singularities [32] and beyond [33]. Consider a compact manifold X of dimension n, a point divisor is an element in the free abelian group generated by the points of X. For an elliptic operator P of order m and a point divisor μ on X, they define the space $L(\mu, P)$ of "meromorphic" solutions of P associated to μ by taking all solutions that are allowed to have some poles at points that enter μ with a positive degree and are required to have zeros at the points that enter μ with a negative degree. Also, the multiplicities of these poles and zeros are controlled from above and below by quantities involving the corresponding degrees. The Riemann-Roch type formula appeared in [32] is a link between the dimensions of the space $L(\mu, P)$ and the 'dual' one $L(\mu^{-1}, P^*)$, where μ^{-1} is the dual divisor of μ and

²A topological approach to defining such mappings on **any** Riemannian co-compact coverings can be found in Appendix A.

 P^* is the adjoint operator of P:

$$\dim L(\mu, P) = \dim L(\mu^{-1}, P^*) + \operatorname{ind} P + \deg_P(\mu), \tag{1.3}$$

where ind P is the Fredholm index of P and $\deg_P(\mu)$ is the degree of μ that is written in terms of binomial coefficients involving m, n and μ . The classical Riemann-Roch formula is a special case of (1.3) if P is the $\overline{\partial}$ operator on a compact Riemann surface X. Later in [33], Gromov and Shubin proved a much more general version of (1.3) for solutions of general elliptic equations with singularities supported on arbitrary compact nowhere dense sets of a manifold (e.g., a Sierpinsky carpet). In this version, the analogs of point divisors are rigged divisors.³ A version of (1.3) on non-compact manifolds is also given in [32, 33].

On the other hand, the classical Liouville theorem says that a harmonic function that grows polynomially is a harmonic polynomial and so, the space of all harmonic functions in \mathbb{R}^n that are bounded by $C(1 + |x|)^N$ is of finite dimension:

$$h_{n,N} := \binom{n+N}{n} - \binom{n+N-2}{n}.$$
(1.4)

This leads to a natural problem concerning the finite dimensionality of the spaces of solutions of an assigned polynomial growth, estimates of their dimensions, and descriptions of the structures of these solutions for more general elliptic operators on certain non-compact manifolds. In the flat case \mathbb{R}^n , M. Avellaneda-F.H. Lin [8] and J. Moser-M. Struwe [59] answered the question for any second order divergence form elliptic equation with periodic coefficients by using tools of homogenization theory: they proved that the dimensions of the spaces of all solutions of these equations of polynomial growth of order at most N are equal to $h_{n,N}$. However, homogenization techniques have significant limitations, e.g., they

³A rigged divisor μ is a tuple $(D^+, L^+; D^-, L^-)$, where D^+, D^- are disjoint nowhere dense compact sets and L^+, L^- are finite-dimensional spaces of distributions supported on D^+, D^- , respectively.

work at the bottom of the spectrum only. Using Floquet theory and duality arguments, in [50], P. Kuchment and Y. Pinchover obtained Liouville type results for a wide range of elliptic periodic operators on abelian co-compact coverings. The main result of [50] says that the Liouville type result happens iff the corresponding Fermi surface is finite, which normally happens at a spectral edge. This is another example of threshold effects: the dimensions are calculated explicitly based on the lowest order term of a non-zero Taylor expansion term of the dispersion relations at a spectral edge.

A solution of polynomial growth of order N can be considered as a solution that is allowed to have a "pole at infinity" of order at most N, while a solution that is "zero at infinity" with multiplicity at least N can be regarded as a solution with a rate of decay of polynomial of order N. So one may think a Liouville type result as a Riemann-Roch type result for divisors located at infinity. It is thus natural to try to combine these theorems.

Consider $p \in [1, \infty]$, $N \in \mathbb{R}$ and an abelian cover X whose deck group is $G(=\mathbb{Z}^d)$. Suppose that μ is a rigged divisor (defined in Section 5.3) and A is a periodic elliptic operator on X. We denote by $L_p(\mu, A, N)$ the space of all solutions u of the equation Au = 0 with zeros enforced and poles allowed by the divisor μ , and of polynomial growth (in L^p -sense) of order N at infinity (see Section 5.4). Now we can give a rough statement (somewhat vague at this point) to outline some of our main results in Chapter 5.

Theorem 1.0.1. Assume that $N \ge 0$, $p \in [1, \infty]$ and μ is a rigged divisor on X. We define p' := p/(p-1).

(i) If 0 is in the resolvent set of A, then the following Liouville-Riemann-Roch equality holds:

$$\dim L_p(\mu, A, N) = \dim L_{p'}(\mu^{-1}, A^*, -N) + \deg_A(\mu).$$

(ii) Suppose that 0 is in the spectrum of A and the operator A satisfies the Liouville property. Let us denote by $V_N^p(A)$ the space $L_p(\mu_0, A, N)$, where μ_0 is the trivial

divisor $(\emptyset, 0; \emptyset, 0)$. Assume either $p = \infty$, $N \ge 0$ or $p \in (1, \infty)$, N > d/p. Then under some local conditions on the dispersion relation of A, we have the following Liouville-Riemann-Roch inequality:

$$\dim L_p(\mu, A, N) \ge \dim V_N^p(A) + \deg_A(\mu) + \dim L_{p'}(\mu^{-1}, A, -N).$$
(1.5)

The inequality (1.5) can be strict in some cases, e.g., when $A = -\Delta$ on \mathbb{R}^3 , N = 0 with some μ . However, when μ contains only poles, the equality in (1.5) occurs.

Here $\deg_A(\mu)$ is the integer depending on μ and the differential operator A (see Section 5.3).

In particular, we are able to show that the Liouville-Riemann-Roch inequalities hold for 'generic' periodic Schrödinger operators at their gap edges and for two-dimensional periodic Schrödinger operators with honeycomb lattice potentials. An immediate application is the non-triviality of the space $L_{\infty}(\mu, A, N)$ in many situations, e.g., when 0 is at the bottom of the spectrum of $A = -\Delta + V$ and $h_{d,N} + \deg_A(\mu) > 0$. This also implies the fact that one can always find a non-zero solution u of polynomial-growth of Awith prescribed zeros as long as N is large enough. The latter fact can be considered as an analog for noncompact abelian coverings of the following well-known property which could be deduced from the classical Riemann-Roch equality: the existence of a non-trivial meromorphic function on a compact Riemann surface with prescribed zeros, provided that poles of sufficiently high orders are allowed.

The results of this dissertation have been published in:

 (Joint with Peter Kuchment, Andy Raich), Green's function asymptotics near the internal edges of spectra of periodic elliptic operators. Spectral gap interior, To appear in Journal of Spectral Theory (2016), arXiv:1508.06703.

- 2. Green's function asymptotics of periodic elliptic operators on abelian coverings of compact manifolds, Submitted, arXiv:1511.00276.
- 3. A short note on additive functions on Riemannian co-compact coverings, Preprint, arXiv:1511.00185.
- 4. (Joint with Peter Kuchment), *A Liouville-Riemann-Roch theorem on abelian coverings*, in preparation.

The results of this dissertation have been presented in the following talks:

- Green's function asymptotics of periodic elliptic operators on abelian coverings of compact manifolds, Contributed talk, Ohio River Analysis Meeting, University of Kentucky, Kentucky, 03/12/2016.
- Green's function asymptotics of periodic elliptic operators on abelian coverings of compact manifolds, Invited talk, Seminar in Applied Mathematics, Department of Mathematics and Statistics, Texas Tech University, TX, 03/02/2016.
- 3. *Green's function asymptotics near the internal edges of spectra of periodic elliptic operators. Spectral gap interior.*, Invited talk, Special Session on Analysis of Differential and Integral Equations, AMS Fall Southeastern Sectional Meeting, University of Memphis, TN, 10/18/2015.
- Green's function asymptotics near the internal edges of spectra of periodic elliptic operators. Spectral gap interior., Contributed talk, Prairie Analysis Seminar, Kansas State University, Manhattan, KS, 09/26/2015.
- Green's function asymptotics near the internal edges of spectra of periodic elliptic operators. Spectral gap interior., Mathematical Physics and Harmonic Analysis Seminar, Texas A&M University, 03/06/2015.

2. FLOQUET-BLOCH THEORY FOR PERIODIC ELLIPTIC OPERATORS

2.1 Periodic elliptic operators

Let a linear differential expression

$$Lu(x) = L(x, D)u = \sum_{|\alpha| \le m} a_{\alpha}(x)D^{\alpha}u(x)$$
(2.1)

of order $m \ge 1$ be given in \mathbb{R}^d . Here if $\alpha = (\alpha_1, \ldots, \alpha_d)$ is a multi-index, the notation D^{α} stands for $D_1^{\alpha_1} \ldots D_d^{\alpha_d}$, where $D_k := -i\partial_k = \frac{1}{i}\frac{\partial}{\partial x_k}$. All coefficients a_{α} are smooth functions on \mathbb{R}^d and they are periodic with respect to the integer lattice \mathbb{Z}^d in \mathbb{R}^d , i.e., $\forall x \in \mathbb{R}^d, n \in \mathbb{Z}^d$: $a_{\alpha}(x+n) = a_{\alpha}(x)$.

The formal adjoint expression (or transpose) is defined as

$$L^{t}u(x) = L(x, D)^{t}u = \sum_{|\alpha| \le m} D^{\alpha}(\overline{a_{\alpha}(x)}u(x)),$$

i.e., $\langle Lu, v \rangle = \langle u, L^t v \rangle$ for any $u, v \in C_c^{\infty}(\mathbb{R}^d)$, where $\langle \cdot, \cdot \rangle$ denotes the L^2 -inner product in \mathbb{R}^d . If $L = L^t$, then L is called a formally self-adjoint differential expression and the operator L with the domain $C_c^{\infty}(\mathbb{R}^d)$ is symmetric.

The full symbol of L is the polynomial (in $\xi \in \mathbb{R}^d$) $L(x,\xi) = \sum_{|\alpha| \le m} a_{\alpha}(x)\xi^{\alpha}$, while the principal symbol of L is the polynomial $L_0(x,\xi) = \sum_{|\alpha|=m} a_{\alpha}(x)\xi^{\alpha}$. Note that $(L^t)_0(x,\xi) = \overline{L_0(x,\xi)}$ and hence, the principal symbol of a formally self-adjoint expression is always real.

The scalar differential expression L is said to be (uniformly) elliptic if its principal

symbol satisfies the inequality

$$|L_0(x,\xi)| \ge c|\xi|^m, \quad c > 0.$$

Under these assumptions, the differential expression (2.1) is defined for $u \in L^2(\mathbb{R}^d)$ in the distribution sense. To consider L as a linear operator in $L^2(\mathbb{R}^d)$, we have to choose a domain ${}^1 \mathcal{D} \subset L^2(\mathbb{R}^d)$ of L such that $Lu \in L^2(\mathbb{R}^d)$ for any $u \in \mathcal{D}$. We denote by $L_{|\mathcal{D}}$ the corresponding operator (or a *realization*) of the differential expression L. There are two important realizations of L. The maximal operator L_{\max} is the linear operator $L_{|\mathcal{D}_{\max}}$, where its domain \mathcal{D}_{\max} is the largest possible one, i.e.,

$$D_{\max} := \{ u \in L^2(\mathbb{R}^d) \mid Lu \in L^2(\mathbb{R}^d) \}.$$

The minimal operator L_{\min} is defined to be the closure of the operator $L_{|C_c^{\infty}(\mathbb{R}^d)}$. Since L is uniformly elliptic and its coefficients are smooth, $D_{\min} = H^m(\mathbb{R}^d)$. When L is a formally self-adjoint expression, it follows that the maximal operator L_{\max} is the adjoint L_{\min}^* of the minimal operator L_{\min} . If $L_{\min} = L_{\max}$ or equivalently, $D_{\max} = D_{\min}$, then L_{\min} is a self-adjoint operator and in this case, L is said to be essentially self-adjoint on $C_c^{\infty}(\mathbb{R}^d)$. In other words, L has a unique self-adjoint realization. It is known [71] that when $L = L^t$ and the coefficients of L are smooth and periodic, the differential expression L is essentially self-adjoint on $C_c^{\infty}(\mathbb{R}^d)$. We also use the same notation L to denote the self-adjoint operator $L_{\min} = L_{\max} = L_{|H^m(\mathbb{R}^d)}$. For simplicity, we also say that L is self-adjoint in this case.

We now describe some important examples of periodic elliptic operators, which will come up in the next chapters.

¹I.e., a dense linear subspace of $L^2(\mathbb{R}^d)$.

(a) The one-electron model of solid state physics can be described by the *periodic Schrödinger* operator in $L^2(\mathbb{R}^d)$

$$L = -\Delta + V(x),$$

where the electric potential V is sufficiently smooth ² and periodic with respect to the group \mathbb{Z}^d . The domain of this operator is the Sobolev space $H^2(\mathbb{R}^d)$. This Schrödinger operator is self-adjoint whenever the potential V is real. Note that many of the techniques and results do not require self-adjointness, i.e., they can also hold for complex-valued potentials V.

(b) One can also consider the magnetic Schrödinger operators

$$L = (-i\nabla + A(x))^2 + V(x),$$

where A(x) and V(x) are periodic magnetic and electric potentials. More generally, the Schrödinger operators with the presence of periodic metrics

$$L = -\nabla \cdot g(x)\nabla + iA(x) \cdot \nabla + V(x)$$

are also of interest. Usually, these operators are more complicated to study.

(c) *Periodic elliptic operators of higher than second-order* are also worthy of studying. A typical example is the polyharmonic Schrödinger operator

$$L = (-\Delta)^{m+1} + V(x),$$

where V is a real and periodic potential and m is a positive integer. We would like to remark that unlike the second-order case, one may encounter more difficulties with

 $^{^{2}}L^{\infty}$ suffices (see e.g., [66]).

these operators since they may fail even the weakest version of unique continuation.

(d) In this chapter, we mainly discuss Z^d-periodic elliptic operators on R^d. Most of the techniques and some of the results in this thesis can be generalized to the case of periodic elliptic operators on abelian coverings of compact Riemannian manifolds (see Chapter 4 and Chapter 5).

2.2 Floquet-Bloch theory

Notation 2.2.1.

- (a) Let $W = [0, 1]^d \subset \mathbb{R}^d$ be the unit cube, which is a **fundamental domain** of \mathbb{R}^d with respect to the lattice \mathbb{Z}^d (Wigner-Seitz cell).
- (b) The dual (or reciprocal) lattice is $2\pi\mathbb{Z}^d$ and the Brillouin zone is its fundamental domain $B = [-\pi, \pi]^d$.
- (c) The *d*-dimensional tori with respect to the lattices \mathbb{Z}^d and $2\pi\mathbb{Z}^d$ are denoted by $\mathbb{T}^d := \mathbb{R}^d/\mathbb{Z}^d$ and $(\mathbb{T}^*)^d := \mathbb{R}^d/2\pi\mathbb{Z}^d$, respectively.

Definition 2.2.2. For any $k \in \mathbb{C}^d$, the subspace $H_k^s(W) \subset H^s(W)$ consists of restrictions to W of functions $f \in H_{loc}^s(\mathbb{R}^d)$ that satisfy for any $\gamma \in \mathbb{Z}^d$ the **Floquet-Bloch condition** (also known as automorphicity condition or cyclic condition)

$$f(x+\gamma) = e^{ik\cdot\gamma}f(x) \quad \text{for a.e} \quad x \in W.$$
(2.2)

Here H^s denotes the standard Sobolev space of order s. Note that when s = 0, the above space coincides with $L^2(W)$ for any k. In this definition, the vector k is called the **quasimomentum**³.

³The name comes from solid state physics [6].

Due to periodicity, the operator L(x, D) preserves condition (2.2) and thus it defines an operator L(k) in $L^2(W)$ with the domain $H_k^m(W)$. In this model, L(k) is realized as a k-independent differential expression L(x, D) acting on functions in W with boundary conditions depending on k (which can be identified with sections of a linear bundle over the torus \mathbb{T}^d). An alternative definition of L(k) is as the operator L(x, D + k) in $L^2(\mathbb{T}^d)$ with the domain $H^m(\mathbb{T}^d)$. In the latter model, L(k) acts on the k-independent domain of periodic functions on W as follows:

$$e^{-ik \cdot x} L(x, D) e^{ik \cdot x} = \sum_{|\alpha| \le m} a_{\alpha}(x) (D+k)^{\alpha}.$$
(2.3)

Due to ellipticity and embedding theorems (see [47, Theorem 2.1]), the operators $L(k) = L(x, D + k) : H^m(\mathbb{T}^d) \to L^2(\mathbb{T}^d)$ are Fredholm for $k \in \mathbb{C}^d$. In addition, note that the condition (2.2) is invariant under translations of k by elements of the dual lattice $2\pi\mathbb{Z}^d$. Moreover, the operator L(k) is unitarily equivalent to $L(k + 2\pi\gamma)$, for any $\gamma \in \mathbb{Z}^d$. In particular, when dealing with real values of k, it suffices to restrict k to the Brillouin zone $[-\pi, \pi]^d$ (or any its shifted copy).

Fourier transform is a major tool of studying linear constant coefficient PDEs, due to their invariance with respect to all shifts. The periodicity of the operator L suggests that it is natural to apply the Fourier transform with respect to the period group \mathbb{Z}^d to blockdiagonalize L. In fact, it is an analog of the Fourier transform on the group \mathbb{Z}^d of periods. The group Fourier transform we have just mentioned is the so called **Floquet transform** \mathcal{F} (see e.g., [26, 47, 48]). Now let us consider a sufficiently fast decaying function f(x)(to begin with, compactly supported functions) on \mathbb{R}^d .

Definition 2.2.3. The Floquet transform \mathcal{F}

$$f(x) \to \widehat{f}(k, x)$$

maps a function f on \mathbb{R}^d into a function \hat{f} defined on $\mathbb{R}^d \times \mathbb{R}^d$ in the following way:

$$\widehat{f}(k,x) := \sum_{\gamma \in \mathbb{Z}^d} f(x+\gamma) e^{-ik \cdot (x+\gamma)}$$

From the above definition, one can see that \widehat{f} is \mathbb{Z}^d -periodic in the *x*-variable and satisfies a cyclic condition with respect to *k*:

$$\begin{cases} \widehat{f}(k, x + \gamma) = \widehat{f}(k, x), \quad \forall \gamma \in \mathbb{Z}^d \\ \widehat{f}(k + 2\pi\gamma, x) = e^{-2\pi i \gamma \cdot x} \widehat{f}(k, x), \quad \forall \gamma \in \mathbb{Z}^d \end{cases}.$$

Thus, it suffices to consider the Floquet transform \widehat{f} as a function defined on $[-\pi, \pi]^d \times \mathbb{T}^d$. Usually, we will regard \widehat{f} as a function $\widehat{f}(k, \cdot)$ in k-variable in $[-\pi, \pi]^d$ with values in the function space $L^2(\mathbb{T}^d)$.

We list some well-known results of the Floquet transform (see e.g., [47, 48]):

Lemma 2.2.4.

I. The transform \mathcal{F} is an isometry of $L^2(\mathbb{R}^d)$ onto

$$\int_{[-\pi,\pi]^d}^{\oplus} L^2(\mathbb{T}^d) = L^2([-\pi,\pi]^d, L^2(\mathbb{T}^d))$$

and of $H^2(\mathbb{R}^d)$ into

$$\int_{[-\pi,\pi]^d}^{\oplus} H^2(\mathbb{T}^d) = L^2([-\pi,\pi]^d, H^2(\mathbb{T}^d)).$$

II. The inversion \mathcal{F}^{-1} is given by the formula

$$f(x) = (2\pi)^{-d} \int_{[-\pi,\pi]^d} e^{ik \cdot x} \widehat{f}(k,x) \, \mathrm{d}k, \quad x \in \mathbb{R}^d.$$
(2.4)

By using cyclic conditions of \hat{f} , we obtain an alternative inversion formula

$$f(x) = (2\pi)^{-d} \int_{[-\pi,\pi]^d} e^{ik \cdot x} \widehat{f}(k, x - \gamma) \,\mathrm{d}k, \quad x \in W + \gamma.$$
(2.5)

III. The action of any \mathbb{Z}^d -periodic elliptic operator L (not necessarily self-adjoint) in $L^2(\mathbb{R}^d)$ under the Floquet transform \mathcal{F} is given by

$$\mathcal{F}L(x,D)\mathcal{F}^{-1} = \int_{[-\pi,\pi]^d}^{\oplus} L(x,D+k) \,\mathrm{d}k = \int_{[-\pi,\pi]^d}^{\oplus} L(k) \,\mathrm{d}k,$$

where L(k) is defined in (2.3).

Equivalently,

$$\widehat{Lf}(k) = L(k)\widehat{f}(k), \quad \forall f \in H^2(\mathbb{R}^d).$$

- *IV.* (A Paley-Wiener theorem for \mathcal{F} .) Let $\phi(k, x)$ be a function defined on $\mathbb{R}^d \times \mathbb{R}^d$ such that for each k, it belongs to the Sobolev space $H^s(\mathbb{T}^d)$ for $s \in \mathbb{R}^+$ and satisfies the cyclic condition in k-variable. Then
 - (a) Suppose the mapping $k \to \phi(k, \cdot)$ is a C^{∞} -map from \mathbb{R}^d into the Hilbert space $H^s(\mathbb{T}^d)$. Then $\phi(k, x)$ is the Floquet transform of a function $f \in H^s(\mathbb{R}^d)$ such that for any compact set K in \mathbb{R}^d and any N > 0, the norm $||f||_{H^s(K+\gamma)} \leq C_N |\gamma|^{-N}$. In particular, by Sobolev's embedding theorem, if s > d/2, then the

pointwise estimation holds:

$$|f(x)| \le C_N (1+|x|)^{-N}, \quad \forall N > 0.$$

(b) Suppose the mapping $k \to \phi(k, \cdot)$ is an analytic map from \mathbb{R}^d into the Hilbert space $H^s(\mathbb{T}^d)$. Then $\phi(k, x)$ is the Floquet transform of a function $f \in H^s(\mathbb{R}^d)$ such that for any compact set K in \mathbb{R}^d , one has $||f||_{H^s(K+\gamma)} \leq Ce^{-C|\gamma|}$. In particular, by Sobolev's embedding theorem, if s > d/2, then the pointwise estimation holds:

$$|f(x)| \le Ce^{-C|x|}$$

From Lemma 2.2.4, one can deduce the well-known result (see [20, 47, 66]) that the spectrum of L is the union of all the spectra of L(k) when k runs over the Brillouin zone, i.e.

$$\sigma(L) = \bigcup_{k \in [-\pi,\pi]^d} \sigma(L(k)).$$
(2.6)

We now remind some notions that play a crucial role in studying periodic PDEs (see e.g., [47,48]).

Definition 2.2.5.

(a) A **Bloch solution** of the equation L(x, D)u = 0 is a solution of the form

$$u(x) = e^{ik \cdot x}\phi(x),$$

where the function ϕ is 1-periodic in each variable x_j for j = 1, ..., d. The vector k is the **quasimomentum** and $z = e^{ik} = (e^{ik_1}, ..., e^{ik_d})$ is the **Floquet multiplier** of the solution. In our formulation, allowing quasimomenta k to be complex is essential.

(b) The (complex) **Bloch variety** B_L of the operator L consists of all pairs $(k, \lambda) \in \mathbb{C}^{d+1}$ such that λ is an eigenvalue of the operator L(k):

$$B_L = \{ (k, \lambda) \in \mathbb{C}^{d+1} : \lambda \in \sigma(L(k)) \}.$$

In another word, the pair (k, λ) belongs to B_L if and only if the equation $Lu = \lambda u$ in \mathbb{R}^d has a non-zero Bloch solution u with a quasimomentum k. Also, the Bloch variety B_L is also called the **dispersion relation/curve**, i.e., the graph of the multivalued function $\lambda(k)$.

- (c) The (complex) Fermi surface F_{L,λ} of the operator L at the energy level λ ∈ C consists of all quasimomenta k ∈ C^d such that the equation L(k)u = λu has a nonzero solution. Equivalently, k ∈ F_{L,λ} means the existence of a nonzero periodic solution u of the equation L(k)u = λu. In other words, Fermi surfaces are level sets of the dispersion relation. By definitions, F_{L,λ} is 2πZ^d-periodic.
- (d) We denote by B_{L,ℝ} and F_{L,λ,ℝ} the real Bloch variety B_L ∩ ℝ^{d+1} and the real Fermi surface F_{L,λ} ∩ ℝ^d, respectively.
- (e) Whenever λ = 0, we will write F_L and F_{L,ℝ} instead of F_{L,0} and F_{L,0,ℝ}, correspondingly. This is convenient, since being at the spectral level λ, we could consider the operator L − λI instead of L and thus, F_{L,λ} = F_{L-λ} and F_{L,λ,ℝ} = F_{L-λ,ℝ}. In other words, we will be able to assume, w.l.o.g. that λ = 0.

Some important properties of Bloch variety and Fermi surface are stated in the next proposition (see e.g., [47,48,50]).

Proposition 2.2.6.

- (a) The Fermi surface and the Bloch variety are the zero level sets of some entire $(2\pi\mathbb{Z}^d$ -periodic in k) functions of finite orders on \mathbb{C}^d and \mathbb{C}^{d+1} respectively.
- (b) The Bloch variety is a $2\pi\mathbb{Z}^d$ -periodic, complex analytic subvariety of \mathbb{C}^{d+1} of codimension one.
- (c) The real Fermi surface $F_{L,\lambda}$ either has zero measure in \mathbb{R}^d or coincides with the whole \mathbb{R}^d .
- (d) $(k,\lambda) \in B_L$ if and only if $(-k,\bar{\lambda}) \in B_{L^*}$. In other words, $F_{L,\lambda} = -F_{L^*,\bar{\lambda}}$ and $F_{L,\mathbb{R}} = -F_{L^*,\mathbb{R}}$.

The analytical and geometrical properties of dispersion relations encode significant information about spectral features of the operator. For example, the absolute continuity of the spectrum of a self-adjoint periodic elliptic operator (which is true for a large class of periodic Schrödinger operators) can be reformulated as the absence of flat components in its Bloch variety, which is also equivalent to a seemingly stronger fact that the Fermi surface at each energy level has zero measure (due to Proposition 2.2.6).

If L is self-adjoint, then by (2.3), L(k) is self-adjoint in $L^2(\mathbb{T}^d)$ and has domain $H^m(\mathbb{T}^d)$ for each $k \in \mathbb{R}^d$. Due to the ellipticity of L, each L(k) is bounded from below and has compact resolvent. This forces each of the operators L(k), $k \in \mathbb{R}^d$ to have discrete spectrum in \mathbb{R} . Therefore, we can label its eigenvalues in non-decreasing order:

$$\lambda_1(k) \le \lambda_2(k) \le \dots \qquad (2.7)$$

Hence, we can single out continuous **band functions** $\lambda_j(k)$ for each $j \in \mathbb{N}$ [80]. The range of the band function λ_j constitutes exactly **the band** $[\alpha_j, \beta_j]$ of the spectrum of L in (2.8) (e.g., see Figure 2.1). Hence, when L is self-adjoint, the spectrum of the operator L in $L^2(\mathbb{R}^d)$ has a **band-gap structure** [20, 47, 66], i.e., it is the union of a sequence of closed bounded intervals (**bands** or **stability zones** of the operator *L*) $[\alpha_j, \beta_j] \subset \mathbb{R} (j = 1, 2, ...)$:

$$\sigma(L) = \bigcup_{j=1}^{\infty} [\alpha_j, \beta_j], \qquad (2.8)$$

such that $\alpha_j \leq \alpha_{j+1}$, $\beta_j \leq \beta_{j+1}$ and $\lim_{j\to\infty} \alpha_j = \infty$. The bands can (and do) overlap when d > 1, but they may leave open intervals in between, called **spectral gaps**. Thus, a spectral gap is an interval of the form (β_j, α_{j+1}) for some $j \in \mathbb{N}$ for which $\alpha_{j+1} > \beta_j$. A **finite spectral gap** is of the form (β_j, α_{j+1}) for some $j \in \mathbb{N}$ such that $\alpha_{j+1} > \beta_j$, and the **semifinite spectral gap** is the open interval $(-\infty, \alpha_1)$, which contains all real numbers below the bottom of the spectrum of L.



Figure 2.1: An example of $\sigma(L)$.

From Proposition 2.2.6 and the proof of [47, Lemma 4.5.1] (see also [80]), the band functions $\lambda(k)$ are piecewise analytic on \mathbb{C}^d .

Remark 2.2.7.

- (a) It is worthwhile to mention that in the first three statements of Lemma 2.2.4, one can replace the Brillouin zone [-π, π]^d by any other fundamental domain of R^d with respect to the dual lattice 2πZ^d (due to 2πZ^d-periodicity in quasimomentum k). In the next chapters, we will use this lemma with some fundamental domain of the form [-π, π]^d + k₀, where k₀ is a fixed quasimomentum in R^d.
- (b) It is sometimes useful to employ an alternative version of Floquet transform for which the reader can find more details in Chapter 5. Analogs of the Plancherel and Paley-Wiener theorem for these Floquet transforms are also obtained like we just see in Lemma 2.2.4.
- (c) Our above discussion about Floquet-Bloch theory (e.g., Floquet transform and its properties in Lemma 2.2.4) can be transferred without any major change to the case of periodic elliptic operators on co-compact abelian coverings.

3. GREEN'S FUNCTION ASYMPTOTICS NEAR THE INTERNAL EDGES OF SPECTRA OF PERIODIC ELLIPTIC OPERATORS. SPECTRAL GAP INTERIOR^{*}

3.1 Introduction

The behavior at infinity of the Green function of the Laplacian in \mathbb{R}^n outside and at the boundary of its spectrum is well known. Analogous results below and at the lower boundary of the spectrum have been established for bounded below periodic elliptic operators of the second order in [9, 60] (see also [81] for the discrete version). Due to the band-gap structure of the spectra of such periodic operators, the question arises whether similar results can be obtained at or near the edges of spectral gaps. The corresponding result at the internal edges of the spectrum was established in [52]. The main result of this chapter, Theorem 3.2.5, is the description of such asymptotics near the spectral edge for generic periodic elliptic operators of second-order with real coefficients in dimension $d \ge 2$, if the spectral edge is attained at a symmetry point of the Brillouin zone.

It is well known that outside of the spectrum the Green function decays exponentially at infinity, with the rate of decay controlled by the distance to the spectrum. See, e.g., Combes-Thomas estimates [10, 18]. However, comparison with the formulas for the case of the Laplacian shows that an additional algebraically decaying factor (depending on the dimension) is lost in this approach. Moreover, the exponential decay in general is expected to be anisotropic, while the operator theory approach can provide only isotropic estimates. The result of this chapter provides the exact principal term of asymptotics, thus resolving these issues.

^{*}Reprinted with permission from "Green's function asymptotics near the internal edges of spectra of periodic elliptic operators. Spectral gap interior.", by Minh Kha, Peter Kuchment, and Andrew Raich, to appear in Journal of Spectral Theory. Copyright cc by the European Mathematical Society.

3.2 Assumptions, notation and the main result

Consider a linear second order elliptic operator in \mathbb{R}^d with periodic coefficients

$$L(x,D) = \sum_{k,l=1}^{d} D_k(a_{kl}(x)D_l) + V(x) = D^*A(x)D + V(x).$$
(3.1)

Here $A = (a_{kl})_{1 \le k,l \le d}$, $D = (D_1, \ldots, D_d)$, and $D_k := -i\partial_k = -i\frac{\partial}{\partial x_k}$. All coefficients a_{kl}, V are smooth real-valued functions on \mathbb{R}^d , periodic with respect to the integer lattice \mathbb{Z}^d in \mathbb{R}^d , i.e., $a_{kl}(x+n) = a_{kl}(x)$ and V(x+n) = V(x), $\forall x \in \mathbb{R}^d, n \in \mathbb{Z}^d$. The operator L is assumed to be uniformly elliptic, i.e., the matrix A is symmetric and

$$\sum_{k,l=1}^{d} a_{kl}(x)\xi_k\xi_l \ge \theta |\xi|^2,$$
(3.2)

for some $\theta > 0$ and any $x \in \mathbb{R}^d$, $\xi = (\xi_1, \dots, \xi_d) \in \mathbb{R}^d$. We recall from the previous chapter that the operator L, with the Sobolev space $H^2(\mathbb{R}^d)$ as the domain, is an unbounded, self-adjoint operator in $L^2(\mathbb{R}^d)$. Moreover, the spectrum $\sigma(L)$ of the operator L has the band-gap structure:

$$\sigma(L) = \bigcup_{j=1}^{\infty} [\alpha_j, \beta_j].$$
(3.3)

We consider the open interval $(-\infty, \alpha_1)$, which contains all real numbers below the bottom of the spectrum of *L*, as an infinite spectral gap. However, we will be mostly interested in finite spectral gaps.

In this chapter, we study Green's function asymptotics for the operator L in a spectral gap, near a spectral gap edge. More precisely, consider a finite spectral gap (β_j, α_{j+1}) for some $j \in \mathbb{N}$ and a value $\lambda \in (\beta_j, \alpha_{j+1})$ which is close either to the spectral edge β_j or to the spectral edge α_{j+1} . We would like to study the asymptotic behavior when $|x-y| \to \infty$ of the Green's kernel $G_{\lambda}(x, y)$ of the resolvent operator $R_{\lambda,L} := (L - \lambda)^{-1}$. The case of

the spectral edges (i.e., $\lambda = \alpha_{j+1}$ or $\lambda = \beta_j$) was studied for the similar purpose in [52]. All asymptotics here and also in [52] are deduced from an assumed "generic" spectral edge behavior of the **dispersion relation** of the operator L, which we will briefly review below.

From now on, we fix L as a self-adjoint elliptic operator of the form (3.1), whose band-gap structure is as (2.8). By adding a constant to the operator L if necessary, we can assume that the spectral edge of interest is 0. It is also enough to suppose that the adjacent spectral band is of the form [0, a] for some a > 0 since the case when the spectral edge 0 is the maximum of its adjacent spectral band is treated similarly.

Suppose there is no spectrum for small negative values of λ and hence there is a spectral gap below 0. Thus, there exists at least one band function $\lambda_j(k)$ for some $j \in \mathbb{N}$ such that 0 is the minimal value of this function on the Brillouin zone.

To establish our main result, we need to impose the following analytic assumption on the dispersion curve λ_j as in [52]:

Assumption A

There exists $k_0 \in [-\pi, \pi]^d$ *and a band function* $\lambda_j(k)$ *such that:*

A1 $\lambda_i(k_0) = 0.$

A2 $\min_{k \in \mathbb{R}^d, i \neq j} |\lambda_i(k)| > 0.$

A3 k_0 is the only¹ (modulo $2\pi\mathbb{Z}^d$) minimum of λ_i .

A4 $\lambda_j(k)$ is a Morse function near k_0 , i.e., its Hessian matrix $H := \text{Hess}(\lambda_j)(k_0)$ at

¹Finitely many such points can be also easily handled.

 k_0 is positive definite. In particular, the Taylor expansion of λ_j at k_0 is:

$$\lambda_j(k) = \frac{1}{2}(k - k_0)^t H(k - k_0) + O(|k - k_0|^3).$$

It is known [44] that the conditions A1 and A2 'generically' hold (i.e., they can be achieved by small perturbation of coefficients of the operator) for Schrödinger operators. Although this has not been proven, the conditions A3 and A4 are widely believed (both in the mathematics and physics literature) to hold 'generically'. In other words, it is conjectured that for a 'generic' selfadjoint second-order elliptic operator with periodic coefficients on \mathbb{R}^d each of the spectral gap's endpoints is a unique (modulo the dual lattice $2\pi\mathbb{Z}^d$), nondegenerate extremum of a single band function $\lambda_j(k)$ (see e.g., [50, Conjecture 5.1]). It is known that for a non-magnetic periodic Schrödinger operator, the bottom of the spectrum always corresponds to a non-degenerate minimum of λ_1 [43]. A similar statement is correct for a wider class of 'factorable' operators [13, 14]. The following condition on k_0 will also be needed:

A5 The quasimomentum k_0 is a high symmetry point of the Brillouin zone, i.e., all components of k_0 must be either equal to 0 or to π .

It is known [37] that the condition **A5** is not always satisfied and spectral edges could occur deeply inside the Brillouin zone. However, as it is discussed in [37], in many practical cases (e.g., in the media close to homogeneous) this condition holds.

We would like to introduce a suitable fundamental domain with respect to the dual lattice $2\pi\mathbb{Z}^d$ to work with.

Definition 3.2.1. Consider the quasimomentum k_0 in Assumption A. Due to A5, $k_0 = (\delta_1 \pi, \delta_2 \pi, \dots, \delta_d \pi)$, where $\delta_j \in \{0, 1\}$ for $j \in \{1, \dots, d\}$. We denote by \mathcal{O} the fundamen-

tal domain so that k_0 is its center of symmetry, i.e.,

$$\mathcal{O} = \prod_{j=1}^{d} [(\delta_j - 1)\pi, (\delta_j + 1)\pi].$$

When $k_0 = 0$, \mathcal{O} is just the Brillouin zone.

We now introduce notation that will be used throughout the chapter.

- Notation 3.2.2. (a) Let $z_1 \in \mathbb{C}$, $z_2 \in \mathbb{C}^{d-1}$, $z_3 \in \mathbb{C}^d$ and r_i be positive numbers for i = 1, 2, 3. Then we denote by $B(z_1, r_1)$, $D'(z_2, r_2)$ and $D(z_3, r_3)$ the open balls (or discs) centered at z_1 , z_2 and z_3 whose radii are r_1 , r_2 and r_3 in \mathbb{C} , \mathbb{C}^{d-1} and \mathbb{C}^d respectively.
- (b) The real parts of a complex vector z, or of a complex matrix A are denoted by R(z) and R(A) respectively.
- (c) The standard notation O(|x − y|⁻ⁿ) for a function f defined on ℝ^{2d} means there exist constants C > 0 and R > 0 such that |f(x, y)| ≤ C|x − y|⁻ⁿ whenever |x − y| > R. Also, f(x, y) = o(|x − y|⁻ⁿ) means that

$$\lim_{|x-y| \to \infty} |f(x,y)| / |x-y|^n = 0.$$

(d) We often use the notation $A \leq B$ to mean that the quantity A is less or equal than the quantity B up to some multiplicative constant factor, which does not affect the arguments.

Note that L(z) is non-self-adjoint if $z \notin \mathbb{R}^d$. Note that L(z) - L(0) is an operator of lower order for each $z \in \mathbb{C}^d$. Therefore, for each $z \in \mathbb{C}^d$, the operator L(z) has discrete spectrum and is therefore a closed operator with non-empty resolvent set (see pp.188-190 in [4]). These operators have the same domain $H^2(\mathbb{T}^d)$ and for each $\phi \in H^2(\mathbb{T}^d)$, $L(z)\phi$ is a $L^2(\mathbb{T}^d)$ -valued analytic function of z, due to (4.2.14). Consequently, $\{L(z)\}_{z\in\mathbb{C}^d}$ is an analytic family of type \mathcal{A} in the sense of Kato [39]². Due to A1-A2, $\lambda_j(k_0)$ is a simple eigenvalue of $L(k_0)$. By using analytic perturbation theory for the family $\{L(z)\}_{z\in\mathbb{C}^d}$ (see e.g., [66, Theorem XII.8]), there is an open neighborhood V of k_0 in \mathbb{C}^d and some $\epsilon_0 > 0$ such that

(P1) λ_j is analytic in a neighborhood of the closure of V.

(P2) $\lambda_j(z)$ has algebraic multiplicity one, i.e., it is a simple eigenvalue of L(z) for any $z \in \overline{V}$.

(P3) The only eigenvalue of L(z) contained in the closed disc $\overline{B}(0, \epsilon_0)$ is $\lambda_j(z)$. Moreover, we may also assume that $|\lambda_j(z)| < \epsilon_0$ for each $z \in V$.

(P4) For each $z \in \overline{V}$, let $\phi(z, x)$ be a nonzero \mathbb{Z}^d -periodic function of x such that it is the unique (up to a constant factor) eigenfunction of L(z) with the eigenvalue $\lambda_j(z)$, i.e., $L(z)\phi(z, \cdot) = \lambda_j(z)\phi(z, \cdot)$. We will also use sometimes the notation ϕ_z for the eigenfunction $\phi(z, \cdot)$.

By elliptic regularity, $\phi(z, x)$ is smooth in x. On a neighborhood of \overline{V} , $\phi(z, \cdot)$ is a $H^2(\mathbb{T}^d)$ -valued holomorphic function.

(P5) By condition A4 and the continuity of $\text{Hess}(\lambda_j)$,³ we can assume that for all $z \in V$,

$$2\Re(\operatorname{Hess}(\lambda_j)(z)) > \min \sigma(\operatorname{Hess}(\lambda_j)(k_0)) I_{d \times d}.$$

(P6) V is invariant under complex conjugation. Furthermore, the smooth function

$$F(z) := (\phi(z, \cdot), \phi(\overline{z}, \cdot))_{L^2(\mathbb{T}^d)}$$
(3.4)

²It is also an analytic family in the Banach space of bounded linear operators acting from $H^2(\mathbb{T}^d)$ to $L^2(\mathbb{T}^d)$.

³The Hessian matrix of λ_j .
is non-zero on V, due to analyticity of the mapping $z \mapsto \phi(z, \cdot)$ and the inequality $F(k_0) = \|\phi(k_0)\|_{L^2(\mathbb{T}^d)}^2 > 0.$

The following lemma will be useful when dealing with operators having real and smooth coefficients:

Lemma 3.2.3. (i) For k in \mathbb{R}^d and $i \in \mathbb{N}$,

$$\lambda_i(k) = \lambda_i(-k). \tag{3.5}$$

In other words, each band λ_i of L is an even function on \mathbb{R}^d .

(ii) If $k_0 \in X$, we have $\lambda_i(k+k_0) = \lambda_i(-k+k_0)$ for all k in \mathbb{R}^d and $i \in \mathbb{N}$.

Proof. Let ϕ_k be an eigenfunction of L(k) corresponding to $\lambda_j(k)$. This means that ϕ_k is a periodic solution to the equation

$$L(x,\partial + ik)\phi_k(x) = \lambda_i(k)\phi_k(x).$$
(3.6)

Taking the complex conjugate of (3.6), we get

$$L(x,\partial - ik)\overline{\phi_k}(x) = \lambda_j(k)\overline{\phi_k}(x).$$

Therefore, $\overline{\phi_k}$ is an eigenfunction of L(-k) with eigenvalue $\lambda_j(k)$. This implies the identity (3.5).

(ii) By (i),
$$\lambda_i(k+k_0) = \lambda_i(-k-k_0) = \lambda_i(-k+k_0)$$
 since $2k_0 \in 2\pi\mathbb{Z}^d$.

Corollary 3.2.4. If $\beta \in \mathbb{R}^d$ such that $k_0 + i\beta \in \overline{V}$ then $\lambda_j(k_0 + i\beta) \in \mathbb{R}$.

Proof. Indeed, the statement (ii) of Lemma 3.2.3 implies that the Taylor series of $\lambda(k)$ at k_0 has only even degree terms and real coefficients.

Corollary 3.2.4 allows us to define near $\beta = 0$ the real analytic function $E(\beta) := \lambda_j(k_0 + i\beta)$ near 0. Since its Hessian at 0 is negative-definite (by A4), there exists a connected and bounded neighborhood V_0 of 0 in \mathbb{R}^d such that $k_0 + iV_0 \subseteq V$ and Hess $(E)(\beta)$ is negative-definite whenever β belongs to V_0 . Thus, E is strictly concave on V_0 and $\sup_{\beta \in V_0} E(\beta) = E(0) = 0$, $\nabla E(\beta) = 0$ iff $\beta = 0$. Note that at the bottom of the spectrum (i.e., j = 1), we could take V_0 as the whole Euclidean space \mathbb{R}^d .

By the Morse lemma and the fact that 0 is a nondegenerate critical point of E, there is a smooth change of coordinates $\Phi : U_0 \to \mathbb{R}^d$ so that $0 \in U_0 \subset V_0$, U_0 is connected, $\Phi(0) = 0$ and $E(\Phi^{-1}(a)) = -|a|^2, \forall a \in \Phi(U_0)$. Set $K_\lambda := \{\beta \in U_0 : E(\beta) \ge \lambda\}$ and $\Gamma_\lambda := \{\beta \in U_0 : E(\beta) = \lambda\}$ for each $\lambda \in \mathbb{R}$. Now, we consider λ to be in the set $\{-|a|^2 : a \in \Phi(U_0), a \ne 0\}$. Then K_λ is a strictly convex *d*-dimensional compact body in \mathbb{R}^d , and $\Gamma_\lambda = \partial K_\lambda$ is a compact hypersurface in \mathbb{R}^d . The compactness of K_λ follows from the equation $-|\Phi(\beta)|^2 = E(\beta) \ge \lambda$ which yields that $|\beta| = |\Phi^{-1}(\Phi(\beta))| \le$ $\max\{|\Phi^{-1}(a)| : a \in \Phi(U_0), |a|^2 \le -\lambda\}$. Additionally, $\lim_{\lambda \to 0^-} \max_{\beta \in K_\lambda} |\beta| = 0$.

Let \mathcal{K}_{λ} be the Gauss-Kronecker curvature of Γ_{λ} . Since the Hessian of E is negativedefinite on Γ_{λ} , \mathcal{K}_{λ} is nowhere-zero. For the value of λ described in the previous paragraph and each $s \in \mathbb{S}^{d-1}$, there is a unique vector $\beta_s \in \Gamma_{\lambda}$ such that the value of the Gauss map of the hypersurface Γ_{λ} at this point coincides with s, i.e.

$$\nabla E(\beta_s) = -|\nabla E(\beta_s)|s. \tag{3.7}$$

This is due to the fact that the Gauss map of a compact, connected oriented hypersurface in \mathbb{R}^d , whose Gauss-Kronecker curvature is nowhere zero, is a diffeomorphism onto the sphere \mathbb{S}^{d-1} (see e.g., [77, Theorem 5, p.104] or [27, Corollary 3.1]). Thus, β_s depends smoothly on s. We also see that

$$\lim_{|\lambda| \to 0} \max_{s \in \mathbb{S}^{d-1}} |\beta_s| = 0.$$

Note that β_s could be defined equivalently by using the support functional h of the strictly convex set K_{λ} . Recall that for each direction $s \in \mathbb{S}^{d-1}$,

$$h(s) = \max_{\xi \in K_{\lambda}} \langle s, \xi \rangle.$$

Then β_s is the unique point in Γ_{λ} such that $\langle s, \beta_s \rangle = h(s)$.

By letting $|\lambda|$ close enough to 0, we can make sure that $(-\lambda)^{1/2} = |a|$ for some $a \in \Phi(U_0)$. Then

$$\{k_0 + it\beta_s, (t,s) \in [0,1] \times \mathbb{S}^{d-1}\} \subset V.$$
(3.8)

We can now state the main result of the chapter.

Theorem 3.2.5. Suppose conditions A1-A5 are satisfied. For $\lambda < 0$ sufficiently close to 0 (depending on the dispersion branch λ_j and the operator L), the Green's function G_{λ} of L at λ admits the following asymptotics as $|x - y| \rightarrow \infty$:

$$G_{\lambda}(x,y) = \frac{e^{(x-y)(ik_0-\beta_s)}}{(2\pi|x-y|)^{(d-1)/2}} \frac{|\nabla E(\beta_s)|^{(d-3)/2}}{\det\left(-\mathcal{P}_s \operatorname{Hess}\left(E\right)(\beta_s)\mathcal{P}_s\right)^{1/2}} \frac{\phi_{k_0+i\beta_s}(x)\overline{\phi_{k_0-i\beta_s}(y)}}{(\phi_{k_0+i\beta_s},\phi_{k_0-i\beta_s})_{L^2(\mathbb{T}^d)}} + e^{(y-x)\cdot\beta_s}r(x,y).$$

Here s = (x - y)/|x - y|, \mathcal{P}_s is the projection from \mathbb{R}^d onto the tangent space of the unit sphere \mathbb{S}^{d-1} at the point s, and when |x - y| is large enough, the remainder term r satisfies $|r(x, y)| \leq C|x - y|^{-d/2}$ for some constant C > 0 (independent of s).

This result achieves our stated goal of showing the precise (anisotropic) rates of the

exponential decay of the Green's function and capturing the additional algebraic decay factor.

3.3 Proof of the main theorem **3.2.5** and some remarks

Theorem 3.2.5 is a direct consequence of its local (with respect to the direction of (x - y)) version:

Theorem 3.3.1. Under the hypotheses of Theorem 3.2.5 and when $\lambda \approx 0$, for each $\omega \in \mathbb{S}^{d-1}$, there are a neighborhood \mathcal{V}_{ω} in \mathbb{S}^{d-1} containing ω and a smooth function $e(s) = (e_{s,2}, \ldots, e_{s,d}) : \mathcal{V}_{\omega} \to (T_s \mathbb{S}^{d-1})^{d-1}$, which e(s) is an orthonormal basis of the tangent space $T_s \mathbb{S}^{d-1}$ for each unit vector $s \in \mathcal{V}_{\omega}$, such that following asymptotics

$$G_{\lambda}(x,y) = \frac{e^{(x-y)(ik_{0}-\beta_{s})}}{(2\pi|x-y|)^{(d-1)/2}} \left(\frac{|\nabla E(\beta_{s})|^{(d-3)}}{\det(-e_{s,p} \cdot \operatorname{Hess}(E)(\beta_{s})e_{s,q})_{2 \le p,q \le d}} \right)^{1/2} \times \frac{\phi_{k_{0}+i\beta_{s}}(x)\overline{\phi_{k_{0}-i\beta_{s}}(y)}}{(\phi_{k_{0}+i\beta_{s}},\phi_{k_{0}-i\beta_{s}})_{L^{2}(\mathbb{T}^{d})}} + e^{(y-x)\cdot\beta_{s}}r(x,y),$$
(3.10)

hold for all (x, y) such that $s = (x - y)/|x - y| \in \mathcal{V}_{\omega}$. Furthermore, there is a positive constant $C(\omega)$ depending on ω such that $|r(x, y)| \leq C(\omega)|x - y|^{-d/2}$.

Proof of Theorem 3.2.5.

Proof. Observe that for any orthonormal basis $\{e_{s,l}\}_{2 \le l \le d}$ of the tangent space $T_s \mathbb{S}^{d-1}$,

$$\det\left(-\mathcal{P}_s \operatorname{Hess}\left(E\right)(\beta_s)\mathcal{P}_s\right) = \det\left(-e_{s,p} \cdot \operatorname{Hess}\left(E\right)(\beta_s)e_{s,q}\right)_{2 \le p,q \le d}.$$

Now, using of a finite cover of the unit sphere by neighborhoods \mathcal{V}_{ω_j} in Theorem 3.3.1, one obtains Theorem 3.2.5.

Remark 3.3.2.

The asymptotics (3.9) (or (3.10)) resemble the formula (1.1) in [60, Theorem 1.1] when λ is below the bottom of the spectrum of the operator. Moreover, as in [61, Theorem 1.1], by using the Gauss-Kronecker curvature K_λ, the main result (3.9) could be restated as follows:

$$G_{\lambda}(x,y) = \frac{e^{(x-y)(ik_0-\beta_s)}}{(2\pi|x-y|)^{(d-1)/2}} \frac{1}{|\nabla E(\beta_s)|\mathcal{K}_{\lambda}(\beta_s)^{1/2}} \frac{\phi_{k_0+i\beta_s}(x)\overline{\phi_{k_0-i\beta_s}(y)}}{(\phi_{k_0+i\beta_s},\phi_{k_0-i\beta_s})_{L^2(\mathbb{T}^d)}} + e^{(y-x)\cdot\beta_s}O(|x-y|^{-d/2}).$$

Although (3.9) is an anisotropic formula, it is not hard to obtain from (3.9) an isotropic upper estimate for the Green's function G_λ based on the distance from λ to the spectrum of the operator L,⁴ e.g., there are some positive constants C₁, C₂ (depending only on L and λ_j) and C₃ (which may depend on λ) such that whenever |x - y| > C₃, the following inequality holds:

$$|G_{\lambda}(x,y)| \le C_1 |\lambda|^{(d-3)/4} \frac{e^{-C_2|\lambda|^{1/2}|x-y|}}{|x-y|^{(d-1)/2}}.$$

• If the band edge occurs at finitely many points, rather than a single k_0 , one just needs to combine the asymptotics coming from all these isolated minima.

Now we outline the proof of Theorem 3.3.1. In Section 3.5, we use the tools of Floquet-Bloch theory in Chapter 2 to reduce the problem to that of finding the asymptotics of a scalar integral. The purpose of Section 3.4 is to prepare for Section 3.5, by shifting an integral from the fundamental domain \mathcal{O} along some purely imaginary directions in \mathbb{C}^d . This reduces finding the asymptotics of the Green's function G_{λ} to an auxiliary Green's function $G_{s,\lambda}$ via the formula (3.14). Next, we single out a principal term G_0 of the Green's function $G_{s,\lambda}$ and then represent this kernel G_0 as a scalar integral in (3.19). We also prove

⁴Recall that the spectral edge is assumed to be zero.

that the error kernel $G_{s,\lambda} - G_0$ decays rapidly (see Theorem 3.5.2). Then in (3.22), our reduced Green's function G_0 can be expressed in terms of the two integrals I and J. Here the integral I is mainly responsible for the asymptotics of G_0 and the integral J decays fast enough to be included in the remainder term r(x, y) in the asymptotics (3.9). The first part of Section 3.6 is devoted to achieving the asymptotics of the main integral I (see Theorem 3.6.2) by adapting the method similar to the one used in the discrete case [81], while the second part of Section 3.6 provides an estimate of J (see Proposition 3.6.6). In order to not overload the main text with technicalities, the proofs of some auxiliary statements are postponed till Sections 3.7-3.9.

3.4 On local geometry of the resolvent set

The following proposition shows that for any $s \in \mathbb{S}^{d-1}$, $k_0 + i\beta_s$ is the only complex quasimomentum having the form of $k + it\beta_s$ where $k \in \mathcal{O}, t \in [0, 1]$ such that λ is in the spectrum of the corresponding fiber operator $L(k + it\beta_s)$. In other words, by moving from $k \in \mathcal{O}$ in the direction $i\beta_s$, the first time we **hit the Fermi surface** $F_{L,\lambda}$ (i.e., the spectrum of L(k) meets λ) is at the value of the quasimomentum $k = k_0 + i\beta_s$. This step is crucial for setting up the scalar integral in the next section, which is solely responsible for the main term asymptotics of our Green's function.

Proposition 3.4.1. If $|\lambda|$ is small enough (depending on the dispersion branch λ_j and L), then $\lambda \in \rho(L(k + it\beta_s))$ if and only if $(k, t) \neq (k_0, 1)$.

The proof of this proposition is presented in Subsection 3.9.3.

3.5 A Floquet reduction of the problem

We will use the Floquet transform (see Chapter 2) to reduce our problem to finding asymptotics of a scalar integral expression, which is close to the one arising when dealing with the Green's function of the Laplacian at a small negative level λ . As in [52], the idea

is to show that only the branch of the dispersion relation λ_j appearing in the Assumption A dominates the asymptotics.

3.5.1 The Floquet reduction

The Green's function G_{λ} of L at λ is the Schwartz kernel of the resolvent operator $R_{\lambda} = (L - \lambda)^{-1}$. Fix a $\lambda < 0$ such that the statement of Proposition 3.4.1 holds. For any $s \in \mathbb{S}^{d-1}$ and $t \in [0, 1]$, we consider the following operator with real coefficients on \mathbb{R}^d :

$$L_{t,s} := e^{t\beta_s \cdot x} L e^{-t\beta_s \cdot x}.$$
(3.11)

For simplicity, we write $L_s := L_{1,s}$ and note that $L_{0,s} = L$. Due to self-adjointness of L, the adjoint of $L_{t,s}$ is

$$L_{t,s}^* = L_{-t,s}.$$
 (3.12)

By definition, $L_{t,s}(k) = L(k + it\beta_s)$ for any k in \mathbb{C}^d and therefore, (2.6) yields

$$\sigma(L_{t,s}) = \bigcup_{k \in \mathcal{O}} \sigma(L(k + it\beta_s)) \supseteq \{\lambda_j(k + it\beta_s)\}_{k \in \mathcal{O}}.$$
(3.13)

The Schwartz kernel $G_{s,\lambda}$ of the resolvent operator $R_{s,\lambda} := (L_s - \lambda)^{-1}$ is

$$G_{s,\lambda}(x,y) = e^{\beta_s \cdot x} G_\lambda(x,y) e^{-\beta_s \cdot y} = e^{\beta_s \cdot (x-y)} G_\lambda(x,y).$$
(3.14)

Thus, instead of finding asymptotics of G_{λ} , we can focus on the asymptotics of $G_{s,\lambda}$.

By (3.13) and Proposition 3.4.1, λ is not in the spectrum of $L_{t,s}$ for any $s \in \mathbb{S}^{d-1}$ and $t \in [0, 1)$. Let us consider

$$R_{t,s,\lambda}f := (L_{t,s} - \lambda)^{-1}f, \quad f \in L^2_{comp}(\mathbb{R}^d),$$

where L^2_{comp} stands for compactly supported functions in L^2 .

Applying Lemma 2.2.4, we have

$$\widehat{R_{t,s,\lambda}f}(k) = (L_{t,s}(k) - \lambda)^{-1}\widehat{f}(k), \quad (t,k) \in [0,1) \times \mathcal{O}.$$

We consider the sesquilinear form

$$(R_{t,s,\lambda}f,\varphi) = (2\pi)^{-d} \int_{\mathcal{O}} \left((L_{t,s}(k) - \lambda)^{-1} \widehat{f}(k), \widehat{\varphi}(k) \right) \mathrm{d}k,$$

where $\varphi \in L^2_{comp}(\mathbb{R}^d)$.

In the next lemma (see Subsection 3.9.3), we show the weak convergence of $R_{t,s,\lambda}$ in L^2_{comp} as $t \nearrow 1$ and introduce the limit operator $R_{s,\lambda} = \lim_{t\to 1^-} R_{t,s,\lambda}$. The limit operator $R_{s,\lambda}$ is central in our study of the asymptotics of the Green's function.

Lemma 3.5.1. Let $d \ge 2$. Under Assumption A, the following equality holds:

$$\lim_{t \to 1^{-}} (R_{t,s,\lambda}f,\varphi) = (2\pi)^{-d} \int_{\mathcal{O}} \left(L_s(k) - \lambda)^{-1} \widehat{f}(k), \widehat{\varphi}(k) \right) \mathrm{d}k.$$
(3.15)

The integral in the right hand side of (3.15) is absolutely convergent for f, φ in $L^2_{comp}(\mathbb{R}^d)$. Thus, the Green's function $G_{s,\lambda}$ is the integral kernel of the operator $R_{s,\lambda}$ defined as follows

$$\widehat{R_{s,\lambda}f}(k) = (L_s(k) - \lambda)^{-1}\widehat{f}(k).$$
(3.16)

3.5.2 Singling out the principal term in $R_{s,\lambda}$

By (3.16), the Green's function $G_{s,\lambda}$ is the integral kernel of the operator $R_{s,\lambda}$ with the domain $L^2_{comp}(\mathbb{R}^d)$. The inversion formula (2.4) gives

$$R_{s,\lambda}f(x) = (2\pi)^{-d} \int_{\mathcal{O}} e^{ik \cdot x} (L_s(k) - \lambda)^{-1} \widehat{f}(k, x) \, \mathrm{d}k, \quad x \in \mathbb{R}^d.$$

The purpose of this part is to single out the part of the above integral that is responsible for the leading term of the Green's function asymptotics.

To find the Schwartz kernel of $R_{s,\lambda}$, it suffices to consider functions $f \in C_c^{\infty}(\mathbb{R}^d)$. Our first step is to localize the integral around the point k_0 . Let us consider a connected neighborhood V of k_0 on which there exist nonzero \mathbb{Z}^d -periodic (in x) functions $\phi_z(x), z \in$ V satisfying 1) $L(z)\phi_z = \lambda_j(z)\phi_z$ and 2) each ϕ_z spans the eigenspace corresponding to the eigenvalue $\lambda_j(z)$ of the operator L(z). According to (**P3**), $\lambda_j(V) \subseteq B(0, \epsilon_0)$ and $\partial B(0, \epsilon_0) \subseteq \rho(L(z))$ when $z \in V$. For such z, let P(z) be the Riesz projection of L(z)that projects $L^2(\mathbb{T}^d)$ onto the eigenspace spanned by ϕ_z , i.e.,

$$P(z) = -\frac{1}{2\pi i} \oint_{|\mu|=\epsilon_0} (L(z) - \alpha)^{-1} \,\mathrm{d}\mu.$$

Taking the adjoint, we get

$$P(z)^* = -\frac{1}{2\pi i} \oint_{|\mu|=\epsilon_0} (L(\overline{z}) - \mu)^{-1} d\mu = P(\overline{z}),$$

which is the Riesz projection from $L^2(\mathbb{T}^d)$ onto the eigenspace spanned by $\phi_{\overline{z}}$. Recall that due to (3.8), by choosing $|\lambda|$ small enough, there exists $r_0 > 0$ (independent of s) such that $k \pm i\beta_s \in V$ for $k \in \overline{D}(k_0, r_0) \cap \mathbb{R}^d$. We denote $P_s(k) := P(k + i\beta_s)$ for such real k. Then $P_s(k)$ is the projector onto the eigenspace spanned by $\phi(k+i\beta_s)$ and $P_s(k)^* = P(k-i\beta_s)$. Additionally, due to (**P6**),

$$P_s(k)g = \frac{(g,\phi(k-i\beta_s))_{L^2(\mathbb{T}^d)}}{(\phi(k+i\beta_s),\phi(k-i\beta_s))_{L^2(\mathbb{T}^d)}}\phi(k+i\beta_s), \quad \forall g \in L^2(\mathbb{T}^d).$$
(3.17)

Let η be a cut-off smooth function on \mathcal{O} such that $\operatorname{supp}(\eta) \subseteq D(k_0, r_0)$ and $\eta = 1$ around k_0 .

We decompose $\widehat{f} = \eta \widehat{f} + (1-\eta) \widehat{f}$. When $k \neq k_0$, the operator $L_s(k) - \lambda$ is invertible by

Proposition 3.4.1. Hence, the following function is well-defined and smooth with respect to (k, x) on $\mathbb{R}^d \times \mathbb{R}^d$:

$$\widehat{u_g}(k,x) = (L_s(k) - \lambda)^{-1} (1 - \eta(k)) \widehat{f}(k,x)$$

Using Lemma 2.2.4, smoothness of \hat{u}_g implies that u_g has rapid decay in x. Now we want to solve

$$(L_s(k) - \lambda)\widehat{u}(k) = \eta(k)\widehat{f}(k).$$
(3.18)

Let $Q_s(k) = I - P_s(k)$ and we denote the ranges of projectors $P_s(k)$, $Q_s(k)$ by $R(P_s(k))$, $R(Q_s(k))$ respectively. We are interested in decomposing the solution \hat{u} into a sum of the form $\hat{u}_1 + \hat{u}_2$ where $\hat{u}_1 = P_s(k)\hat{u}_1$ and $\hat{u}_2 = Q_s(k)\hat{u}_2$. Let $\hat{f}_1 = P_s(k)\eta(k)\hat{f}$ and $\hat{f}_2 = Q_s(k)\eta(k)\hat{f}$. Observe that since the Riesz projection $P_s(k)$ commutes with the operator $L_s(k)$ and $R(P_s(k))$ is invariant under the action of $L_s(k)$, we have $Q_s(k)L_s(k)P_s(k) = P_s(k)L_s(k)Q_s(k) = 0$ and $Q_s(k)L_s(k)Q_s(k) = L_s(k)Q_s(k)$. Thus, the problem of solving (3.18) can be reduced to the following block-matrix structure form

$$\left(\begin{array}{c|c} (L_s(k) - \lambda)P_s(k) & 0\\ \hline 0 & (L_s(k) - \lambda)Q_s(k) \end{array}\right) \left(\begin{array}{c} \widehat{u_1} \\ \widehat{u_2} \end{array}\right) = \left(\begin{array}{c} \widehat{f_1} \\ \widehat{f_2} \end{array}\right).$$

When k is close to k_0 ,

$$B(0,\epsilon_0) \cap \sigma(L_s(k)|_{R(Q_s(k))}) = B(0,\epsilon_0) \cap \sigma(L(k+i\beta_s)) \setminus \{\lambda_j(k+i\beta_s)\} = \emptyset.$$

Since $\lambda = \lambda_j(k_0 + i\beta_s) \in B(0, \epsilon_0)$, λ must belong to $\rho(L_s(k)|_{R(Q_s(k))})$. Hence, the operator function $\widehat{u}_2(k) = (L_s(k) - \lambda)^{-1}Q_s(k)\widehat{f}_2(k)$ is well-defined and smooth in k and hence by Lemma 2.2.4 again, u_2 has rapid decay when $|x| \to \infty$. More precisely, we have

the following claim:

Theorem 3.5.2. For each $s \in \mathbb{S}^{d-1}$, let $K_s(x, y)$ be the Schwartz kernel of the operator T_s acting on $L^2(\mathbb{R}^d)$ as follows:

$$T_s = \mathcal{F}^{-1}\left(\int_{\mathcal{O}}^{\oplus} T_s(k) \,\mathrm{d}k\right) \mathcal{F},$$

where \mathcal{F} is the Floquet transform (see Definition 2.2.3) and

$$T_s(k) = (1 - \eta(k))(L_s(k) - \lambda)^{-1} + \eta(k)((L_s(k) - \lambda)|_{R(Q_s(k))})^{-1}Q_s(k)$$

Then the kernel $K_s(x, y)$ is continuous away from the diagonal and furthermore, as $|x - y| \to \infty$, we have

$$\sup_{s \in \mathbb{S}^{d-1}} |K_s(x, y)| = O(|x - y|^{-N}), \quad \forall N > 0.$$

The proof of this claim shall be provided in Section 3.7.

The u_1 term contributes the leading asymptotics for the Schwartz kernel $G_{s,\lambda}$. Therefore, we only need to solve the equation $(L_s(k) - \lambda)P_s(k)\hat{u_1} = \hat{f_1}$ on the one-dimensional range of $P_s(k)$.

Applying (3.17), we can rewrite

$$\widehat{f}_1(k) = \frac{\eta(k)(f, \phi(k-i\beta_s))_{L^2(\mathbb{T}^d)}}{(\phi(k+i\beta_s), \phi(k-i\beta_s))_{L^2(\mathbb{T}^d)}} \phi(k+i\beta_s),$$

so that equation becomes

$$(L_s(k) - \lambda) \frac{(\widehat{u}_1, \phi(k - i\beta_s))_{L^2(\mathbb{T}^d)} \phi(k + i\beta_s)}{(\phi(k + i\beta_s), \phi(k - i\beta_s))_{L^2(\mathbb{T}^d)}} = \frac{\eta(k)(\widehat{f}, \phi(k - i\beta_s))_{L^2(\mathbb{T}^d)} \phi(k + i\beta_s)}{(\phi(k + i\beta_s), \phi(k - i\beta_s))_{L^2(\mathbb{T}^d)}}$$

$$\frac{(\lambda_j(k+i\beta_s)-\lambda)(\widehat{u_1},\phi(k-i\beta_s))_{L^2(\mathbb{T}^d)}\phi(k+i\beta_s)}{(\phi(k+i\beta_s),\phi(k-i\beta_s))_{L^2(\mathbb{T}^d)}} = \frac{\eta(k)(\widehat{f},\phi(k-i\beta_s))_{L^2(\mathbb{T}^d)}\phi(k+i\beta_s)}{(\phi(k+i\beta_s),\phi(k-i\beta_s))_{L^2(\mathbb{T}^d)}}$$

In addition to the equation $\widehat{u_1} = P_s(k)\widehat{u_1}, \widehat{u_1}$ must also satisfy

$$(\lambda_j(k+i\beta_s)-\lambda)(\widehat{u_1},\phi(k-i\beta_s))_{L^2(\mathbb{T}^d)} = \eta(k)(\widehat{f},\phi(k-i\beta_s))_{L^2(\mathbb{T}^d)}$$

Thus, we define

$$\widehat{u_1}(k,\cdot) := \frac{\eta(k)\phi(k+i\beta_s,\cdot)(f,\phi(k-i\beta_s))_{L^2(\mathbb{T}^d)}}{(\phi(k+i\beta_s),\phi(k-i\beta_s))_{L^2(\mathbb{T}^d)}(\lambda_j(k+i\beta_s)-\lambda)}$$

By the inverse Floquet transform (2.4),

$$u_1(x) = (2\pi)^{-d} \int_{\mathcal{O}} e^{ik \cdot x} \frac{\eta(k)\phi(k+i\beta_s, x)(\hat{f}, \phi(k-i\beta_s))_{L^2(\mathbb{T}^d)}}{(\phi(k+i\beta_s), \phi(k-i\beta_s))_{L^2(\mathbb{T}^d)}(\lambda_j(k+i\beta_s)-\lambda)} \,\mathrm{d}k,$$

for any $x \in \mathbb{R}^d$.

3.5.3 A reduced Green's function.

We are now ready for setting up **a reduced Green's function** G_0 , whose asymptotic behavior reflects exactly the leading term of the asymptotics of the Green's function $G_{s,\lambda}$. We introduce $G_0(x, y)$ (roughly speaking) as the Schwartz kernel of the restriction of the operator $R_{s,\lambda}$ onto the one-dimensional range of P_s (which is the direct integral of idempotents $P_s(k)$) as follows:

$$u_1(x) = \int_{\mathbb{R}^d} G_0(x, y) f(y) dy, \quad x \in \mathbb{R}^d,$$

where f is in $L^2_{comp}(\mathbb{R}^d)$.

So,

We recall from (3.4) that $F(k+i\beta_s)$ is the inner product $(\phi(k+i\beta_s), \phi(k-i\beta_s))_{L^2(\mathbb{T}^d)}$. As in [52], we notice that

$$\begin{aligned} u_1(x) &= (2\pi)^{-d} \int_{\mathcal{O}} \int_{\mathbb{T}^d} e^{ik \cdot x} \eta(k) \widehat{f}(k, y) \frac{\overline{\phi(k - i\beta_s, y)} \phi(k + i\beta_s, x)}{F(k + i\beta_s)(\lambda_j(k + i\beta_s) - \lambda)} \, \mathrm{d}y \, \mathrm{d}k \\ &= (2\pi)^{-d} \int_{\mathcal{O}} \eta(k) \int_{[0,1]^d} \sum_{\gamma \in \mathbb{Z}^d} f(y - \gamma) e^{ik \cdot (x + \gamma - y)} \frac{\overline{\phi(k - i\beta_s, y)} \phi(k + i\beta_s, x)}{F(k + i\beta_s)(\lambda_j(k + i\beta_s) - \lambda)} \, \mathrm{d}y \, \mathrm{d}k \\ &= (2\pi)^{-d} \int_{\mathcal{O}} \eta(k) \sum_{\gamma \in \mathbb{Z}^d} \int_{[0,1]^d - \gamma} f(y) e^{ik \cdot (x - y)} \frac{\overline{\phi(k - i\beta_s, y + \gamma)} \phi(k + i\beta_s, x)}{F(k + i\beta_s)(\lambda_j(k + i\beta_s) - \lambda)} \, \mathrm{d}y \, \mathrm{d}k \end{aligned}$$

$$= (2\pi)^{-d} \int_{\mathcal{O}} \eta(k) \sum_{\gamma \in \mathbb{Z}^d} \int_{[0,1]^d - \gamma} f(y) e^{ik \cdot (x-y)} \frac{\overline{\phi(k - i\beta_s, y)} \phi(k + i\beta_s, x)}{F(k + i\beta_s)(\lambda_j(k + i\beta_s) - \lambda)} \, \mathrm{d}y \, \mathrm{d}k$$
$$= (2\pi)^{-d} \int_{\mathbb{R}^d} f(y) \left(\int_{\mathcal{O}} \eta(k) e^{ik \cdot (x-y)} \frac{\overline{\phi(k - i\beta_s, y)} \phi(k + i\beta_s, x)}{F(k + i\beta_s)(\lambda_j(k + i\beta_s) - \lambda)} \, \mathrm{d}k \right) \mathrm{d}y.$$

Therefore, our reduced Green's function is

$$G_0(x,y) = (2\pi)^{-d} \int_{\mathcal{O}} \eta(k) e^{ik \cdot (x-y)} \frac{\phi(k+i\beta_s, x)\phi(k-i\beta_s, y)}{F(k+i\beta_s)(\lambda_j(k+i\beta_s) - \lambda)} \,\mathrm{d}k.$$
(3.19)

3.6 Asymptotics of the Green's function

Let (e_1, \ldots, e_d) be the standard orthonormal basis in \mathbb{R}^d . Fixing $\omega \in \mathbb{S}^{d-1}$, we would like to show that the asymptotics (3.10) will hold for all (x, y) such that x - y belongs to a conic neighborhood containing ω . Without loss of generality, suppose that $\omega \neq e_1$.

Now let \mathcal{R}_s be the rotation in \mathbb{R}^d such that $\mathcal{R}_s(s) = e_1$ and \mathcal{R}_s leaves the orthogonal complement of the subspace spanned by $\{s, e_1\}$ invariant. We define $e_{s,j} := \mathcal{R}_s^{-1}(e_j)$, for all $j = 2, \ldots, d$. Then, $\langle s, e_{s,p} \rangle = \langle e_1, e_p \rangle = 0$ and $\langle e_{s,p}, e_{s,q} \rangle = \langle e_p, e_q \rangle = \delta_{p,q}$ for p, q > 1. In other words,

 $\{s, e_{s,2}, \ldots, e_{s,d}\}$ is an orthonormal basis of \mathbb{R}^d .

Then around ω , we pick a compact coordinate patch \mathcal{V}_{ω} , so that the $\mathbb{R}^{d(d-1)}$ -valued

function $e(s) = (e_{s,l})_{2 \le l \le d}$ is smooth in a neighborhood of \mathcal{V}_{ω} .

We use the same notation for \mathcal{R}_s and its \mathbb{C} -linear extension to \mathbb{C}^d .

3.6.1 The asymptotics of the leading term of the Green's function

We introduce the function $\rho(k, x, y)$ on $D(k_0, r_0) \times \mathbb{R}^d \times \mathbb{R}^d$ as follows:

$$\rho(k, x, y) = \frac{\phi(k + i\beta_s, x)\overline{\phi(k - i\beta_s, y)}}{F(k + i\beta_s)}$$

where F is defined in (3.4) and $D(k_0, r_0)$ is described in Subsection 3.5.2.

Due to Proposition 3.9.6, the function ρ is in $C^{\infty}(\overline{D(k_0, r)} \times \mathbb{R}^d \times \mathbb{R}^d)$. For each (x, y), the Taylor expansion around k_0 of $\rho(k)$ gives

$$\rho(k, x, y) = \rho(k_0, x, y) + \rho'(k, x, y)(k - k_0), \qquad (3.20)$$

where $\rho' \in C^{\infty}(\overline{D(k_0, r_0)} \times \mathbb{R}^d \times \mathbb{R}^d, \mathbb{C}^d)$. Note that for $z \in V$, $\phi(z, x)$ is \mathbb{Z}^d -periodic in x and thus, ρ and ρ' are $\mathbb{Z}^d \times \mathbb{Z}^d$ -periodic in (x, y). Since our integrals are taken with respect to k, it is safe to write $\rho(k_0)$ instead of $\rho(k_0, x, y)$. We often omit the variables x, yin ρ if no confusion can arise.

Let $\mu(k) := \eta(k + k_0)$ be a cut-off function supported near 0, where η is introduced in Subsection 3.5.2. We define

$$I := (2\pi)^{-d} \int_{\mathcal{O}} e^{i(k-k_0) \cdot (x-y)} \frac{\mu(k-k_0)}{\lambda_j (k+i\beta_s) - \lambda} \, \mathrm{d}k,$$

$$J := (2\pi)^{-d} \int_{\mathcal{O}} e^{i(k-k_0) \cdot (x-y)} \frac{\mu(k-k_0)(k-k_0)\rho'(k,x,y)}{\lambda_j (k+i\beta_s) - \lambda} \, \mathrm{d}k.$$
(3.21)

Hence, we can represent the reduced Green's function as

$$G_0(x,y) = e^{ik_0 \cdot (x-y)} (\rho(k_0)I + J).$$
(3.22)

The rest of this subsection is devoted to computing the asymptotics of the main integral I, which gives the leading term in asymptotic expansion of the reduced Green's function $G_0(x, y)$ as $|x - y| \to \infty$.

By making the change of variables $\xi = (\xi_1, \xi') = \mathcal{R}_s(k - k_0)$, we have

$$I = (2\pi)^{-d} \int_{\mathbb{R}^d} e^{i|x-y|\xi_1} \frac{\mu(\xi_1,\xi')}{(\lambda_j \circ \mathcal{R}_s^{-1})(\xi + \mathcal{R}_s(k_0 + i\beta_s)) - \lambda} \,\mathrm{d}\xi.$$
 (3.23)

We introduce the following function defined on some neighborhood of 0 in \mathbb{C}^d :

$$W_s(z) := (\lambda_j \circ \mathcal{R}_s^{-1})(-iz + \mathcal{R}_s(k_0 + i\beta_s)) - \lambda$$

It is holomorphic near 0 (on $i\mathcal{R}_s(V)$) and $W_s(0) = 0$. Then $W_s(iz)$ is the analytic continuation to the domain $\mathcal{R}_s(V)$ of the denominator of the integrand in (3.23). For a complex vector $z = (z_1, \ldots, z_d) \in \mathbb{C}^d$, we write $z = (z_1, z')$, where $z' = (z_2, \ldots, z_d)$.

The following proposition provides a factorization of W_s that is crucial for computing the asymptotics of the integral I.

Proposition 3.6.1. There exist r > 0 and $\epsilon > 0$ (independent of $s \in V_{\omega}$), such that W_s has the decomposition

$$W_s(z) = (z_1 - A_s(z'))B_s(z), \quad \forall z = (z_1, z') \in B(0, r) \times D'(0, \epsilon).^5$$
(3.24)

Here the functions A_s , B_s are holomorphic in $D'(0, \epsilon)$ and $B(0, r) \times D'(0, \epsilon)$ respectively such that $A_s(0) = 0$ and B_s is non-vanishing on $B(0, r) \times D'(0, \epsilon)$. Also, these functions and their derivatives depend continuously on s. Moreover for $z' \in D'(0, \epsilon)$,

$$A_s(z') = \frac{1}{2}z' \cdot Q_s z' + O(|z'|^3), \qquad (3.25)$$

⁵See Notation 3.2.2 in Section 3.2 for the definitions of B(0, r) and $D'(0, \epsilon)$.

where $O(|z'|^3)$ is uniform in s when $z' \to 0$ and Q_s is the positive definite $(d-1) \times (d-1)$ matrix

$$Q_s = -\frac{1}{|\nabla E(\beta_s)|} \left(e_{s,p} \cdot \operatorname{Hess}\left(E\right)(\beta_s) e_{s,q} \right)_{2 \le p,q \le d}.$$
(3.26)

Proof. By Cauchy-Riemann equations for W_s and (3.7),

$$\frac{\partial W_s}{\partial z_1}(0) = \frac{\partial W_s}{\partial \xi_1}(0) = -i\nabla\lambda_j(k_0 + i\beta_s) \cdot \mathcal{R}_s^{-1} e_1 = -\nabla E(\beta_s) \cdot s = |\nabla E(\beta_s)| > 0.$$
(3.27)

Thus 0 is a simple zero of W_s . Due to smoothness in s of W_s and β_s , we have

$$c := \min_{s \in \mathcal{V}_{\omega}} \frac{\partial W_s}{\partial z_1}(0) \ge \min_{s \in \mathbb{S}^{d-1}} |\nabla E(\beta_s)| > 0.$$
(3.28)

Applying the Weierstrass preparation theorem (see Theorem 3.9.2), we obtain the decomposition (3.24) on a neighborhood of 0.

To show that this neighborhood can be chosen such that it does not depend on s, we have to chase down how the neighborhood is constructed in the proof of [38, Theorem 7.5.1] (only the first three lines of the proof there matter) and then show that all steps in this construction can be done independently of s.

In the first step of the construction, we need r > 0 such that $W_s(z_1, 0') \neq 0$ when $0 < |z_1| < 2r$. The mapping $(s, z) \mapsto \frac{\partial W_s}{\partial z_1}(z) = -i\nabla\lambda_j(-i\mathcal{R}_s^{-1}z + k_0 + i\beta_s) \cdot s$ is jointly continuous on $\mathcal{V}_\omega \times \mathcal{R}_s(V)$ and the value of this mapping at z = 0 is greater or equal than c due to (3.27) and (3.28). Therefore, $\left|\frac{\partial W_s}{\partial z_1}(z)\right| > c/2$ in some open neighborhood $X_s \times Y_s$ of (s, 0) in $\mathcal{V}_\omega \times \mathbb{C}^d$. By compactness, $\mathcal{V}_\omega \subseteq \bigcup_{k=1}^N X_{s_k}$ for a finite collection of points s_1, \ldots, s_N on the sphere. Let Y be the intersection of all Y_{s_k} and let r > 0 be such that $D(0, 2r) \subseteq Y$. Note that r is independent of s. We claim r has the desired property. Observe that for |z| < 2r, we have $\left|\frac{\partial W_s}{\partial z_1}(z)\right| > \frac{c}{2}$ for any s in \mathcal{V}_ω . For a proof by

contradiction, suppose that there is some z_1 such that $0 < |z_1| < 2r$ and $W_s(z_1, 0') = 0 = W_s(0, 0')$ for some s. Applying Rolle's theorem to the function $t \in [0, 1] \mapsto W_s(tz_1, 0')$ yields $\frac{\partial W_s}{\partial z_1}(tz_1, 0') = 0$ for some $t \in (0, 1)$. Consequently, $(tz_1, 0') \notin D(0, 2r)$ while $|tz_1| < |z_1| < 2r$ (contradiction!).

For the second step of the construction, we want some $\delta > 0$ (independent of s) such that $W_s(z) \neq 0$ when $|z_1| = r, |z'| < \delta$. This can be done in a similar manner. Let $S(0,r) \subset \mathbb{C}$ be the circle with radius r. Now we consider the smooth mapping W: $(s, z_1, z') \mapsto W_s(z_1, z')$ where $z_1 \in S(0, r)$. Its value at each point $(s, z_1, 0')$ is equal to $W_s(z_1, 0')$, which is non-zero due to the choice of r in the first step of the construction. Thus, it is also non-zero in some open neighborhood $\tilde{X}_{s,z_1} \times \tilde{Y}_{s,z_1} \times \tilde{Z}_{s,z_1}$ of $(s, z_1, 0')$ in $\mathcal{V}_\omega \times S(0, r) \times \mathbb{C}^{d-1}$. We select points $s_1, \ldots, s_M \in \mathcal{V}_\omega$ and $\gamma_1, \ldots, \gamma_M \in S(0, r)$ so that the union of all $\tilde{X}_{s_k,\gamma_k} \times \tilde{Y}_{s_k,\gamma_k}, 1 \leq k \leq M$ covers the compact set $\mathcal{V}_\omega \times S(0, r)$. Next we choose $\delta > 0$ so that $D'(0, \delta)$ is contained in the intersection of these \tilde{Z}_{s_k,z_k} . Note that δ is independent of s and also z_1 . Of course $W_s(z_1, z') \neq 0$ for all s and $z \in \{|z_1| = r, |z'| < \delta\}$.

Also, from the proof of [38, Theorem 7.5.1], the function A_s is defined via the following formula

$$z_1 - A_s(z') = \exp\left(\frac{1}{2\pi i} \int_{|\omega|=r} \left(\frac{\partial W_s(\omega, z')}{\partial \omega} / W_s(\omega, z')\right) \log(z_1 - \omega) \,\mathrm{d}\omega\right).$$
(3.29)

The mappings $(s, z') \mapsto A_s(z')$ and $(s, z) \mapsto B_s(z)$ are jointly continuous due to (3.24) and (3.29). There exists $0 < \epsilon \leq \delta$ such that $\max_{s \in \mathcal{V}_{\omega}} |A_s(z')| < r$ whenever $|z'| < \epsilon$. We have the identity (3.24) on $B(0, r) \times D'(0, \epsilon)$. Now, we show that this is indeed the neighborhood that has the desired properties. Since $|z'| < \epsilon$ implies that the points $z = (A_s(z'), z') \in B(0, r) \times D'(0, \epsilon)$, we can evaluate (3.24) at these points to obtain

$$W_s(A_s(z'), z') = 0, \quad z' \in D'(0, \epsilon).$$
 (3.30)

By differentiating (3.30), we have

$$\frac{\partial W_s}{\partial z_p}(A_s(z'), z') + \frac{\partial W_s}{\partial z_1}(A_s(z'), z')\frac{\partial A_s}{\partial z_p}(z') = 0, \quad \text{for } p = 2, \dots, d.$$
(3.31)

Observe that from the above construction, the term $\frac{\partial W_s}{\partial z_1}(A_s(z'), z')$ is always non-zero whenever $|z'| < \epsilon$. Consequently, all first-order derivatives of A_s are jointly continuous in (s, z). Similarly, we deduce by induction on $n \in \mathbb{N}^d$ that all derivatives of the function A_s depend continuously on s since after taking differentiation of the equation (3.30) up to order n, the n-order derivative term always goes with the nonzero term $\frac{\partial W_s}{\partial z_1}(A_s(z'), z')$ and the remaining terms in the sum are just lower order derivatives. Hence the same conclusion holds for all derivatives of B_s by differentiating (3.24).

In particular, set z' = 0 in (3.31) to obtain

$$\frac{\partial W_s}{\partial z_p}(0) + \frac{\partial W_s}{\partial z_1}(0)\frac{\partial A_s}{\partial z_p}(0) = 0, \quad \text{for } p = 2, \dots, d.$$
(3.32)

Note that for p > 1,

$$\frac{\partial W_s}{\partial z_p}(z) = -i\nabla\lambda_j(-i\mathcal{R}_s^{-1}z + k_0 + i\beta_s) \cdot \mathcal{R}_s^{-1}e_p.$$
(3.33)

By substituting z = 0,

$$\frac{\partial W_s}{\partial z_p}(0) = -i\nabla\lambda_j(k_0 + i\beta_s) \cdot \mathcal{R}_s^{-1}e_p$$

$$= -\nabla E(\beta_s) \cdot e_{s,p} = -|\nabla E(\beta_s)|s \cdot e_{s,p} = 0.$$
(3.34)

(3.27), (3.32) and (3.34) imply

$$\frac{\partial A_s}{\partial z_p}(0) = 0, \quad \text{for } p = 2, \dots, d.$$
(3.35)

Taking a partial derivative with respect to z_q (q > 1) of (3.33) at z = 0, we see that

$$\frac{\partial^2 W_s}{\partial z_p \partial z_q}(0) = \sum_{m=1}^d -\nabla \left(\frac{\partial \lambda_j}{\partial z_m} (k_0 + i\beta_s)\right) \cdot \mathcal{R}_s^{-1} e_p (\mathcal{R}_s^{-1} e_q)_m$$
$$= -\sum_{m,n=1}^d \frac{\partial^2 \lambda_j}{\partial z_m \partial z_n} (k_0 + i\beta_s) (e_{s,p})_m (e_{s,q})_n$$
$$= e_{s,q} \cdot \text{Hess} (E) (\beta_s) e_{s,p}.$$
(3.36)

A second differentiation of (3.31) at $z = (A_s(z'), z')$ gives

$$0 = \left(\frac{\partial^2 W_s}{\partial z_p \partial z_q}(z) + \frac{\partial W_s}{\partial z_1}(z) \frac{\partial^2 A_s}{\partial z_p \partial z_q}(z')\right) + \left(\frac{\partial^2 W_s}{\partial z_1 \partial z_q}(z) \frac{\partial A_s}{\partial z_p}(z') + \frac{\partial^2 W_s}{\partial z_p \partial z_1}(z) \frac{\partial A_s}{\partial z_q}(z') + \frac{\partial^2 W_s}{\partial z_1^2}(z) \frac{\partial A_s}{\partial z_p}(z') \frac{\partial A_s}{\partial z_q}(z')\right).$$
(3.37)

At z = 0, the sum in the second bracket of (3.37) is zero due to (3.35). Thus,

$$\frac{\partial^2 A_s}{\partial z_p \partial z_q}(0) = -\left(\frac{\partial W_s}{\partial z_1}(0)\right)^{-1} \frac{\partial^2 W_s}{\partial z_p \partial z_q}(0) \quad (2 \le p, q \le d).$$
(3.38)

Together with (3.27) and (3.36), the above equality becomes

$$\frac{\partial^2 A_s}{\partial z_p \partial z_q}(0) = -\frac{1}{|\nabla E(\beta_s)|} \Big(e_{s,p} \cdot \operatorname{Hess}\left(E\right)(\beta_s) e_{s,q} \Big)_{2 \le p,q \le d} = Q_s.$$
(3.39)

Consequently, by (3.35) and (3.39), the Taylor expansion of A_s at 0 implies (3.25).

Finally, the remainder term $O(|z'|^3)$ in the Taylor expansion (3.25), denoted by $R_{s,3}(z')$,

can be estimated as follows:

$$\begin{aligned} |R_{s,3}(z')| &\lesssim |z'|^3 \max_{|\alpha|=3, 0 \leq t \leq 1} \left| \frac{\partial^{\alpha} A_s}{\partial z^{\alpha}}(tz') \right| \\ &\lesssim |z'|^3 \max_{|\alpha|=3, |y| \leq |z'|} \left| \frac{\partial^{\alpha} A_s}{\partial z^{\alpha}}(y) \right|. \end{aligned}$$

Due to the continuity of third-order derivatives of A_s on $\mathcal{V}_{\omega} \times D'(0, \epsilon)$,

$$\lim_{|z'| \to 0} \max_{s \in \mathcal{V}_{\omega}} \frac{|R_{s,3}(z')|}{|z'|^3} < \infty.$$
(3.40)

This proves the last claim of this proposition.

We can now let the size of the support of $\eta \ (\subseteq O)$ be small enough such that the decomposition (3.24) in Proposition 3.6.1 holds on the support of μ , i.e., $\operatorname{supp}(\mu) \subseteq B(0,r) \times D'(0,\epsilon)$. Therefore, from (3.23), we can represent the integral I as follows:

$$I = (2\pi)^{-d} \int_{\mathbb{R}^d} e^{i|x-y|\xi_1} \frac{\mu(\xi_1,\xi')}{W_s(i\xi)} \,\mathrm{d}\xi_1 \,\mathrm{d}\xi' = (2\pi)^{-d} \int_{|\xi'| < \epsilon} \int_{\mathbb{R}} \frac{e^{i|x-y|\xi_1} \tilde{\mu}_s(\xi_1,\xi')}{i\xi_1 - A_s(i\xi')} \,\mathrm{d}\xi_1 \,\mathrm{d}\xi',$$
(3.41)

where $\tilde{\mu}_s(\xi) = \mu(\xi)(B_s(i\xi))^{-1}$. We extend $\tilde{\mu}_s$ to a smooth compactly supported function on \mathbb{R}^d by setting $\tilde{\mu}_s = 0$ outside its support. Since all derivatives of $\tilde{\mu}_s$ depend continuously on *s*, they are uniformly bounded in *s*. Let $\nu_s(t, \xi')$ be the Fourier transform in the variable ξ_1 of the function $\tilde{\mu}_s(-\xi_1, \xi')$ for each $\xi' \in \mathbb{R}^{d-1}$, i.e.,

$$\nu_s(t,\xi') = \int_{-\infty}^{+\infty} e^{it\xi_1} \tilde{\mu}_s(\xi_1,\xi') \,\mathrm{d}\xi_1.$$

By applying the Lebesgue Dominated Convergence Theorem, the function ν_s is continuous in (s, t, ξ') on $\mathcal{V}_{\omega} \times \mathbb{R}^d$. For such ξ' , $\nu_s(\cdot, \xi')$ is a Schwartz function in t on \mathbb{R} . Due to Lemma 3.9.1, for any N > 0, $\nu_s(t, \xi') = O(|t|^{-N})$ uniformly in s and ξ' as $t \to \infty$. We also choose

 ϵ small enough such that whenever $|\xi'| < \epsilon$, the absolute value of the remainder term $O(|\xi'|^3)$ in (3.25) is bounded from above by $\frac{1}{4}\xi' \cdot Q_s\xi'$. Note that ϵ is still independent of s, because the term $O(|\xi'|^3)/|\xi'|^3$ is uniformly bounded by the quantity in (3.40). Meanwhile, each positive definite matrix Q_s dominates the positive matrix $\gamma_{\omega}I_{(d-1)\times(d-1)}$, where $\gamma_{\omega} > 0$ is the smallest among all the eigenvalues of all matrices Q_s ($s \in \mathcal{V}_{\omega}$). This implies that if $0 < |\xi'| < \epsilon$, then

$$\Re(i\xi_1 - A_s(i\xi')) = -\Re(A_s(i\xi')) = -\Re(-\frac{1}{2}\xi' \cdot Q_s\xi' + O(|\xi'|^3))$$
$$= \frac{1}{2}\xi' \cdot Q_s\xi' - \Re(O(|\xi'|^3)) > \frac{1}{4}\gamma_{\omega}|\xi'|^2 > 0.$$

We thus can obtain the following integral representation for a factor in the integrand of I (see (3.41)):

$$\frac{1}{i\xi_1 - A_s(i\xi')} = \int_{-\infty}^0 e^{(i\xi_1 - A_s(i\xi'))w} \,\mathrm{d}w, \quad (\xi_1, \xi') \in \mathbb{R} \times (D'(0, \epsilon) \setminus \{0\}).$$
(3.42)

Therefore,

$$I = \frac{1}{(2\pi)^d} \int_{|\xi'| < \epsilon} \int_{-\infty}^0 e^{-wA_s(i\xi')} \int_{-r}^r e^{i(w+|x-y|)\xi_1} \tilde{\mu}_s(\xi_1,\xi') \,\mathrm{d}\xi_1 \,\mathrm{d}w \,\mathrm{d}\xi'$$

$$= \frac{1}{(2\pi)^d} \int_{|\xi'| < \epsilon} \int_{-\infty}^{|x-y|} e^{(-t+|x-y|)A_s(i\xi')} \nu_s(t,\xi') \,\mathrm{d}t \,\mathrm{d}\xi'.$$
(3.43)

Now our remaining task is to prove the following asymptotics of the integral *I*:

Theorem 3.6.2. We have

$$I = \frac{|\nabla E(\beta_s)|^{(d-3)/2} |x-y|^{-(d-1)/2}}{(2\pi)^{(d-1)/2} \det\left(-e_{s,p} \cdot \operatorname{Hess}\left(E\right)(\beta_s)e_{s,q}\right)_{2 \le p,q \le d}^{1/2}} + O(|x-y|^{-d/2}).$$
(3.44)

Here the term $O(|x-y|^{-d/2})$ is uniform in $s \in \mathcal{V}_{\omega}$ as $|x-y| \to \infty$.

The next lemma reduces the leading term of the right hand side of (3.44) to a scalar integral as follows

Lemma 3.6.3.

$$\int_{\mathbb{R}^{d-1}} \int_{\mathbb{R}} \exp\left(-\frac{1}{2}x' \cdot Q_s x'\right) \nu_s(t,0) \,\mathrm{d}t \,\mathrm{d}x' = \frac{(2\pi)^{(d+1)/2} |\nabla E(\beta_s)|^{(d-3)/2}}{\det\left(-e_{s,p} \cdot \operatorname{Hess}\left(E\right)(\beta_s)e_{s,q}\right)_{2 \le p,q \le d}^{1/2}}.$$

Proof. By applying the Fourier inversion formula to ν_s , we get

$$\frac{1}{2\pi} \int_{\mathbb{R}} \nu_s(t,0) \, \mathrm{d}t = \tilde{\mu}_s(0) = (B_s(0))^{-1} = (\frac{\partial W_s}{\partial z_1}(0))^{-1} = \frac{1}{|\nabla E(\beta_s)|}.$$
 (3.45)

Here (3.27) is used in the last equality. Thus,

$$\begin{split} &\int_{\mathbb{R}^{d-1}} \int_{\mathbb{R}} \exp\left(-\frac{1}{2}x' \cdot Q_s x'\right) \nu_s(t,0) \, \mathrm{d}t \, \mathrm{d}x' \\ &= \frac{1}{(\det Q_s)^{1/2}} \int_{\mathbb{R}} \int_{\mathbb{R}^{d-1}} \exp\left(-\frac{1}{2}|u'|^2\right) \nu_s(t,0) \, \mathrm{d}u' \, \mathrm{d}t = \frac{(2\pi)^{(d-1)/2}}{(\det Q_s)^{1/2}} \int_{\mathbb{R}} \nu_s(t,0) \, \mathrm{d}t \\ &= \frac{(2\pi)^{(d+1)/2}}{(\det Q_s)^{1/2} |\nabla E(\beta_s)|} = \frac{(2\pi)^{(d+1)/2} |\nabla E(\beta_s)|^{(d-3)/2}}{\det (-e_{s,p} \cdot \operatorname{Hess}\left(E\right)(\beta_s) e_{s,q}\right)_{2 \le p,q \le d}^{1/2}}. \end{split}$$

Note that we use the change of variables $u' := Q_s^{1/2} x'$, (3.45), (3.26) in the first, the third and the last equality respectively.

For clarity, we introduce the notation $x_0 := |x - y|$. The purpose of the following two lemmas is to truncate some unnecessary (rapidly decreasing) parts of the main integrals we are interested in.

Lemma 3.6.4. *i)* For any $\alpha \in (0, 1)$ and n > 0, one has

$$\sup_{s \in \mathcal{V}_{\omega}} \int_{|\xi'| < \epsilon} \int_{(-\infty, -x_0^{\alpha}) \cup (x_0^{\alpha}, x_0)} \exp\left((x_0 - t)A_s(i\xi')\right) \nu_s(t, \xi') \,\mathrm{d}t \,\mathrm{d}\xi' = O(x_0^{-n})$$

and

$$\sup_{s \in \mathcal{V}_{\omega}} \int_{\mathbb{R}^{d-1}} \int_{|t| > x_0^{\alpha}} \exp\left(-\frac{1}{2}x' \cdot Q_s x'\right) \nu_s(t,0) \, \mathrm{d}t \, \mathrm{d}x' = O(x_0^{-n}).$$

ii) For any $\beta < 1/2$, n > 0 and each fixed $t \in [-x_0/2, x_0/2]$, one obtains

$$\sup_{s \in \mathcal{V}_{\omega}} \int_{\epsilon \sqrt{x_0 - t} > |x'| \ge x_0^{\beta}} \left| \exp\left((x_0 - t) A_s\left(\frac{ix'}{\sqrt{x_0 - t}}\right) \right) \right| \mathrm{d}x' = O(x_0^{-n})$$

and

$$\sup_{s \in \mathcal{V}_{\omega}} \int_{|x'| \ge x_0^{\beta}} \exp\left(-\frac{1}{2}x' \cdot Q_s x'\right) \mathrm{d}x' = O(x_0^{-n}).$$

Proof. i) We recall that $\sup_{s,\xi'} |\nu_s(t,\xi')| = O(|t|^{-n})$ for any n > 0. Observe that when $t \le x_0, |e^{(x_0-t)A_s(i\xi')}| \le 1$. Thus, we have

$$\sup_{s \in \mathcal{V}_{\omega}} \left| \int_{|\xi'| < \epsilon} \int_{(-\infty, -x_0^{\alpha}) \cup (x_0^{\alpha}, x_0)} \exp\left((x_0 - t) A_s(i\xi') \right) \nu_s(t, \xi') \, \mathrm{d}t \, \mathrm{d}\xi' \right|
\lesssim \int_{|\xi'| < \epsilon} \int_{(-\infty, -x_0^{\alpha}) \cup (x_0^{\alpha}, x_0)} |t|^{-n/\alpha - 1} \, \mathrm{d}t \, \mathrm{d}\xi' \lesssim \int_{|t| > x_0^{\alpha}} |t|^{-n/\alpha - 1} \, \mathrm{d}t = O(x_0^{-n}).$$
(3.46)

Since $\left|\exp\left(-\frac{1}{2}x' \cdot Q_s x'\right)\right| \le 1$, the second integral in this part also decays rapidly by the same argument.

ii) When $t < x_0$, we can substitute $\xi' = x'(x_0 - t)^{-1/2}$ into (3.25) to obtain

$$(x_0 - t) \cdot A_s \left(\frac{ix'}{\sqrt{x_0 - t}}\right) = -\frac{1}{2}x' \cdot Q_s x' + O\left(\frac{|x'|^3}{\sqrt{x_0 - t}}\right).$$
 (3.47)

Due to our choice of ϵ and the definition of γ_{ω} , we get the following estimate when

$$\begin{aligned} |x'| &< \epsilon \sqrt{x_0 - t}: \\ \sup_{s \in \mathcal{V}_{\omega}} \left| \exp\left((x_0 - t) A_s \left(\frac{ix'}{\sqrt{x_0 - t}} \right) \right) \right| &= \sup_{s \in \mathcal{V}_{\omega}} \exp\left(-\frac{1}{2} x' \cdot Q_s x' + O\left(\frac{|x'|^3}{\sqrt{x_0 - t}} \right) \right) \\ &\leq \exp\left(-\frac{1}{4} \gamma_{\omega} |x'|^2 \right). \end{aligned}$$

$$(3.48)$$

Hence, the two integrals in the statement can be estimated from above by:

$$\int_{|x'| \ge x_0^{\beta}} \exp\left(-\frac{1}{4}\gamma_{\omega}|x'|^2\right) dx' \lesssim \int_{x_0^{\beta}}^{\infty} \exp\left(-\frac{1}{4}\gamma_{\omega}r^2\right) r^{d-2} dr$$
$$\lesssim \int_{x_0^{\beta}}^{\infty} r^{-n/\beta - 1 - (d-2)} r^{d-2} dr = O(x_0^{-n}).$$

	-	-	

Lemma 3.6.5. *If* $\alpha \in (0, 1)$ *, we have*

$$\sup_{s \in \mathcal{V}_{\omega}} \int_{|t| \le x_0^{\alpha}} \int_{|x'| < \epsilon \sqrt{x_0 - t}} \left(\left(1 - \frac{t}{x_0} \right)^{-(d-1)/2} - 1 \right) \exp\left((x_0 - t) A_s \left(\frac{ix'}{\sqrt{x_0 - t}} \right) \right) \\ \times \nu_s \left(t, \frac{x'}{\sqrt{x_0 - t}} \right) \mathrm{d}x' \, \mathrm{d}t = O(x_0^{2\alpha - 1}).$$

Proof. As we argued in the proof of Lemma 3.6.4 (ii), this integral is majorized by

$$\int_{-x_0^{\alpha}}^{x_0^{\alpha}} \int_{|x'| < \epsilon\sqrt{x_0 - t}} \exp\left(-\frac{1}{4}x' \cdot Q_s x'\right) \cdot \left|\nu_s\left(t, \frac{x'}{\sqrt{x_0 - t}}\right)\right| \cdot \left|\left(1 - \frac{t}{x_0}\right)^{-(d-1)/2} - 1\right| \, \mathrm{d}x' \, \mathrm{d}t.$$

It suffices to estimate the factor $\left(1 - \frac{t}{x_0}\right)^{-(d-1)/2} - 1$ since ν_s is uniformly bounded on $\mathbb{R} \times D'(0, \epsilon)$. But this is straightforward, since

$$\int_{-x_0^{\alpha}}^{x_0^{\alpha}} \left| \left(1 - \frac{t}{x_0} \right)^{-(d-1)/2} - 1 \right| dt \le 2x_0^{\alpha} \left(\left(1 - x_0^{\alpha-1} \right)^{-(d-1)/2} - 1 \right) = O(x_0^{2\alpha-1}).$$

This finishes the proof of this lemma.

Proof of Theorem 3.6.2.

Proof. Thanks to Lemma 3.6.3, it is enough to prove the relation

$$I = (2\pi)^{-d} x_0^{-(d-1)/2} \int_{\mathbb{R}^{d-1}} \int_{\mathbb{R}} \exp\left(-\frac{1}{2}x' \cdot Q_s x'\right) \nu_s(t,0) \,\mathrm{d}t \,\mathrm{d}x' + O(x_0^{-d/2}).$$

Due to Lemma 3.6.4 (i) with $\alpha=1/4,$ we only need to show that

$$\widetilde{I} = x_0^{-(d-1)/2} \int_{\mathbb{R}^{d-1}} \int_{|t| \le x_0^{1/4}} \exp\left(-\frac{1}{2}x' \cdot Q_s x'\right) \nu_s(t,0) \,\mathrm{d}t \,\mathrm{d}x' + O(x_0^{-d/2})$$

where

$$\widetilde{I} = \int_{|\xi'| < \epsilon} \int_{|t| \le x_0^{1/4}} \exp\left((x_0 - t)A_s(i\xi')\right) \nu_s(t,\xi') \,\mathrm{d}t \,\mathrm{d}\xi'.$$

Then we substitute $x' = \xi' \sqrt{x_0 - t}$ to the integral \widetilde{I} to get

$$\widetilde{I} = x_0^{-(d-1)/2} \int_{|t| \le x_0^{1/4}} \int_{|x'| < \epsilon \sqrt{x_0 - t}} \left(1 - \frac{t}{x_0} \right)^{-(d-1)/2} \exp\left((x_0 - t) A_s \left(\frac{ix'}{\sqrt{x_0 - t}} \right) \right) \\ \times \nu_s \left(t, \frac{x'}{\sqrt{x_0 - t}} \right) \mathrm{d}x' \mathrm{d}t.$$

By Lemma 3.6.5 with $\alpha=1/4,$ we have

$$\int_{|t| \le x_0^{1/4}} \int_{|x'| < \epsilon \sqrt{x_0 - t}} \exp\left((x_0 - t)A_s\left(\frac{ix'}{\sqrt{x_0 - t}}\right)\right) \nu_s\left(t, \frac{x'}{\sqrt{x_0 - t}}\right) \mathrm{d}x' \,\mathrm{d}t$$

= $x_0^{(d-1)/2} \widetilde{I} + O(x_0^{-1/2}).$ (3.49)

Next, it is clear that for $|t| \leq x_0^{1/4}$, one has

$$\left|\nu_{s}\left(t,\frac{x'}{\sqrt{x_{0}-t}}\right)-\nu_{s}(t,0)\right| \leq \frac{|x'|}{\sqrt{x_{0}-t}} \sup_{s,\xi'} |\nabla_{\xi'}\nu_{s}(t,\xi')| \lesssim \frac{|x'|}{\sqrt{x_{0}}} \sup_{s,\xi'} |\nabla_{\xi'}\nu_{s}(t,\xi')|.$$

Also, from the definition of the function ν_s , it follows that $\sup_{s,\xi'} |\nabla_{\xi'} \nu_s(t,\xi')| = O(|t|)^{-n}$ for any n > 0. Consequently,

$$\int_{|t| \le x_0^{1/4}} \int_{|x'| < \epsilon \sqrt{x_0 - t}} \left| \exp\left((x_0 - t) A_s\left(\frac{ix'}{\sqrt{x_0 - t}} \right) \right) \left(\nu_s\left(t, \frac{x'}{\sqrt{x_0 - t}} \right) - \nu_s(t, 0) \right) \right| dx' dt \\
\lesssim \frac{1}{\sqrt{x_0}} \int_{\mathbb{R}^{d-1}} \exp\left(-\frac{1}{4} \gamma_\omega |x'|^2 \right) |x'| dx' \cdot \int_{\mathbb{R}} \sup_{s,\xi'} |\nabla_{\xi'} \nu_s(t, \xi')| dt = O(x_0^{-1/2}).$$
(3.50)

Using (3.49), (3.50) and Lemma 3.6.4 (i), it remains to derive the relation

$$\int_{|t| \le x_0^{1/4}} \int_{|x'| < \epsilon \sqrt{x_0 - t}} \exp\left((x_0 - t)A_s\left(\frac{ix'}{\sqrt{x_0 - t}}\right)\right) \nu_s(t, 0) \, \mathrm{d}x' \, \mathrm{d}t$$

$$= \int_{|t| \le x_0^{1/4}} \int_{\mathbb{R}^{d-1}} \exp\left(-\frac{1}{2}x' \cdot Q_s x'\right) \nu_s(t, 0) \, \mathrm{d}x' \, \mathrm{d}t + O(x_0^{-1/2}).$$
(3.51)

Due to Lemma (3.6.4) (ii) with $\beta = 1/6$, we obtain

$$\sup_{s \in \mathcal{V}_{\omega}} \int_{\epsilon \sqrt{x_0 - t} > |x'| \ge x_0^{1/6}} \left| \exp\left((x_0 - t) A_s\left(\frac{ix'}{\sqrt{x_0 - t}}\right) \right) \right| \mathrm{d}x' = O(x_0^{-n}),$$
$$\sup_{s \in \mathcal{V}_{\omega}} \int_{|x'| \ge x_0^{1/6}} \exp\left(-\frac{1}{2}x' \cdot Q_s x' \right) \mathrm{d}x' = O(x_0^{-n}).$$

On the other hand,

$$\sup_{s \in \mathcal{V}_{\omega}} \int_{|x'| < x_0^{1/6}} \left| \exp\left((x_0 - t) A_s \left(\frac{ix'}{\sqrt{x_0 - t}} \right) \right) - \exp\left(-\frac{1}{2} x' \cdot Q_s x' \right) \right| dx'$$

$$= \sup_{s \in \mathcal{V}_{\omega}} \int_{|x'| < x_0^{1/6}} \exp\left(-\frac{1}{2} x' \cdot Q_s x' \right) \left| \exp\left(O\left(\frac{|x'|^3}{\sqrt{x_0}} \right) \right) - 1 \right| dx'$$

$$\lesssim \int_{|x'| < x_0^{1/6}} \exp\left(-\frac{1}{2} \gamma_{\omega} |x'|^2 \right) \frac{|x'|^3}{\sqrt{x_0}} dx' = O(x_0^{-1/2}).$$

Hence, we deduce

$$\int_{|x'|<\epsilon\sqrt{x_0-t}} \exp\left((x_0-t)A_s\left(\frac{ix'}{\sqrt{x_0-t}}\right)\right) \mathrm{d}x' - \int_{\mathbb{R}^{d-1}} \exp\left(-\frac{1}{2}x'\cdot Q_sx'\right) \mathrm{d}x' = O(x_0^{-1/2})$$

for each $t \in [-x_0^{1/4}, x_0^{1/4}]$. Finally, we multiply the above relation with $\nu_s(t, 0)$ and then integrate over the interval $[-x_0^{1/4}, x_0^{1/4}]$. Since $\sup_s |\nu_s(t, 0)|$ is integrable over \mathbb{R} , the right hand side is still $O(x_0^{-1/2})$. Thus, we derive (3.51) as we wish.

3.6.2 Estimates of the integral J

In this part, we want to show that the expression J decays as $O(|x - y|^{-d/2})$. Thus, taking into account (3.44), we conclude that J does not contribute to the leading term of the reduced Green's function.

In (3.20), we set the coordinate functions of ρ' as (ρ_1, \ldots, ρ_d) . Let us introduce the smooth function $\mu^{(l)}(k, x, y) = \rho_l(k + k_0, x, y)\mu(k)$ for any $k \in \mathbb{R}^d$. The support of $\mu^{(l)}$ (as a function of k for each pair (x, y)) is contained in the support of μ and $\mu^{(l)}(k, \cdot, \cdot)$ is $\mathbb{Z}^d \times \mathbb{Z}^d$ -periodic. We denote the components of a vector k in \mathbb{R}^d as (k_1, \ldots, k_d) . Observe that J is the sum of integrals J_l $(1 \le l \le d)$ if we define

$$J_{l} := (2\pi)^{-d} \int_{\mathcal{O}} e^{i(k-k_{0})\cdot(x-y)} \frac{\mu^{(l)}(k-k_{0},x,y)(k-k_{0})_{l}}{\lambda_{j}(k+i\beta_{s}) - \lambda} \,\mathrm{d}k.$$
(3.52)

Proposition 3.6.6. As $|x - y| \to \infty$, we have $J_1 = O(|x - y|^{-(d+1)/2})$ and $J_l = O(|x - y|^{-d/2})$ if l > 1. In particular, $J = O(|x - y|^{-d/2})$.

Proof. Indeed, to treat these integrals, we need to re-examine the calculation in the previous subsection done for the integral I. After applying the orthogonal transformation \mathcal{R}_s

on each integral J_l , we rewrite them under the form of (3.41) as

$$J_{l} = (2\pi)^{-d} \int_{|\xi'| < \epsilon} \int_{\mathbb{R}} e^{i|x-y|\xi_{1}} \frac{\tilde{\mu}_{s}^{(l)}(\xi_{1},\xi',x,y)\xi_{l}}{i\xi_{1} - A_{s}(i\xi')} \,\mathrm{d}\xi_{1} \,\mathrm{d}\xi', \qquad (3.53)$$

where $\tilde{\mu}_{s}^{(l)}(\xi, x, y)$ is $\mu^{(l)}(\xi, x, y)(B_{s}(i\xi))^{-1}$ on the support of $\mu^{(l)}$ and vanishes elsewhere. Let $\nu_{s}^{(l)}(t, \xi', x, y)$ be the Fourier transform in ξ_{1} of $\tilde{\mu}_{s}^{(l)}(\xi_{1}, \xi', x, y)$. If the parameter *s* is viewed as another argument of our functions here, then $\nu_{s}^{(l)}(\cdot, \xi', x, y)$ is a Schwartz function for each quadruple (s, ξ', x, y) . It is elementary to check that the Fourier transform $\nu_{s}^{(l)}(t, \xi', x, y)$ is jointly continuous on $\mathcal{V}_{\omega} \times \mathbb{R} \times \mathbb{R}^{d-1} \times \mathbb{R}^{d} \times \mathbb{R}^{d}$ due to the corresponding property of $\tilde{\mu}_{s}^{(l)}(\xi, x, y)$. Periodicity in (x, y) of $\nu_{s}^{(l)}$ and Lemma 3.9.1 imply the following decay:

$$\lim_{t \to \infty} |t|^N \sup_{(s,\xi',x,y) \in \mathcal{V}_\omega \times \overline{D'(0,\epsilon)} \times \mathbb{R}^d \times \mathbb{R}^d} |\nu_s^{(l)}(t,\xi',x,y)| = 0, \quad N \ge 0.$$
(3.54)

In particular,

$$\max_{1 \le l \le d} \sup_{(s,t,\xi',x,y) \in \mathcal{V}_{\omega} \times \mathbb{R} \times \overline{D'(0,\epsilon)} \times \mathbb{R}^d \times \mathbb{R}^d} |\nu_s^{(l)}(t,\xi',x,y)| < \infty$$
(3.55)

and

$$S := \max_{1 \le l \le d} \int_{\mathbb{R}} \sup_{(s,\xi',x,y) \in \mathcal{V}_{\omega} \times \overline{D'(0,\epsilon)} \times \mathbb{R}^d \times \mathbb{R}^d} |\nu_s^{(l)}(t,\xi',x,y)| \, \mathrm{d}t < \infty.$$
(3.56)

Recall that when $0 < |\xi'| < \epsilon$, $\Re(A_s(i\xi')) < 0$ and thus from (3.55),

$$\lim_{t \to -\infty} e^{(-t+|x-y|)A_s(i\xi')} \nu_s^{(1)}(t,\xi',x,y) = 0.$$
(3.57)

<u>Case 1</u>: *l* = 1.

Using (3.42), (3.57) and integration by parts, we obtain

$$J_{1} = \frac{1}{(2\pi)^{d}} \int_{|\xi'| < \epsilon} \int_{-\infty}^{0} e^{-wA_{s}(i\xi')} \int_{-r}^{r} \xi_{1} e^{i(w+|x-y|)\xi_{1}} \tilde{\mu}_{s}^{(1)}(\xi_{1},\xi',x,y) \,\mathrm{d}\xi_{1} \,\mathrm{d}w \,\mathrm{d}\xi$$

$$= -\frac{i}{(2\pi)^{d}} \int_{|\xi'| < \epsilon} \int_{-\infty}^{|x-y|} e^{(-t+|x-y|)A_{s}(i\xi')} \frac{\mathrm{d}}{\mathrm{d}t} \nu_{s}^{(1)}(t,\xi',x,y) \,\mathrm{d}t \,\mathrm{d}\xi'$$

$$= -\frac{i}{(2\pi)^{d}} \int_{|\xi'| < \epsilon} \left(\nu_{s}^{(1)}(|x-y|,\xi',x,y) + \int_{-\infty}^{|x-y|} A_{s}(i\xi') e^{(-t+|x-y|)A_{s}(i\xi')} \times \nu_{s}^{(1)}(t,\xi',x,y) \,\mathrm{d}t \right) \,\mathrm{d}\xi'.$$
(3.58)

Recall the notation $x_0 = |x - y|$. The term

$$\int_{|\xi'|<\epsilon} \nu_s^{(1)}(x_0,\xi',x,y) \,\mathrm{d}\xi'$$

decays rapidly in x_0 , due to (3.54). We decompose the other term

$$\int_{-\infty}^{x_0} A_s(i\xi') e^{(x_0-t)A_s(i\xi')} \nu_s^{(1)}(t,\xi',x,y) \,\mathrm{d}t$$

into two parts, where the first integral is taking over $(x_0/2, x_0]$ and the second one over $(-\infty, x_0/2]$. The first part decays rapidly, as in Lemma 3.6.4 (i). Now we need to prove that the second part decays as $O(x_0^{(d+1)/2})$. To do this, we use the change of variables $x' = \xi' \sqrt{x_0 - t}$ to rewrite the remaining integral as

$$x_{0}^{(d+1)/2} \int_{|\xi'| < \epsilon} \int_{-\infty}^{x_{0}/2} A_{s}(i\xi') e^{(-t+x_{0})A_{s}(i\xi')} \nu_{s}^{(1)}(t,\xi',x,y) \, \mathrm{d}t \, \mathrm{d}\xi'$$

$$= \int_{-\infty}^{x_{0}/2} \left(1 - \frac{t}{x_{0}}\right)^{-(d+1)/2} \int_{|x'| < \epsilon\sqrt{x_{0}-t}} \left(-\frac{1}{2}x' \cdot Q_{s}x' + O\left(\frac{|x'|^{3}}{\sqrt{x_{0}-t}}\right)\right) \qquad (3.59)$$

$$\times \exp\left(-\frac{1}{2}x' \cdot Q_{s}x' + O\left(\frac{|x'|^{3}}{\sqrt{x_{0}-t}}\right)\right) \nu_{s}^{(1)}\left(t,\left(\frac{x'}{\sqrt{x_{0}-t}}\right),x,y\right) \, \mathrm{d}x' \, \mathrm{d}t.$$

From (3.56), we derive

$$\int_{-\infty}^{x_0/2} \left(1 - \frac{t}{x_0}\right)^{-(d+1)/2} \sup_{(s,\xi',x,y)\in\mathcal{V}_{\omega}\times\overline{D'(0,\epsilon)}\times\mathbb{R}^d\times\mathbb{R}^d} |\nu_s^{(1)}(t,\xi',x,y)| \,\mathrm{d}t \le 2^{\frac{(d+1)}{2}}S.$$
(3.60)

On the other hand, we recall that

$$\Re\left(-\frac{1}{2}x'\cdot Q_s x' + O\left(\frac{|x'|^3}{\sqrt{x_0-t}}\right)\right) \le -\frac{1}{4}\gamma_{\omega}|x'|^2.$$

The exponential term is majorized as follows:

$$\left| \left(-\frac{1}{2}x' \cdot Q_s x' + O\left(\frac{|x'|^3}{\sqrt{x_0 - t}}\right) \right) \exp\left(-\frac{1}{2}x' \cdot Q_s x' + O\left(\frac{|x'|^3}{\sqrt{x_0 - t}}\right) \right) \right|$$
$$\leq \left(\frac{1}{2}x' \cdot Q_s x' + O(\epsilon |x'|^2) \right) \exp\left(-\frac{1}{4}\gamma_{\omega} |x'|^2 \right).$$

Consequently,

$$\int_{|x'|<\epsilon\sqrt{x_0-t}} \left| \left(-\frac{1}{2}x' \cdot Q_s x' + O\left(\frac{|x'|^3}{\sqrt{x_0-t}}\right) \right) \exp\left(-\frac{1}{2}x' \cdot Q_s x' + O\left(\frac{|x'|^3}{\sqrt{x_0-t}}\right) \right) \right| dx' \\
\lesssim \int_{\mathbb{R}^{d-1}} |x'|^2 \exp\left(-\frac{1}{4}\gamma_{\omega}|x'|^2\right) dx' < \infty.$$
(3.61)

Combining (3.58) through (3.61), we deduce $J_1 = O(x_0^{-(d+1)/2})$.

<u>Case 2</u>: *l* > 1.

Using (3.42) and decomposing J_l into two parts as in **Case 1**, we get

$$J_{l} = \frac{1}{(2\pi)^{d}} \int_{|\xi'| < \epsilon} \int_{-\infty}^{0} \xi_{l} e^{-wA_{s}(i\xi')} \int_{-r}^{r} e^{i(w+|x-y|)\xi_{1}} \tilde{\mu}_{s}^{(l)}(\xi_{1},\xi',x,y) \,\mathrm{d}\xi_{1} \,\mathrm{d}w \,\mathrm{d}\xi$$

$$= \frac{1}{(2\pi)^{d}} \int_{|\xi'| < \epsilon} \int_{-\infty}^{|x-y|} \xi_{l} e^{(-t+|x-y|)A_{s}(i\xi')} \nu_{s}^{(l)}(t,\xi',x,y) \,\mathrm{d}t \,\mathrm{d}\xi'$$

$$= \frac{1}{(2\pi)^{d}} \int_{|\xi'| < \epsilon} \int_{-\infty}^{|x-y|/2} \xi_{l} e^{(-t+|x-y|)A_{s}(i\xi')} \nu_{s}^{(l)}(t,\xi',x,y) \,\mathrm{d}t \,\mathrm{d}\xi' + o(|x-y|^{-d/2}).$$

(3.62)

By changing the variables as before,

$$x_{0}^{d/2} \int_{|\xi'| < \epsilon} \int_{-\infty}^{x_{0}/2} \xi_{l} e^{(-t+x_{0})A_{s}(i\xi')} \nu_{s}^{(l)}(t,\xi',x,y) \, \mathrm{d}t \, \mathrm{d}\xi'$$

$$= \int_{-\infty}^{x_{0}/2} \left(1 - \frac{t}{x_{0}}\right)^{-d/2} \int_{|x'| < \epsilon\sqrt{x_{0}-t}} x_{l}' \exp\left(-\frac{1}{2}x' \cdot Q_{s}x' + O\left(\frac{|x'|^{3}}{\sqrt{x_{0}-t}}\right)\right) \qquad (3.63)$$

$$\times \nu_{s}^{(l)} \left(t, \left(\frac{x'}{\sqrt{x_{0}-t}}\right), x, y\right) \, \mathrm{d}x' \, \mathrm{d}t.$$

In a similar manner, we obtain

$$\begin{split} \int_{-\infty}^{x_0/2} \left(1 - \frac{t}{x_0}\right)^{-d/2} \int_{|x'| < \epsilon \sqrt{x_0 - t}} \left| x_l' \exp\left(-\frac{1}{2}x' \cdot Q_s x' + O\left(\frac{|x'|^3}{\sqrt{x_0 - t}}\right)\right) \right| \\ & \times \left| \nu_s^{(l)} \left(t, \left(\frac{x'}{\sqrt{x_0 - t}}\right), x, y\right) \right| \mathrm{d}x' \mathrm{d}t \\ & \le 2^{d/2} S \int_{\mathbb{R}^{d-1}} |x'| \exp\left(-\frac{1}{4}\gamma_\omega |x'|^2\right) \mathrm{d}x' < \infty. \end{split}$$

This final estimate and (3.62)-(3.63) imply $J_l = O(x_0^{-d/2})$.

3.7 The full Green's function asymptotics

The main purpose of this section is to give a detailed proof of Theorem 3.5.2. Essentially, this theorem is needed for showing that full Green's function $G_{s,\lambda}$ has the same asymptotics as the reduced Green's function G_0 as $|x - y| \to \infty$. First, we recall that for each unit vector s, $T_s(k) = (1 - \eta(k))(L_s(k) - \lambda)^{-1} + \eta(k)((L_s(k) - \lambda)|_{R(Q_s(k))})^{-1}Q_s(k)$ and the operator T_s is unitarily equivalent (via the Floquet transform) to the direct integral of the operators $T_s(k)$ over \mathcal{O} . Now we observe that the kernel of each projector $P_s(k)$ (see Subsection 3.5.2) is the smooth function:

$$\frac{\phi(k+i\beta_s,x)\overline{\phi(k-i\beta_s,y)}}{F(k+i\beta_s)}$$

for each k in the support of η . Thus, $(1 - \eta(k))P_s(k)$ is a finite rank smoothing operator on \mathbb{T}^d . Moreover, we also have $(L_s(k) - \lambda)T_s(k) = T_s(k)(L_s(k) - \lambda) = I - \eta(k)P_s(k)$. Each $T_s(k)$ is a parametrix (i.e., an inverse modulo a smoothing operator) of the elliptic operator $L_s(k) - \lambda$ when $(s, k) \in \mathbb{S}^{d-1} \times \mathcal{O}$. This suggests to study parametrices of the family of elliptic operators $L_s(k) - \lambda$ simultaneously.

3.7.1 Parameter-dependent periodic pseudodifferential operators

First, we briefly recall some basic definitions of **periodic** (or **toroidal**) pseudodifferential operators (i.e., Ψ DO on the torus \mathbb{T}^d). We also introduce some useful classes of symbols with parameters and describe some of their properties that we will use.

There are several approaches to defining pseudodifferential operators on the torus. The standard approach based on Hörmander's symbol classes (see e.g., [72]) uses local smooth structure on the torus \mathbb{T}^d and thus ignores the group structure on \mathbb{T}^d . An alternative approach uses Fourier series with the difference calculus and avoids using local coordinate charts on \mathbb{T}^d (the details in [67, Chapter 4])⁶. To make a distinction, Ruzhansky and Turunen in [67] refer to the symbols in the first approach as **Euclidean symbols** and the symbols in the latter one as **toroidal symbols** (see [67, Section 4.5]). We recall their definitions for only the Kohn-Nirenberg symbol classes, which we need here:

⁶A different approach to periodic ΨDOs is introduced by A. Sobolev [73].

Definition 3.7.1. Let m be a real number.

(a) The class $S^m(\mathbb{T}^d \times \mathbb{R}^d)$ consists of all smooth functions $\sigma(x,\xi)$ on $\mathbb{T}^d \times \mathbb{R}^d$ such that for any multi-indices α, β ,

$$|D_{\xi}^{\alpha}D_{x}^{\beta}\sigma(x,\xi)| \le C_{\alpha\beta}(1+|\xi|)^{m-|\alpha|},$$

for some constant $C_{\alpha,\beta}$ that depends only on α,β . Symbols in $S^m(\mathbb{T}^d \times \mathbb{R}^d)$ are called **Euclidean symbols** of order m on \mathbb{T}^d .

(b) The class $S^m(\mathbb{T}^d \times \mathbb{Z}^d)$ consists of all functions $\sigma(x,\xi)$ on $\mathbb{T}^d \times \mathbb{Z}^d$ such that for each $\xi \in \mathbb{Z}^d$, $\sigma(.,\xi) \in C^{\infty}(\mathbb{T}^d)$ and for any multi-indices α, β ,

$$|\Delta_{\xi}^{\alpha} D_x^{\beta} \sigma(x,\xi)| \le C_{\alpha\beta} (1+|\xi|)^{m-|\alpha|},$$

for some constant $C_{\alpha,\beta}$ that depends only on α,β . Here we recall the definition of the forward difference operator Δ_{ξ}^{α} with respect to the variable ξ [67]. Let f be a complex-valued function defined on \mathbb{Z}^d and $1 \leq j \leq d$. Then we define

$$\Delta_j f(\xi) := f(\xi_1, \dots, \xi_{j-1}, \xi_j + 1, \xi_{j+1}, \dots, \xi_d) - f(\xi),$$

and for any multi-index α ,

$$\Delta_{\xi}^{\alpha} := \Delta_1^{\alpha_1} \dots \Delta_d^{\alpha_d}.$$

Symbols in $S^m(\mathbb{T}^d \times \mathbb{Z}^d)$ are called **toroidal symbols** of order m on \mathbb{T}^d .

(c) The intersection of all the classes $S^m(\mathbb{T}^d \times \mathbb{R}^d)$ $(S^m(\mathbb{T}^d \times \mathbb{Z}^d))$ is denoted by $S^{-\infty}(\mathbb{T}^d \times \mathbb{R}^d)$ $(S^{-\infty}(\mathbb{T}^d \times \mathbb{Z}^d))$, which are also called **smoothing symbols**.

Due to [67, Theorem 4.5.3], a symbol is toroidal of order m if and only if it could be extended in ξ to an Euclidean symbol of the same order m. Such an extension is unique

modulo a smoothing symbol. Consequently, we will use the notation $S^m(\mathbb{T}^d)$ for both classes $S^m(\mathbb{T}^d \times \mathbb{R}^d)$ and $S^m(\mathbb{T}^d \times \mathbb{Z}^d)$. The two approaches are essentially equivalent in defining pseudodifferential operators on \mathbb{T}^d whenever the symbol is in the class $S^m(\mathbb{T}^d)$. Following [36], this motivates us to define periodic pseudodifferential operators as follows:

Definition 3.7.2. Given a symbol $\sigma(x,\xi) \in S^m(\mathbb{T}^d)$, we denote by $Op(\sigma)$ the corresponding periodic pseudodifferential operator defined by

$$(Op(\sigma)f)(x) := \sum_{\xi \in \mathbb{Z}^d} \sigma(x,\xi) \tilde{f}(\xi) e^{2\pi i \xi \cdot x}, \qquad (3.64)$$

where $\tilde{f}(\xi)$ is the Fourier coefficient of f at ξ . The right hand side of (3.64) converges absolutely if, for instance, $f \in C^{\infty}(\mathbb{T}^d)$.

We also use the notation $Op(S^m(\mathbb{T}^d))$ for the set of all periodic pseudodifferential operators $Op(\sigma)$ with $\sigma \in S^m(\mathbb{T}^d)$.

Since we must deal with parameters s and k, we introduce a suitable class of symbols depending on parameters $(s, k) \in \mathbb{S}^{d-1} \times \mathcal{O}$.

Definition 3.7.3. The parameter-dependent class $\tilde{S}^m(\mathbb{T}^d)$ consists of symbols $\sigma(s, k; x, \xi)$ satisfying the following conditions:

- For each $(s,k) \in \mathbb{S}^{d-1} \times \mathcal{O}$, the function $\sigma(s,k;\cdot,\cdot)$ is a symbol in the class $S^m(\mathbb{T}^d)$.
- Consider any multi-indices α, β, γ . Then for each $s \in \mathbb{S}^{d-1}$, the function $\sigma(s, \cdot; \cdot, \cdot)$ is smooth on $\mathcal{O} \times \mathbb{T}^d \times \mathbb{R}^d$, and furthermore,

$$\sup_{s\in\mathbb{S}^{d-1}} |D_k^{\alpha} D_{\xi}^{\beta} D_x^{\gamma} \sigma(s,k;x,\xi)| \le C_{\alpha\beta\gamma} (1+|\xi|)^{m-|\alpha|-|\beta|},$$

for some constant $C_{\alpha\beta\gamma} > 0$ that is independent of s, k, x, and ξ .

Thus, taking derivatives of a symbol in $\tilde{S}^m(\mathbb{T}^d)$ with respect to k improves decay in ξ . We also denote

$$\tilde{S}^{-\infty}(\mathbb{T}^d) := \bigcap_{m \in \mathbb{R}} \tilde{S}^m(\mathbb{T}^d).$$

Definition 3.7.4. For each $m \in \mathbb{R} \cup \{-\infty\}$, we denote by $Op(\tilde{S}^m(\mathbb{T}^d))$ the set of all families of periodic pseudodifferential operators $\{Op(\sigma(s,k;\cdot,\cdot))\}_{(s,k)\in\mathbb{S}^{d-1}\times\mathcal{O}}$, where σ runs over the class $\tilde{S}^m(\mathbb{T}^d)$.

Example 3.7.5.

- Suppose that $|\lambda|$ is small enough so that $\max_{s \in \mathbb{S}^{d-1}} |\beta_s| < 1$. Then the family of symbols $\{(1 + (\xi + k + i\beta_s)^2)^{m/2}\}_{(s,k)}$ belongs to the class $\tilde{S}^m(\mathbb{T}^d)$ for any $m \in \mathbb{R}$.
- If $a_{\alpha}(x) \in C^{\infty}(\mathbb{T}^d)$ and $m \ge 0$, then

$$\{\sum_{|\alpha| \le m} a_{\alpha}(x)(\xi + k + i\beta_s)^{\alpha}\}_{(s,k)} \in \tilde{S}^m(\mathbb{T}^d).$$

- The family of elliptic operators $\{(L_s(k) \lambda)\}_{(s,k)}$ is in $Op(\tilde{S}^2(\mathbb{T}^d))$.
- If $a = \{a(s,k;x,\xi)\}_{(s,k)} \in \tilde{S}^{l}(\mathbb{T}^{d}) \text{ and } b = \{b(s,k;x,\xi)\}_{(s,k)} \in \tilde{S}^{m}(\mathbb{T}^{d}) \text{ then } ab = \{ab(s,k;x,\xi)\}_{(s,k)} \in \tilde{S}^{l+m}(\mathbb{T}^{d}).$
- $a(s,k;x,\xi) \in \tilde{S}^{l}(\mathbb{T}^{d})$ implies $D_{k}^{\alpha}D_{\xi}^{\beta}D_{x}^{\gamma}a(s,k;x,\xi) \in \tilde{S}^{l-|\alpha|-|\beta|}(\mathbb{T}^{d}).$

The following result will be needed in the next subsection:

Theorem 3.7.6. There exists a family of parametrices $\{A_s(k)\}_{(s,k)}$ in the class $Op(\tilde{S}^{-2}(\mathbb{T}^d))$ for the family of elliptic operators $\{(L_s(k) - \lambda)\}_{(s,k)}$.

The reader can refer to Section 3.8 for the proof of this result as well as some other basic properties of parameter-dependent toroidal Ψ DOs.

3.7.2 Decay of the Schwartz kernel of T_s

Lemma 3.7.7. For all k on a sufficiently small neighborhood of the support of η , λ (< 0) is in the resolvent of the operator $L_s(k)Q_s(k)$ acting on $L^2(\mathbb{T}^d)$. Furthermore, for such k, we have the following identity:

$$((L_s(k) - \lambda)|_{R(Q_s(k))})^{-1}Q_s(k) = \lambda^{-1}P_s(k) + (L_s(k)Q_s(k) - \lambda)^{-1}.$$
(3.65)

Proof. In the block-matrix form, $(L_s(k)Q_s(k) - \lambda)$ is

$$\left(\begin{array}{c|c} -\lambda P_s(k) & 0\\ \hline 0 & (L_s(k) - \lambda)|_{R(Q_s(k))} \end{array}\right).$$
(3.66)

This gives the first claim of this lemma. The inverse of (3.66) is

$$\left(egin{array}{c|c} -\lambda^{-1}P_s(k) & 0 \ \hline 0 & ((L_s(k)-\lambda)|_{R(Q_s(k))})^{-1} \end{array}
ight).$$

which proves the identity (3.65).

The identity (3.65) implies that for each (s, k), the operator

$$\eta(k)((L_s(k) - \lambda)|_{R(Q_s(k))})^{-1}Q_s(k)$$

is a periodic pseudodifferential operator in $S^{-2}(\mathbb{T}^d)$. Thus, each of the operators $T_s(k)$ is also in $S^{-2}(\mathbb{T}^d)$ and its symbol is smooth in (s,k) since $P_s(k)$ and $Q_s(k)$ are smooth in (s,k). Actually, more information about the family of operators $\{T_s(k)\}_{(s,k)}$ and their Schwartz kernels can be obtained.

At first, we want to introduce a class of family of operators whose kernels behave
nicely.

Definition 3.7.8. We denote by S the set consisting of families of smoothing operators $\{U_s(k)\}_{(s,k)}$ acting on \mathbb{T}^d so that the following properties hold:

- For any m₁, m₂ ∈ ℝ, the operator U_s(k) is smooth in k as a B(H^{m₁}(T^d), H^{m₂}(T^d))-valued function⁷.
- The following uniform condition holds for any multi-index α :

$$\sup_{s,k} \|D_k^{\alpha} U_s(k)\|_{B(H^{m_1}(\mathbb{T}^d), H^{m_2}(\mathbb{T}^d))} < \infty.$$

We remark that if the family of smoothing operators $\{U_s(k)\}_{(s,k)}$ is in $Op(\tilde{S}^{-\infty}(\mathbb{T}^d))$, then this family also belongs to S.

In order to obtain information on Schwartz kernels of a family of operators in S, we need to use the following standard lemma on Schwartz kernels of integral operators acting on \mathbb{T}^d .

Lemma 3.7.9. Let A be a bounded operator in $L^2(\mathbb{T}^d)$. Suppose that the range of A is contained in $H^m(\mathbb{T}^d)$, where m > d/2 and in addition,

$$\|Af\|_{H^m(\mathbb{T}^d)} \le C \|f\|_{H^{-m}(\mathbb{T}^d)}$$

for all $f \in L^2(\mathbb{T}^d)$.

Then A is an integral operator whose kernel $K_A(x, y)$ is bounded and uniformly continuous on $\mathbb{T}^d \times \mathbb{T}$ and the following estimate holds:

$$|K_A(x,y)| \le \gamma_0 C,\tag{3.67}$$

⁷We remind the reader that B(E, F) denotes the space of all bounded linear operators from the Banach space E to F.

where γ_0 is a constant depending only on d and m.

The fact can be found in [1, Lemma 2.2].

Now we can state a useful property of Schwartz kernels of a family of operators in S.

Corollary 3.7.10. If $\{U_s(k)\}_{(s,k)}$ is a family of smoothing operators in S, then the Schwartz kernel $K_{U_s}(k, x, y)$ of the operator $U_s(k)$ satisfies

$$\sup_{s,k,x,y} |D_k^{\alpha} K_{U_s}(k,x,y)| < \infty,$$

for any multi-index α .

Proof. We pick any m > d/2. Then by Definition 3.7.8, we have

$$\sup_{s,k} \|D_k^{\alpha} U_s(k)f\|_{H^m(\mathbb{T}^d)} \le C_{\alpha} \|f\|_{H^{-m}(\mathbb{T}^d)}.$$

Applying Lemma 3.7.9, the estimates (3.67) hold for kernels $D_k^{\alpha} K_{U_s}(k, x, y)$ of the operators $D_k^{\alpha} U_s(k)$ uniformly in (s, k).

We now go back to the family of operators $T_s(k)$.

Proposition 3.7.11. There is a family of periodic pseudodifferential operators $\{A_s(k)\}_{(s,k)}$ in $Op(\tilde{S}^{-2}(\mathbb{T}^d))$ such that the family of operators $\{T_s(k) - A_s(k)\}_{(s,k)}$ belongs to S.

Proof. Due to Theorem 3.7.6, there is a family of operators $\{A_s(k)\}_{(s,k)}$ in $Op(\tilde{S}^{-2}(\mathbb{T}^d))$ and a family of operators $\{R_s(k)\}_{(s,k)}$ in $Op(\tilde{S}^{-\infty}(\mathbb{T}^d))$ such that

$$(L_s(k) - \lambda)A_s(k) = I - R_s(k).$$

Since $T_s(k)(L_s(k) - \lambda) = I - \eta(k)P_s(k)$, we deduce that

$$T_s(k) = A_s(k) - \eta(k)P_s(k)A_s(k) + T_s(k)R_s(k).$$
(3.68)

Now it remains to show that the two families of smoothing operators $\{T_s(k)R_s(k)\}_{(s,k)}$ and $\{\eta(k)P_s(k)A_s(k)\}_{(s,k)}$ are in S. Let us fix any two real numbers m_1, m_2 and a multiindex α . Notice that $(L_s(k) - \lambda)$ is analytic in k as a $B(H^{m_2}(\mathbb{T}^d), H^{m_2-2}(\mathbb{T}^d))$ -valued function and also,

$$\sup_{s,k} \|D_k^{\alpha}(L_s(k)-\lambda)\|_{B(H^{m_2}(\mathbb{T}^d),H^{m_2-2}(\mathbb{T}^d))} < \infty.$$

Due to Lemma 3.7.7,

$$T_s(k) = (1 - \eta(k))(L_s(k) - \lambda)^{-1} + \eta(k)\lambda^{-1}P_s(k) + \eta(k)(L_s(k)Q_s(k) - \lambda)^{-1}.$$

Thus, $T_s(k)$ is smooth in k as a $B(H^{m_2-2}(\mathbb{T}^d), H^{m_2}(\mathbb{T}^d))$ -valued function and moreover,

$$\sup_{s,k} \|D_k^{\alpha} T_s(k)\|_{B(H^{m_2-2}(\mathbb{T}^d), H^{m_2}(\mathbb{T}^d))} < \infty.$$
(3.69)

Since $\{R_s(k)\}$ is in $Op(\tilde{S}^{-\infty}(\mathbb{T}^d))$, $R_s(k)$ is smooth in k as a $B(H^{m_1}(\mathbb{T}^d), H^{m_2-2}(\mathbb{T}^d))$ -valued function and furthermore,

$$\sup_{s,k} \|D_k^{\alpha} R_s(k)\|_{B(H^{m_1}(\mathbb{T}^d), H^{m_2-2}(\mathbb{T}^d))} < \infty.$$
(3.70)

By (3.69), (3.70) and Leibnitz's rule, we deduce that $T_s(k)R_s(k)$ is smooth in k as a $B(H^{m_1}(\mathbb{T}^d), H^{m_2}(\mathbb{T}^d))$ -valued function and the corresponding uniform estimate also holds. Hence, we conclude that the family $\{T_s(k)R_s(k)\}_{(s,k)}$ belongs to S. Meanwhile, since $\{\eta(k)P_s(k)\}_{(s,k)}$ is in S and $\{D_k^{\alpha}A_s(k)\}_{(s,k)}$ is a toroidal pseudodifferential operator of order $2 - |\alpha|$ for any multi-index α , we could repeat the above argument to show that the family $\{\eta(k)P_s(k)A_s(k)\}_{(s,k)}$ is also in S.

We need the following important estimate of Schwartz kernels of operators $T_s(k)$:

Corollary 3.7.12. Let $K_s(k, x, y)$ be the Schwartz kernel of the operator $T_s(k)$. Let N > d - 2. If α is a multi-index such that $|\alpha| = N$, then each $D_k^{\alpha} K_s(k, x, y)$ is a continuous function on $\mathbb{T}^d \times \mathbb{T}^d$ and the following estimate also holds uniformly with respect to (x, y):

$$\sup_{(s,k)\in\mathbb{S}^{d-1}\times\mathcal{O}}|D_k^{\alpha}K_s(k,x,y)|<\infty.$$

Proof. Due to Proposition 3.7.11, the operator $T_s(k)$ is a sum of operators $A_s(k)$ and $U_s(k)$ such that $\{A_s(k)\}_{(s,k)} \in Op(\tilde{S}^{-2}(\mathbb{T}^d))$ and $\{U_s(k)\}_{(s,k)} \in S$. In particular,

$$K_s(k, x, y) = K_{A_s}(k, x, y) + K_{U_s}(k, x, y).$$

Recall that in the distributional sense, the Schwartz kernel $K_{A_s}(k, x, y)$ of the periodic pseudodifferential operator $A_s(k)$ is given by

$$\sum_{\xi \in \mathbb{Z}^d} \sigma(s,k;x,\xi) e^{2\pi i \xi \cdot (x-y)},$$

where $\sigma(s, k; x, \xi)$ is the symbol of the operator $A_s(k)$.

Since $\{\sigma(s,k;x,\xi)\}_{(s,k)}$ is in $\tilde{S}^{-2}(\mathbb{T}^d)$,

$$|e^{2\pi i\xi \cdot (x-y)} D_k^{\alpha} \sigma(s,k;x,\xi)| \lesssim (1+|\xi|)^{-2-N}$$

Since -(2+N) < -d, the sum

$$\sum_{\xi \in \mathbb{Z}^d} D_k^{\alpha} \sigma(s,k;x,\xi) e^{2\pi i \xi \cdot (x-y)}$$

converges absolutely and moreover,

$$\sup_{(s,k,x,y)\in\mathbb{S}^{d-1}\times\mathcal{O}\times\mathbb{T}^d\times\mathbb{T}^d}|D_k^{\alpha}K_{A_s}(k,x,y)| \lesssim \sum_{\xi\in\mathbb{Z}^d}(1+|\xi|)^{-(d+1)}<\infty.$$

Combining this with Corollary 3.7.10, we complete the proof.

Notation 3.7.13. Let ψ be a function on \mathbb{R}^d and γ be a vector in \mathbb{R}^d , then $\tau_{\gamma}\psi$ is the γ -shifted version of ψ . Namely, it is defined as follows:

$$\tau_{\gamma}\psi(\cdot)=\psi(\cdot+\gamma).$$

We denote by \mathcal{P} the subset of $C_0^{\infty}(\mathbb{R}^d)$ consisting of all functions ψ such that its support is connected, and if γ is a **non-zero vector in** \mathbb{Z}^d , then the support of $\tau_{\gamma}\psi$ does not intersect with the support of ψ .

Definition 3.7.14. Since \mathbb{R}^d is the universal covering space of \mathbb{T}^d , we can consider the covering map

$$\pi: \mathbb{R}^d \to \mathbb{R}^d / \mathbb{Z}^d = \mathbb{T}^d.$$

In particular, $\pi(x + \gamma) = \pi(x)$ for any $x \in \mathbb{R}^d$ and $\gamma \in \mathbb{Z}^d$.

A standard fundamental domain (with respect to the covering map π) is of the form $[0,1]^d + \gamma$ for some vector γ in \mathbb{R}^d . Thus, a standard fundamental domain is a fundamental domain of \mathbb{R}^d with respect to the lattice \mathbb{Z}^d .

Using Definition 2.2.3 of the Floquet transform \mathcal{F} , we can obtain the following formula:

Lemma 3.7.15. Let ϕ and θ be any two smooth functions in \mathcal{P} . Then the Schwartz kernel

 $K_{s,\phi,\theta}$ of the operator $\phi T_s \theta$ satisfies the following identity for any $(x,y) \in \mathbb{R}^d \times \mathbb{R}^d$:

$$K_{s,\phi,\theta}(x,y) = \frac{1}{(2\pi)^d} \int_{\mathcal{O}} e^{ik \cdot (x-y)} \phi(x) K_s(k,\pi(x),\pi(y)) \theta(y) \,\mathrm{d}k.$$

Proof. Since both $\phi, \theta \in \mathcal{P}$, there are standard fundamental domains W_{ϕ} and $W_{\theta} \subset \mathbb{R}^d$ so that

$$\operatorname{supp}(\phi) \subset \mathring{W}_{\phi}, \quad \operatorname{supp}(\theta) \subset \mathring{W}_{\theta}.$$

Then, it suffices to show that $\langle \phi T_s \theta f,g\rangle$ equals

$$\frac{1}{(2\pi)^d} \int_{W_{\phi}} \int_{W_{\theta}} \int_{\mathcal{O}} e^{ik \cdot (x-y)} (\phi \overline{g})(x) K_s(k, \pi(x), \pi(y))(\theta f)(y) \, \mathrm{d}k \, \mathrm{d}y \, \mathrm{d}x,$$

for any f, g in $C^{\infty}(\mathbb{R}^d)$.

We observe that

$$\begin{split} &\langle \phi T_s \theta f, g \rangle = \langle \mathcal{F} \phi T_s \theta f, \mathcal{F} g \rangle \\ = & \frac{1}{(2\pi)^d} \left\langle (\mathcal{F} \phi \mathcal{F}^{-1}) \left(\int_{\mathcal{O}}^{\oplus} T_s(k) \, \mathrm{d}k \right) \mathcal{F}(\theta f), \mathcal{F} g \right\rangle \\ = & \frac{1}{(2\pi)^d} \left\langle \left(\int_{\mathcal{O}}^{\oplus} T_s(k) \, \mathrm{d}k \right) \mathcal{F}(\theta f), \mathcal{F}(\overline{\phi} g) \right\rangle. \end{split}$$

Since $\theta \in \mathcal{P}$, for any y in W_{θ} , we have

$$\mathcal{F}(\theta f)(k, \pi(y)) = (\theta f)(y)e^{-ik \cdot y}.$$

Similarly,

$$\mathcal{F}(\overline{\phi}g)(k,\pi(x)) = (\overline{\phi}g)(x)e^{-ik\cdot x}, \quad \forall x \in W_{\phi}.$$

We also have

$$\left(\int_{\mathcal{O}}^{\oplus} T_s(k) \,\mathrm{d}k\right) (\mathcal{F}(\theta f))(k, \pi(x)) = T_s(k)(\mathcal{F}(\theta f)(k, \cdot))(\pi(x)).$$

Consequently,

$$\left\langle \left(\int_{\mathcal{O}}^{\oplus} T_s(k) \, \mathrm{d}k \right) \mathcal{F}(\theta f), \mathcal{F}(\overline{\phi}g) \right\rangle$$

= $\int_{\mathcal{O}} \int_{W_{\phi}} T_s(k) (\mathcal{F}(\theta f)(k, \cdot))(\pi(x)) \overline{(\phi g)(x)} e^{-ik \cdot x} \, \mathrm{d}x \, \mathrm{d}k$
= $\int_{\mathcal{O}} \int_{W_{\phi}} \int_{W_{\theta}} K_s(k, \pi(x), \pi(y)) \mathcal{F}(\theta f)(k, \pi(y))(\phi \overline{g})(x) e^{ik \cdot x} \, \mathrm{d}y \, \mathrm{d}x \, \mathrm{d}k$
= $\int_{\mathcal{O}} \int_{W_{\phi}} \int_{W_{\theta}} e^{ik \cdot (x-y)} K_s(k, \pi(x), \pi(y))(\theta f)(y)(\phi \overline{g})(x) \, \mathrm{d}y \, \mathrm{d}x \, \mathrm{d}k.$

Using Fubini's theorem to rewrite the above integral, we have the desired identity. \Box

Proposition 3.7.16. Consider any two smooth compactly supported functions ϕ and θ on \mathbb{R}^d such that their supports are disjoint. Then the kernel $K_{s,\phi,\theta}(x,y)$ is continuous on $\mathbb{R}^d \times \mathbb{R}^d$ and moreover, it satisfies the following decay:

$$\sup_{s} |K_{s,\phi,\theta}(x,y)| \le C_N |\phi(x)\theta(y)| \cdot |x-y|^{-N},$$

for any N > d - 2. Here, the constant C_N is independent of ϕ and θ .

Proof. By using partitions of unity, any smooth compactly supported function can be written as a finite sum of smooth functions in the set \mathcal{P} . Thus, we can assume without loss of generality that both ϕ and θ belong to \mathcal{P} .

First, observe that for any $(k, n) \in \mathcal{O} \times \mathbb{Z}^d$,

$$T_s(k+2\pi n) = \mathcal{M}_n^{-1} T_s(k) \mathcal{M}_n,$$

where \mathcal{M}_n is the multiplication operator on $L^2(\mathbb{T}^d)$ by the exponential function $e^{2\pi i n \cdot x}$. Hence,

$$\nabla_k^{\alpha} K_s(k+2\pi n, \pi(x), \pi(y)) = e^{-2\pi i n \cdot \pi(x)} \nabla_k^{\alpha} K_s(k, \pi(x), \pi(y)) e^{2\pi i n \cdot \pi(y)},$$

for any multi-index α . Since $e^{2\pi i n \cdot x} = e^{2\pi i n \cdot \pi(x)}$ for any $x \in \mathbb{R}^d$, we obtain

$$e^{i(k+2\pi n)\cdot(x-y)}\nabla_k^{\alpha}K_s(k+2\pi n,\pi(x),\pi(y)) = e^{ik\cdot(x-y)}\nabla_k^{\alpha}K_s(k,\pi(x),\pi(y)).$$
(3.71)

Applying Lemma 3.7.15, we then use integration by parts (all boundary terms vanish when applying integration by parts due to (3.71)) to derive that for any $|\alpha| = N$,

$$(2\pi)^d (i(x-y))^{\alpha} K_{s,\phi,\theta}(x,y) = \phi(x)\theta(y) \int_{\mathcal{O}} e^{ik \cdot (x-y)} \nabla_k^{\alpha} K_s(k,\pi(x),\pi(y)) \,\mathrm{d}k$$

Suppose N > d - 2. Then by applying Corollary 3.7.12, the above integral is absolutely convergent and it is also uniformly bounded in (s, x, y). Consequently, the kernel $K_{s,\phi,\theta}(x, y)$ is continuous. Furthermore,

$$\sup_{s} |K_{s,\phi,\theta}(x,y)| \lesssim |\phi(x)\theta(y)| \cdot \min_{|\alpha|=N} |(x-y)^{\alpha}|^{-1} \lesssim |\phi(x)\theta(y)| \cdot |x-y|^{-N}.$$

We now have enough tools to approach our goal:

Proof of Theorem 3.5.2.

Proof. Let us fix a point (s, x) in $\mathbb{S}^{d-1} \times \mathbb{R}^d$. Now we consider a point y = x + st, where t is a real number. When |t| > 0, we can choose two cut-off functions ϕ and θ such that ϕ and θ equal 1 on some neighborhoods of x and y, respectively, and also, the supports of

these two functions are disjoint. Then, Proposition 3.7.16 implies that the kernel $K_s(x, y)$ is continuous at (x, y) since it coincides with $K_{s,\phi,\theta}$ on a neighborhood of (x, y). This yields the first statement about the continuity off diagonal of K_s . Again, by Proposition 3.7.16, we obtain

$$\sup_{s} |K_{s}(x,y)| = \sup_{s} |K_{s,\phi,\theta}(x,y)| \le C_{N} |x-y|^{-N},$$

which proves the last statement.

3.8 Some results on parameter-dependent toroidal Ψ DOs

The aim in this section is to provide some results needed to complete the proof of Theorem 3.7.6. We adopt the approach of [36] to periodic elliptic differential operators.

The next two theorems are straightforward modifications of the proofs for non-parameter toroidal Ψ DOs:

Theorem 3.8.1. (*The asymptotic summation theorem*) Given families of symbols $b_l \in \tilde{S}^{m-l}(\mathbb{T}^d)$, where each family $b_l = \{b_l(s,k)\}_{(s,k)}$ for l = 0, 1, ..., there exists a family of symbols b in $\tilde{S}^m(\mathbb{T}^d)$ such that

$$\{b(s,k) - \sum_{i < l} b_i(s,k)\}_{(s,k)} \in \tilde{S}^{m-l}(\mathbb{T}^d).$$
(3.72)

We will write $b \sim \sum_{l} b_{l}$ if b satisfies (3.72).

Proof. Step 1. Let $n = m + \epsilon$ for some $\epsilon > 0$. Then

$$|b_l(s,k;x,\xi)| \le C_l(1+|\xi|)^{m-l} = \frac{C_l(1+|\xi|)^{n-l}}{(1+|\xi|)^{\epsilon}}.$$

Thus, there is a sequence $\{\eta_l\}_{l\geq 1}$ such that $\eta_l \to +\infty$ and

$$|b_l(s,k;x,\xi)| < \frac{1}{2^l}(1+|\xi|)^{n-l}$$

for $|\xi| > \eta_l$. Let $\rho \in C^{\infty}(\mathbb{R})$ satisfy that $0 \le \rho \le 1$, $\rho(t) = 0$ whenever |t| < 1 and $\rho(t) = 1$ whenever |t| > 2. We define:

$$b(s,k;x,\xi) = \sum_{l} \rho\left(\frac{|\xi|}{\eta_l}\right) b_l(s,k;x,\xi).$$

Since only a finite number of summands are non-zero on any compact subset of $\mathbb{T}^d \times \mathbb{R}^d$, $b(s, \cdot; \cdot, \cdot) \in C^{\infty}(\mathcal{O} \times \mathbb{T}^d \times \mathbb{R}^d)$. Moreover, $b(s, k) - \sum_{r < l} b_r(s, k)$ is equal to:

$$\sum_{r < l} \left(\rho\left(\frac{|\xi|}{\eta_r}\right) - 1 \right) b_r(s,k) + \rho\left(\frac{|\xi|}{\eta_l}\right) b_l(s,k) + \sum_{r > l} \rho\left(\frac{|\xi|}{\eta_r}\right) b_r(s,k).$$

The first summand is compactly supported while the second summand is in $S^{m-l}(\mathbb{T}^d)$. Now let $\epsilon < 1$. Then, the third summand is bounded from above by

$$\sum_{r>l} \frac{1}{2^r} (1+|\xi|)^{n-r} \le (1+|\xi|)^{n-l-1} \le (1+|\xi|)^{m-l}.$$

Consequently,

$$\sup_{s \in \mathbb{S}^{d-1}} \left| b(s,k) - \sum_{r < l} b_r(s,k) \right| \le C(1+|\xi|)^{m-l}.$$

Step 2. For $|\alpha| + |\beta| + |\gamma| \le N$, one can choose η_l such that

$$\sup_{s \in \mathbb{S}^{d-1}} \left| D_k^{\alpha} D_{\xi}^{\beta} D_x^{\gamma} b_l(s,k;x,\xi) \right| \le \frac{1}{2^l} (1+|\xi|)^{n-l-|\alpha|-|\beta|}$$

for $\eta_l < |\xi|$. The same argument as in Step 1 implies that

$$\sup_{s \in \mathbb{S}^{d-1}} \left| D_k^{\alpha} D_{\xi}^{\beta} D_x^{\gamma}(b(s,k) - \sum_{r < l} b_r(s,k)) \right| \le C_N (1 + |\xi|)^{m-l-|\alpha|-|\beta|}.$$
(3.73)

Step 3. The sequence of η_l 's in Step 2 depends on N. We denote this sequence by $\eta_{l,N}$ to indicate this dependence on N. By induction, we can assume that for all $l, \eta_{l,N} \leq \eta_{l,N+1}$. Applying the Cantor diagonal process to this family of sequences, i.e., let $\eta_l = \eta_{l,l}$ then b has the property (3.73) for every N.

Theorem 3.8.2. (The composition formula) Let $a = \{a(s,k)\}$ be a family of symbols in $\tilde{S}^{l}(\mathbb{T}^{d})$ and $Q(x,D) = \sum_{|\alpha| \leq m} a_{\alpha}(x)D^{\alpha}$ be a differential operators of order $m \geq 0$ with smooth periodic coefficients $a_{\alpha}(x)$. Then the family of periodic pseudodifferential operators $\{Q(x, D + k + i\beta_{s})Op(a(s,k))\}_{(s,k)} \in Op(\tilde{S}^{l+m}(\mathbb{T}^{d}))$. Indeed, we have:

$$Q(x, D + k + i\beta_s)Op(a(s, k)) = Op((Q \circ a)(s, k)),$$

where

$$(Q \circ a)(s,k;x,\xi) = \sum_{|\alpha| \le m} \frac{1}{\alpha!} D_{\xi}^{\alpha} Q(x,\xi+k+i\beta_s) D_x^{\alpha} a(s,k;x,\xi)$$
(3.74)

Proof. The composition formula (3.74) is obtained for each (s, k) is standard in pseudodifferential operator theory (see e.g., [36, 67, 72]). We only need to check that the family of symbols $\{(Q \circ a)(s, k; x, \xi)\}_{(s,k)}$ is in $\tilde{S}^{l+m}(\mathbb{T}^d)$. But this fact follows easily from (3.74) and Leibnitz's formula.

We now finish the proof of Theorem 3.7.6.

Theorem 3.8.3. (*The inversion formula*) *There exists a family of symbols* $a = \{a(s,k)\}_{(s,k)}$

in $\tilde{S}^{-2}(\mathbb{T}^d)$ and a family of symbols $r = \{r(s,k)\}_{(s,k)}$ in $\tilde{S}^{-\infty}(\mathbb{T}^d)$ such that

$$(L_s(k) - \lambda)Op(a(s, k)) = I - Op(r(s, k))$$

Proof. Let

$$L_0(s,k;x,\xi) := \sum_{|\alpha|=2} a_\alpha(x)(\xi+k+i\beta_s)^\alpha,$$
$$\|a\|_\infty := \sum_{|\alpha|=2} \|a_\alpha(\cdot)\|_{L^\infty(\mathbb{T}^d)},$$

and

$$M := \max_{(s,k) \in \mathbb{S}^{d-1} \times \mathcal{O}} \left(|k|^2 + \theta^{-1} ||a||_{\infty} |\beta_s|^2 + \theta^{-1} \right),$$

where θ is the ellipticity constant in (3.2). Whenever $|\xi| > (2M)^{1/2}$,

$$\begin{aligned} |L_0(s,k;x,\xi)| &\ge \Re(L_0(s,k;x,\xi)) \ge \theta |\xi+k|^2 - \sum_{|\alpha|=2} a_\alpha(x) (\beta_s)^\alpha \\ &\ge \theta \left(\frac{|\xi|^2}{2} - |k|^2\right) - \|a\|_\infty |\beta_s|^2 > 1 \end{aligned}$$

Let $\rho \in C^{\infty}(\mathbb{R})$ be a function satisfying $\rho(t) = 0$ when $|t| < (2M)^{1/2}$ and $\rho(t) = 1$ when $|t| > 2M^{1/2}$. We define the function

$$a_0(s,k)(x,\xi) = \rho(|\xi|) \frac{1}{L_0(s,k;x,\xi)}.$$
(3.75)

Then $a_0 := \{a_0(s,k)\}_{(s,k)}$ is well-defined and belongs to $\tilde{S}^{-2}(\mathbb{T}^d)$. The next lemma is the final piece we need to complete the proof of the theorem.

Lemma 3.8.4. (i) If $b = \{b(s,k)\}_{(s,k)} \in \tilde{S}^{l}(\mathbb{T}^{d})$ then $b - (L - \lambda) \circ (a_{0}b) \in \tilde{S}^{l-1}(\mathbb{T}^{d})$.

(ii) There exists a sequence of families of symbols $a_l = \{a_l(s,k)\}_{(s,k)}$ in $\tilde{S}^{-2-l}(\mathbb{T}^d), l = 0, 1, \ldots$ and a sequence of families of symbols $r_l = \{r_l(s,k)\}_{(s,k)}$ in $\tilde{S}^{-l}(\mathbb{T}^d), l = 0, 1, \ldots$

such that a_0 is the family of symbols in (3.75), $r_0(s,k) = 1$ for every (s,k) and for all l,

$$(L-\lambda) \circ a_l = r_l - r_{l+1}.$$

Proof. (i) Let $p(s,k) = (L(s,k) - \lambda)(x,\xi) - L_0(s,k;x,\xi)$ so that $p = \{p(s,k)\}_{(s,k)} \in \tilde{S}^1(\mathbb{T}^d)$ and hence, $p \circ (a_0b)$ is in $\tilde{S}^{l-1}(\mathbb{T}^d)$ due to Theorem 3.8.2. Moreover, $b - L_0a_0b = (1 - \rho(|\xi|))b$ is a family of symbols whose ξ -supports are compact and thus it is in $\tilde{S}^{-\infty}(\mathbb{T}^d)$. We can now derive again from the composition formula (3.74) when $P := L_0$ that

$$(L - \lambda) \circ (a_0 b) = L_0 \circ (a_0 b) + p \circ (a_0 b) = L_0 a_0 b + \dots = b + \dots,$$

where the dots are the terms in $\tilde{S}^{l-1}(\mathbb{T}^d)$.

(ii) Recursively, let $a_l = a_0 r_l$ and $r_{l+1} = r_l - (L - \lambda) \circ a_l$. By part (i), $r_{l+1} \in \tilde{S}^{-(l+1)}(\mathbb{T}^d)$.

Now let a be the asymptotic sum of the families of symbols a_l , i.e., $a \sim \sum_l a_l$. Then

$$(L-\lambda) \circ a \sim \sum_{l} (L-\lambda) \circ a_{l} = \sum_{l} r_{l} - r_{l+1} = r_{0} = 1$$

which implies that $1-(L-\lambda)\circ a \sim 0$. In other words, this means that $r := 1-(L-\lambda)\circ a \in \tilde{S}^{-\infty}(\mathbb{T}^d)$. Hence, there exists a family of symbols a in $\tilde{S}^{-2}(\mathbb{T}^d)$ and a family of symbols r in $\tilde{S}^{-\infty}(\mathbb{T}^d)$ satisfying $(L-\lambda)\circ a = 1-r$. Finally, an application of Theorem 3.8.2 completes the proof of Theorem 3.7.6.

3.9 Some auxiliary statements

3.9.1 A lemma on the principle of non-stationary phase

Lemma 3.9.1. Let M be a compact manifold (with or without boundary) and $a : \mathbb{R} \times M \to \mathbb{C}$ be a smooth function with compact support. Then for any N > 0, there exists a constant

 $C_N > 0$ so that the following estimate holds for any non-zero $t \in \mathbb{R}$:

$$\sup_{x \in M} \left| \int_{-\infty}^{\infty} e^{ity} a(y, x) \,\mathrm{d}y \right| \le C_N |t|^{-N}.$$
(3.76)

Here C_N depends only on N, the diameter R of the y-support of a and $\sup_{x,y} |\partial_y^N a|$.

Proof. Let $t \neq 0$. Applying integration by parts repeatedly (*N*-times), it follows that

$$\left|\int_{-\infty}^{\infty} e^{ity} a(y,x) \,\mathrm{d}y\right| = |t|^{-N} \left|\int_{-\infty}^{\infty} e^{ity} \partial_y^N a(y,x) \,\mathrm{d}y\right| \le R \sup_{x,y} |\partial_y^N a| \cdot |t|^{-N}.$$

3.9.2 The Weierstrass preparation theorem

Theorem 3.9.2. Let f(t, z) be an analytic function of $(t, z) \in \mathbb{C}^{1+n}$ in a neighborhood of (0, 0) such that (0, 0) is a simple zero of f, i.e.:

$$f(0,0) = 0, \quad \frac{\partial f}{\partial t}(0,0) \neq 0.$$

Then there is a unique factorization

$$f(t,z) = (t - A(z))B(t,z),$$

where A, B are analytic in a neighborhood of 0 and (0,0) respectively. Moreover, $B(0,0) \neq 0$ and A(0) = 0.

The proof of a more general version of this theorem could be found in [38, Theorem 7.5.1].

3.9.3 Proofs of Proposition 3.4.1 and Lemma 3.5.1

Remark 3.9.3. Consider a domain \mathcal{D} of \mathbb{C}^d and let $f : \mathcal{D} \to \mathbb{C}$ be a holomorphic function. For $z \in \mathbb{C}^d$, write z = x + iy where $x, y \in \mathbb{R}^d$. Now we fix a vector β in \mathbb{R}^d and denote $\mathcal{D}_{\beta} = (\mathcal{D} - i\beta) \cap \mathbb{R}^d$. If this intersection is non-empty, we may consider the restriction $k \to f(k + i\beta)$ as a real analytic function defined on a subdomain \mathcal{D}_{β} of \mathbb{R}^d . Thanks to Cauchy-Riemann equations of f, we do not need to make any distinction between derivatives of f with respect to x (when f is viewed as a real analytic one) or z(when f is considered as a complex analytic one) at every point in \mathcal{D}_{β} since

$$\frac{\partial f}{\partial x_l}(k+i\beta) = \frac{\partial f}{\partial z_l}(k+i\beta) = -i\frac{\partial f}{\partial y_l}(k+i\beta), \quad 1 \le l \le d.$$

For higher order derivatives, we use induction and the above identity to obtain

$$\partial_x^{\alpha} f(k+i\beta) = \partial_z^{\alpha} f(k+i\beta) = (-i)^{|\alpha|} \partial_u^{\alpha} f(k+i\beta),$$

for any multi-index α . We use these facts implicitly for the function λ_j . When dealing with the analytic function $f = \lambda_j$ in this part, denote $\partial^{\alpha} \lambda_j$ to indicate either its x or z-derivatives.

We also want to mention this simple relation between derivatives of λ_j and E:

$$\partial^{\alpha} E(\beta) = \partial^{\alpha}_{\mu} \lambda_{i}(k_{0} + i\beta) = i^{|\alpha|} \partial^{\alpha} \lambda_{i}(k_{0} + i\beta).$$

Proof of Proposition 3.4.1.

Proof. We recall from Section 2 that V is an open neighborhood of k_0 in \mathbb{C}^d such that the properties (**P1**)-(**P6**) are satisfied. Note that V depends only on the local structure at k_0 of the dispersion branch λ_j of L. Denote $\mathcal{O}_s = \{k + it\beta_s : k \in \mathcal{O}, t \in [0, 1]\}$ for each

 $s \in \mathbb{S}^{d-1}$. For C > 0 (which is defined later), set $M_{s,C} = \mathcal{O}_s \cap \{z \in \mathbb{C}^d : |\Re(z) - k_0| < C\}$ and $N_{s,C} = \mathcal{O}_s \setminus M_{s,C}$ (e.g., see Figure 3.1). For C and $|\lambda|$ small enough, we can suppose $M_{s,C} \in V$ since β_s is small too. We also assume that $|\lambda| \leq \epsilon_0.^8$



Figure 3.1: An illustration of the regions $M_{s,C}$ and $N_{s,C}$ when $k_0 \equiv 0$.

For any point $z = k + it\beta_s \in M_{s,C}$, we want to show if $\lambda \in \sigma(L(z))$, it forces $z = k_0 + i\beta_s$. By (P3), this is the same as showing the equation $\lambda_j(z) = \lambda$ has no solution z in $M_{s,C}$ except for the trivial solution $z = k_0 + i\beta_s$. Suppose for contradiction $\lambda_j(k + it\beta_s) = \lambda = \lambda(\beta_s)$ for some $t \in [0, 1]$ and k in $\{k \in \mathcal{O} \mid 0 < |k - k_0| < C\}$. By Taylor expanding around $k_0 + it\beta_s$, there is some $\gamma \in (0, 1)$ such that

$$\lambda - \lambda_j (k_0 + it\beta_s) = \left((k - k_0) \cdot \nabla \lambda_j (k_0 + it\beta_s) + \sum_{|\alpha|=3} \frac{(k - k_0)^{\alpha}}{\alpha!} \partial^{\alpha} \lambda_j (k_0 + it\beta_s) \right) + \frac{1}{2} (k - k_0) \cdot \operatorname{Hess} (\lambda_j) (k_0 + it\beta_s) (k - k_0) + \sum_{|\alpha|=4} \frac{(k - k_0)^{\alpha}}{\alpha!} \partial^{\alpha} \lambda_j (\gamma (k - k_0) + k_0 + it\beta_s).$$
(3.77)

⁸Recall the definition of ϵ_0 from (**P3**).

If $|\alpha|$ is odd, then by Remark 3.9.3 and the fact that E is real, we have

$$\partial^{\alpha}\lambda_j(k_0 + it\beta_s) = \frac{1}{i^{|\alpha|}}\partial^{\alpha}E(t\beta_s) \in i\mathbb{R}.$$

Taking the real part of the equation (3.77) to get

$$E(\beta_s) - E(t\beta_s) = -\frac{1}{2}(k - k_0) \cdot \operatorname{Hess}(E)(t\beta_s)(k - k_0) + \sum_{|\alpha|=4} \frac{(k - k_0)^{\alpha}}{\alpha!} \times \Re(\partial^{\alpha}\lambda_j(\gamma(k - k_0) + k_0 + it\beta_s)).$$

The left-hand side is bounded above by $(1 - t)\lambda \leq 0$ because of the concavity of E $(E(t\beta_s) \geq tE(\beta_s) = t\lambda)$. On the other hand, by (P5),

$$-\frac{1}{2}(k-k_0) \cdot \operatorname{Hess}\left(E\right)(t\beta_s)(k-k_0) \ge \frac{1}{4}|k-k_0|^2 \min \sigma(\operatorname{Hess}\left(\lambda_j\right)(k_0))$$
$$\sum_{|\alpha|=4} \frac{(k-k_0)^{\alpha}}{\alpha!} \Re\left(\partial^{\alpha}\lambda_j(\gamma(k-k_0)+k_0+it\beta_s)\right) \le C(d)|k-k_0|^4 \max_{z\in\overline{V},|\alpha|=4}|\partial^{\alpha}\lambda_j(z)|.$$

We simply choose $C^2 < \frac{\min \sigma(\operatorname{Hess} (\lambda_j)(k_0))}{C(d) \max_{z \in \overline{V}, |\alpha|=4} |\partial^{\alpha} \lambda_j(z)|}$ to get a contradiction if $k \neq k_0$.

For the remaining part, we just need to treat points $k + it\beta_s$ in $N_{s,C}$. We have $\lambda \in \rho(L(k)), \forall k \in \mathbb{R}^d$. The idea is to adapt the upper-semicontinuity of the spectrum of an analytic family of type \mathcal{A} on \mathbb{C}^d , following [39]. For any $k \in \mathcal{O}$ and $z \in \mathbb{C}^d$, the composed operators $(L(k+z) - L(k))(L(k) - \lambda)^{-1}$ are closed and defined on $L^2(\mathbb{T}^d)$ and by closed graph theorem, these are bounded operators. Clearly,

$$L(k+z) - \lambda = (1 + (L(k+z) - L(k))(L(k) - \lambda)^{-1})(L(k) - \lambda).$$

Thus, λ is in the resolvent of L(k+z) if the operator $1 + (L(k+z) - L(k))(L(k) - \lambda)^{-1}$ is invertible. Hence, it is enough to show that there is some positive constant τ such that for any $k \in \mathcal{O}$ and $|z| < \tau$,

$$\|(L(k+z) - L(k))(L(k) - \lambda)^{-1}\|_{op} < 1/2, \quad |k - k_0| \ge C,$$
(3.78)

where the operator norm on $L^2(\mathbb{T}^d)$ is denoted by $\|\cdot\|_{op}$. Indeed, if $|\lambda|$ is small enough so that we have $\max_{s\in\mathbb{S}^{d-1}}|\beta_s| < \tau$ and then (3.78) implies that $\lambda \in \rho(L(k+it\beta_s))$ for any $t \in [0,1]$.

Finally, we will use some energy estimates of linear elliptic equations and spectral theory to obtain (3.78). Observe that,

$$L(k+z) - L(k) = z \cdot A(x)(D+k) + (D+k) \cdot A(x)z + z \cdot A(x)z.$$

For $v \in H^1(\mathbb{T}^d)$ and |z| < 1, there is some constant $C_1 > 0$ (independent of z) such that

$$\|(z \cdot A(x)(D+k) + (D+k) \cdot A(x)z + z \cdot A(x)z)v\|_{L^2(\mathbb{T}^d)} \le C_1 |z| \cdot \|v\|_{H^1(\mathbb{T}^d)}.$$
 (3.79)

Set $v := (L(k) - \lambda)^{-1}u$ for $u \in L^2(\mathbb{T}^d)$. Ellipticity of L(k) yields $v \in H^2(\mathbb{T}^d)$ and in particular, we obtain (3.79) for such v. Testing the equation $(L(k) - \lambda)v = u$ with the function v, we derive the standard energy estimate

$$||Dv||_{L^2(\mathbb{T}^d)} \le C_2(||v||_{L^2(\mathbb{T}^d)} + ||u||_{L^2(\mathbb{T}^d)}).$$
(3.80)

Note that both C_1 and C_2 in (3.79) and (3.80) are independent of k and λ since we take k in the bounded set \mathcal{O} and consider $|\lambda|$ to be small enough.

Suppose that $|\lambda|$ is less than one-half of the length of the gap between the dispersion

branches λ_j and λ_{j-1} . Due to functional calculus of the self-adjoint operator L(k), we get

$$\|(L(k) - \lambda)^{-1}\|_{op} = \operatorname{dist}(\lambda, \sigma(L(k)))^{-1} = \min\{(\lambda_j(k) - \lambda), (\lambda - \lambda_{j-1}(k))\}^{-1}.$$

Now let $\delta_1 = -\frac{1}{2} \max \lambda_{j-1}(k) > 0$ and $\delta_2 = \min_{k \in \mathcal{O}, |k-k_0| \ge C} \lambda_j(k)$. Then due to A3, $\delta_2 > 0$. Moreover,

$$\lambda - \lambda_{j-1}(k) \ge \lambda - \max_{k \in \mathcal{O}} \lambda_{j-1}(k) > \delta_1,$$

and

$$\lambda_j(k) - \lambda \ge \min_{k \in \mathcal{O}, |k-k_0| \ge C} \lambda_j(k) - \lambda > \delta_2.$$

Hence,

$$\|(L(k) - \lambda)^{-1}\|_{op} < \delta := \min\{\delta_1, \delta_2\}^{-1}.$$
(3.81)

In other words, $||v||_{L^2(\mathbb{T}^d)} \leq \delta ||u||_{L^2(\mathbb{T}^d)}$. Applying this fact together with (3.79) and (3.80), we have

$$\begin{aligned} \| (L(k+z) - L(k))(L(k) - \lambda)^{-1} u \|_{L^{2}(\mathbb{T}^{d})} &\leq |z| C_{1} \|v\|_{H^{1}(\mathbb{T}^{d})} \\ &\leq |z| C_{1} C_{2}(\|v\|_{L^{2}(\mathbb{T}^{d})} + \|u\|_{L^{2}(\mathbb{T}^{d})}) \\ &\leq |z| C_{1} C_{2}(1+\delta) \|u\|_{L^{2}(\mathbb{T}^{d})}. \end{aligned}$$

Now (3.78) is a consequence of the above estimate if we let

$$\tau \le \min\left(\frac{1}{2C_1C_2(1+\delta)}, 1\right).$$

Proof of Lemma 3.5.1.

Proof. From Proposition 2.2.6, the complex Bloch variety $\Sigma := B_L$ of the operator L is

an analytic subset of codimension one in \mathbb{C}^{d+1} . By [47, 80], there exist an entire scalar function $h(k, \mu)$ and an entire operator-valued function $I(k, \mu)$ on \mathbb{C}^{d+1} such that

(1) h vanishes only on Σ and has simple zeros on Σ , i.e., its normal derivative is not zero at all smooth parts of Σ .

(2) In $\mathbb{C}^{d+1} \setminus \Sigma$, $(L(k) - \mu)^{-1} = h(k, \mu)^{-1} I(k, \mu)$.

In particular, $(L_{t,s}(k) - \lambda)^{-1} = h(k + it\beta_s, \lambda)^{-1}I(k + it\beta_s, \lambda)$ for $k \in \mathbb{R}^d$ and $t \in [0, 1)$ by Proposition 3.4.1. Due to **Assumption A** and **(P2)**, if $k_0 + it\beta_s \in V$, the k-variable function $h(k, \lambda)^{-1}$ is equal (up to a non-vanishing analytic factor) to $(\lambda_j(k + it\beta_s) - \lambda)^{-1}$ on an open disc $D(k_0, 2\varepsilon) \subseteq V$ in \mathbb{C}^d for some $\varepsilon > 0$. Hence, we can write the sesquilinear form for such values of k as

$$(R_{t,s,\lambda}f,\varphi) = R_1 + R_2,$$

where

$$R_1 = (2\pi)^{-d} \int_{\mathcal{O} \cap D(k_0,\varepsilon)} \frac{\left(M(k,\lambda)\widehat{f}(k),\widehat{\varphi}(k)\right)}{\lambda_j(k+it\beta_s) - \lambda} \,\mathrm{d}k$$

and

$$R_2 = (2\pi)^{-d} \int_{\mathcal{O} \setminus D(k_0,\varepsilon)} \left(L(k+it\beta_s) - \lambda)^{-1} \widehat{f}(k), \widehat{\varphi}(k) \right) \mathrm{d}k.$$

Here $M(k, \lambda)$ is a $L^2(\mathbb{T}^d)$ -valued analytic function on $D(k_0, \varepsilon)$ when $|\lambda|$ is small. Since f and φ have compact supports, their Floquet transforms $\widehat{f}(k), \widehat{\varphi}(k)$ are analytic with respect to k. To prove the equality (3.15), we apply the Lebesgue Dominated Convergence Theorem. For R_1 , it suffices to show that the denominator in the integrand when $t \to 1^-$ is integrable over $D(k_0, \varepsilon)$ for $d \ge 2$. Indeed,

$$\left|\lambda_{j}(k+i\beta_{s})-\lambda\right| \geq \delta \left|i\nabla E(\beta_{s})\cdot(k-k_{0})-\frac{1}{2}(k-k_{0})\cdot\operatorname{Hess}\left(E\right)(\beta_{s})(k-k_{0})\right|$$

for some $\delta > 0$ if ε is chosen small enough so that in the Taylor expansion of λ_j at $k_0 + i\beta_s$, the remainder term $O(|k-k_0|^3)$ is dominated by the quadratic term $|k-k_0|^2$. Furthermore,

$$\left| i \nabla E(\beta_s) \cdot (k - k_0) - \frac{1}{2} (k - k_0) \cdot \operatorname{Hess}(E)(\beta_s)(k - k_0) \right|^2 \ge C(|\langle k - k_0, s \rangle|^2 + |k - k_0|^4),$$

for some constant C > 0 (independent of k). Now let $v := (k - k_0)$ and so the right hand side of the above estimate is just $|\langle v, s \rangle|^2 + |v|^4$ (up to a constant factor). One can apply Hölder's inequality to obtain

$$|\langle v, s \rangle|^2 + |v|^4 \ge |\langle v, s \rangle|^2 + |v'|^4 \ge C |\langle v, s \rangle|^{3/2} |v'|,$$

where $v = (v_1, v') \in \mathbb{R} \times \mathbb{R}^{d-1}$. Thus, we deduce

$$|\lambda_j(k+i\beta_s) - \lambda|^{-1} \le C |\langle v, s \rangle|^{-3/4} |v'|^{-1/2}.$$
(3.82)

Since the function $|x|^{-n}$ is integrable near 0 in \mathbb{R}^d if and only if n < d, $|v'|^{-1/2}$ and $|\langle v, s \rangle|^{-3/4}$ are integrable near 0 in \mathbb{R}^{d-1} and \mathbb{R} respectively. Therefore, the function in the right hand side of (3.82) is integrable near 0.

The integrability of R_2 as $t \to 1^-$ follows from the estimation (3.78) in the proof of Proposition 3.4.1. Indeed,

$$\|(L(k+it\beta_{s})-\lambda)^{-1}\|_{op} = \|(1-(L(k+it\beta_{s})-L(k))(\lambda-L(k))^{-1})^{-1}(\lambda-L(k))^{-1}\|_{op}$$

$$\leq \frac{\|(L(k)-\lambda)^{-1}\|_{op}}{1-\|(L(k+it\beta_{s})-L(k))(\lambda-L(k))^{-1}\|_{op}}.$$
(3.83)

By decreasing $|\lambda|$, if necessary, and repeating the arguments when showing (3.78) and

(3.81) we derive:

$$1 - \|(L(k + it\beta_s) - L(k))(\lambda - L(k))^{-1}\|_{op} \ge 1/2, \quad \forall k \in \mathcal{O} \setminus D(k_0, \varepsilon)$$
(3.84)

and

$$\sup_{k \in \mathcal{O} \setminus D(k_0,\varepsilon)} \| (L(k) - \lambda)^{-1} \|_{op} < \infty.$$
(3.85)

Thanks to (3.83), (3.84), (3.85), Cauchy-Schwarz inequality and Lemma 2.2.4, we have:

$$\sup_{t\in[0,1]} \left| \left(L(k+it\beta_s) - \lambda)^{-1} \widehat{f}(k), \widehat{\varphi}(k) \right) \right| \leq 2 \| (L(k) - \lambda)^{-1} \|_{op} \cdot \| \widehat{f}(k) \|_{L^2(\mathbb{T}^d)} \| \widehat{\varphi}(k) \|_{L^2(\mathbb{T}^d)} \\ \lesssim \| \widehat{f}(k) \|_{L^2(\mathbb{T}^d)} \| \widehat{\varphi}(k) \|_{L^2(\mathbb{T}^d)} \quad , \forall k \in \mathcal{O} \setminus D(k_0, \varepsilon)$$

and

$$\int_{\mathcal{O}\setminus D(k_0,\varepsilon)} \|\widehat{f}(k)\|_{L^2(\mathbb{T}^d)} \|\widehat{\varphi}(k)\|_{L^2(\mathbb{T}^d)} \,\mathrm{d}k \le \|f\|_{L^2(\mathbb{R}^d)} \|\varphi\|_{L^2(\mathbb{R}^d)} < \infty$$

This completes the proof of our lemma.

3.9.4 Regularity of eigenfunctions $\phi(z, x)$

In this subsection, we study the regularity of the eigenfunctions $\phi(z, x)$ of the operator L(z) with corresponding eigenvalue $\lambda_j(z)$ (see (P4)). It is known that for each $z \in V$, the eigenfunction $\phi(z, x)$ is smooth in x. We will claim that these eigenfunctions are smooth in (z, x) when z is near to k_0 . The idea is that initially, $\phi(z, \cdot)$ is an analytic section of the Hilbert bundle $V \times H^2(\mathbb{T}^d)$ and then by ellipticity, it is also an analytic section of the bundle $V \times H^m(\mathbb{T}^d)$ for any m > 0 (for statements related to Fredholm morphisms between analytic Banach bundles, see e.g., [82]) and hence smoothness will follow.

For the sake of completeness, we provide the proof of the above claim by applying standard bootstrap arguments in the theory of elliptic differential equations.

Lemma 3.9.4. The function $\partial_x^{\alpha} \phi(z, x)$ is jointly continuous on $V \times \mathbb{R}^d$ for any multi-index α .

Proof. By periodicity, it suffices to restrict x to \mathbb{T}^d . Let $K := \overline{V}$. Due to (P4), the function $z \mapsto \phi(z, \cdot)$ is a $H^2(\mathbb{T}^d)$ -valued analytic on some neighborhood of K. Thus,

$$\sup_{z \in K} \|\phi(z, \cdot)\|_{H^2(\mathbb{T}^d)} < \infty.$$

Then, we can apply bootstrap arguments for the equation

$$L(z)\phi(z,\cdot) = \lambda_j(z)\phi(z,\cdot)$$

to see that $M_m := \sup_{z \in K} \|\phi(z, \cdot)\|_{H^m(\mathbb{T}^d)}$ is finite for any nonnegative integer m.

Now we consider z and z' in K. Let $\phi_{z,z'}(x) := \phi(z,x) - \phi(z',x)$. Then, $\phi_{z,z'}$ is a (classical) solution of the equation

$$L(z)\phi_{z,z'} = f_{z,z'},$$

where $f_{z,z'} := (\lambda_j(z)\phi(z,\cdot) - \lambda_j(z')\phi(z',\cdot)) + (L(z') - L(z))\phi(z',\cdot).$

By induction, we will show that for any $m \ge 0$,

$$\|\phi_{z,z'}\|_{H^m(\mathbb{T}^d)} \lesssim |z-z'|.$$
 (3.86)

The case m = 0 is clear because (P4) implies that $z \mapsto \|\phi(z, \cdot)\|_{L^2(\mathbb{T}^d)}$ is Lipschitz continuous.

Next, we assume that the estimate (3.86) holds for m. As in (3.79),

$$\|(L(z) - L(z'))\phi(z', \cdot)\|_{H^m(\mathbb{T}^d)} \lesssim |z - z'| \cdot \|\phi(z', \cdot)\|_{H^{m+1}(\mathbb{T}^d)} \lesssim M_{m+1}|z - z'|.$$
(3.87)

Using triangle inequalities, the estimates (3.86), (3.87) and analyticity of λ_j , we get

$$\|f_{z,z'}\|_{H^{m}(\mathbb{T}^{d})} \lesssim \|\lambda_{j}(z)\phi(z,\cdot) - \lambda_{j}(z')\phi(z',\cdot)\|_{H^{m}(\mathbb{T}^{d})} + \|(L(z) - L(z'))\phi(z',\cdot)\|_{H^{m}(\mathbb{T}^{d})}$$

$$\lesssim |\lambda_{j}(z)| \cdot \|\phi_{z,z'}\|_{H^{m}(\mathbb{T}^{d})} + M_{m}|\lambda_{j}(z) - \lambda_{j}(z')| + M_{m+1}|z - z'|$$

$$\lesssim |z - z'|.$$

(3.88)

Notice that for any $m \ge 0$, the following standard energy estimate holds (see e.g., [22, 28, 53]):

$$\|\phi_{z,z'}\|_{H^{m+2}(\mathbb{T}^d)} \lesssim \|f_{z,z'}\|_{H^m(\mathbb{T}^d)} + \|\phi_{z,z'}\|_{L^2(\mathbb{T}^d)}.$$
(3.89)

Combining (3.88) and (3.89), we deduce that $\|\phi_{z,z'}\|_{H^{m+2}(\mathbb{T}^d)} \leq |z-z'|$. Hence, (3.86) holds for m+2. This finishes our induction.

Applying the Sobolev embedding theorem, $\|\phi_{z,z'}\|_{C^m(\mathbb{T}^d)} \lesssim |z-z'|$ for any $m \ge 0$. In other words, $\phi \in C(K, C^m(\mathbb{T}^d))$ for any m. Since $C(K \times \mathbb{T}^d) = C(K, C(\mathbb{T}^d))$, this completes the proof.

Notation 3.9.5. Consider a *z*-parameter family of linear partial differential operators $\{L(z)\}$ where $z \in \mathbb{R}^d$. Suppose $L(x, \xi, z)$ is the symbol of L(z). Whenever it makes sense, the differential operator $\frac{\partial L(z)}{\partial z_l}$ is the one whose symbol is $\frac{\partial L}{\partial z_l}(x, \xi, z)$ for any $l \in \{1, 2, ..., d\}$. Proposition 3.9.6. Assume D is an open disc centered at k_0 in \mathbb{R}^d such that $D \pm i\beta_s \Subset V$ for any $s \in \mathbb{S}^{d-1}$. Then all eigenfunctions $\phi(k \pm i\beta_s, x)$ are smooth on a neighborhood of $\overline{D} \times \mathbb{R}^d$. Furthermore, all derivatives of $\phi(k \pm i\beta_s, x)$ are bounded on $\overline{D} \times \mathbb{R}^d$ uniformly in *s*, *i.e.*, for any multi-indices α, β :

$$\sup_{(s,k,x)\in\mathbb{S}^{d-1}\times\overline{D}\times\mathbb{R}^d} |\partial_k^{\alpha}\partial_x^{\beta}\phi(k\pm i\beta_s,x)| < \infty.$$

Proof. Pick any open disc D' in \mathbb{R}^d so that $\overline{D} \pm i\beta_s \subset D' \pm i\beta_s \subseteq V$. We will prove that

all eigenfunctions are smooth on the domain $D' \times \mathbb{R}^d$. Also, it is enough to consider the function $\phi(k + i\beta_s)$ since the other one is treated similarly.

First, we show that $\frac{\partial \phi}{\partial k_l}(k + i\beta_s, x)$ is continuous for any $1 \leq l \leq d$. By Lemma 3.9.4, the function $(k, x) \mapsto \phi(k + i\beta_s, x)$ is continuous on $D' \times \mathbb{T}^d$. We consider any two complex-valued test functions $\varphi \in C_c^{\infty}(D')$ and $\psi \in C^{\infty}(\mathbb{T}^d)$. Testing the equation of the eigenfunction $\phi(k + i\beta_s, x)$ with ψ and $\frac{\partial \varphi}{\partial k_l}$, we derive

$$\int_{D'} \int_{\mathbb{T}^d} (L(k+i\beta_s) - \lambda_j(k+i\beta_s))\phi(k+i\beta_s, x)\overline{\psi(x)}\frac{\partial\varphi}{\partial k_l}(k) \,\mathrm{d}x \,\mathrm{d}k = 0.$$

Observe that $L(k + i\beta_s)^* = L(k - i\beta_s)$ and $\left(\frac{\partial L(k - i\beta_s)}{\partial k_l}\right)^* = \frac{\partial L(k + i\beta_s)}{\partial k_l}$. We integrate by parts to derive

$$0 = \int_{D'} \left(\left(L(k+i\beta_s) - \lambda_j(k+i\beta_s) \right) \phi(k+i\beta_s, x), \psi(x) \right)_{L^2(\mathbb{T}^d)} \frac{\partial \varphi}{\partial k_l}(k) \, \mathrm{d}k$$

$$= \int_{D'} \left(\phi(k+i\beta_s, x), \left(L(k-i\beta_s) - \overline{\lambda_j}(k+i\beta_s) \right) \psi(x) \right)_{L^2(\mathbb{T}^d)} \frac{\partial \varphi}{\partial k_l}(k) \, \mathrm{d}k$$

$$= \int_{D'} \left(-\frac{\partial \phi}{\partial k_l}(k+i\beta_s, x), \left(L(k-i\beta_s) - \overline{\lambda_j}(k+i\beta_s) \right) \psi(x) \right)_{L^2(\mathbb{T}^d)} \varphi(k) \, \mathrm{d}k$$

$$- \int_{D'} \left(\phi(k+i\beta_s, x), \frac{\partial L(k-i\beta_s)}{\partial k_l} \psi(x) - \frac{\partial \overline{\lambda_j}}{\partial k_l}(k+i\beta_s) \psi(x) \right)_{L^2(\mathbb{T}^d)} \varphi(k) \, \mathrm{d}k$$

$$= \int_{D'} \left(\left(-L(k+i\beta_s) + \lambda_j(k+i\beta_s) \right) \frac{\partial \phi}{\partial k_l}(k+i\beta_s, x), \psi(x) \right)_{L^2(\mathbb{T}^d)} \varphi(k) \, \mathrm{d}k$$

$$- \int_{D'} \left(\left(\frac{\partial L(k+i\beta_s)}{\partial k_l} - \frac{\partial \lambda_j}{\partial k_l}(k+i\beta_s) \right) \phi(k+i\beta_s, x), \psi(x) \right)_{L^2(\mathbb{T}^d)} \varphi(k) \, \mathrm{d}k.$$
(3.90)

We introduce

$$\begin{split} \phi_l(k,x) &:= \frac{\partial \phi}{\partial k_l} (k+i\beta_s,x), \\ G(k) &:= -L(k+i\beta_s) + \lambda_j (k+i\beta_s), \\ H(k,x) &:= \left(\frac{\partial L(k+i\beta_s)}{\partial k_l} - \frac{\partial \lambda_j}{\partial k_l} (k+i\beta_s)\right) \phi(k+i\beta_s,x). \end{split}$$

By invoking the previous lemma, the Lipschitz continuity of the $C^{m+2}(\mathbb{T}^d)$ -valued function $\phi(k + i\beta_s, \cdot)$ implies that the mapping $k \mapsto H(k, \cdot)$ must be Lipschitz as a $C^m(\mathbb{T}^d)$ valued function on $\overline{D'}$ for any $m \ge 0$. On the other hand, the $H^2(\mathbb{T}^d)$ -valued function $\phi_l(k, \cdot)$ is also Lipschitz on D' due to (**P4**). Hence, both $(G(k)\phi_l(k, \cdot), \psi)_{L^2(\mathbb{T}^d)}$ and $(H(k, \cdot), \psi)_{L^2(\mathbb{T}^d)}$ are continuous on D' for any test function ψ . The continuity let us conclude from (3.90) that for every $k \in D'$, $\phi_l(k, \cdot)$ is a weak solution of the equation

$$G(k)\phi_l(k,x) = H(k,x).$$
 (3.91)

We interpret (3.91) in the classical sense since all the coefficients of this equation are smooth. Consider any k_1, k_2 in D' and subtract the equation corresponding to k_1 from the one corresponding to k_2 to obtain the equation for the oscillation function $\phi_l(k_1, \cdot) - \phi_l(k_2, \cdot)$:

$$G(k_1)(\phi_l(k_1, x) - \phi_l(k_2, x)) = (G(k_2) - G(k_1))\phi_l(k_2, x) + (H(k_1, x) - H(k_2, x)).$$

Note that due to regularities of λ_j , H and the fact that the differential operator G(k) depends analytically on k, we get

$$\|H(k_1, \cdot) - H(k_2, \cdot)\|_{H^m(\mathbb{T}^d)} + \|(G(k_1) - G(k_2))\phi_l(k_2, \cdot)\|_{H^m(\mathbb{T}^d)} = O(|k_1 - k_2|), \quad \forall m \in \mathbb{N}.$$

Combining this with the uniform boundedness in k of the supremum norms of all coefficients of the differential operator $G(k_1)$, we obtain

$$\|\phi_l(k_1, \cdot) - \phi_l(k_2, \cdot)\|_{H^m(\mathbb{T}^d)} = O(|k_1 - k_2|),$$

by using energy estimates as in the proof of Lemma 3.9.4. An application of the Sobolev

embedding theorem shows that $\partial_x^\beta \phi_l(k, x)$ is continuous on $D' \times \mathbb{T}^d$ for any multi-index β .

To deduce continuity of higher derivatives $\partial_x^{\beta} \partial_k^{\alpha} \phi(k+i\beta_s)$ ($|\alpha| > 1$, $|\beta| \ge 0$), we induct on $|\alpha|$ and repeat the arguments of the $|\alpha| = 1$ case.

Finally, the last statement of this proposition also follows since all of our estimates hold uniformly in s.

Observation 3.9.7. 1. The property (P4) is crucial in order to bootstrap regularities of eigenfunctions $\phi(k \pm i\beta_s)$.

2. If one just requires $\phi(k \pm i\beta_s) \in C^m(\overline{D} \times \mathbb{R}^d)$ for certain m > 0 then the smoothness on coefficients of L could be relaxed significantly (see [28, 53]).

3.10 Concluding remarks

- 1. The condition that the potentials *A*, *V* are infinitely differentiable is an overkill. The Fredholm property of the corresponding Floquet operators is essential, which can be obtained under much weaker assumptions.
- 2. The main result of this chapter assumes the central symmetry (evenness) of the relevant branch of the dispersion curve $\lambda_j(k)$, which does not hold for instance for operators with periodic magnetic potentials [24, 69]. Note that the result of [52] at the spectral edge does not require such a symmetry. It seems that in the inside-thegap situation one also should not need such a symmetry. However, we have not been able to do so here, and thus were limited to the case of high symmetry points of the Brillouin zone.
- 3. In the case when λ is below the whole spectrum, the result of this chapter implies [60, Theorem 1.1] for self-adjoint operators.

4. GREEN'S FUNCTION ASYMPTOTICS OF PERIODIC ELLIPTIC OPERATORS ON ABELIAN COVERINGS OF COMPACT MANIFOLDS.

4.1 Introduction

Many classical properties of solutions of periodic Schrödinger operators on Euclidean spaces were generalized successfully to solutions of periodic Schrödinger operators on coverings of compact manifolds (see e.g., [3, 15, 16, 45, 50, 56, 74, 75]). Hence, a question arises of whether one can obtain analogs of the results of Chapter 3 and the paper [52] as well. The main theorems 4.3.1 and 4.3.4 of this chapter provide such results for operators on abelian coverings of compact Riemannian manifolds. The results are in line with Gromov's idea that the large scale geometry of a co-compact normal covering is captured mostly by its deck transformation group (see e.g., [17, 31, 68]). For instance, the dimension of the covering manifold does not enter explicitly to the asymptotics. Rather, the torsion-free rank d of the abelian deck transformation group influences these asymptotics significantly. One can find a similar effect in various results involving analysis on Riemannian co-compact normal coverings such as the long time asymptotic behaviors of the heat kernel on a noncompact abelian Riemannian covering [46], or the analogs of Liouville's theorem [50] (see also [68] for an excellent survey on analysis on co-compact coverings).

We discuss now the main thrust of this chapter.

Let X be a noncompact Riemannian manifold that is a normal abelian covering of a compact Riemannian manifold M with the deck transformation group G. For any function u on X and any $g \in G$, we denote by u^g the "shifted" function

$$u^g(x) = u(g \cdot x),$$

for any $x \in X$. Consider a bounded below second-order elliptic operator L on the manifold X with real and smooth coefficients. We assume that L is a **periodic** operator on X, i.e., the following invariance condition holds:

$$Lu^g = (Lu)^g,$$

for any $g \in G$ and $u \in C_c^{\infty}(X)$. The operator L, with the Sobolev space $H^2(X)$ as its domain, is an unbounded self-adjoint operator in $L^2(X)$.

As one could expect from Remark 2.2.7, the following fact for such operators on cocompact abelian coverings is well-known (see e.g., [16, 20, 47, 48, 66, 74, 75]):

Theorem 4.1.1. The spectrum of the above operator L in $L^2(X)$ has a **band-gap struc***ture*:

$$\sigma(L) = \bigcup_{j=1}^{\infty} [\alpha_j, \beta_j],$$

such that $\alpha_j \leq \alpha_{j+1}$, $\beta_j \leq \beta_{j+1}$ and $\lim_{j\to\infty} \alpha_j = \infty$.

The bands can overlap when the dimension of the covering X is greater than 1.

In this chapter, we study Green's function asymptotics for the operator L at an energy level $\lambda \in \mathbb{R}$, such that λ belongs to the union of all closures of finite spectral gaps¹. We divide this into two cases:

- Case I: (Spectral gap interior) The level λ is in a finite spectral gap (β_j, α_{j+1}) such that λ is close either to the spectral edge β_j or to the spectral edge α_{j+1}.
- Case II: (Spectral edge case) The level λ coincides with one of the spectral edges of some finite spectral gap, i.e., λ = α_{j+1} (lower edge) or λ = β_j (upper edge) for some j ∈ N.

¹All of the results still hold for the case when λ does not exceed the bottom of the spectrum, i.e. for the semi-infinite gap.

In Case I, the Green's function $G_{\lambda}(x, y)$ is the Schwartz kernel of the resolvent operator $R_{\lambda,L} := (L-\lambda)^{-1}$, while in Case II, it is the Schwartz kernel of the weak limit of resolvent operators $R_{\lambda,L} := (L-\lambda \pm \varepsilon)^{-1}$ as $\varepsilon \to 0$ (the sign \pm depends on whether λ is an upper or a lower spectral edge). Note that in the flat case $X = \mathbb{R}^d$, Green's function asymptotics of periodic elliptic operators were obtained in Chapter 3 for Case I, and in [52] for Case II. As in the previous chapter (see also [52]), we will deduce all asymptotics from an assumed "generic" spectral edge behavior of the dispersion relation of the operator L.

The organization of the chapter is as follows. In Subsection 4.2.1, we will review some general notions and results about group actions on abelian coverings, and then in Subsection 4.2.2, we introduce additive and multiplicative functions defined on an abelian covering, which will be needed for writing down the main formulae of Green's function asymptotics. Subsection 4.2.3 contains not only a brief introduction to periodic elliptic operators on abelian coverings, but also the necessary notations and assumptions for formulating the asymptotics. The main results of this chapter are stated in Section 4.3. In Section 4.4, the Floquet-Bloch theory is recalled and the problem is reduced to studying a scalar integral. Some auxiliary statements that appeared in Chapter 3 and [52] are collected in Section 4.5 for reader's convenience, and the final proofs of the main results are provided in Section 4.6. Section 4.7 provides the proofs of some technical claims that were postponed from previous sections.

4.2 Notions and preliminary results

4.2.1 Group actions and abelian coverings

Let X be a noncompact smooth Riemannian manifold of dimension n equipped with an **isometric, properly discontinuous, free**, and **co-compact** action of an **finitely generated abelian discrete group** G. The action of an element $g \in G$ on $x \in X$ is denoted by $g \cdot x$. Due to our conditions, the orbit space M = X/G is a **compact** smooth Riemannian

manifold of dimension n when equipped with the metric pushed down from X. We assume that X and M are connected. Thus, we are dealing with a normal abelian covering of a compact manifold

$$X \xrightarrow{\pi} M(=X/G),$$

where G is the deck group of the covering π .

Let $d_X(\cdot, \cdot)$ be the distance metric on the Riemannian manifold X. It is known that X is a complete Riemannian manifold since it is a Riemannian covering of a compact Riemannian manifold M (see e.g., [17]). Thus, for any two points p and q in X, $d_X(p,q)$ is the length of a length minimizing geodesic connecting these two points.

Let S be any finite generating set of the deck group G. We define the word length $|g|_S$ of $g \in G$ to be the number of generators in the shortest word representing g as a product of elements in S:

$$|g|_{S} = \min\{n \in \mathbb{N} \mid g = s_{1} \dots s_{n}, s_{i} \in S \cup S^{-1}\}.$$

The word metric d_S on G with respect to S is the metric on G defined by the formula

$$d_S(g,h) = |g^{-1}h|_S$$

for any $g, h \in G$.

We introduce a notion in geometric group theory due to Gromov that we will need here (see e.g., [57, 62]).

Definition 4.2.1. Let Y, Z be metric spaces. A map $f : Y \to Z$ is called a **quasi-isometry**, if the following conditions are satisfied:

• There are constants $C_1, C_2 > 0$ such that

$$C_1^{-1}d_Y(x,y) - C_2 \le d_Z(f(x), f(y)) \le C_1 d_Y(x,y) + C_2$$

for all $x, y \in Y$.

 The image f(Y) is a net in Z, i.e., there is some constant C > 0 so that if z ∈ Z, then there exists y ∈ Y such that d_Z(f(y), z) < C.

We remark that given any two finite generating sets S_1 and S_2 of G, the two word metrics d_{S_1} and d_{S_2} on G are equivalent (see e.g., [62, Theorem 1.3.12]).

The next result, which directly follows from the Švarc-Milnor lemma (see e.g., [57, Lemma 2.8], [62, Proposition 1.3.13]), establishes a quasi-isometry between the word metric $d_S(\cdot, \cdot)$ of the deck group G and the distance metric $d_X(\cdot, \cdot)$ of the Riemannian covering X of a closed connected Riemannian manifold M.

Proposition 4.2.2. *For any* $x \in X$ *, the map*

$$(G, d_S) \to (X, d_X)$$

 $g \mapsto g \cdot x$

given by the action of the deck transformation group G on X is a quasi-isometry.

Since G is a finitely generated abelian group, its torsion free subgroup is a free abelian subgroup \mathbb{Z}^d of finite index. Hence, we obtain a normal \mathbb{Z}^d -covering

$$X \to M'(=X/\mathbb{Z}^d),$$

and a normal covering of M with a finite number of sheets

$$M' \to M.$$

Then M' is still a compact Riemannian manifold. By switching to the normal subcovering $X \xrightarrow{\mathbb{Z}^d} M'$, we **assume from now on that the deck group** G is \mathbb{Z}^d and substitute M' for M. This will not reduce generality of our results ².

- Notation 4.2.3. (a) Hereafter, we choose the symmetric set $\{-1, 1\}^d$ to be the generating set S of \mathbb{Z}^d . Then the function $z = (z_1, \ldots, z_d) \mapsto \sum_{j=1}^d |z_j|$ is the word length function $|\cdot|_S$ on \mathbb{Z}^d associated with S.
- (b) For a general Riemannian manifold Y, we denote by μ_Y the Riemannian measure of Y. We use the notation $L^2(Y)$ for the Lebesgue function space $L^2(Y, \mu_Y)$. Also, the notation $L^2_{comp}(Y)$ stands for the subspace of $L^2(Y)$ consisting of compactly supported functions. It is worth mentioning that in our case, the Riemannian measure μ_X is the lifting of the Riemannian measure μ_M to X. Thus, μ_X is a G-invariant Riemannian measure on X.
- (c) We recall that a fundamental domain F(M) for M in X (with respect to the action of G) is an open subset of X such that for any g ≠ e, F(M) ∩ g ⋅ F(M) = Ø and the subset

$$X \setminus \bigcup_{g \in G} g \cdot F(M)$$

has measure zero. One can refer to [7] for constructions of such fundamental domains. Henceforth, we use the notation F(M) to stand for a fixed fundamental domain for M in X.

²The same reduction holds for any **finitely generated virtually abelian** deck group G.

Remark 4.2.4. The closure of F(M) contains at least one point in X from every orbit of G, i.e.,

$$X = \bigcup_{g \in G} g \cdot \overline{F(M)}.$$
(4.1)

Thus, if $F: X \to \mathbb{R}$ is the lifting of an integrable function $f: M \to \mathbb{R}$ to X, then

$$\int_{M} f(x)d\mu_M(x) = \int_{\overline{F(M)}} F(x)d\mu_X(x).$$
(4.2)

4.2.2 Additive and multiplicative functions on abelian coverings

To formulate our main results in Section 4.3, we need to introduce an analog of exponential type functions on the noncompact covering X.

We begin with a notion of additive and multiplicative functions on X (see [56])³.

Definition 4.2.5. • A real smooth function u on X is said to be **additive** if there is a homomorphism $\alpha : G \to \mathbb{R}$ such that

$$u(g \cdot x) = u(x) + \alpha(g), \text{ for all } (g, x) \in G \times X.$$

A real smooth function v on X is said to be multiplicative if there is a homomorphism β from G to the multiplicative group R \ {0} such that

$$v(g \cdot x) = \beta(g)v(x), \text{ for all } (g, x) \in G \times X.$$

Let m ∈ N. A function α (resp. β) that maps X to ℝ^m is called a vector-valued additive (resp. multiplicative) function on X if every component of α (resp. β) is also additive (resp. multiplicative) on X.

³The definition can apply to any covering manifold with a discrete deck group.

Following [50, 56], we can define explicitly some additive and multiplicative functions for which the group homomorphisms α , β appearing in Definition 4.2.5 are trivial.

Definition 4.2.6. Let f be a nonnegative function in $C_c^{\infty}(X)$ such that f is strictly positive on $\overline{F(M)}$. For any $j = 1, \ldots, d$, we define the following function

$$E_j(x) = \sum_{g \in \mathbb{Z}^d} \exp\left(-g_j\right) f(g \cdot x).$$

We also put $E(x) := (E_1(x), ..., E_d(x)).$

Then E_j is a positive function satisfying the multiplicative property $E_j(g \cdot x) = \exp(g_j)E_j(x)$, for any $g = (g_1, \ldots, g_d) \in \mathbb{Z}^d$. The multiplicative function E plays a similar role to the one played by the exponential function e^x on the Euclidean space \mathbb{R}^d .

By taking logarithms, we obtain an additive function on X, which leads to the next definition.

Definition 4.2.7. We denote by *h* the smooth \mathbb{R}^d -valued function on *X*

$$h(x) := (\log E_1(x), \cdots, \log E_n(x)).$$

Then $h = (h_1, \ldots, h_d)$ with $h_j(x) = \log E_j(x)$. Thus, h satisfies the following additivity:

$$h(g \cdot x) = h(x) + g$$
, for all $(g, x) \in G \times X$. (4.3)

Here we use the natural embedding $G = \mathbb{Z}^d \subset \mathbb{R}^d$.

Clearly, the definitions of functions E and h depend on the choice of the function f and the fundamental domain F(M). So, there is no canonical choice for constructing additive and multiplicative functions. Nevertheless, a more invariant approach to defining additive functions on co-compact coverings can be found in Appendix A, for instance. The following important comparison between the Riemannian metric and the distance from the additive function h in Definition 4.2.7 will be needed later.

Proposition 4.2.8. There are some positive constants R_h (depending on h) and C > 1such that whenever $d_X(x, y) \ge R_h$, we have

$$C^{-1} \cdot d_X(x,y) \le |h(x) - h(y)| \le C \cdot d_X(x,y).$$

Here $|\cdot|$ *is the Euclidean distance on* \mathbb{R}^d *, and the constant* C *is independent of the choice of* h*.*

As a consequence, the pseudo-distance $d_h(x, y) := |h(x) - h(y)| \to \infty$ if and only if $d_X(x, y) \to \infty$.

The proof of this statement is given in Section 4.7.

Definition 4.2.9. For any additive function h, \mathcal{A}_h is the set consisting of unit vectors $s \in \mathbb{S}^{d-1}$ such that there exist two points x and y satisfying $d_X(x, y) > R_h$ and

$$s = (h(x) - h(y))/|h(x) - h(y)|.$$

The set A_h is called the **admissible set of the additive function** h, and its elements are **admissible directions** of h.

For the proof of the following proposition, one can see in Section 4.7.

Proposition 4.2.10. For any additive function h on X, one has

$$\mathbb{Q}^d \cap \mathbb{S}^{d-1} = \{ g/|g| \mid g \in \mathbb{Z}^d \setminus \{0\} \} \subset \mathcal{A}_h.$$
(4.4)

Hence, the admissible set \mathcal{A}_h of h is dense in the sphere \mathbb{S}^{d-1} . In particular, when d = 2, \mathcal{A}_h is the whole unit circle \mathbb{S}^1 .
Remark 4.2.11. When the dimension n of X is less than (d-1)/2 (e.g., if d > 5 and X is the standard two dimensional jungle gym JG^2 in \mathbb{R}^d , see [63]), the (d-1)-dimensional Lebesgue measure of the admissible set \mathcal{A}_h of any additive function h on X is zero. To see this, we first denote by X_h the 2n-dimensional smooth manifold $\{(x, y) \in X \times X \mid d_X(x, y) > R_h\}$, and then consider the smooth mapping:

$$\Psi: X_h \to \mathbb{S}^{d-1}$$
$$(x, y) \mapsto \frac{h(x) - h(y)}{|h(x) - h(y)|}$$

Then \mathcal{A}_h is the range of Ψ . Since $\dim X_h < \dim \mathbb{S}^{d-1}$, every point in the range of Ψ is critical and thus, \mathcal{A}_h has measure zero by Sard's theorem.

Example 4.2.12.

Here is a family of non-trivial examples of additive functions in the flat case, i.e., when the covering space X is ℝ^d and the base is the d-dimensional torus T^d. Let d ≥ 1 and φ be a real smooth function in ℝ^d such that φ is ℤ^d-periodic. It is shown in [5] that there exists a unique map F_φ = ((F_φ)₁,...,(F_φ)_d) : ℝ^d → ℝ^d satisfying F_φ(0) = 0, the additive condition (4.3), i.e., F_φ(x + n) = F_φ(x) + n for any (x, n) ∈ ℝ^d × ℤ^d, and the equation

$$\Delta(F_{\varphi})_i = \nabla \varphi \cdot \nabla(F_{\varphi})_i,$$

for any $1 \leq i \leq d$. Note that F_{φ} is just the identity mapping in the trivial case when $\varphi = 0$. Moreover, it is also known [5] that when d = 2, F_{φ} is a diffeomorphism of \mathbb{R}^d onto itself. In particular, for any \mathbb{Z}^2 -periodic function φ , $|F_{\varphi}(x) - F_{\varphi}(y)| \geq C_{\varphi}|x-y|$ for any $x, y \in \mathbb{R}^2$ for some $C_{\varphi} > 0$. However, when $d \geq 3$, F_{φ} may admit a critical point for some \mathbb{Z}^d -periodic function φ .

Let X → M, Y → N be normal Z^{d₁} and Z^{d₂} coverings of compact Riemannian manifolds M and N respectively. Then X × Y → M × N is also a normal Z^{d₁+d₂} covering of M×N. Consider any R^{d₁}-valued function h₁ (resp. R^{d₂}-valued function h₂) defined on X (resp. Y). Let us denote by h₁ ⊕ h₂ the following R^{d₁+d₂}-valued function on X × Y:

$$(h_1 \oplus h_2)(x, y) = (h_1(x), h_2(y)), \quad (x, y) \in X \times Y.$$

Then it is clear that $h_1 \oplus h_2$ is additive (resp. multiplicative) on $X \times Y$ if and only if both functions h_1 and h_2 are additive (resp. multiplicative). Moreover, $\mathcal{A}_{h_1 \oplus h_2} \subseteq$ $\{(a_1 \cdot \mathcal{A}_{h_1}, a_2 \cdot \mathcal{A}_{h_2}) \mid 0 < a_1, a_2 < 1 \text{ and } a_1^2 + a_2^2 = 1\}.$

4.2.3 Some notions and assumptions

Let L be a **bounded from below, real** and **symmetric second-order elliptic**⁴ operator on X with **smooth**⁵ coefficients such that the operator **commutes with the action of** G. An operator that commutes with the action of G is called a G-periodic (or sometimes **periodic**) operator for brevity.

Notice that on a Riemannian co-compact covering, any G-periodic elliptic operator with smooth coefficients is **uniformly elliptic** in the sense that

$$|L_0^{-1}(x,\xi)| \le C|\xi|^{-2}, \quad (x,\xi) \in T^*X, \xi \ne 0.$$

Here $|\xi|$ is the Riemannian length of (x,ξ) and $L_0(x,\xi)$ is the principal symbol of L.

The periodic operator L can be pushed down to an elliptic operator L_M on M and thus,

L is the lifting of an elliptic operator L_M to X. By a slight abuse of notation, we will use

⁴The ellipticity is understood in the sense of the nonvanishing of the principal symbol of the operator L on the cotangent bundle of the underlying manifold (with the zero section removed).

⁵The smoothness condition is assumed for avoiding lengthy technicalities and it can be relaxed.

the same notation L for both elliptic operators acting on X and M.

Under these assumptions on L, the symmetric operator L with the domain $C_c^{\infty}(X)$ is essentially self-adjoint in $L^2(X)$, i.e., the minimal operator L_{min} coincides with the maximal operator L_{max} (see Chapter 2 or [71] for notation L_{min} and L_{max}). This fact can be found in [7, Proposition 3.1], for instance ⁶. Hence, there exists a unique self-adjoint extension in the Hilbert space $L^2(X)$ of L, which we denote also by L. Since L is a uniformly elliptic operator on the manifold X of bounded geometry, its domain is the Sobolev space $H^2(X)$ [71, Proposition 4.1], and henceforward, we always work with this self-adjoint operator L.

From now on, we fix an additive function h (see Definition 4.2.7). The following lemma is a preparation for the next definition.

Lemma 4.2.13. For any $k \in \mathbb{C}^d$, we have

$$e^{-ik \cdot h(x)}L(x,D)e^{ik \cdot h(x)} = L(x,D) + B(k),$$

where B(k) is a smooth differential operator of order 1 on X that commutes with the action of the deck group G. Thus by pushing down, the differential operators $e^{-ik \cdot h(x)}L(x, D)e^{ik \cdot h(x)}$ and B(k) can be considered also as differential operators on M. Moreover, given any $m \in \mathbb{R}$, the mapping

$$k \mapsto e^{-ik \cdot h(x)} L(x, D) e^{ik \cdot h(x)}$$

is analytic in k as a $B(H^{m+2}(M), H^m(M))$ -valued function.

Proof. It is standard that the commutator $[L, e^{ik \cdot h(x)}]$ is a differential operator of order 1

⁶In [7], Atiyah proves for symmetric elliptic operators acting on Hermitian vector bundles over any general co-compact covering manifold (not necessary to be a Riemannian covering). Later, in [16], Brüning and Sunada extend Atiyah's arguments to the case including compact quotient space X/G with singularities.

on X. Now one can write

$$B(k) = e^{-ik \cdot h(x)} L e^{ik \cdot h(x)} - L = e^{-ik \cdot h(x)} [L, e^{ik \cdot h(x)}]$$

to see that B(k) is also a smooth differential operator of order 1. Also, one can check that B(k) commutes with the action of G by using G-periodicity of the operator L and additivity of h. This proves the first claim of the lemma. From a standard fact (see e.g., [36, Theorem 2.2]), the operator $e^{-ik \cdot h(x)} L e^{ik \cdot h(x)}$ defined on X can be written as a sum $\sum_{|\alpha| \leq 2} k^{\alpha} L_{\alpha}$, where L_{α} is a G-periodic differential operator on X of order $2 - |\alpha|$ which is independent of k. By pushing the above sum down to a sum of operators on M, the claim about analyticity in k is then obvious.

Definition 4.2.14. For any $k \in \mathbb{C}^d$, we denote by L(k) the elliptic operator

$$e^{-ik\cdot h(x)}L(x,D)e^{ik\cdot h(x)}$$

in $L^2(M)$ with the domain the Sobolev space $H^2(M)$.

The operator L(k) is self-adjoint in $L^2(M)$ for each $k \in \mathbb{R}^d$, with the domain $H^2(M)$. Due to ellipticity of L, each of the operators L(k) ($k \in \mathbb{R}^d$) has discrete real spectrum and thus, we can single out continuous and piecewise-analytic band functions $\lambda_j(k)$ for each $j \in \mathbb{N}$ as before. By Lemma 4.2.13, the operators L(k) are perturbations of the self-adjoint operator L(0) by lower order operators B(k) for each $k \in \mathbb{C}^d$. Consequently, the spectra of the operators L(k) on M are all discrete (see [4, pp.180-190]). In a similar manner to the flat case, we have:

Theorem 4.2.15. [16, 20, 45, 47, 66, 74, 75] *The spectrum of* L *is the union of all the spectra of* L(k) *when* k *runs over the Brillouin zone (or any its shifted copy), i.e., for any*

quasimomentum $k_0 \in \mathbb{R}^d$:

$$\sigma(L) = \bigcup_{k \in k_0 + [-\pi,\pi]^d} \sigma(L(k)) = \bigcup_{k \in k_0 + [-\pi,\pi]^d} \{\lambda_j(k) \mid j \in \mathbb{N}\}.$$
(4.5)

We recall the notions of Bloch variety and Fermi surface from Chapter 2.

Definition 4.2.16. A Bloch solution with quasimomentum k of the equation L(x, D)u = 0 is a solution of the form

$$u(x) = e^{ik \cdot h(x)}\phi(x),$$

where *h* is any fixed additive function on *X* and the function ϕ is invariant under the action of the deck transformation group *G*.⁷

Using the above definition of Bloch solution with quasimomentum k, one can define the Bloch variety and the Fermi surfaces of the operator L as in Definition 2.2.5.

Without loss of generality, it is enough to assume henceforth that 0 is the spectral edge of interest (by adding a constant into the operator L if necessary) and there is a spectral gap below this spectral edge 0. Therefore, 0 is the lower spectral edge of some spectral band ⁸, i.e., 0 is the minimal value of some band function $\lambda_j(k)$ for some $j \in \mathbb{N}$ over the Brillouin zone.

From now on, we will impose Assumption A in Chapter 3 on the band function λ_i .

Remark 4.2.17.

(a) For the flat case, the main theorem in [44] shows that the conditions A1 and A2 are 'generically' satisfied, i.e., they can be achieved by small perturbation of the potential of a periodic Schrödinger operator. The same proof in [44] still works for periodic Schrödinger operators on a general abelian covering.

⁷It is easy to see that this definition is independent of the choice of h.

⁸The upper spectral edge case is treated similarly.

- (b) We shall only use this condition A5 for the *spectral gap interior* case.
- (c) Due to results of [43] (in the flat case $X = \mathbb{R}^d$) and of [45] (in the general case), all these assumptions A1-A5 hold at the bottom of the spectrum for non-magnetic Schrödinger operators.

Here are some notations that will be used throughout this chapter.

Notation 4.2.18. For any two functions f and g defined on $X \times X$, if there exist constants C > 0 and R > 0 such that $|f(x, y)| \le C|g(x, y)|$ whenever $d_X(x, y) > R$, we write f(x, y) = O(g(x, y)).

We say that a set W in \mathbb{C}^d is symmetric if for any $z \in W$, we have $\overline{z} \in W$.

The following proposition will play a crucial role in establishing Theorem 4.3.1. We omit the proof since it requires no change from our discussion in Chapter 3.

Proposition 4.2.19. There exists an $\epsilon_0 > 0$ and a symmetric open subset $V \subset \mathbb{C}^d$ containing the quasimomentum k_0 from the Assumption A such that the band function λ_j in Assumption A has an analytic continuation into a neighborhood of \overline{V} , and the following properties hold for any z in a symmetric neighborhood of \overline{V} :

(P1) $\lambda_j(z)$ is a simple eigenvalue of L(z).

(P2) $|\lambda_j(z)| < \epsilon_0 \text{ and } \overline{B}(0,\epsilon_0) \cap \sigma(L(z)) = \{\lambda_j(z)\}.$

(P3) There is a nonzero G-periodic function ϕ_z defined on X such that

$$L(z)\phi_z = \lambda_j(z)\phi_z.$$

Moreover, $z \mapsto \phi_z$ can be chosen analytic as a $H^2(M)$ -valued function. (P4) $2\Re(\text{Hess}(\lambda_j)(z)) > \min \sigma(\text{Hess}(\lambda_j)(k_0)) \cdot I_{d \times d}$. (P5) $F(z) := (\phi(z, \cdot), \phi(\overline{z}, \cdot))_{L^2(M)} \neq 0$. Let $\mathcal{V} = \{\beta \in \mathbb{R}^d \mid k_0 + i\beta \in \overline{V}\}$. Now we introduce the function $E(\beta) := \lambda_j(k_0 + i\beta)$, which is defined on \mathcal{V} . The next lemma (see also Chapter 3) is the only place in this chapter where the condition A5 is used.

Lemma 4.2.20. Assume the condition A5. Then E is a real-valued function. By reducing the neighborhood V in Proposition 4.2.19 if necessary, the function E can be assumed real analytic and strictly concave function from V to \mathbb{R} such that its Hessian at any point β in V is negative-definite.

For $\lambda \in \mathbb{R}$, we put $\Gamma_{\lambda} := \{\beta \in \mathcal{V} : E(\beta) = \lambda\}.$

Due to Lemma 4.2.20, there exists a diffeomorphism β from \mathbb{S}^{d-1} onto Γ_{λ} such that

$$\nabla E(\beta_s) = -|\nabla E(\beta_s)|s.$$

In addition, $\lim_{|\lambda|\to 0} \max_{s\in\mathbb{S}^{d-1}} |\beta_s| = 0$. By letting $|\lambda|$ be sufficiently small, we will suppose that there is an $r_0 > 0$ (independent of s) such that

$$\{k + it\beta_s \mid (t,s) \in [0,1] \times \mathbb{S}^{d-1}, \ |k - k_0| \le r_0\} \subset V.$$
(4.6)

4.3 The main results

We recall that h is a fixed additive function (see Definition 4.2.7).

First, we consider the case when λ is inside a gap and is near to one of the edges of the gap. The following result is an analog for abelian coverings of compact Riemannian manifolds of Theorem 3.2.5.

Theorem 4.3.1. (Spectral gap interior)

Suppose that $d \ge 2$ and the conditions A1-A5 are satisfied. For $\lambda < 0$ sufficiently close to 0 (depending on the dispersion branch λ_j and the operator L), the Green's function G_{λ} of L at λ admits the following asymptotics as $d_X(x, y) \to \infty$:

$$G_{\lambda}(x,y) = \frac{e^{(h(x)-h(y))(ik_{0}-\beta_{s})}}{(2\pi|h(x)-h(y)|)^{(d-1)/2}} \cdot \frac{|\nabla E(\beta_{s})|^{(d-3)/2}}{\det\left(-\mathcal{P}_{s}\operatorname{Hess}\left(E\right)(\beta_{s})\mathcal{P}_{s}\right)^{1/2}} \times \frac{\phi_{k_{0}+i\beta_{s}}(x)\overline{\phi_{k_{0}-i\beta_{s}}(y)}}{(\phi_{k_{0}+i\beta_{s}},\phi_{k_{0}-i\beta_{s}})L^{2}(M)} + e^{(h(y)-h(x))\cdot\beta_{s}}r(x,y).$$

$$(4.7)$$

Here

$$s = (h(x) - h(y))/|h(x) - h(y)| \in \mathcal{A}_h,$$

and \mathcal{P}_s is the projection from \mathbb{R}^d onto the tangent space of the unit sphere \mathbb{S}^{d-1} at the point *s*. Also, there is a constant C > 0 (independent of *s* and of the choice of *h*) such that the remainder term *r* satisfies

$$|r(x,y)| \le d_X(x,y)^{-d/2},$$

when $d_X(x, y)$ is large enough.

By using rational admissible directions (see (4.4)) in the formula (4.7), the large scale behaviors of the Green's function along orbits of the *G*-action admit the following nice form in which the additive function *h* is absent.

Corollary 4.3.2. Under the same notations and hypotheses of Theorem 4.3.1 and suppose that $\lambda < 0$ is close enough to 0, as $|g| \to \infty$ ($g \in \mathbb{Z}^d$), we have

$$G_{\lambda}(x,g\cdot x) = \frac{e^{g\cdot(ik_{0}-\beta_{g/|g|})}}{(2\pi|g|)^{(d-1)/2}} \cdot \frac{|\nabla E(\beta_{g/|g|})|^{(d-3)/2}}{\det\left(-\mathcal{P}_{g/|g|}\operatorname{Hess}\left(E\right)(\beta_{g/|g|})\mathcal{P}_{g/|g|}\right)^{1/2}} \times \frac{\phi_{k_{0}+i\beta_{g/|g|}}(x)\overline{\phi_{k_{0}-i\beta_{g/|g|}}(g\cdot x)}}{(\phi_{k_{0}+i\beta_{g/|g|}},\phi_{k_{0}-i\beta_{g/|g|}})_{L^{2}(M)}} + e^{g\cdot\beta_{s}}O(|g|^{-d/2}).$$

$$(4.8)$$

We also give another interpretation of Theorem 4.3.1 in the special case $X = \mathbb{R}^2$ as follows:

Corollary 4.3.3. Let φ be any real, \mathbb{Z}^2 -periodic and smooth function on \mathbb{R}^2 , and we recall

the notation F_{φ} from Example 4.2.12. Let *s* be any unit vector in \mathbb{R}^2 and $y \in \mathbb{R}^2$. Then as $|t| \to \infty$ $(t \in \mathbb{R})$, the Green's function G_{λ} of *L* at $\lambda \ (\approx 0)$ has the following asymptotics

$$G_{\lambda}(F_{\varphi}^{-1}(ts+F_{\varphi}(y)),y) = \frac{e^{ts\cdot(ik_{0}-\beta_{s})}}{(2\pi|\nabla E(\beta_{s})|\cdot\det(-\mathcal{P}_{s}\operatorname{Hess}(E)(\beta_{s})\mathcal{P}_{s})\cdot|t|)^{1/2}} \times \frac{\phi_{k_{0}+i\beta_{s}}(F_{\varphi}^{-1}(ts+F_{\varphi}(y)))\overline{\phi_{k_{0}-i\beta_{s}}(y)}}{(\phi_{k_{0}+i\beta_{s}},\phi_{k_{0}-i\beta_{s}})_{L^{2}(\mathbb{T}^{2})}} + e^{ts\cdot\beta_{s}}O(|t|^{-1}).$$

We now switch to the case when λ is on the boundary of the spectrum. The following result is a generalization of [52, Theorem 2].

Theorem 4.3.4. (Spectral edge case)

Let $d \ge 3$, the operator L satisfy the assumptions A1-A4, and $R_{-\varepsilon} = (L + \varepsilon)^{-1}$ for a small $\varepsilon > 0$ denote the resolvent of L near the spectral edge $\lambda = 0$ (which exists, due to Assumption A). Then:

i) For any $\phi, \varphi \in L^2_{comp}(X)$, as $\varepsilon \to 0$, we have:

$$\langle R_{-\varepsilon}\phi,\varphi\rangle \to \langle R\phi,\varphi\rangle.$$

for an operator $R: L^2_{comp}(X) \to L^2_{loc}(X)$.

ii) The Schwartz kernel G(x, y) of the operator R, which we call the Green's function of L (at the spectral edge 0), has the following asymptotics when $d_X(x, y) \to \infty$:

$$G(x,y) = \frac{\Gamma(\frac{d-2}{2})e^{i(h(x)-h(y))\cdot k_0}}{2\pi^{d/2}\sqrt{\det H}|H^{-1/2}(h(x)-h(y))|^{d-2}} \cdot \frac{\phi_{k_0}(x)\overline{\phi_{k_0}(y)}}{\|\phi_{k_0}\|_{L^2(M)}^2} \times \left(1+O\left(d_X(x,y)^{-1}\right)\right) + O\left(d_X(x,y)^{1-d}\right),$$
(4.9)

where H is the Hessian matrix of λ_j at k_0 .

Remark 4.3.5.

- (a) An interesting feature in the main results is that the dimension n of the covering manifold X does not explicitly enter into the asymptotics (4.7) and (4.9) (especially, see also (4.8)). Anyway, it certainly influences the geometry of the dispersion curves and therefore the asymptotics too. However, as the Riemannian distance of x and y becomes larger, one can see that in the asymptotics, the role of the dimension n is rather limited, while the influence of the rank d of the torsion-free subgroup of the deck group G is stronger.
- (b) Note that for a periodic elliptic operator of second order on ℝ^d, at the bottom of its spectrum, the operator is known to be critical when the dimension d ≤ 2 (see [56,60,64]). So, the assumption d ≥ 3 is needed in Theorem 4.3.4.

Proving Theorem 4.3.4 by generalizing [52, Theorem 2] is similar to showing Theorem 4.3.1 by generalizing Theorem 3.2.5. Thus, after finishing the proof of Theorem 4.3.1, we will sketch briefly the proof of Theorem 4.3.4 in Section 4.6.

We outline the general strategy of both the proofs of Theorem 4.3.1 and Theorem 4.3.4. As in Chapter 3 and [52], the idea is to show that only one branch of the dispersion relation λ_j appearing in the Assumption A will control the asymptotics.

- Step 1: We use the Floquet transform to reduce the problems of finding asymptotics of Green's functions to the problems of obtaining asymptotics of some integral expressions with respect to the quasimomentum *k*.
- Step 2: We localize these expressions around the quasimomentum k₀ and then we cut an "infinite-dimensional" part of the operator to deal only with the multiplication operator by the dispersion branch λ_i.

 Step 3: The dispersion curve around this part is almost a paraboloid according to the assumption A4, thus, we can reduce this piece of operator to the normal form in the free case. In this step, we obtain some scalar integral expressions which are close to the ones arising when dealing with the Green's function of the Laplacian operator at the level λ. Our remaining task is devoted to computing the asymptotics of these scalar integrals.

4.4 A Floquet-Bloch reduction of the problem

In this section, we will consider the Green's function $G_{\lambda}(x, y)$ at the level λ in Case I (i.e., *Spectral gap interior*).

4.4.1 The Floquet transforms on abelian coverings and a Floquet reduction of the problem

Notation 4.4.1. We introduce the following fundamental domain \mathcal{O} (with respect to the dual lattice $2\pi\mathbb{Z}^d$):

$$\mathcal{O} = k_0 + [-\pi, \pi]^d.$$

If k_0 is a high symmetry point of the Brillouin zone (i.e., k_0 satisfies the assumption **A5**), then \mathcal{O} is the fundamental domain we defined in Definition 3.2.1.

The following transform will play the role of the Fourier transform for the periodic case. Indeed, it is a version of the Fourier transform on the group \mathbb{Z}^d of periods.

Definition 4.4.2. The **Floquet transform** \mathcal{F} (which depends on the choice of h)

$$f(x) \to \widehat{f}(k, x)$$

maps a compactly supported function f on X into a function \hat{f} defined on $\mathbb{R}^d \times X$ in the following way:

$$\widehat{f}(k,x) := \sum_{\gamma \in \mathbb{Z}^d} f(\gamma \cdot x) e^{-ih(\gamma \cdot x) \cdot k}.$$

From the above definition, one can see that \hat{f} is \mathbb{Z}^d -periodic in the *x*-variable and satisfies a cyclic condition with respect to *k*:

$$\begin{cases} \widehat{f}(k,\gamma\cdot x) = \widehat{f}(k,x), \quad \forall\gamma\in\mathbb{Z}^d\\ \widehat{f}(k+2\pi\gamma,x) = e^{-2\pi i\gamma\cdot h(x)}\widehat{f}(k,x), \quad \forall\gamma\in\mathbb{Z}^d \end{cases}$$

Thus, it suffices to consider the Floquet transform \hat{f} as a function defined on $\mathcal{O} \times M$. Usually, we will regard \hat{f} as a function $\hat{f}(k, \cdot)$ in k-variable in \mathcal{O} with values in the function space $L^2(M)$.

We recall some properties of the Floquet transform on abelian coverings as in Lemma 2.2.4. Note that the proof for the abelian covering case does not require any change from the proof for the flat case.

I. The transform \mathcal{F} is an isometry of $L^2(X)$ onto

$$\int_{\mathcal{O}}^{\oplus} L^2(M) = L^2(\mathcal{O}, L^2(M))$$

and of $H^2(X)$ onto

$$\int_{\mathcal{O}}^{\oplus} H^2(M) = L^2(\mathcal{O}, H^2(M)).$$

II. The following two equivalent inversion formulae \mathcal{F}^{-1} are given by

$$f(x) = (2\pi)^{-d} \int_{\mathcal{O}} e^{ik \cdot h(x)} \widehat{f}(k, x) \, \mathrm{d}k, \quad x \in X.$$
(4.10)

and

$$f(x) = (2\pi)^{-d} \int_{\mathcal{O}} e^{ik \cdot h(x)} \widehat{f}(k, \gamma^{-1} \cdot x) \, \mathrm{d}k, \quad x \in \gamma \cdot \overline{F(M)}.$$
(4.11)

III. (*Block-diagonalization*) The action of any periodic elliptic operator P in $L^2(X)$ under the Floquet transform \mathcal{F} is given by

$$\mathcal{F}P(x,D)\mathcal{F}^{-1} = \int_{\mathcal{O}}^{\oplus} P(k) \,\mathrm{d}k,$$

where $P(k)(x, D) = e^{-ik \cdot h(x)} P(x, D) e^{ik \cdot h(x)}$. In other words, $\widehat{Pf}(k) = P(k)\widehat{f}(k)$, for any $f \in H^2(X)$.

The next statement is proven in Proposition 3.4.1 for the flat case. We omit the proof.

Proposition 4.4.3. If $|\lambda|$ is small enough (depending on the dispersion branch λ_j and L), then for any $(t, s) \in [0, 1] \times \mathbb{S}^{d-1}$, we have $\lambda \in \sigma(L(k+it\beta_s))$ if and only if $(k, t) = (k_0, 1)$.

As in the previous chapter, the main ingredients in the proof of this statement are the upper-semicontinuity of the spectra of the analytic family $\{L(k)\}_{k\in\mathbb{C}^d}$ and the fact that E is a **real** function, whose Hessian is negative definite (Proposition 4.2.20).

We consider the following real, smooth linear elliptic operators on X:

$$L_{t,s} = e^{t\beta_s \cdot h(x)} L e^{-t\beta_s \cdot h(x)}, \quad (t,s) \in [0,1] \times \mathbb{S}^{d-1}.$$

Notice that these operators are G-periodic, and when pushing $L_{t,s}$ down to M, we get the operator $L(-i\beta_s)$. We also use the notation L_s for $L_{1,s}$. By Floquet-Bloch theory for the operator $L_{t,s}$, we obtain

$$\sigma(L_{t,s}) = \bigcup_{k \in \mathcal{O}} \sigma(L_{t,s}(k)) = \bigcup_{k \in \mathcal{O}} \sigma(L(k + it\beta_s)) \supseteq \{\lambda_j(k + it\beta_s)\}_{k \in \mathcal{O}}.$$
 (4.12)

We now fix a real number λ such that the statement of Proposition 4.4.3 holds. By (4.12) and Proposition 4.4.3, λ is in the resolvent set of $L_{t,s}$ for any $(t,s) \in [0,1) \times \mathbb{S}^{d-1}$. Let $R_{t,s,\lambda}$ be the resolvent operator $(L_{t,s} - \lambda)^{-1}$. From the block-diagonalize property of the Floquet transform, for any $f \in L^2_{comp}(X)$, we have

$$\widehat{R_{t,s,\lambda}f}(k) = (L_{t,s}(k) - \lambda)^{-1}\widehat{f}(k), \quad (t,k) \in [0,1) \times \mathcal{O}.$$

Hence, the sesquilinear form $(R_{t,s,\lambda}f,\varphi)$ is equal to

$$(2\pi)^{-d} \int_{\mathcal{O}} \left((L_{t,s}(k) - \lambda)^{-1} \widehat{f}(k), \widehat{\varphi}(k) \right) \mathrm{d}k$$

where $\varphi \in L^2_{comp}(X)$.

In the next lemma, the weak convergence as $t \nearrow 1$ of the operator $R_{t,s,\lambda}$ in $L^2_{comp}(X)$ is proved and thus, we can introduce the limit operator $R_{s,\lambda} := \lim_{t \to 1^-} R_{t,s,\lambda}$.

Lemma 4.4.4. Let $d \ge 2$. Under Assumption A, for f, φ in $L^2_{comp}(X)$, the following equality holds:

$$\lim_{t \to 1^{-}} (R_{t,s,\lambda}f,\varphi) = (2\pi)^{-d} \int_{\mathcal{O}} \left(L_s(k) - \lambda)^{-1} \widehat{f}(k), \widehat{\varphi}(k) \right) \mathrm{d}k.$$
(4.13)

The integral in the right hand side of (4.13) is absolutely convergent.

This lemma is a direct corollary of the analyticity of the Bloch variety (see Proposition 2.2.6), Proposition 4.4.3 and the Lebesgue Dominated Convergence Theorem as being shown in Lemma 3.5.1. We skip the proof.

For any $(t,s) \in [0,1) \times \mathbb{S}^{d-1}$, let $G_{t,s,\lambda}$ be the Green's function of $L_{t,s}$ at λ , which is

the kernel of $R_{t,s,\lambda}$. Thus,

$$G_{t,s,\lambda}(x,y) = e^{t\beta_s \cdot (h(x) - h(y))} G_\lambda(x,y).$$

Taking the limit and applying Lemma 4.4.4, we conclude that the function

$$G_{s,\lambda}(x,y) := e^{\beta_s \cdot (h(y) - h(x))} G_\lambda(x,y)$$

is the integral kernel of the operator $R_{s,\lambda}$ defined as follows:

$$\widehat{R_{s,\lambda}f}(k) = (L_s(k) - \lambda)^{-1}\widehat{f}(k).$$
(4.14)

Hence, the problem of finding asymptotics of G_{λ} is now equivalent to obtaining asymptotics of any function $G_{s,\lambda}$, where s is an admissible direction in \mathcal{A}_h .

In addition, by (4.10) and (4.14), the function $G_{s,\lambda}$, which is also the Green's function of the operator L_s at λ , is the integral kernel of the operator $R_{s,\lambda}$ that acts on $L^2_{comp}(X)$ in the following way:

$$R_{s,\lambda}f(x) = (2\pi)^{-d} \int_{\mathcal{O}} e^{ik \cdot h(x)} (L_s(k) - \lambda)^{-1} \widehat{f}(k, x) \, \mathrm{d}k, \quad x \in X.$$
(4.15)

This accomplishes Step 1 in our strategy of the proof.

4.4.2 Isolating the leading term in $R_{s,\lambda}$ and a reduced Green's function

The purpose of this part is to complete Step 2, i.e., to localize the part of the integral in (4.15), that is responsible for the leading term of the Green's function asymptotics.

Definition 4.4.5. For any $z \in V$, we denote by P(z) the spectral projector $\chi_{B(0,\varepsilon_0)}(L(z))$,

i.e.,

$$P(z) = -\frac{1}{2\pi i} \oint_{|\alpha|=\epsilon_0} (L(z) - \alpha)^{-1} \,\mathrm{d}\alpha.$$

By (P2), P(z) projects $L^2(M)$ onto the eigenspace spanned by ϕ_z . We also put Q(z) := I - P(z) and denote by R(P(z)), R(Q(z)) the ranges of the projectors P(z), Q(z) correspondingly.

Using (P6) and the fact that $P(k+i\beta_s)^* = P(k-i\beta_s)$, we can deduce that if $|k-k_0| \le r_0$ (see (3.8)), the following equality holds

$$P(k+i\beta_s)u = \frac{(u,\phi_{k-i\beta_s})_{L^2(M)}}{(\phi_{k+i\beta_s},\phi_{k-i\beta_s})_{L^2(M)}}\phi_{k+i\beta_s}, \quad \forall u \in L^2(M).$$
(4.16)

Let η be a cut-off smooth function on \mathcal{O} supported on $\{k \in \mathcal{O} \mid |k - k_0| < r_0\}$ and equal to 1 around k_0 .

According to (4.15), for any $f \in C_c^{\infty}(X)$, we want to find u such that

$$(L_s(k) - \lambda)\widehat{u}(k) = \widehat{f}(k).$$

Then the Green's function $G_{s,\lambda}$ satisfies

$$\int_X G_{s,\lambda}(x,y)f(y) \,\mathrm{d}\mu_X(y) = \mathcal{F}^{-1}\widehat{u}(k,x) = u(x),$$

where \mathcal{F} is the Floquet transform introduced in Definition 4.4.2.

By Proposition 4.4.3, the operator $L_s(k) - \lambda$ is invertible for any k such that $k \neq k_0$. Hence, we can decompose $\hat{u}(k) = \hat{u_0}(k) + (L_s(k) - \lambda)^{-1}(1 - \eta(k))\hat{f}(k)$, where $\hat{u_0}$ satisfies the equation

$$(L_s(k) - \lambda)\widehat{u_0}(k) = \eta(k)\widehat{f}(k).$$

Observe that R(P(z)) and R(Q(z)) are invariant subspaces for the operator L(z) for

any $z \in V$. Thus, if u_1, u_2 are functions such that $\hat{u}_1(k) = P(k + i\beta_s)\hat{u}_0(k)$ and $\hat{u}_2(k) = Q(k + i\beta_s)\hat{u}_0(k)$, we must have

$$(L_s(k) - \lambda)P(k + i\beta_s)\widehat{u}_1(k) = \eta(k)P(k + i\beta_s)\widehat{f}(k)$$
(4.17)

and

$$(L_s(k) - \lambda)Q(k + i\beta_s)\widehat{u}_2(k) = \eta(k)Q(k + i\beta_s)\widehat{f}(k).$$

Due to (**P2**), when k is close to k_0 , $\lambda = \lambda_j(k_0 + i\beta_s)$ must belong to the resolvent of the operator $L_s(k)|_{R(Q(k+i\beta_s))}$. Hence, we can write $\hat{u}_2(k) = \eta(k)(L_s(k) - \lambda)^{-1}Q(k + i\beta_s)\hat{f}(k)$. Therefore, $\hat{u}(k)$ equals

$$\widehat{u_1}(k) + \left((1 - \eta(k))(L_s(k) - \lambda)^{-1} + \eta(k)((L_s(k) - \lambda)|_{R(Q(k + i\beta_s))})^{-1}Q(k + i\beta_s) \right) \widehat{f}(k).$$

The next theorem shows that we can ignore the infinite-dimensional part of the operator $R_{s,\lambda}$, i.e., the second term in the above sum of two operators.

Theorem 4.4.6. Define

$$T_s(k) := (1 - \eta(k))(L_s(k) - \lambda)^{-1} + \eta(k)((L_s(k) - \lambda)|_{R(Q(k + i\beta_s))})^{-1}Q(k + i\beta_s).$$

Let T_s be the operator acting on $L^2(X)$ as follows:

$$T_s = \mathcal{F}^{-1}\left((2\pi)^{-d} \int_{\mathcal{O}}^{\oplus} T_s(k) \,\mathrm{d}k\right) \mathcal{F}.$$

Then the Schwartz kernel $K_s(x, y)$ of the operator T_s is continuous away from the diagonal of X, and moreover, it is also rapidly decaying in a uniform way with respect to

 $s \in \mathbb{S}^{d-1}$, *i.e.*, for any N > 0,

$$\sup_{s \in \mathbb{S}^{d-1}} |K_s(x, y)| = O(d_X(x, y)^{-N}).$$

A proof using microlocal analysis is provided in Section 4.7.

Now let $V_s := R_{s,\lambda} - T_s$. Then the Schwartz kernel $G_0(x, y)$ of the operator V_s satisfies the following relation:

$$\int_X G_0(x,y)f(y) \,\mathrm{d}\mu_X(y) = \mathcal{F}^{-1}\widehat{u_1}(k,x) = u_1(x). \tag{4.18}$$

In what follows, we will find an integral representation of the kernel G_0 . We will see that G_0 provides the leading term of the asymptotics of the kernel $G_{s,\lambda}$. For this reason, G_0 is called **a reduced Green's function**.

To find u_1 , we use the equation (4.17) and apply (4.16) to deduce

$$(\lambda_j(k+i\beta_s)-\lambda)(\widehat{u}_1(k),\phi_{k-i\beta_s})_{L^2(M)} = \eta(k)(\widehat{f}(k),\phi_{k-i\beta_s})_{L^2(M)}.$$

Using $\widehat{u_1}(k) = P(k + i\beta_s)\widehat{u_1}(k)$ and (4.16) again, the above identity becomes

$$\widehat{u_1}(k,x) := \frac{\eta(k)\phi_{k+i\beta_s}(x)(\widehat{f}(k),\phi_{k-i\beta_s})_{L^2(M)}}{(\phi_{k+i\beta_s},\phi_{k-i\beta_s})_{L^2(M)}(\lambda_j(k+i\beta_s)-\lambda)}, \quad k \neq k_0.$$

By the inverse Floquet transform (4.10), for any $x \in X$,

$$u_1(x) = (2\pi)^{-d} \int_{\mathcal{O}} e^{ik \cdot h(x)} \frac{\eta(k)\phi_{k+i\beta_s}(x)(\widehat{f}(k), \phi_{k-i\beta_s})_{L^2(M)}}{(\phi_{k+i\beta_s}, \phi_{k-i\beta_s})_{L^2(M)}(\lambda_j(k+i\beta_s) - \lambda)} \,\mathrm{d}k.$$

Now we repeat some calculations in Chapter 3 to have

$$\begin{split} u_1(x) &= \frac{1}{(2\pi)^d} \int_{\mathcal{O}} \int_M \frac{e^{ik \cdot h(x)} \eta(k) \widehat{f}(k, y) \overline{\phi_{k-i\beta_s}(y)} \phi_{k+i\beta_s}(x)}{(\phi_{k+i\beta_s}, \phi_{k-i\beta_s})_{L^2(M)} (\lambda_j(k+i\beta_s) - \lambda)} \, \mathrm{d}\mu_M(y) \, \mathrm{d}k \\ &= \frac{1}{(2\pi)^d} \int_{\mathcal{O}} \int_{\overline{F(M)}} \sum_{\gamma \in G} \frac{e^{ik \cdot (h(x) - h(\gamma^{-1} \cdot y))} \eta(k) \overline{\phi_{k-i\beta_s}(y)} \phi_{k+i\beta_s}(x)}{(\phi_{k+i\beta_s}, \phi_{k-i\beta_s})_{L^2(M)} (\lambda_j(k+i\beta_s) - \lambda)} \, \mathrm{d}\mu_X(y) \, \mathrm{d}k \\ &= \frac{1}{(2\pi)^d} \int_{\mathcal{O}} \sum_{\gamma \in G} \int_{\gamma \cdot \overline{F(M)}} f(y) \frac{e^{ik \cdot (h(x) - h(y))} \eta(k) \overline{\phi_{k-i\beta_s}(\gamma^{-1} \cdot y)} \phi_{k+i\beta_s}(x)}{(\phi_{k+i\beta_s}, \phi_{k-i\beta_s})_{L^2(M)} (\lambda_j(k+i\beta_s) - \lambda)} \, \mathrm{d}\mu_X(y) \, \mathrm{d}k \\ &= \frac{1}{(2\pi)^d} \int_X f(y) \left(\int_{\mathcal{O}} \frac{e^{ik \cdot (h(x) - h(y))} \eta(k) \overline{\phi_{k-i\beta_s}(y)} \phi_{k+i\beta_s}(x)}{(\phi_{k+i\beta_s}, \phi_{k-i\beta_s})_{L^2(M)} (\lambda_j(k+i\beta_s) - \lambda)} \, \mathrm{d}k \right) \, \mathrm{d}\mu_X(y). \end{split}$$

In the second equality above, we use the identity (4.2).

Consequently, from (4.18), we conclude that our reduced Green's function is

$$G_0(x,y) = \frac{1}{(2\pi)^d} \int_{\mathcal{O}} e^{ik \cdot (h(x) - h(y))} \eta(k) \frac{\phi_{k+i\beta_s}(x) \overline{\phi_{k-i\beta_s}(y)}}{(\phi_{k+i\beta_s}, \phi_{k-i\beta_s})_{L^2(M)} (\lambda_j(k+i\beta_s) - \lambda)} \,\mathrm{d}k.$$
(4.19)

4.5 Some auxiliary statements

In this part, we provide the analogs of some results from Chapter 3 and [52], which do not require any significant change in the proofs when dealing with the case of abelian coverings. Instead of repeating the details, we will make brief comments.

The first result studies the local smoothness in (z, x) of the eigenfunctions $\phi_z(x)$ of the operator L(z) with the eigenvalue $\lambda_j(z)$.

Lemma 4.5.1. Suppose that $B \subset \mathbb{R}^d$ is the open ball centered at k_0 with radius r_0 (see (4.6)). Then for each $s \in \mathbb{S}^{d-1}$, the functions $\phi_{k\pm i\beta_s}(x)$ are smooth on a neighborhood of $\overline{B} \times M$ in $\mathbb{R}^d \times M$. In addition, for any multi-index α , the functions $D_k^{\alpha} \phi_{k\pm i\beta_s}(x)$ are also

jointly continuous in (s, k, x). In particular, we have

$$\sup_{(s,k,x)\in\mathbb{S}^{d-1}\times\overline{B}\times M}|D_k^{\alpha}\phi_{k\pm i\beta_s}(x)|<\infty.$$

To obtain Lemma 4.5.1, one can modify the proof of Proposition 3.9.6 without any significant change. Indeed, the three main ingredients in the proof are the smoothness in z of the family of operators $\{L(z)\}_{z\in V}$ acting between Sobolev spaces (Lemma 4.2.13), the property (**P3**) for bootstrapping regularity of eigenfunctions in k, and the standard coercive estimates of elliptic operators L(z) on the compact manifold M (see e.g., [76, estimate (11.29)]) for bootstrapping regularity in x.

The next result is the asymptotics of the scalar integral expression obtained from the integral representation (4.19) of the reduced Green's function G_0 .

Proposition 4.5.2. Suppose that $d \ge 2$ and B is the open ball defined in Proposition 4.5.1. Let $\eta(k)$ be a smooth cut off function around the point k_0 , and $\{\mu_s(k, x, y)\}_{s\in\mathbb{S}^{d-1}}$ be a family of smooth \mathbb{C}^d -valued functions defined on $\overline{B} \times M \times M$. We also use the same notation $\mu_s(k, x, y)$ for its lift to $\overline{B} \times X \times X$. For each quadruple $(s, a, x, y) \in$ $\mathbb{S}^{d-1} \times \mathbb{R}^d \times X \times X$, we define

$$I(s,a) := \frac{1}{(2\pi)^d} \int_{\mathcal{O}} e^{ik \cdot a} \frac{\eta(k)}{\lambda_j (k+i\beta_s) - \lambda} \, \mathrm{d}k$$

and

$$J(s, a, x, y) := \frac{1}{(2\pi)^d} \int_{\mathcal{O}} e^{ik \cdot a} \frac{\eta(k)(k-k_0) \cdot \mu_s(k, x, y)}{\lambda_j(k+i\beta_s) - \lambda} \, \mathrm{d}k.$$

Assume that the size of the support of η is small enough. Fix a direction $s \in \mathbb{S}^{d-1}$ and

consider all vectors a such that $s = \frac{a}{|a|}$. Then when |a| is large enough, we have

$$I(s,a) = \frac{e^{ik_0 \cdot a} |\nabla E(\beta_s)|^{(d-3)/2}}{(2\pi|a|)^{(d-1)/2} \det\left(-\mathcal{P}_s \operatorname{Hess}\left(E\right)(\beta_s)\mathcal{P}_s\right)^{1/2}} + O(|a|^{-d/2})$$
(4.20)

and

$$\sup_{(x,y)\in X\times X} |J(s,a,x,y)| = O(|a|^{-d/2}).$$
(4.21)

Moreover, if all derivatives of $\mu_s(k, x, y)$ with respect to k are uniformly bounded in $s \in \mathbb{S}^{d-1}$, then all the terms $O(\cdot)$ in (4.20) and (4.21) are also uniform in $s \in \mathbb{S}^{d-1}$ when $|a| \to \infty$.

The proof of Proposition 4.5.2 can be extracted from Section 3.6. The main ingredient (see Proposition 3.6.1) is an application of the Weierstrass Preparation Lemma in several complex variables to have a factorization of the denominator $\lambda_j(k + i\beta_s) - \lambda$ of the integrands of I, J into a form that is close to the normal form in the free case. This trick was used in [81] in the discrete setting.

The next result [52, Theorem 3.3] will be needed in the proof of Theorem 4.3.4.

Proposition 4.5.3. Assume $d \ge 3$. Let $a \in \mathbb{R}^d$. Let η be a smooth function satisfying the assumptions of Proposition 4.5.2, and let $\mu(k, x, y)$ be a smooth *G*-periodic function from a neighborhood of $\overline{B} \times X \times X$ to \mathbb{C}^d . Then the following asymptotics hold when $|a| \to \infty$:

$$\frac{1}{(2\pi)^d} \int_{\mathcal{O}} e^{ik \cdot a} \frac{\eta(k)}{\lambda_j(k)} \, \mathrm{d}k = \frac{\Gamma(\frac{d}{2}-1)e^{ik_0 \cdot a}}{2\pi^{d/2} (\det H)^{1/2} |H^{-1/2}(a)|^{d-2}} (1+O(|a|^{-1}),$$

and

$$\sup_{x,y\in X} \left| \int_{\mathcal{O}} e^{ik\cdot a} \frac{\eta(k)(k-k_0)\cdot\mu(k,x,y)}{\lambda_j(k)} \,\mathrm{d}k \right| = O(|a|^{-d+1}).$$

Here the notation $\Gamma(z)$ means the Gamma function $\Gamma(z) = \int_0^\infty x^{z-1} e^{-x} dx$.

4.6 **Proofs of the main results**

Proof of Theorem 4.3.1.

Proof. We fix an admissible direction s of the additive function h and consider any $x, y \in X$ such that

$$\frac{h(x) - h(y)}{|h(x) - h(y)|} = s \in \mathcal{A}_h.$$

As we discussed in Section 4.4, the Green's function G_{λ} satisfies

$$G_{\lambda}(x,y) = e^{\beta_s \cdot (h(y) - h(x))} G_{s,\lambda}(x,y), \qquad (4.22)$$

where $G_{s,\lambda}$ is the Schwartz kernel of the resolvent operator R_s . Also, $R_{s,\lambda} = V_s + T_s$. Due to Theorem 4.4.6, the Schwartz kernel of T_s decays rapidly (uniformly in s) when $d_X(x,y)$ is large enough. Hence, to find the asymptotics of the kernel of $R_{s,\lambda}$, it suffices to consider the kernel G_0 of the operator V_s . Define

$$a := h(x) - h(y)$$
 (4.23)

and

$$\tilde{\mu}_{\omega}(k,p,q) := \frac{\phi_{k+i\beta_{\omega}}(p)\overline{\phi_{k-i\beta_{\omega}}(q)}}{(\phi_{k+i\beta_{\omega}},\phi_{k-i\beta_{\omega}})_{L^{2}(M)}}, \quad (\omega,p,q) \in \mathbb{S}^{d-1} \times M \times M$$

By Lemma 4.5.1, $\tilde{\mu}_{\omega}$ is a smooth function on $\overline{B} \times M \times M$. By Taylor expanding around k_0 , $\tilde{\mu}_{\omega}(k, p, q) = \tilde{\mu}_{\omega}(k_0, p, q) + (k - k_0) \cdot \mu_{\omega}(k, p, q)$ for some smooth \mathbb{C}^d -valued function $\mu_{\omega}(k, p, q)$ defined on $\overline{B} \times M \times M$. From Lemma 4.5.1 and the definition of $\tilde{\mu}_{\omega}$,

$$\sup_{(\omega,k,x)\in\mathbb{S}^{d-1}\times\overline{B}\times M}\left|D_{k}^{\alpha}\tilde{\mu}_{\omega}(k,x,y)\right|<\infty,$$

for any multi-index α . Thus, all derivatives of μ_{ω} with respect to k are also uniformly

bounded in $\omega \in \mathbb{S}^{d-1}$. We now can rewrite (4.19) as follows:

$$\begin{aligned} G_0(x,y) &= \frac{1}{(2\pi)^d} \int_{\mathcal{O}} e^{ik \cdot a} \frac{\eta(k)}{\lambda_j(k+i\beta_s) - \lambda} \left(\tilde{\mu}_s(k_0, x, y) + (k-k_0) \cdot \mu_s(k, x, y) \right) dk \\ &= I(s,a) \frac{\phi_{k_0+i\beta_s}(x) \overline{\phi_{k_0-i\beta_s}(y)}}{(\phi_{k_0+i\beta_s}, \phi_{k_0-i\beta_s})_{L^2(M)}} + J(s, a, x, y). \end{aligned}$$

Here the integrals I(s, a) and J(s, a, x, y) are defined in Proposition 4.5.2. Applying Proposition 4.5.2, we obtain the following asymptotics whenever |a| is large enough:

$$G_{0}(x,y) = \left(\frac{e^{ik_{0}\cdot a}|\nabla E(\beta_{s})|^{(d-3)/2}}{(2\pi|a|)^{(d-1)/2}\det\left(-\mathcal{P}_{s}\operatorname{Hess}\left(E\right)(\beta_{s})\mathcal{P}_{s}\right)^{1/2}} + O(|a|^{-d/2})\right) \times \frac{\phi_{k_{0}+i\beta_{s}}(x)\overline{\phi_{k_{0}-i\beta_{s}}(y)}}{(\phi_{k_{0}+i\beta_{s}},\phi_{k_{0}-i\beta_{s}})_{L^{2}(M)}} + O(|a|^{-d/2}),$$
(4.24)

where all the terms $O(\cdot)$ are uniform in s. Due to (4.23) and Proposition 4.2.8, $O(|a|^{\ell}) = O(d_X(x,y)^{\ell})$ for any $\ell \in \mathbb{Z}$, provided that $d_X(x,y) > R_h$. Hence, by choosing the constant R_h larger if necessary, we can assume that when $d_X(x,y) > R_h$, the asymptotics (4.24) would follow. Finally, we substitute (4.23) to the asymptotics (4.24) and then use (4.22) to deduce Theorem 4.3.1.

Proof of Theorem 4.3.4.

Proof. We recall that $\lambda = \lambda_j(k_0) = 0$ and $R_{-\varepsilon}$ is the resolvent operator $(L + \varepsilon)^{-1}$ when $\varepsilon > 0$ is small enough. We will repeat the Floquet reduction approach in Section 4.4. Given any $f, \varphi \in L^2_{comp}(X)$, the sesquilinear form $\langle R_{-\varepsilon}f, \varphi \rangle$ is

$$(2\pi)^{-d} \int_{\mathcal{O}} \left((L(k) + \varepsilon)^{-1} \widehat{f}(k), \widehat{\varphi}(k) \right) \mathrm{d}k.$$

The first conclusion of this theorem is achieved by a similar argument in [52, Lemma 2.3]. Hence the operator $R = \lim_{\epsilon \to 0^+} R_{-\epsilon}$ is defined by the identity $\widehat{Rf}(k) = R(k)\widehat{f}(k)$ and the Green's function G is the Schwartz kernel of the operator R. To single out the principal term in R, we first choose a neighborhood $V \subset O$ of k_0 such that when $k \in V$, there is a non-zero G-periodic eigenfunction $\phi_k(x)$ of the operator L(k) with the corresponding eigenvalue $\lambda_j(k)$ and moreover, the mapping $k \mapsto \phi_k(\cdot)$ is analytic in k as a $H^2(M)$ valued function. For such $k \in V$, let us denote by P(k) the spectral projector of L(k) that projects $L^2(M)$ onto the eigenspace spanned by ϕ_k . The notation R(I - P(k)) stands for the range of the projector I - P(k). Then we pick η as a smooth cut off function around k_0 such that $\operatorname{supp}(\eta) \Subset V$. Define the operator

$$T := \frac{1}{(2\pi)^d} \int_{\mathcal{O}}^{\oplus} T(k) \, \mathrm{d}k,$$

where

$$T(k) := (1 - \eta(k))L(k)^{-1} + \eta(k)(L(k)|_{R(I - P(k))})^{-1}(I - P(k)).$$

As in Theorem 4.4.6, the Schwartz kernel K(x, y) of T is rapidly decaying as $d_X(x, y) \rightarrow \infty$. Thus, the asymptotics of the Green's function G are the same as the asymptotics of the Schwartz kernel G_0 of the operator R - T. To find G_0 , we repeat the arguments in Section 4.4 to derive the formula

$$G_0(x,y) = \frac{1}{(2\pi)^d} \int_{\mathcal{O}} e^{ik \cdot (h(x) - h(y))} \frac{\eta(k)}{\lambda_j(k)} \frac{\phi_k(x)\overline{\phi_k(y)}}{\|\phi_k\|_{L^2(M)}^2} dk, \quad x, y \in X$$

As in the proof of Theorem 4.3.1, we set a := h(x) - h(y) and rewrite the smooth function

$$\frac{\phi_k(x)\overline{\phi_k(y)}}{\|\phi_k\|_{L^2(M)}^2} = \frac{\phi_{k_0}(x)\overline{\phi_{k_0}(y)}}{\|\phi_{k_0}\|_{L^2(M)}^2} + (k - k_0) \cdot \mu(k, x, y),$$

for some smooth *G*-periodic function $\mu : \overline{B} \times X \times X \to \mathbb{C}^d$. Now by applying Proposition 4.5.3, the proof is completed.

4.7 **Proofs of technical statements**

4.7.1 Proof of Proposition 4.2.8

Fixing a point $x_0 \in X$, we let

$$K := \overline{F(M)},$$

$$R := \max_{x \in K} d_X(x_0, x),$$

and

$$\widetilde{R_h} := \max_{(x,y) \in K \times K} |h(x) - h(y)|.$$

Due to Proposition 4.2.2 and the fact that $|\cdot|_S$ is equivalent to $|\cdot|$ on \mathbb{Z}^d , there exist $C_1 > 1$ and $C_2 > 0$ such that

$$C_1^{-1} \cdot d_X(g_1 \cdot x_0, g_2 \cdot x_0) - C_2 \le |g_1 - g_2| \le C_1 \cdot d_X(g_1 \cdot x_0, g_2 \cdot x_0) + C_2,$$

for any $g_i \in \mathbb{Z}^d$, i = 1, 2.

Now we consider any two points x, y in X. By (4.1), we can select \tilde{x}, \tilde{y} in K such that $x = g_1 \cdot \tilde{x}$ and $y = g_2 \cdot \tilde{y}$ for some $g_1, g_2 \in \mathbb{Z}^d$. Since \mathbb{Z}^d acts by isometries, we get

$$d_X(g_1 \cdot x_0, g_1 \cdot \tilde{x}) = d_X(x_0, \tilde{x})$$
 and $d_X(g_2 \cdot x_0, g_2 \cdot \tilde{y}) = d_X(x_0, \tilde{y}).$ (4.25)

By (A.3), we have

$$h(x) - h(y) = h(\tilde{x}) - h(\tilde{y}) + g_1 - g_2.$$

Using triangle inequalities and (4.25), we obtain

$$|h(x) - h(y)| \le \widetilde{R_h} + |g_1 - g_2| \le C_1 \cdot d_X(g_1 \cdot x_0, g_2 \cdot x_0) + \widetilde{R_h} + C_2$$

$$\le C_1 \cdot d_X(x, y) + C_1 \cdot (d_X(x_0, \tilde{x}) + d_X(x_0, \tilde{y})) + \widetilde{R_h} + C_2$$

$$\le C_1 \cdot d_X(x, y) + (2C_1R + \widetilde{R_h} + C_2).$$

Likewise,

$$|h(x) - h(y)| \ge |g_1 - g_2| - \widetilde{R_h} \ge C_1^{-1} \cdot d_X(g_1 \cdot x_0, g_2 \cdot x_0) - (\widetilde{R_h} + C_2)$$

$$\ge C_1^{-1} \cdot d_X(x, y) - (C_1^{-1} \cdot (d_X(x_0, \tilde{x}) + d_X(x_0, \tilde{y})) + \widetilde{R_h} + C_2)$$

$$\ge C_1^{-1} \cdot d_X(x, y) - (2C_1R + \widetilde{R_h} + C_2).$$

The statement follows if we put $C := 2C_1$ and $R_h := 2C_1(2C_1R + \widetilde{R_h} + C_2)$.

4.7.2 Proof of Proposition 4.2.10

By Definition 4.2.7, any rational point in the unit sphere \mathbb{S}^{d-1} is an admissible direction of the additive function h and thus we have (4.4). By using the stereographic projection, one can see that the subset $\mathbb{Q}^d \cap \mathbb{S}^{d-1}$ is dense in \mathbb{S}^{d-1} . Hence, the density of \mathcal{A}_h follows. Now we consider the case d = 2. For any point $x_0 \in X$, we denote by $\mathcal{A}_h(x_0)$ the subset of \mathcal{A}_h consisting of unit vectors s such that there exists a point x in $\{x \in X \mid d_X(x, x_0) >$ $R_h\}$ satisfying either $h(x) - h(x_0) = |h(x) - h(x_0)|s$ or $h(x_0) - h(x) = |h(x) - h(x_0)|s$. It is enough to prove that for any x_0 , $\mathcal{A}_h(x_0) = \mathbb{S}^1$. Without loss of generality, we suppose that $h(x_0) = 0$. Let Y be the range of the continuous function $x \mapsto \frac{h(x)}{|h(x)|}$, which is defined on the connected set $\{x \in X \mid d_X(x, x_0) > R_h\}$. Then Y is a connected subset that contains $\mathbb{Q}^2 \cap \mathbb{S}^1$ since $h(n \cdot x_0) = n$ for any $n \in \mathbb{Z}^d$. Suppose for contradiction, there is a unit vector s such that $s \notin \mathcal{A}_h(x_0)$ and hence, $Y \subseteq \mathbb{S}^1 \setminus \{\pm s\}$. Thus, Y cannot be connected, which is a contradiction.

4.7.3 Proof of Theorem 4.4.6

It suffices to prove the following claim:

Theorem 4.7.1. Let ϕ and θ be two functions in $C_c^{\infty}(X)$ such that the metric distance on X between the supports of these two functions is bigger than R_h . Let $K_{s,\phi,\theta}$ be the Schwartz kernel of the operator $\phi T_s \theta$. Then $K_{s,\phi,\theta}$ is continuous and rapidly decaying (uniformly in s) on $X \times X$, i.e., for any N > 0, we have

$$\sup_{s \in \mathbb{S}^{d-1}} |K_{s,\phi,\theta}(x,y)| \le C(1 + d_X(x,y))^{-N},$$

for some positive constant $C = C(N, \|\phi\|_{\infty}, \|\theta\|_{\infty})$.

Let $K_s(k, x, y)$ be the Schwartz kernel of the operator $T_s(k)$. The next lemma is an analog for abelian coverings of Lemma 3.7.15.

Lemma 4.7.2. Let ϕ and θ be any two compactly supported functions on X such that $\operatorname{supp}(\phi) \cap \operatorname{supp}(\theta) = \emptyset$. Then the following identity holds for any $(x, y) \in X \times X$:

$$K_{s,\phi,\theta}(x,y) = \frac{1}{(2\pi)^d} \int_{\mathcal{O}} e^{ik \cdot (h(x) - h(y))} \phi(x) K_s(k,\pi(x),\pi(y)) \theta(y) dk$$

where π is the covering map $X \to M$.

Proof. Let \mathcal{P} be the subset of $C_c^{\infty}(X)$ consisting of all functions ψ whose support is connected, and if $\gamma \in G$ such that $\operatorname{supp} \psi^{\gamma} \cap \operatorname{supp} \psi \neq \emptyset$ then γ is the identity element of the deck group G. Since any compactly supported function on X can be decomposed as a finite sum of functions in \mathcal{P} , we can assume that both ϕ and θ belong to \mathcal{P} . Then the rest is similar to the proof of Lemma 3.7.15.

Another key ingredient in proving Theorem 4.7.1 is the following result:

Proposition 4.7.3. Let dim M = n. Then for any multi-index α such that $|\alpha| \ge n$, $D_k^{\alpha}K_s(k, x, y)$ is a continuous function on $M \times M$. Furthermore, we have

$$\sup_{(s,k,x,y)\in\mathbb{S}^{d-1}\times\mathcal{O}\times M\times M}|D_k^{\alpha}K_s(k,x,y)|<\infty$$

Before providing the proof of Proposition 4.7.3, let us use it to prove Theorem 4.7.1. *Proof of Theorem 4.7.1*.

Proof. The exponential function $e^{2\pi i\gamma \cdot h(x)}$ is *G*-periodic for any $\gamma \in G$, and hence, it is also defined on *M*. We use the same notation $e^{2\pi i\gamma \cdot h(x)}$ for the corresponding multiplication operator on $L^2(M)$. Then we can write

$$T_s(k+2\pi\gamma) = e^{-2\pi i\gamma \cdot h(x)} T_s(k) e^{2\pi i\gamma \cdot h(x)}, \quad (k,\gamma) \in \mathcal{O} \times G$$

It follows that for any multi-index α ,

$$e^{i(k+2\pi\gamma)\cdot(h(x)-h(y))}\nabla_{k}^{\alpha}K_{s}(k+2\pi\gamma,\pi(x),\pi(y)) = e^{ik\cdot(h(x)-h(y))}\nabla_{k}^{\alpha}K_{s}(k,\pi(x),\pi(y)).$$
(4.26)

Now we apply integration by parts to the identity in Lemma 4.7.2 to obtain

$$i^{N}(h(x) - h(y))^{\alpha}K_{s,\phi,\theta}(x,y) = \frac{\phi(x)\theta(y)}{(2\pi)^{d}} \int_{\mathcal{O}} e^{ik \cdot (h(x) - h(y))} \nabla_{k}^{\alpha}K_{s}(k,\pi(x),\pi(y)) \,\mathrm{d}k.$$
(4.27)

Note that due to (4.26), when using integration by parts, we do not have any boundary term. If $|\alpha| \ge n$, then the above integral is uniformly bounded in (s, x, y) by Proposition 4.7.3. When $\phi(x)\theta(y) \ne 0$, we have $d_X(x, y) > R_h$ and so, $h(x) \ne h(y)$ by Proposition 4.2.8. Therefore, the kernel $K_{s,\phi,\theta}(x, y)$ is continuous on $X \times X$. Now fix (x, y) such that $\phi(x)\theta(y) \ne 0$. Next we choose $\ell_0 \in \{1, \ldots, d\}$ such that $|h_{\ell_0}(x) -$ $h_{\ell_0}(y)| = \max_{1 \le \ell \le d} |h_\ell(x) - h_\ell(y)| > 0$. Fix any $N \ge n$. Let $\alpha = (\alpha_1, \ldots, \alpha_d) = N(\delta_{1,\ell_0}, \ldots, \delta_{d,\ell_0})$, where $\delta_{\cdot, \cdot}$ is the Kronecker delta. Then $|(h(x) - h(y))^{\alpha}|^{-1} = |h_{\ell_0}(x) - h_{\ell_0}(y)|^{-N} \le d^{N/2}|h(x) - h(y)|^{-N}$. Consequently, from (4.27), we derive a positive constant C (independent of x, y) such that

$$\sup_{s \in \mathbb{S}^{d-1}} |K_{s,\phi,\theta}(x,y)| \le C |\phi(x)\theta(y)| |(h(x)-h(y))^{\alpha}|^{-1} \le C d^{N/2} ||\phi||_{\infty} ||\theta||_{\infty} ||h(x)-h(y)|^{-N}.$$

Using Proposition 4.2.8, the above estimate becomes

$$\sup_{(s,x,y)\in\mathbb{S}^{d-1}\times X\times X} (1+d_X(x,y))^N |K_{s,\phi,\theta}(x,y)| < \infty,$$

which yields the conclusion.

Back to Proposition 4.7.3, we first introduce several notions. Let S(M) be the space of Schwartz functions on M. The first notion is about the order of an operator on the Sobolev scale (see e.g. [67, Definition 5.1.1]).

Definition 4.7.4. A linear operator $A : S(M) \to S(M)$ is said to be **of order** $\ell \in \mathbb{R}$ on the Sobolev scale $(H^m(M))_{m \in \mathbb{R}}$ if for every $m \in \mathbb{R}$ it can be extended to a bounded linear operator $A_{m,m-\ell} \in B(H^m(M), H^{m-\ell}(M))$. In this situation, we denote by the same notation A any of the operators $A_{m,m-\ell}$.

A typical example of an operator of order ℓ on the Sobolev scale is any pseudodifferential operator of order ℓ acting on M.

Definition 4.7.5. Given $\ell \in \mathbb{R}$. We denote by $S_{\ell}(M)$ the set consisting of families of operators $\{B_s(k)\}_{(s,k)\in\mathbb{S}^{d-1}\times\mathcal{O}}$ acting on M so that the following properties hold:

• For any $(s,k) \in \mathbb{S}^{d-1} \times \mathcal{O}$, $B_s(k)$ is of order ℓ on the Sobolev scale $(H^p(M))_{p \in \mathbb{R}}$.

- For any p ∈ ℝ, the operator B_s(k) is smooth in k as a B(H^p(M), H^{p-ℓ}(M))-valued function.
- For any multi-index α, D^α_kB_s(k) is of order ℓ−|α| on the Sobolev scale (H^p(M))_{p∈ℝ} and moreover, for any p ∈ ℝ, the following uniform condition holds

$$\sup_{(s,k)\in\mathbb{S}^{d-1}\times\mathcal{O}}\|D_k^{\alpha}B_s(k)\|_{B(H^p(M),H^{p-\ell+|\alpha|}(M))}<\infty.$$

It is worth giving a separate definition for the class $S_{-\infty}(M) = \bigcap_{\ell \in \mathbb{R}} S_{\ell}(M)$ as follows:

Definition 4.7.6. We denote by $\mathcal{S}_{-\infty}(M)$ the set consisting of families of smoothing operators $\{U_s(k)\}_{(s,k)\in\mathbb{S}^{d-1}\times\mathcal{O}}$ acting on M so that the following properties hold:

- For any m₁, m₂ ∈ ℝ, the operator U_s(k) is smooth in k as a B(H^{m₁}(M), H^{m₂}(M))-valued function.
- The following uniform condition holds for any multi-index α :

$$\sup_{(s,k)\in\mathbb{S}^{d-1}\times\mathcal{O}}\|D_k^{\alpha}U_s(k)\|_{B(H^{m_1}(M),H^{m_2}(M))}<\infty.$$

We now introduce the class $\tilde{S}^{\ell}(\mathbb{T}^n)$ of parameter-dependent toroidal symbols on the *n*-dimensional torus ⁹.

Definition 4.7.7. The parameter-dependent class $\tilde{S}^{\ell}(\mathbb{T}^n)$ consists of symbols $\sigma(s,k;x,\xi)$ satisfying the following conditions:

For each (s, k) ∈ S^{d-1} × O, the function σ(s, k; ·, ·) is a symbol of order ℓ on the torus Tⁿ (see e.g., Definition 3.7.3 in Chapter 3).

⁹Note that for the case n = d, the class of parameter-dependent toroidal symbols was introduced in Definition 3.7.3. Nevertheless, the techniques and results on parameter-dependent toroidal pseudodifferential operators obtained in Section 3.8 of Chapter 3 still apply for the general case $n \ge 1$.

Consider any multi-indices α, β, γ and any s ∈ S^{d-1}. Then the function σ(s, ·; ·, ·) is smooth on O×Tⁿ×Rⁿ. Furthermore, for some positive constant C_{αβγ} (independent of s,k,x,ξ), we have

$$\sup_{s\in\mathbb{S}^{d-1}} |D_k^{\alpha} D_{\xi}^{\beta} D_x^{\gamma} \sigma(s,k;x,\xi)| \le C_{\alpha\beta\gamma} (1+|\xi|)^{m-|\alpha|-|\beta|}.$$

We also define

$$\tilde{S}^{-\infty}(\mathbb{T}^n) := \bigcap_{\ell \in \mathbb{R}} \tilde{S}^{\ell}(\mathbb{T}^n).$$

The class of pseudodifferential operators on the torus \mathbb{T}^n is also provided in the next definition.

Definition 4.7.8.

Given a symbol σ(x, ξ) of order l on the torus Tⁿ, the corresponding periodic pseudodifferential operator Op(σ) is defined by

$$\left(Op(\sigma)f\right)(x):=\sum_{\xi\in\mathbb{Z}^n}\sigma(x,\xi)\tilde{f}(\xi)e^{2\pi i\xi\cdot x},$$

where $\tilde{f}(\xi)$ is the Fourier coefficient of f at ξ .

For any l∈ ℝ ∪ {-∞}, the set of all families of periodic pseudodifferential operators {Op(σ(s, k; ·, ·))}_{(s,k)∈S^{d-1}×O}, where σ runs over the class Š^ℓ(Tⁿ), is denoted by Op(Š^ℓ(Tⁿ)).

Remarks 4.7.9.

(a) It is straightforward to check from definitions and the Leibnitz rule that for any $\ell_1, \ell_2 \in \mathbb{R} \cup \{-\infty\}$, if $\{A_s(k)\}_{(s,k)\in\mathbb{S}^{d-1}\times\mathcal{O}}$, $\{B_s(k)\}_{(s,k)\in\mathbb{S}^{d-1}\times\mathcal{O}}$ are two families of operators

in the class $S_{\ell_1}(M)$ and $S_{\ell_2}(M)$, respectively, then the family $\{A_s(k)B_s(k)\}_{(s,k)\in\mathbb{S}^{d-1}\times\mathcal{O}}$ belongs to $S_{\ell_1+\ell_2}(M)$.

- (b) If the family of operators {B_s(k)}_{(s,k)∈S^{d-1}×O} belongs to the class S_ℓ(M) then by definition, the family of operators {D^α_kB_s(k)}_{(s,k)∈S^{d-1}×O} is in the class S_{ℓ-|α|}(M) for any multi-index α.
- (c) $\mathcal{S}_{-\infty}(\mathbb{T}^n)$ is the class \mathcal{S} introduced in Definition 3.7.8.
- (d) Given a family of symbols {σ(s, k; ·, ·)}_{(s,k)∈S^{d-1}×O} ∈ Š^ℓ(Tⁿ), it follows from definitions here and boundedness on Sobolev spaces of periodic pseudodifferential operators (see e.g., [67, Corollary 4.8.3]) that the corresponding family of periodic pseudodifferential operators {Op(σ(s, k; ·, ·))}_{(s,k)∈S^{d-1}×O} is in the class S_ℓ(Tⁿ). In other words, Op(Š^ℓ(Tⁿ)) ⊆ S_ℓ(Tⁿ) for any ℓ ∈ ℝ ∪ {−∞}.

Roughly speaking, the next lemma says that we can deduce regularity of the Schwartz kernel of an operator provided that it acts "nicely" on Sobolev spaces.

Lemma 4.7.10. Let A be a bounded operator in $L^2(M)$, where M is a compact ndimensional manifold. Suppose that the range of A is contained in $H^m(M)$, where m > n/2 and in addition,

$$\|Af\|_{H^m(M)} \le C \|f\|_{H^{-m}(M)} \tag{4.28}$$

for all $f \in L^2(M)$.

Then A is an integral operator whose kernel $K_A(x, y)$ is bounded and uniformly continuous on $M \times M$ and the following estimate holds:

$$|K_A(x,y)| \le \gamma_0 C,\tag{4.29}$$

where γ_0 is a constant depending only on *n* and *m*.

Proof. For the Euclidean case, this fact is shown in [1, Lemma 2.2]. To prove this on a general compact manifold, we simply choose a finite cover $\mathcal{U} = \{U_p\}$ of M with charts $U_p \cong \mathbb{R}^n$. Then fix a smooth partition of unity $\{\varphi_p\}$ with respect to the cover \mathcal{U} , i.e., $\operatorname{supp} \varphi_p \Subset U_p$. We decompose $A = \sum_{p,q} \varphi_p A \varphi_q$. Given any $f \in L^2(M)$, the estimate (4.28) will imply the estimate $\|\varphi_p A \varphi_q f\|_{H^m(U_p)} \leq C \|f\|_{H^{-m}(U_q)}$ for any p, q. Hence, we obtain the conclusion of the lemma for the kernel of each operator $\varphi_p A \varphi_q$, and thus for kernel of A too.

In what follows, we will show a nice behavior of kernels of families of operators in the class $S_{\ell}(M)$ following from an application of the previous lemma.

Corollary 4.7.11. Let $\ell \in \mathbb{R} \cup \{-\infty\}$. If $\{B_s(k)\}_{(s,k)}$ is a family of operators in $S_\ell(M)$, then the Schwartz kernel $K_{B_s}(k, x, y)$ of the operator $B_s(k)$ satisfies

$$\sup_{s,k,x,y} |D_k^{\alpha} K_{B_s}(k,x,y)| < \infty,$$

for any multi-index α satisfying $|\alpha| \ge n + \ell + 2$.

Proof. For such $|\alpha| \ge n + \ell + 2$, we pick some integer $m \in (n/2, (-\ell + |\alpha|)/2]$. Then by Definition 4.7.5, we have

$$\sup_{s,k} \|D_k^{\alpha} B_s(k) f\|_{H^m(M)} \le C_{\alpha} \|f\|_{H^{-m}(M)}.$$

Applying Lemma 4.7.10, the estimates (4.29) hold for kernels $D_k^{\alpha} K_{B_s}(k, x, y)$ of the operators $D_k^{\alpha} B_s(k)$ uniformly in (s, k).

The next theorem shows the inversion formula (i.e., the existence of a family of parametrices) in the case of \mathbb{T}^n . The proof of this theorem just comes straight from the proof of Theorem 3.7.6. We omit the details. **Theorem 4.7.12.** Let $r \in \mathbb{N}$. Let us consider a family of $2r^{th}$ order elliptic operators $\{(\mathcal{Q}_s(k)\}_{(s,k)\in\mathbb{S}^{d-1}\times\mathcal{O}} \text{ on the torus } \mathbb{T}^n$. Assume that this family is in $Op(\tilde{S}^{2r}(\mathbb{T}^n))$ and moreover, for each $(s,k)\in\mathbb{S}^{d-1}\times\mathcal{O}$, the symbol $\sigma(s,k;x,\xi)$ of the operator $\mathcal{Q}_s(k)$ is of the form

$$\sigma(s,k;x,\xi) = L_0(s,k;x,\xi) + \tilde{\sigma}(s,k;x,\xi),$$

where the families of parameter-dependent symbols $\{L_0(s,k;x,\xi)\}_{(s,k)}, \{\tilde{\sigma}(s,k;x,\xi)\}_{(s,k)}$ are in the class $\tilde{S}^{2r}(\mathbb{T}^n)$ and $\tilde{S}^{2r-1}(\mathbb{T}^n)$, respectively. Moreover, suppose that there is some constant A > 0 such that whenever $|\xi| > A$, we have

$$|L_0(s,k;x,\xi)| \ge 1, \quad (s,k,x) \in \mathbb{S}^{d-1} \times \mathcal{O} \times \mathbb{T}^n.$$

We call $L_0(s, k; x, \xi)$ the "leading part" of the symbol $\sigma(s, k; x, \xi)$.

Then there exists a family of parametrices $\{\mathcal{A}_s(k)\}_{(s,k)}$ in $Op(\tilde{S}^{-2r}(\mathbb{T}^n))$ such that

$$\mathcal{Q}_s(k)\mathcal{A}_s(k) = I - \mathcal{R}_s(k)$$

where $\mathcal{R}_s(k)$ is some family of smoothing operators in the class $\mathcal{S}_{-\infty}(\mathbb{T}^n)$.

To build a family of parametrices on a compact manifold, we will follow closely the strategy in [36] by working on open subsets of the torus first and then gluing together to get the final global result.

Theorem 4.7.13. There exists a family of operators $\{A_s(k)\}_{(s,k)\in\mathbb{S}^{d-1}\times\mathcal{O}}$ in $\mathcal{S}_{-2}(M)$ and a family of operators $\{R_s(k)\}_{(s,k)\in\mathbb{S}^{d-1}\times\mathcal{O}}$ in $\mathcal{S}_{-\infty}(M)$ such that

$$(L_s(k) - \lambda)A_s(k) = I - R_s(k).$$

Proof. Let V_p (p = 1, ..., N) be a finite covering of the compact manifold M by evenly

covered coordinate charts. We also choose an open covering U_p (p = 1, ..., N) that refine the covering $\{V_p\}$ such that $\overline{U_p} \subset V_p$ for any p. We can assume that each V_p is an open subset of $(0, 2\pi)^n$ in \mathbb{R}^n and hence, we can view each V_p as an open subset of the torus \mathbb{T}^n .

To simplify the notation, we will suppress the index p = 1, ..., N which specifies the open sets V_p , U_p until the final steps of the proof. Let us denote by i_U , r_U the inclusion mapping from $i_U : U \to \mathbb{T}^n$ and the restriction mapping $r_U : C^{\infty}(\mathbb{T}^n) \to C^{\infty}(U)$, correspondingly. We also use the same notation $L_s(k) - \lambda$ for its restrictions to the coordinate charts V, U if no confusion arises. Then $(L_s(k) - \lambda)r_U$ can be considered as an operator on \mathbb{T}^n .

Let us first establish the following localized version of the inversion formula

Lemma 4.7.14. There are two families of symbols $\{a(s,k;x,\xi)\}_{(s,k)\in\mathbb{S}^{d-1}\times\mathcal{O}}$ in $\tilde{S}^{-2}(\mathbb{T}^n)$ and $\{r(s,k;x,\xi)\}_{(s,k)\in\mathbb{S}^{d-1}\times\mathcal{O}}$ in $\tilde{S}^{-\infty}(\mathbb{T}^n)$ so that

$$(L_s(k) - \lambda)r_U\mathcal{A}_s(k) = r_U(I - \mathcal{R}_s(k)),$$

where $\mathcal{A}_s(k) = Op(a(s,k;\cdot,\cdot)), \mathcal{R}_s(k) = Op(r(s,k;\cdot,\cdot)).$

Proof. We denote by $(L_s(k) - \lambda)^T$ the transpose operator of $(L_s(k) - \lambda)$ on V. Now let ν be a function in $C_c^{\infty}(V)$ such that $\nu = 1$ in a neighborhood of \overline{U} and $0 \le \nu \le 1$. Define

$$\mathcal{Q}_s(k) = (L_s(k) - \lambda)(L_s(k) - \lambda)^T \nu + (1 - \nu)\Delta^2.$$

Observe that each operator $Q_s(k)$ is a globally defined 4^{th} order differential operator on \mathbb{T}^n with the following principal symbol

$$\nu(x)|\sigma_0(s,k;x,\xi)|^2 + (1-\nu(x))|\xi|^4.$$

Here $\sigma_0(s, k; x, \xi)$ is the non-vanishing symbol of the elliptic operator $L_s(k) - \lambda$. Thus, each operator $\mathcal{Q}_s(k)$ is an elliptic differential operator on \mathbb{T}^n . In order to apply Theorem 4.7.12 to the family $\{\mathcal{Q}_s(k)\}_{(s,k)}$, we need to study its family of symbols $\{\sigma(s, k; x, \xi)\}_{(s,k)}$.

On the evenly covered chart V, we can assume that the operator $L_s(k) - \lambda$ is of the form

$$\sum_{|\alpha| \le 2} a_{\alpha}(x) (D + (k + i\beta_s)^T \cdot \nabla \tilde{h})^{\alpha},$$

for some functions $a_{\alpha} \in C^{\infty}(V)$ and \tilde{h} is a smooth function obtained from the additive function h through some coordinate transformation on the chart V. Similarly, since $(L_s(k) - \lambda)^T = L(k - i\beta_s) - \lambda$, one can write the operator $(L_s(k) - \lambda)^T$ on V as follows:

$$\sum_{|\alpha| \le 2} \tilde{a}_{\alpha}(x) (D + (k - i\beta_s)^T \cdot \nabla \tilde{h})^{\alpha},$$

for some functions $\tilde{a}_{\alpha} \in C^{\infty}(V)$. Then, on \mathbb{T}^n , the operator $\mathcal{Q}_s(k)$ has the following form:

$$\sum_{|\alpha|,|\beta|\leq 2} a_{\alpha}(x)\tilde{a}_{\beta}(x)(D+(k+i\beta_s)^T\cdot\nabla\tilde{h})^{\alpha}(D+(k-i\beta_s)^T\cdot\nabla\tilde{h})^{\beta}\nu(x)+(1-\nu(x))\Delta^2.$$

Put

$$L_0^{(1)}(s,k;x,\xi) := \sum_{|\alpha|=2} a_{\alpha}(x)(\xi + (k+i\beta_s)^T \cdot \nabla \tilde{h})^{\alpha},$$
$$L_0^{(2)}(s,k;x,\xi) := \sum_{|\beta|=2} \tilde{a}_{\beta}(x)(\xi + (k-i\beta_s)^T \cdot \nabla \tilde{h})^{\beta}$$

and

$$L_0(s,k;x,\xi) = \nu(x)L_0^{(1)}(s,k;x,\xi)L_0^{(2)}(s,k;x,\xi) + (1-\nu(x))|\xi|^4.$$
(4.30)
Then the symbol $\sigma(s,k;x,\xi)$ of the operator $\mathcal{Q}_s(k)$ can be written as

$$L_0(s,k;x,\xi) + \tilde{\sigma}(s,k;x,\xi),$$

where the family of symbols $\{\tilde{\sigma}(s,k;x,\xi)\}_{(s,k)}$ is in the class $\tilde{S}^3(\mathbb{T}^n)$. Using the boundedness of $\nabla \tilde{h}$ and coefficients a_{α} on the support of ν , we deduce that the family of the symbols of $\{\mathcal{Q}_s(k)\}_{(s,k)}$ is in $\tilde{S}^4(\mathbb{T}^n)$. Thus, our remaining task is to find a constant A > 0such that whenever $|\xi| > A$, we obtain $|L_0(s,k;x,\xi)| > 1$. Note that by ellipticity, there are positive constants θ_1 , θ_2 such that

$$\sum_{|\alpha|=2} a_{\alpha}(x) \ge \theta_1 |\xi|^2$$

and

$$\sum_{|\alpha|=2} \tilde{a}_{\alpha}(x) \ge \theta_2 |\xi|^2.$$

We define

$$\|a\|_{\infty} := \sum_{|\alpha|=|\beta|=2} \|a_{\alpha}(\cdot)\|_{L^{\infty}(\operatorname{supp}(\nu))} + \|\tilde{a}_{\beta}(\cdot)\|_{L^{\infty}(\operatorname{supp}(\nu))}$$

and

$$A_p := \max_{(s,k,x)\in\mathbb{S}^{d-1}\times\mathcal{O}\times\operatorname{supp}(\nu)} \left(|k^T \cdot \nabla \tilde{h}|^2 + \theta_p^{-1} ||a||_{\infty} |\beta_s^T \cdot \nabla \tilde{h}|^2 + \theta_p^{-1} \right), \quad p = 1, 2.$$

Suppose that $|\xi|^2 > 2 \max_{p=1,2} A_p$, then for any p = 1, 2, we have

$$\begin{split} \sqrt{\nu(x)} |L_0^{(p)}(s,k;x,\xi)| &\geq \Re\left(\sqrt{\nu(x)}L_0^{(p)}(s,k;x,\xi)\right) \\ &\geq \sqrt{\nu(x)} \left(\theta_p |\xi + k^T \cdot \nabla \tilde{h}|^2 - \sum_{|\alpha|=2} a_\alpha(x)(\beta_s^T \cdot \nabla \tilde{h})^\alpha\right) \\ &\geq \sqrt{\nu(x)} \left(\theta_p \left(\frac{|\xi|^2}{2} - |k^T \cdot \nabla \tilde{h}|^2\right) - \|a\|_\infty |\beta_s^T \cdot \nabla \tilde{h}|^2\right) \\ &\geq \sqrt{\nu(x)}. \end{split}$$

Thus, due to (4.30), if $|\xi|^2 > 2 \max_{p=1,2} A_p + 1$ then $|L_0(s,k;x,\xi)| \ge (\sqrt{\nu(x)})^2 + (1 - \nu(x))|\xi|^4 \ge 1$ as we wish. Now we are able to apply Theorem 4.7.12 to the family of operators $\{\mathcal{Q}_s(k)\}_{(s,k)}$, i.e., there are families of operators $\{\mathcal{B}_s(k)\}_{(s,k)} \in Op(\tilde{S}^{-4}(\mathbb{T}^n))$ and $\{\mathcal{R}_s(k)\}_{(s,k)} \in \mathcal{S}_{-\infty}(\mathbb{T}^n)$ such that $\mathcal{Q}_s(k)\mathcal{B}_s(k) = I - \mathcal{R}_s(k)$.

Let $\mathcal{A}_s(k) := (L_s(k) - \lambda)^T \nu \mathcal{B}_s(k)$. Since $\nu = 1$ on a neighborhood of \overline{U} , we obtain

$$r_U(I - \mathcal{R}_s(k)) = r_U \mathcal{Q}_s(k) \mathcal{B}_s(k)$$

= $r_U \left((L_s(k) - \lambda) (L_s(k) - \lambda)^T \nu \mathcal{B}_s(k) + (1 - \nu) \Delta^2 \mathcal{B}_s(k) \right)$
= $r_U (L_s(k) - \lambda) (L_s(k) - \lambda)^T \nu \mathcal{B}_s(k)$
= $(L_s(k) - \lambda) r_U (L_s(k) - \lambda)^T \nu \mathcal{B}_s(k) = (L_s(k) - \lambda) r_U \mathcal{A}_s(k).$

In addition, $\{\mathcal{A}_s(k)\}_{(s,k)} \in Op(\tilde{S}^{-2}(\mathbb{T}^n))$ according to the composition formula in Theorem 3.8.2. Hence, the lemma is proved.

Let $\mu_p \in C_c^{\infty}(U_p)$ (p = 1, ..., N) be a smooth partition of unity with respect to the cover $\{U_p\}_{p=1,...,N}$ and for any p = 1, ..., N, let ν_p be a smooth function in $C_c^{\infty}(U_p)$ such that it equals one on a neighborhood of supp (μ_p) . By Lemma 4.7.14, there are families of

operators $\{\mathcal{A}_s^{(p)}(k)\}_{(s,k)} \in Op(\tilde{S}^{-2}(\mathbb{T}^n))$ and $\{\mathcal{R}_s^{(p)}(k)\}_{(s,k)} \in \mathcal{S}_{-\infty}(\mathbb{T}^n)$ such that

$$(L_s(k) - \lambda) r_{U_p} \mathcal{A}_s^{(p)}(k) = r_{U_p} (I - \mathcal{R}_s^{(p)}(k)).$$
(4.31)

Due to pseudolocality, $(1 - \nu_p)\mathcal{A}_s^{(p)}(k)\mu_p \in \mathcal{S}_{-\infty}(\mathbb{T}^n)$. This implies that $r_{U_p}\mathcal{A}_s^{(p)}(k)\mu_p - \nu_p\mathcal{A}_s^{(p)}(k)\mu_p \in \mathcal{S}_{-\infty}(\mathbb{T}^n)$, and thus,

$$(L_s(k) - \lambda)r_{U_p}\mathcal{A}_s^{(p)}(k)\mu_p - (L_s(k) - \lambda)\nu_p\mathcal{A}_s^{(p)}(k)\mu_p \in \mathcal{S}_{-\infty}(\mathbb{T}^n).$$

By (4.31), $\mu_p - (L_s(k) - \lambda)r_{U_p}\mathcal{A}_s^{(p)}(k)\mu_p \in \mathcal{S}_{-\infty}(\mathbb{T}^n)$. Hence,

$$\mu_p I - (L_s(k) - \lambda) \nu_p \mathcal{A}_s^{(p)}(k) \mu_p \in \mathcal{S}_{-\infty}(\mathbb{T}^n).$$

Since both operators $\mu_p I$ and $(L_s(k) - \lambda)\nu_p \mathcal{A}_s^{(p)}(k)\mu_p$ are globally defined on the manifold M, it follows that

$$\sum_{p} \left(\mu_p I - (L_s(k) - \lambda) \nu_p \mathcal{A}_s^{(p)}(k) \mu_p \right) \in \mathcal{S}_{-\infty}(M).$$
(4.32)

Because $Op(\tilde{S}^{-2}(\mathbb{T}^n) \subset S_{-2}(\mathbb{T}^n)$ (see Remark 4.7.9), each family $\{\mathcal{A}_s^{(p)}(k)\}_{(s,k)}$ is in the class $S_{-2}(\mathbb{T}^n)$ for every p. Since $\{\nu_p \mathcal{A}_s^{(p)}(k)\mu_p\}_{(s,k)}$ is globally defined on M, we also have $\{\nu_p \mathcal{A}_s^{(p)}(k)\mu_p\}_{(s,k)} \in S_{-2}(M)$ for any p. Now define $A_s(k) := \sum_p \nu_p \mathcal{A}_s^{(p)}(k)\mu_p$ and $R_s(k) := I - (L_s(k) - \lambda)A_s(k)$. Then $\{A_s(k)\}_{(s,k)} \in S_{-2}(M)$ and moreover, due to (4.32), the family of operators $\{R_s(k)\}_{(s,k)}$ is in $\mathcal{S}_{-\infty}(M)$.

The statement of the following lemma is standard.

Lemma 4.7.15. Let \mathcal{M} be a compact metric space, \mathcal{D} be a domain in \mathbb{R}^m ($m \in \mathbb{N}$) and H_1 , H_2 be two infinite-dimensional separable Hilbert spaces. Let $\{T_s\}_{s\in\mathcal{M}}$ be a family of smooth maps from \mathcal{D} to $B(H_1, H_2)$ such that for any multi-index α , the map $(s, d) \mapsto D_d^{\alpha} T_s(d)$ is continuous from $\mathcal{M} \times \mathcal{D}$ to $B(H_1, H_2)$. Suppose that there is a family of maps $\{V_s\}_{s \in \mathcal{M}}$ from \mathcal{D} to $B(H_2, H_1)$ such that $V_s(d)T_s(d) = 1_{H_1}$ and $T_s(d)V_s(d) =$ 1_{H_2} for any $(s, d) \in \mathcal{M} \times \mathcal{D}$. Then for each $s \in \mathcal{M}$, the map $d \in \mathcal{D} \mapsto V_s(d)$ is smooth as a $B(H_2, H_1)$ -valued function. Furthermore for any multi-index α , the map $(s, k) \mapsto D_d^{\alpha} V_s(d)$ is continuous on $\mathcal{M} \times \mathcal{D}$ as a $B(H_2, H_1)$ -valued function.

We now go back to the family of operators $\{T_s(k)\}_{(s,k)\in\mathbb{S}^{d-1}\times\mathcal{O}}$. The next statement is the main ingredient in establishing Proposition 4.7.3.

Proposition 4.7.16. There is a family of operators $\{B_s(k)\}_{(s,k)}$ in $S_{-2}(M)$ such that the family of operators $\{T_s(k) - B_s(k)\}_{(s,k)}$ belongs to $S_{-\infty}(M)$.

Proof. Using Theorem 4.7.13, we can find a family $\{A_s(k)\}_{(s,k)} \in S_{-2}(M)$ and a family $\{R_s(k)\}_{(s,k)} \in S_{-\infty}(M)$ such that

$$(L_s(k) - \lambda)A_s(k) = I - R_s(k).$$

From the definition of $T_s(k)$, we obtain $T_s(k)(L_s(k) - \lambda) = I - \eta(k)P(k + i\beta_s)$.

Using these equalities, we deduce

$$T_s(k) = A_s(k) - \eta(k)P(k+i\beta_s)A_s(k) + T_s(k)R_s(k).$$

We recall from Section 4.4 that $P(k+i\beta_s)$ projects $L^2(M)$ onto the eigenspace spanned by the eigenfunction $\phi_{k+i\beta_s}$. Hence, its kernel is the following function

$$\frac{\phi(k+i\beta_s)(x)\overline{\phi(k-i\beta_s)(y)}}{(\phi(k+i\beta_s),\phi(k-i\beta_s))_{L^2(M)}}$$

which is smooth due to Lemma 4.5.1. Thus, the family of operators $\{\eta(k)P(k+i\beta_s)\}_{(s,k)}$

is in $\mathcal{S}_{-\infty}(M)$. Also, the family of operators $\{\eta(k)Q(k+i\beta_s)\}_{(s,k)}$ belongs to $\mathcal{S}_0(M)$.

We put $B_s(k) := A_s(k) - \eta(k)P(k + i\beta_s)A_s(k)$, then $\{B_s(k)\}_{(s,k)} \in S_{-2}(M)$. Since $T_s(k) - B_s(k) = T_s(k)R_s(k)$, the remaining task is to check that the family of operators $\{T_s(k)R_s(k)\}_{(s,k)}$ belongs to the class $S_{-\infty}(M)$.

Let us consider any two real numbers m_1 and m_2 . By Lemma 4.2.13, the operators $L_s(k) - \lambda$ and $L_s(k)Q(k + i\beta_s) - \lambda$ are smooth in k as $B(H^{m_2}(M), H^{m_2-2}(M))$ -valued functions such that their derivatives with respect to k are jointly continuous in (s, k). On the other hand, we can rewrite (see Lemma 3.65):

$$T_s(k) = (1 - \eta(k))(L_s(k) - \lambda)^{-1} + \eta(k)\lambda^{-1}P(k + i\beta_s) + \eta(k)(L_s(k)Q(k + i\beta_s) - \lambda)^{-1}.$$

Hence, by Lemma 4.7.15, $T_s(k)$ is smooth in k as a $B(H^{m_2-2}(M), H^{m_2}(M))$ -valued function and its derivatives with respect to k are jointly continuous in (s, k). Therefore, for any multi-index α , we have

$$\sup_{(s,k)\in\mathbb{S}^{d-1}\times\mathcal{O}}\|D_k^{\alpha}T_s(k)\|_{B(H^{m_2-2}(M),H^{m_2}(M))}<\infty.$$

Moreover since $\{R_s(k)\}_{(s,k)} \in S_{-\infty}(M)$, $R_s(k)$ is smooth as a $B(H^{m_1}(M), H^{m_2-2}(M))$ -valued function and for any multi-index α ,

$$\sup_{(s,k)\in\mathbb{S}^{d-1}\times\mathcal{O}}\|D_k^{\alpha}R_s(k)\|_{B(H^{m_1}(M),H^{m_2-2}(M))}<\infty$$

From the Leibniz rule, the composition $T_s(k)R_s(k)$ is smooth as a $B(H^{m_1}(M), H^{m_2}(M))$ -valued function and for any multi-index α , the following uniform condition also holds

$$\sup_{(s,k)\in\mathbb{S}^{d-1}\times\mathcal{O}}\|D_k^{\alpha}(T_s(k)R_s(k))\|_{B(H^{m_1}(M),H^{m_2}(M))}<\infty.$$

Consequently, $\{T_s(k)R_s(k)\}_{(s,k)\in\mathbb{S}^{d-1}\times\mathcal{O}}\in\mathcal{S}_{-\infty}(M)$ as we wish.

We finish this subsection.

Proof of Proposition 4.7.3.

Proof. Proposition 4.7.16 provides us with the decomposition $T_s(k) = B_s(k) + C_s(k)$, where $\{B_s(k)\}_{(s,k)} \in S_{-2}(M)$ and $\{C_s(k)\}_{(s,k)} \in S_{-\infty}(M)$. Let $K_{B_s(k)}$, $K_{C_s(k)}$ be the Schwartz kernels of $B_s(k)$ and $C_s(k)$, correspondingly. It follows from applying Corollary 4.7.11 that for any multi-index α satisfying $|\alpha| \ge n$, the kernel $D_k^{\alpha} B_s(k)$ is continuous on $M \times M$ and

$$\sup_{(s,k,x,y)} |D_k^{\alpha} K_{B_s(k)}(k,x,y)| < \infty.$$

A similar conclusion also holds for the kernels $D_k^{\alpha}C_s(k)$ and thus, for $D_k^{\alpha}K_s(k, x, y)$ too.

4.8 Concluding remarks

- The asymptotics (4.7) and (4.9) can be described in terms of the Albanese map and the Albanese pseudo-distance (see [46, Section 2]), provided that the additive function *h* is chosen to be harmonic (see also Appendix A).
- The main results in this chapter can be easily carried over to the case when the band edge occurs at finitely many quasimomenta k_0 in the Brillouin zone (instead of assuming the condition A3) by summing the asymptotics coming from all these non-degenerate isolated extrema.

It was shown recently in [25] that for a wide class of two dimensional periodic elliptic second-order operators (including the class of operators we consider in this paper and periodic magnetic Schrödinger operators in 2D), the extrema of any spectral band function (not necessarily spectral edges) are attained on a finite set of values of the quasimomentum in the Brillouin zone. • The proofs of our results go through verbatim for periodic elliptic second-order operators acting on vector bundles over the abelian covering *X*.

5. A LIOUVILLE-RIEMANN-ROCH THEOREM ON ABELIAN COVERINGS.

5.1 Introduction

The classical Riemann-Roch formula is an important result that connects the dimension of a space of holomorphic functions and the dimension of a space of meromorphic (1,0)-forms on a compact Riemann surface. These spaces are linked with the notion of "divisors", which provide constraints limiting where poles (i.e., singularities) and zeros can be and of what order (i.e., multiplicities). The Riemann-Roch formula gives an equality between the difference of these two dimensions (i.e., the index of the $\bar{\partial}$ operator) and the degree of the divisor, i.e. the total sum of multiplicities.

Since then, many generalizations of the Riemann-Roch formula have been discovered from various viewpoints. Among these, Gromov and Shubin established their Riemann-Roch type theorems for elliptic differential operators on compact and non-compact manifolds in [32,33]. Namely, these results link the dimensions of spaces of solutions of elliptic equations on compact manifolds with the prescribed divisor of allowed poles and required zeros. As in the classical case, the differences between these dimensions are the same as the degrees of the corresponding divisors. Roughly speaking, one can say that the degree of a divisor is the difference between the indices of the operators with and without required zeros and singularities.

On the other hand, Liouville type theorems count the dimension of the space of solutions growing polynomially of a given order at infinity. Such solutions can be considered as ones with poles at infinity. This dimension can be interpreted as an index, since in this situation the co-dimension of the range turns out the be equal to zero.

The obvious similarity of the questions (as well as some results and proofs) suggest to one to seek a combination of these results. This is what we attempt in this chapter. In Sections 5.2 and 5.3, we introduce the necessary preliminaries concerning Liouville and Riemann-Roch type theorems, reminding some notions and results from [49, 50] and Gromov-Shubin's works [32, 33]. In Section 5.4, we obtain the main results that combine the Liouville and Riemann-Roch statements. It is interesting that the combination is nontrivial, and the results are not easily predictable. Proofs of these main results are provided in Section 5.5. In Section 5.6, we provide the realizations of the main theorems in several specific situations. Some technical details and proofs are delegated to Section 5.7.

Besides the non-trivial combination of the Liouville and Riemann-Roch theorems, some extensions of these individual theorems are also obtained.

5.2 Some preliminaries for Liouville type results

5.2.1 Some preliminaries for periodic elliptic operators on abelian coverings

Let X be a noncompact smooth Riemannian manifold of dimension n equipped with an isometric, properly discontinuous, free, and co-compact action of a finitely generated abelian discrete group G. The compact orbit space is M := X/G.

Remark 5.2.1. No harm will be done, if the reader assumes that $X = \mathbb{R}^d$ and M is the torus $\mathbb{T}^d = \mathbb{R}^d / \mathbb{Z}^d$. The results are new in this case as well. The only warning is that in this situation the dimension of X and the rank of the group \mathbb{Z}^d coincide, while this is not required in general.

Let μ_M be the Riemannian measure of M and μ_X be its lifting to X. Thus, μ_X is a G-invariant. We also use the notation $L^2(X)$ (see Chapter 4) to denote the space of L^2 -functions on X with respect to μ_X .

We consider the G-invariant bilinear ¹ duality

$$\langle \cdot, \cdot \rangle : C_c^{\infty}(X) \times C^{\infty}(X) \to \mathbb{C}, \quad \langle f, g \rangle = \int_X f(x)g(x) \,\mathrm{d}\mu_X.$$
 (5.1)

¹One can also consider the sesquilinear form to obtain analogous results.

By continuity, we can extend it to a G-invariant bilinear non-degenerate duality

$$\langle \cdot, \cdot \rangle : L^2(X) \times L^2(X) \to \mathbb{C}.$$
 (5.2)

Let A be an elliptic operator of order m on X with smooth coefficients. In this chapter, we will be assuming that A is G-periodic. Note that then A can be pushed down to an elliptic operator on M. We will assume in most cases (except rare non-self-adjoint considerations) that the operator A is bounded below.

The formal adjoint operator (transpose) A^* (with respect to the bilinear duality (5.1)) has similar properties. In particular, A^* is also a periodic elliptic operator of order m on X.

Note that since G is a finitely generated abelian group, G is the direct sum of a finite abelian group and \mathbb{Z}^d , where d is the rank of the torsion free subgroup of G. In what follows, without any effect on the results, we could replace M by the compact Riemannian manifold X/\mathbb{Z}^d and thus, we can work with \mathbb{Z}^d as our new deck group². Therefore, we assume henceforward that $G = \mathbb{Z}^d$, where $d \in \mathbb{N}$.

We remind the reader that the reciprocal lattice G^* of the deck group $G = \mathbb{Z}^d$ is $(2\pi\mathbb{Z})^d$ and the Brillouin zone $B = [-\pi, \pi]^d$ is chosen as its fundamental domain in this chapter. The quotient \mathbb{R}^d/G^* is the dual torus $(\mathbb{T}^*)^d$. So, G^* -periodic functions on \mathbb{R}^d can be naturally identified with functions on $(\mathbb{T}^*)^d$.

For any **quasimomentum** $k \in \mathbb{C}^d$, let γ_k be the **character** of the deck group G defined as $\gamma_k(g) = e^{ik \cdot g}$ (for a given character, the corresponding quasimomentum is defined modulo the reciprocal lattice). If k is real, γ_k is unitary and vice versa. Abusing the notations slightly, we will sometimes identify each unitary character of \mathbb{Z}^d with a quasimomentum k in the dual group $(\mathbb{T}^*)^d$.

²This is exactly what we do in Chapter 4: One can always eliminate the torsion part of G by switching to a subcovering $X \to X/\mathbb{Z}^d$.

We denote by $L_k^2(X)$ the space of all γ_k -automorphic function f(x) in $L_{loc}^2(X)$, i.e. such that

$$f(g \cdot x) = \gamma_k(g)f(x), \text{ for a.e } x \in X \text{ and } g \in G.$$
(5.3)

It is convenient at this moment to introduce, given a quasimomentum k, the following line (i.e., one-dimensional) vector bundle E_k over M:

Definition 5.2.2. Given any $k \in \mathbb{C}^d$, we denote by E_k the bundle associated with the principal *G*-bundle *X* over *M* with the character γ_k . Namely, we consider the free left action of *G* on the Cartesian product $X \times \mathbb{C}$ given by

$$g \cdot (x, z) = (g \cdot x, \gamma_k(g)z), \quad (g, x, z) \in G \times X \times \mathbb{C}.$$

Now E_k is defined as the orbit space of this action and the canonical projection $X \times \mathbb{C} \to M$ descends to the surjective mapping $E_k \to M$.

From the definition, E_k is a vector bundle over M with fiber \mathbb{C} (see e.g., [54]) and any L^2 -section of the bundle E_k must satisfy the quasi-periodic property (5.3). Now the space $L_k^2(X)$ can be identified with the space of all L^2 -sections of the bundle E_k .

This construction can be easily generalized to Sobolev spaces:

Definition 5.2.3. For any quasimomentum $k \in \mathbb{C}^d$ and any real number s, we denote by $H_k^s(X)$ the closed subspace of $H_{loc}^s(X)$ consisting of γ_k -automorphic functions. Then $H_k^s(X)$ is a Hilbert space equipped with a natural inner product arising from the inner product of the Sobolev space $H^s(\overline{\mathcal{F}})$, where \mathcal{F} is any fixed fundamental domain for the action of the group G on X. The topology of the Hilbert space $H_k^s(X)$ does not depend on the choice of the fundamental domain \mathcal{F} .

Equivalently, the space $H_k^s(X)$ can be identified with $H^s(E_k)$, the space of all H^s -sections of the bundle E_k .

For any k, the periodic operator A maps $H_k^m(X)$ into $L_k^2(X)$. Thus, the restrictions of A to these subspaces $H_k^m(X)$ define elliptic operators A(k) on the spaces of sections of the bundles E_k over the compact manifold M. When A is self-adjoint and k is real, the operator A(k), with the space $H^m(E_k)$ as the domain, is an unbounded, bound from below self-adjoint operator in $L^2(E_k)$.

In this chapter, we use a version of the Floquet transform introduced in the previous chapters as follows

$$f(x) \mapsto \mathbf{F}f(k,x) = \sum_{g \in G} f(g \cdot x)\overline{\gamma_k(g)} = \sum_{g \in G} f(g \cdot x)e^{-ik \cdot g}, \quad k \in \mathbb{C}^d.$$
(5.4)

For reader's convenience, we recall some basic properties of this Floquet transform in Section 5.7. As one would expect, this transform decomposes the original operator A on the non-compact manifold X into a direct integral of operators A(k) acting on sections of line bundles E_k over torus, which is compact:

$$A = \int_{(\mathbb{T}^*)^d}^{\oplus} A(k) \, \mathrm{d}k, \quad L^2(X) = \int_{(\mathbb{T}^*)^d}^{\oplus} L^2(E_k) \, \mathrm{d}k.$$
(5.5)

Here the measure dk is the normalized Haar measure of $(\mathbb{T}^*)^d$, which can be also considered as the normalized Lebesgue measure on the Brillouin zone B. Therefore, the union of the spectra of operators A(k) over the torus $(\mathbb{T}^*)^d$ is the spectrum of the periodic operator A. Finally, the notions of Bloch variety, dispersion relations, Fermi surfaces of the operator A are defined similarly as in Chapter 2 and Chapter 4 (see Definition 2.2.5). Note that most of the fundamental properties of these notions in the flat case still hold in the abelian covering case, e.g., Proposition 2.2.6.

5.2.2 Floquet-Bloch solutions and Liouville theorem on abelian coverings

In this section, we introduce the notions of Bloch and Floquet solutions of periodic PDEs and then state the Liouville theorem of [50].

Definition 5.2.4. For any $g \in G$ and quasimomentum $k \in \mathbb{C}^d$, we denote by $\Delta_{g;k}$ the "k-twisted" version of the first difference operator acting on functions on the covering X as follows

$$\Delta_{g;k}u(x) = e^{-ik \cdot g}u(g \cdot x) - u(x).$$
(5.6)

The iterated "twisted" finite differences of order N with quasimomentum k are defined as

$$\Delta_{g_1,\dots,g_N;k} = \Delta_{g_1;k}\dots\Delta_{g_N;k}, \quad \text{for} \quad g_1,\dots,g_N \in G.$$
(5.7)

Definition 5.2.5. A function u on X is a **Floquet function of order** N with quasimomentum k if any finite difference of order N + 1 with quasimomentum k annihilates u. Also, a **Bloch function with quasimomentum** k is a Floquet function of order 0 with quasimomentum k.

According to this definition, a Bloch function u(x) with quasimomentum k is a γ_k automorphic function on X, i.e., $u(g \cdot x) = e^{ik \cdot g}u(x)$ for any $g \in G$. Thus, if u is a continuous Bloch function with a *real* quasimomentum, then $u \in L^2_{loc}(X)$ and for any compact subset K of X, the sequence $\{||u||_{L^2(gK)}\}_{g \in G}$ is bounded (i.e., belongs to $\ell^{\infty}(G)$).

It is also known [50] that u(x) is a Floquet function of order N with quasimomentum k if and only if u can be represented in the form

$$u(x) = e_k(x) \left(\sum_{|j| \le N} [x]^j p_j(x) \right),$$

where $j = (j_1, \ldots, j_d) \in \mathbb{Z}_+^d$, and the functions p_j are *G*-periodic. Here for any $j \in \mathbb{Z}^d$,

we define

$$|j| := |j_1| + \dots |j_d|, \tag{5.8}$$

while $e_k(x)$ and $[x]^j$ are analogs of the exponential e^{ikx} and the monomial x^j on \mathbb{R}^d (see [50] for details). For convenience, in Section 5.7, we collect some basic facts of Floquet functions on abelian coverings. In the flat case $X = \mathbb{R}^d$, a Floquet function of order N with quasimomentum k is the product of the plane wave e^{ikx} and a polynomial of degree N with G-periodic coefficients.

An important consequence of this representation is that any Floquet function $u(x) \in L^2_{loc}(X)$ of order N with a real quasimomentum satisfies the following L^2 -growth estimate

$$||u||_{L^2(gK)} \le C(1+|g|)^N, \quad \forall g \in G \quad \text{and} \quad K \Subset X.$$

Here |g| is defined according to (5.8), and we have used the Švarc-Milnor lemma from geometric group theory (see e.g., [57, Lemma 2.8]) to conclude that on a Riemannian co-compact covering X, the Riemannian distance between any compact subset K and its g-translation gK is comparable with |g|.

If u is continuous, the above L^2 -growth estimate can be replaced by the corresponding L^{∞} -growth estimate.

We now need to introduce the spaces of polynomially growing solutions of the equation $Au = \lambda u$. To simplify the notations, we will assume from now on that $\lambda = 0$, since, as we discussed before, we can deal with the operator $A - \lambda$ instead of A (see also Definition 2.2.5).

Definition 5.2.6. Let $K \Subset X$ be a domain such that X is the union of all G-translations of K, i.e.,

$$X = \bigcup_{g \in G} gK.$$
(5.9)

For any $s, N \in \mathbb{R}$ and $1 \le p \le \infty$, we define the vector spaces

$$V_N^p(X) := \left\{ u \in C^{\infty}(X) \mid \{ \|u\|_{L^2(gK)} \cdot \langle g \rangle^{-N} \}_{g \in G} \in \ell^p(G) \right\},\$$

and

$$V_N^p(A) := \{ u \in V_N^p(X) \mid Au = 0 \}$$

Here $\langle g \rangle := (1 + |g|^2)^{1/2}$.

It is clear that these spaces are independent of the choice of the compact subset K satisfying (5.9). In particular, one can take as K a fundamental domain for G-action on X. Moreover, we have $V_{N_1}^{p_1}(X) \subseteq V_{N_2}^{p_2}(X)$ and $V_{N_1}^{p_1}(A) \subseteq V_{N_2}^{p_2}(A)$ whenever $N_1 \leq N_2$ and $p_1 \leq p_2$.

Definition 5.2.7. For $N \ge 0$, we say that the Liouville theorem of order (p, N) holds for A, if the space $V_N^p(A)$ is finite dimensional.

Now we can restate one of the main results in [50] as follows

Theorem 5.2.8. [50]

- (i) The following statements are equivalent:
 - (1) The cardinality of the real Fermi surface $F_{A,\mathbb{R}}$ is finite modulo G^* -shifts, i.e., Bloch solutions exist for only finitely many unitary characters γ_k .
 - (2) The Liouville theorem of order (∞, N) holds for A for some $N \ge 0$.
 - (3) The Liouville theorem of order (∞, N) holds for A for all $N \ge 0$.
- (ii) Suppose that the Liouville theorem holds for A. Then for any $N \in \mathbb{N}$, each solution $u \in V_N^{\infty}(A)$ can be represented as a finite sum of Floquet solutions:

$$u(x) = \sum_{k \in F_{A,\mathbb{R}}} \sum_{0 \le j \le N} u_{k,j}(x),$$
(5.10)

where each $u_{k,j}$ is a Floquet solution of order j with a quasimomentum k.

(iii) A crude estimate of the dimension of $V_N^{\infty}(A)$:

$$\dim V_N^{\infty}(A) \le \binom{d+N}{N} \cdot \sum_{k \in F_{A,\mathbb{R}}} \dim \operatorname{Ker} A(k) < \infty.$$

Due to the relation between $F_{A,\mathbb{R}}$ and $F_{A^*,\mathbb{R}}$ (see Proposition 2.2.6), the Liouville theorem holds for A if and only if it also holds for A^* .

5.2.3 Explicit formulas for dimensions of spaces $V_N^{\infty}(A)$

In order to obtain explicit formulas for the dimensions of $V_N^{\infty}(A)$, we need to remind some notions from [50]. Recall that, for each quasimomentum k, A(k) is in the space $\mathcal{L}(H_k^m(X), L_k^2(X))$ of bounded linear operators acting from $H_k^m(X)$ to $L_k^2(X)$. For a real number s, the spaces $H_k^s(X)$ are the fibers of the following analytic G^* -periodic Hilbert vector bundle over \mathbb{C}^d :

$$\mathcal{E}^s := \bigcup_{k \in \mathbb{C}^d} H^s_k(X) = \bigcup_{k \in \mathbb{C}^d} H^s(E_k).$$
(5.11)

Consider a quasimomentum k_0 in $F_{A,\mathbb{R}}$. We can trivialize ³ the vector bundle \mathcal{E}^m , so that in a neighborhood of k_0 , A(k) becomes an analytic family of bounded operators from $H_{k_0}^m$ to $L_{k_0}^2$ (see Subsection 5.7.2). Suppose that the spectra of operators A(k) are discrete for any value of the quasimomentum k.

Assume now that zero is an eigenvalue of the operator $A(k_0)$ with algebraic multiplicity r. Let Υ be a contour in \mathbb{C} separating 0 from the rest of the spectrum of $A(k_0)$. Due to Proposition 5.7.3, we can pick a small neighborhood of k_0 such that the contour Υ does not intersect with $\sigma(A(k))$ for any k in this neighborhood. We denote by $\Pi(k)$ the

³In fact, \mathcal{E}^m is analytically trivializable (see e.g., [47, 82]) although we do not need this fact here.

r-dimensional spectral Riesz projector for the operator A(k), associated with the contour Υ . Now one can pick an orthonormal basis $\{e_j\}_{1 \le j \le r}$ in the range of $\Pi(k_0)$ and define $e_j(k) := \Pi(k)e_j$. Then let us consider the $r \times r$ matrix $\lambda(k)$ of the operator $A(k)\Pi(k)$ in the basis $\{e_j(k)\}$, i.e.,

$$\lambda_{ij}(k) = \langle A(k)e_j(k), e_i(k) \rangle = \langle A(k)\Pi(k)e_j, e_i \rangle.$$
(5.12)

Our considerations in this part will not change if we multiply $\lambda(k)$ by an invertible matrix function analytic in a neighborhood of k_0 .

Remark 5.2.9. An important special case is when r = 1 near k_0 . Then $\lambda(k)$ is just the band function that vanishes at k_0 .

Now, using the Taylor expansion around k_0 , we decompose $\lambda(k)$ into the series of homogeneous matrix valued polynomials:

$$\lambda(k) = \sum_{j \ge 0} \lambda_j (k - k_0), \tag{5.13}$$

where each λ_j is a $\mathbb{C}^{r \times r}$ -valued homogeneous polynomial of degree j in d variables.

For each quasimomentum $k_0 \in F_{A,\mathbb{R}}$, let $\ell_0(k_0)$ be the order of **the first non-zero term** of the Taylor expansion (5.13) around k_0 of the matrix function $\lambda(k)$.

The next result, extracted from [50], provides explicit formulas for dimensions of the spaces $V_N^{\infty}(A)$.

Theorem 5.2.10. [50] Suppose that the real Fermi surface $F_{A,\mathbb{R}}$ is finite (modulo G^* -shifts) and the spectrum $\sigma(A(k))$ is discrete for any quasimomentum k.

(a) For each integer $0 \le N < \min_{k \in F_{A,\mathbb{R}}} \ell_0(k)$, we have

$$\dim V_N^{\infty}(A) = \sum_{k \in F_{A,\mathbb{R}}} m_k \left[\binom{d+N}{d} - \binom{d+N-\ell_0(k)}{d} \right], \quad (5.14)$$

where m_k is the algebraic multiplicity of the zero eigenvalue of the operator A(k).

(b) If for every $k \in F_{A,\mathbb{R}}$, det $\lambda_{\ell_0(k)}$ is not identically equal to zero, the formula (5.14) holds for any $N \ge 0$.

The reader might wonder what does the binomial coefficient $\binom{d+N-\ell_0(k)}{d}$ mean if $d+N-\ell_0(k)$ happens to be negative. We adopt the following agreement:

Definition 5.2.11. If in some formulas throughout this text one encounters a binomial coefficient $\binom{M}{N}$, where the difference M - N is negative, we define its value to be equal to zero.

It is worthwhile to note that the positivity of $\ell_0(k)$ is equivalent to the fact that both algebraic and geometric multiplicities of the zero eigenvalue of the operator A(k) are the same. Also, the non-vanishing of the determinant of $\lambda_{\ell_0(k)}$ implies that $\ell_0(k) > 0$.

5.2.4 A characterization of the spaces $V_N^p(A)$

Notation 5.2.12. For a real number r, we denote by $\lfloor r \rfloor$ the largest integer that is strictly less than r, while [r] denotes the largest integer that is less or equal than r.

The following statement follows from Theorem 5.2.8 (ii):

Lemma 5.2.13. $V_N^{\infty}(A) = V_{[N]}^{\infty}(A)$ for any non-negative real number N.

The proofs of the next two theorems are delegated to the Subsection 5.7.6.

Theorem 5.2.14. For each $1 \le p < \infty$ such that pN > d, one has

$$V^{\infty}_{\lfloor N-d/p \rfloor}(A) \subseteq V^p_N(A).$$

If, additionally, the Fermi surface $F_{A,\mathbb{R}}$ is finite modulo G^* -shifts, then

$$V_N^p(A) = V_{\lfloor N-d/p \rfloor}^\infty(A).$$

Corollary 5.2.15. If the Fermi surface $F_{A,\mathbb{R}}$ is finite modulo G^* -shifts, then Liouville theorem of order (p, N) for a pair (p, N) such that pN > d holds for A if and only if the Liouville theorem of order (∞, N) holds for A for some $N \ge 0$, and thus, according to Theorem 5.2.8, for all $N \ge 0$.

The following theorem could be regarded as a version of the unique continuation property at infinity for the periodic elliptic operator A.

Theorem 5.2.16. Assume that $F_{A,\mathbb{R}}$ is finite (modulo G^* -shifts). Then the space $V_N^p(A)$ is trivial if either one of the following conditions holds:

- (a) $p \neq \infty, N \leq d/p$.
- (b) $p = \infty, N < 0.$

In fact, a more general version of these results is:

Theorem 5.2.17. Let $p \in [1, \infty)$. Let also ϕ be a continuous, positive function defined on \mathbb{R}^+ such that

$$N_{p,\phi} := \sup\left\{ N \in \mathbb{Z} : \int_{0}^{\infty} \phi(r)^{-p} \cdot \langle r \rangle^{pN+d-1} \, \mathrm{d}r < \infty \right\} < \infty$$

We define $\mathcal{V}_{\phi}^{p}(A)$ as the space of all solutions u of A = 0 satisfying the condition

$$\sum_{g\in\mathbb{Z}^d} \|u\|_{L^2(gK)}^p \cdot \phi(|g|)^{-p} < \infty$$

holds for some compact domain K satisfying (5.9).

If $F_{A,\mathbb{R}}$ is finite (modulo G^* -shifts), then one has

- If $N_{p,\phi} \ge 0$, then $\mathcal{V}^p_{\phi}(A) = V^{\infty}_{N_{p,\phi}}(A)$.
- If $N_{p,\phi} < 0$, then $\mathcal{V}^p_{\phi}(A) = \{0\}$.

Note that if $\phi(r) = \langle r \rangle^N$, then $\mathcal{V}^p_{\phi}(A) = V^p_N(A)$.

The proofs of Theorems 5.2.14 and 5.2.16 (provided in Subsection 5.7.6) easily transfer to this general version. We thus skip the proof of Theorem 5.2.17.

Remark 5.2.18. It is worthwhile to note that results of this section do not require the assumption of discreteness of spectra of the operators A(k). This is useful, in particular, when considering overdetermined problems.

5.3 The Gromov-Shubin version of the Riemann-Roch theorem for elliptic operators on noncompact manifolds

We follow here closely the paper [33] by M. Gromov and M. Shubin, addressing only its parts that are relevant for our considerations.

5.3.1 Some notions and preliminaries

Through this section, P will denote a linear elliptic differential expressions with smooth coefficients on a non-compact manifold \mathcal{X} (later on, \mathcal{X} will be the space of an abelian co-compact covering). We denote by P^* its transpose elliptic differential operator, defined via the identity

$$\langle Pu, v \rangle = \langle u, P^*v \rangle, \quad \forall u, v \in C^{\infty}_{c}(\mathcal{X}),$$

where $\langle \cdot, \cdot \rangle$ is the bilinear duality (5.1).

We notice that both P and P^* can be applied as differential expressions to any smooth function on \mathcal{X} and these operations keep the spaces $C^{\infty}(\mathcal{X})$ and $C_c^{\infty}(\mathcal{X})$ invariant.

We assume that P and P^* are defined as operators on domains Dom P and $Dom P^*$, such that

$$C_c^{\infty}(\mathcal{X}) \subseteq \text{Dom} P \subseteq C^{\infty}(\mathcal{X}), \tag{5.15}$$

$$C_c^{\infty}(\mathcal{X}) \subseteq \text{Dom} P^* \subseteq C^{\infty}(\mathcal{X}).$$
 (5.16)

Definition 5.3.1. We denote by Im P and $\text{Im } P^*$ the ranges of P and P^* on their corresponding domains, i.e.

Im
$$P = P(\text{Dom } P)$$
, Im $P^* = P^*(\text{Dom } P^*)$. (5.17)

As usual, Ker P and Ker P^* denote the spaces of solutions of the equations Pu = 0, $P^*u = 0$ in Dom P and Dom P^* respectively.

We also need to define some auxiliary spaces⁴. Namely, assume that we can choose linear subspaces⁵ Dom' P and Dom' P^{*} of $C^{\infty}(\mathcal{X})$ so that

 $(\mathcal{P}1)$

$$C_c^{\infty}(\mathcal{X}) \subseteq \text{Dom}' P \subseteq C^{\infty}(\mathcal{X}), \tag{5.18}$$

$$C_c^{\infty}(\mathcal{X}) \subseteq \text{Dom}' P^* \subseteq C^{\infty}(\mathcal{X}),$$
(5.19)

and

⁴Most of the complications in definitions here and below come from non-compactness of the manifold.

⁵The notation Dom' might confuse the reader, leading her to thinking that this is a different domain of the operator. It is rather an object dual to the domain Dom.

 $(\mathcal{P}2)$

$$\operatorname{Im} P^* \subseteq \operatorname{Dom}' P, \quad \operatorname{Im} P \subseteq \operatorname{Dom}' P^*.$$

(P3) The bilinear pairing $\int_{\mathcal{X}} f(x)g(x) d\mu_X$ (see (5.1)) makes sense for functions from the relevant spaces, to define the pairings

$$\langle \cdot, \cdot \rangle : \operatorname{Dom}' P^* \times \operatorname{Dom} P^* \mapsto \mathbb{C}, \quad \langle \cdot, \cdot \rangle : \operatorname{Dom} P \times \operatorname{Dom}' P \mapsto \mathbb{C},$$

so that

 $(\mathcal{P}4)$ The duality ("integration by parts formula")

$$\langle Pu, v \rangle = \langle u, P^*v \rangle, \quad \forall u \in \text{Dom } P, \ v \in \text{Dom } P^*$$

holds.

We also need an appropriate notion of a polar (annihilator) to a subspace:

Definition 5.3.2. For a subspace $L \subset \text{Dom } P$, its **annihilator** L° is the subspace of Dom' P consisting of all elements of Dom' P that are orthogonal to L with respect to the pairing $\langle \cdot, \cdot \rangle$:

$$L^{\circ} = \{ u \in \text{Dom}' P \mid \langle v, u \rangle = 0, \text{ for any } v \in \text{Dom} P \}.$$

Analogously, M° is the annihilator in $\text{Dom}' P^*$ of a linear subspace $M \subset \text{Dom} P^*$ with respect to $\langle \cdot, \cdot \rangle$.

Following [33], we now introduce an appropriate for our goals notion of Fredholm type operators.

Definition 5.3.3. *P* is a **Fredholm operator on** \mathcal{X} if the following requirements are satisfied:

(i)

$$\dim \operatorname{Ker} P < \infty, \dim \operatorname{Ker} P^* < \infty$$

and

(ii)

$$\operatorname{Im} P = \left(\operatorname{Ker} P^*\right)^\circ.$$

Then the **index** of P is defined as

 $\operatorname{ind} P = \dim \operatorname{Ker} P - \operatorname{codim} \operatorname{Im} P = \dim \operatorname{Ker} P - \dim \operatorname{Ker} P^*.$

5.3.2 Point divisors

We need to recall the rather technical notion of a **rigged divisor** from [33]. However, for reader's sake, we start with more familiar and easier to comprehend particular case of a **point divisor**, which appeared initially in the first version of Gromov-Shubin's analog of the Riemann-Roch formula [32].

Definition 5.3.4. A point divisor μ on X consists of two finite disjoint subsets of X

$$D^{+} = \{x_1, \dots, x_r\}, D^{-} = \{y_1, \dots, y_s\}$$
(5.20)

and two tuplets $0 < p_1, \ldots, p_r$ and $q_1, \ldots, q_s < 0$ of integers⁶. The support of the point divisor μ consists of the points x_1, \ldots, x_r and y_1, \ldots, y_s . We will also write $\mu = x_1^{p_1} \ldots x_r^{p_r} \cdot y_1^{q_1} \ldots y_s^{q_s}$.

⁶In other words, μ is an element of the free abelian group generated by points of \mathcal{X} .

In [32], such a divisor is used to allow solutions of an elliptic equation Pu = 0 to have poles up to certain orders at the points of D^+ and zeros on D^- . Namely,

(i) For any 1 ≤ j ≤ r, there exists an open neighborhood U_j of x_j such that on U_j \{x_j}, one has u = u_s + u_r, where u_r ∈ C[∞](U_j), u_s ∈ C[∞](U_j \ {x_j}) and when x → x_j,

$$u_s(x) = o(|x - x_j|^{m-n-p_j}).$$

(ii) For any $1 \le j \le s$, as $x \to y_j$, one has

$$u(x) = O(|x - y_j|^{|q_j|}).$$

5.3.3 Rigged divisors

The notion of a "rigged" divisor comes from the desire to allow for some infinite sets D_{\pm} , but at the same time to impose only finitely many conditions ("zeros" and "singularities") on the solution.

So, let us take a deep breath and dive into it. First, let us define some distribution spaces:

Definition 5.3.5. For a closed set $C \subset \mathcal{X}$, we denote by $\mathcal{E}'_C(\mathcal{X})$ the space of distributions on \mathcal{X} , whose supports belong to C (i.e., they are zero outside C).

Definition 5.3.6.

A rigged divisor associated with P is a tuple μ = (D⁺, L⁺; D⁻, L⁻), where D[±] are compact nowhere dense disjoint subsets in X and L[±] are finite-dimensional vector spaces of distributions on X supported in D[±] respectively, i.e.,

$$L^+ \subset \mathcal{E}'_{D^+}(\mathcal{X}), \quad L^- \subset \mathcal{E}'_{D^-}(\mathcal{X}).$$

2. The secondary spaces \tilde{L}^{\pm} associated with L^{\pm} are defined as follows:

$$\tilde{L}^{+} = \{ u \mid u \in \mathcal{E}'_{D^{+}}(\mathcal{X}), Pu \in L^{+} \}, \quad \tilde{L}^{-} = \{ u \mid u \in \mathcal{E}'_{D^{-}}(\mathcal{X}), P^{*}u \in L^{-} \}.$$

3. Let $\ell^{\pm} = \dim L^{\pm}$ and $\tilde{\ell}^{\pm} = \dim \tilde{L}^{\pm}$. The **degree** of μ is defined as follows:

$$\deg_P \mu = (\ell^+ - \tilde{\ell}^+) - (\ell^- - \tilde{\ell}^-).$$
(5.21)

4. The inverse of μ is the rigged divisor $\mu^{-1} := (D^-, L^-; D^+, L^+)$ associated with P^* .

Remark 5.3.7.

- Notice that the degree of the divisor involves the operator *P*, so it would have been more prudent to call it "degree of the divisor with respect to the operator *P*," but we'll neglect this, hoping that no confusion will arise.
- Observe that P and P^{*} are injective on \mathcal{E}'_{D^+} and \mathcal{E}'_{D^-} , correspondingly ⁷. Thus,

$$\ell^{\pm} \ge \tilde{\ell}^{\pm}.\tag{5.22}$$

• The sum of the degrees of a divisor μ and of its inverse is zero.

Although we have claimed that point divisors are also rigged divisors, this is not immediately clear when comparing Definitions 5.3.4 and 5.3.6. Namely, we have to assign the spaces L_{\pm} to a point divisor and to check that the definitions are equivalent in this case.

⁷For example, if $u \in \mathcal{E}'_{D^+}$ and Pu = 0 then u is smooth due to elliptic regularity, but then u = 0 everywhere since the complement of D^+ is dense.

This was done⁸ in [33], if one defines the spaces associated with a point divisor as follows:

$$L^{+} = \left\{ \sum_{1 \le j \le r} \sum_{|\alpha| \le p_j - 1} c_j^{\alpha} \delta^{\alpha}(\cdot - x_j) \mid c_j^{\alpha} \in \mathbb{C} \right\}$$

and

$$L^{-} = \left\{ \sum_{1 \le j \le s} \sum_{|\alpha| \le |q_s| - 1} c_j^{\alpha} \delta^{\alpha} (\cdot - y_j) \mid c_j^{\alpha} \in \mathbb{C} \right\},\$$

where δ and δ^{α} denote the Dirac delta function and its derivative corresponding to the multi-index α .

It was also shown in [33]) that the degree
$$\deg_P(\mu)$$
 in this case is $\sum_{1 \le j \le r} \left[\binom{p_j + n - 1}{n} - \binom{p_j + n - 1 - m}{n} \right] - \sum_{1 \le j \le s} \left[\binom{q_j + n - 1}{n} - \binom{q_j + n - 1 - m}{n} \right]$. Here, as before, n is the dimension of the manifold X and m is the order of the operator P .

5.3.4 Gromov-Shubin theorem on noncompact manifolds

To state (a version of) the Gromov-Shubin theorem, we now introduce the spaces of solutions of P with allowed singularities on D^+ and vanishing conditions on D^- .

Notation 5.3.8. For a compact subset K of \mathcal{X} and $u \in C^{\infty}(\mathcal{X} \setminus K)$, we shall write that

$$u \in \operatorname{Dom}_K P,$$

if there is a compact neighborhood \hat{K} of K and $\hat{u} \in \text{Dom } P$ such that $u = \hat{u}$ outside \hat{K} .

Definition 5.3.9. For an elliptic operator P and a rigged divisor $\mu = (D^+, L^+; D^-, L^-)$, the space $L(\mu, P)$ is defined as follows:

 $\underbrace{\{u \in \text{Dom}_{D^+} P \mid \exists \tilde{u} \in \mathcal{D}'(\mathcal{X}) \text{ such that } \tilde{u} = u \text{ on } \mathcal{X} \setminus D^+, P\tilde{u} \in L^+ \text{ and } (u, L^-) = 0\}.}_{\text{8Which is not trivial.}}$

Here $(u, L^-) = 0$ means that u is orthogonal to every element in L^- with respect to the canonical bilinear duality. The distribution \tilde{u} are regularization of $u \in C^{\infty}(\mathcal{X} \setminus D^+)$. Now we can state a variant of Gromov-Shubin's version of the Riemann-Roch theorem.

Theorem 5.3.10. Let P be an elliptic operator such that (5.15) and properties (P1)-(P4) are satisfied. Let also μ be a rigged divisor associated with P. If P is a Fredholm operator on \mathcal{X} , then the following Riemann-Roch inequality holds:

$$\dim L(\mu, P) - \dim L(\mu^{-1}, P^*) \ge \inf P + \deg_P(\mu).$$
(5.23)

If both P and P^{*} are Fredholm on \mathcal{X} , (5.23) becomes the Riemann-Roch equality:

$$\dim L(\mu, P) = \inf P + \deg_P(\mu) + \dim L(\mu^{-1}, P^*).$$
(5.24)

Remark 5.3.11.

- Although the authors of [33] do not state their theorem in the exact form above, the Riemann-Roch inequality (5.23) follows from their proof.
- If one considers the difference dim L(μ, P) dim L(μ⁻¹, P*) as some "index of P in presence of the divisor μ" (say, denote it by ind_μ(P)), the Riemann-Roch equality becomes

$$\operatorname{ind}_{\mu}(P) = \operatorname{ind} P + \deg_{P}(\mu) \tag{5.25}$$

and thus it says that introduction of the divisor changes the index of the operator by $\deg_P(\mu)$.

It is useful for our future considerations to mention briefly some of the ingredients ⁹ of the proof from [33]. To start, Gromov and Shubin [33] define some auxiliary spaces,

⁹The reader interested in the main results only, can skip to Corollary 5.3.13.

which we recall now. Let

$$\Gamma(\mathcal{X}, \mu, P) := \{ u \in C^{\infty}(X \setminus D^{+}) \mid u \in \text{Dom}_{D^{+}} P, \ \exists \tilde{u} \in \mathcal{D}'(\mathcal{X}) \text{ such that} \\ \tilde{u} = u \text{ on } \mathcal{X} \setminus D^{+}, P\tilde{u} \in L^{+} + C^{\infty}(\mathcal{X}) \text{ and, } \langle u, L^{-} \rangle = 0 \},$$

As before, \tilde{u} is a regularization of $u \in C^{\infty}(\mathcal{X} \setminus D^+)$. The space of all such regularizations \tilde{u} is

$$\widetilde{\Gamma}(\mathcal{X},\mu,P) := \{ \widetilde{u} \in \mathcal{D}'(\mathcal{X}) \mid \widetilde{u}_{|\mathcal{X} \setminus D^+} \in \Gamma(\mathcal{X},\mu,P), P\widetilde{u} \in L^+ + C^{\infty}(\mathcal{X}) \}.$$

Let us also introduce the spaces

$$\Gamma_{\mu}(\mathcal{X}, P) = \{ u \in \text{Dom } P \mid \langle u, L^{-} \rangle = 0 \}$$

and

$$\widetilde{\Gamma}_{\mu}(\mathcal{X}, P) = \{ f \in \operatorname{Dom}' P^* \mid \langle f, \tilde{L}^- \rangle = 0 \}.$$

Then for any $u \in \Gamma(\mathcal{X}, \mu, P)$, one can extend by continuity the restriction of Pu on $\mathcal{X} \setminus D^+$ to a smooth function on the whole \mathcal{X} . Let us denote this extension by $\tilde{P}u$. Then \tilde{P} is a linear map from $\Gamma(\mathcal{X}, \mu, P)$ to $\tilde{\Gamma}_{\mu}(\mathcal{X}, P)$. In the same manner, we can also define the corresponding extension \tilde{P}^* as a linear map from $\Gamma(\mathcal{X}, \mu^{-1}, P^*)$ to $\tilde{\Gamma}_{\mu^{-1}}(\mathcal{X}, P^*)$. The important point here is that $L(\mu, P)$ and $L(\mu^{-1}, P^*)$ are the kernels of these two extensions \tilde{P} and \tilde{P}^* , correspondingly. Let us also introduce the duality

$$(\cdot, \cdot) : \Gamma(\mathcal{X}, \mu, P) \times \widetilde{\Gamma}_{\mu^{-1}}(\mathcal{X}, P^*) \to \mathbb{C}$$
 (5.26)

as follows:

$$(u, f) := \langle \tilde{u}, f \rangle, \quad u \in \Gamma(\mathcal{X}, \mu, P), f \in \tilde{\Gamma}_{\mu^{-1}}(\mathcal{X}, P^*),$$

where \tilde{u} is any element in the preimage of $\{u\}$ under the restriction map from $\tilde{\Gamma}(\mathcal{X}, \mu, P)$ to $\Gamma(\mathcal{X}, \mu, P)$. Similarly, we get the duality

$$(\cdot, \cdot) : \widetilde{\Gamma}_{\mu}(\mathcal{X}, P) \times \Gamma(\mathcal{X}, \mu^{-1}, P^*) \to \mathbb{C}$$
 (5.27)

As it was remarked in [33], these dualities are well-defined and non-degenerate and moreover, one has the following relation

$$(\tilde{P}u, v) = (u, \tilde{P}^*v), \quad \forall u \in \Gamma(\mathcal{X}, \mu, P), v \in \Gamma(\mathcal{X}, \mu^{-1}, P^*).$$

Gromov and Shubin then prove the following identity

$$\dim \operatorname{Ker} \tilde{P} = \operatorname{ind} P + \deg_P(\mu) + \operatorname{codim} \operatorname{Im} \tilde{P^*}.$$
(5.28)

by applying the additivity of Fredholm indices to some short exact sequences of the spaces introduced above (see [33, Lemma 3.1] and [33, Remark 3.2]). The assumption that P is Fredholm on \mathcal{X} is important in this consideration.

Furthermore, $(\operatorname{Im} \tilde{P})^{\circ} = \operatorname{Ker} \tilde{P^*}$ ([33, Lemma 3.4]) and hence, $\operatorname{Im} \tilde{P} \subset (\operatorname{Ker} \tilde{P^*})^{\circ}$. Here $(\operatorname{Im} \tilde{P})^{\circ}$ and $(\operatorname{Ker} \tilde{P^*})^{\circ}$ are the annihilators of $\operatorname{Im} \tilde{P}$, $\operatorname{Ker} \tilde{P^*}$ with respect to the dualities (5.26) and (5.27), respectively. By [33, Lemma 3.3], one gets the inequality (3.9) in [33]:

$$\dim \operatorname{Ker} \tilde{P^*} = \operatorname{codim} (\operatorname{Ker} \tilde{P^*})^{\circ} \le \operatorname{codim} \operatorname{Im} \tilde{P}.$$
(5.29)

Then the Riemann-Roch inequality (5.23) follows immediately from (5.28) and (5.29). If P^* is also Fredholm, one can apply (5.23) for P^* and μ^{-1} instead of P and μ to get

$$\dim \operatorname{Ker} \tilde{P} \le \operatorname{ind} P + \deg_P(\mu) + \dim \operatorname{Ker} \tilde{P^*}.$$
(5.30)

Thus, the Riemann-Roch equality (5.24) can be deduced from (5.23) and (5.30). In this case, we also have $\text{Im }\tilde{P} = (\text{Ker }\tilde{P^*})^\circ$ and $\text{Im }\tilde{P} = (\text{Ker }\tilde{P^*})^\circ$ as a byproduct of the proof of (5.24) (see [33, Theorem 2.12]).

Remark 5.3.12. If (5.29) becomes an equality, i.e.,

$$\dim \operatorname{Ker} \tilde{P^*} = \operatorname{codim} \operatorname{Im} \tilde{P},$$

one obtains the Riemann-Roch equality (5.24) for the rigged divisor μ without assuming that P^* is Fredholm on \mathcal{X} . Conversely, if (5.24) holds, then $\operatorname{Im} \tilde{P} = (\operatorname{Ker} \tilde{P^*})^{\circ}$.

As a result, we have the following useful corollary:

Corollary 5.3.13. Let P be Fredholm on \mathcal{X} , Im $P = \text{Dom}' P^*$, and $\mu = (D^+, L^+; D^-, L^-)$ be a rigged divisor on the manifold \mathcal{X} . Then the Riemann-Roch equality (5.24) holds for P and this divisor μ .

Moreover, the space $L(\mu^{-1}, P^*)$ is trivial, if the following additional condition is satisfied: Suppose that u is a smooth function in Dom P such that $\langle Pu, \tilde{L}^- \rangle = 0$. Then there exists a solution v in Dom P of the equation Pv = 0 satisfying $\langle u - v, L^- \rangle = 0$.

In particular, this assumption holds automatically if $D^- = \emptyset$ and $L^- = \{0\}$.

We end this part by recalling an application of Theorem 5.3.10, which is needed later.

Example 5.3.14. [33, Example 4.6] Consider $P = P^* = -\Delta$ on $\mathcal{X} = \mathbb{R}^d$, where $d \ge 3$ and

$$\operatorname{Dom} P = \operatorname{Dom} P^* = \{ u \mid u \in C^{\infty}(\mathbb{R}^d), \Delta u \in C^{\infty}_c(\mathbb{R}^d) \text{ and } \lim_{|x| \to \infty} u(x) = 0 \},$$

$$\operatorname{Dom}' P = \operatorname{Dom}' P^* = C_c^{\infty}(\mathbb{R}^d).$$

Then the operators P and P^* are Fredholm on \mathbb{R}^d , $\operatorname{Ker} P = \operatorname{Ker} P^* = \{0\}$, $\operatorname{Im} P = \operatorname{Im} P^* = C_c^{\infty}(\mathbb{R}^d)$, and thus $\operatorname{ind} P = 0$ (see [33, Example 4.2]).

Let

$$D^+ = \{y_1, \dots, y_k\}, \quad D^- = \{z_1, \dots, z_l\}.$$

with all the points $y_1, \ldots, y_k, z_1, \ldots, z_l$ pairwise distinct. Consider the following distributional spaces: L^+ is the vector space spanned by Dirac delta distributions $\delta(\cdot - y_j)$ supported at the points y_j $(1 \le j \le k)$; L^- is spanned by the first order derivatives $\frac{\partial}{\partial x_{\alpha}}\delta(\cdot - z_j)$ of Dirac delta distributions supported at z_j $(1 \le j \le l, 1 \le \alpha \le d)$.¹⁰

Consider now the rigged divisor $\mu := (D^+, L^+; D^-, L^-)$. Then $\deg_{-\Delta}(\mu) = k - dl$. Furthermore,

$$L(\mu, -\Delta) = \left\{ u \mid u(x) = \sum_{j=1}^{k} \frac{a_j}{|x - y_j|^{d-2}}, a_j \in \mathbb{C}, \text{ and } \nabla u(z_j) = 0, j = 1, \dots, l. \right\},\$$

and

$$L(\mu^{-1}, -\Delta) = \left\{ v \mid v(x) = \sum_{j=1}^{l} \sum_{\alpha=1}^{d} b_{j,\alpha} \frac{\partial}{\partial x_{\alpha}} \left(|x - z_j|^{2-d} \right), b_{j,\alpha} \in \mathbb{C}, \text{ and} \\ u(y_j) = 0, j = 1, \dots, k. \right\}.$$

In this case, the Gromov-Shubin-Riemann-Roch formula (Theorem 5.3.10) is

$$\dim L(\mu, -\Delta) = k - dl + \dim L(\mu^{-1}, -\Delta).$$
(5.31)

¹⁰Note that the secondary spaces \tilde{L}^{\pm} are trivial.

5.4 The main results

In this section, we consider a periodic elliptic operator A on an abelian covering X. We assume that the Liouville property holds for the operator A at the level $\lambda = 0$, i.e., the real Fermi surface of A (see Definition 2.2.5) is finite (modulo G^{*}-shifts) (see Theorem 5.2.8). We consider the following two cases:

5.4.1 Non-empty Fermi surface

In this case, suppose that $F_{A,\mathbb{R}} = \{k_1, \ldots, k_\ell\}$ (modulo G^* -shifts), where $\ell \ge 1$. To present our main results in this case, we need to make the following assumption on the local behavior of the Bloch variety of the operator A around each quasimomentum k_j in the real Fermi surface.

Assumption A

(A1) For any quasimomentum k, the spectrum of the operator A(k) is discrete. Under this assumption, the following lemma can be deduced immediately from Proposition 5.7.3 in Section 5.7 and perturbation theory (see e.g., [39, 66]):

Lemma 5.4.1. For each quasimomentum $k_r \in F_{A,\mathbb{R}}$, there is an open neighborhood V_r of k_r in \mathbb{R}^d and a closed contour $\Upsilon_r \subset \mathbb{C}$, such that

- (a) The neighborhoods V_r are mutually disjoint;
- (b) The contour Υ_r surrounds the eigenvalue 0 and does not contain any other points of the spectrum $\sigma(A(k_r))$;
- (c) The intersection $\sigma(A(k)) \cap \Upsilon_r$ is empty for any $k \in V_r$.

Then, any $k \in V_r$, we can define the Riesz projector $\Pi_r(k)$ associated with A(k)and the contour Υ_r . Thus, $\Pi_r(k)A(k)$ is well-defined for any $k \in V_r$. Let m_r be the algebraic multiplicity of the eigenvalue 0 of the operator $A(k_r)$. The immediate consequence is:

Lemma 5.4.2. The projector $\Pi_r(k)$ depends analytically on $k \in V_r$. In particular, its range $R(\Pi_r(k))$ has the same dimension m_r for all $k \in V_r$ and the union $\bigcup_{k \in V_r} R(\Pi_r(k))$ forms a trivial holomorphic vector bundle over V_r .

(A2) We denote by $A_r(k)$ the matrix representation of the operator $\Pi_r(k)A(k)|_{R(\Pi_r(k))}$ with respect to a fixed holomorphic basis $(f_j(k))_{1 \le j \le m_r}$ of the range $R(\Pi_r(k))$ when $k \in V_r$. Then $A_r(k)$ is an invertible matrix except only for $k = k_r$. We equip \mathbb{C}^{m_r} with the maximum norm and impose the following integrability condition:

$$\sup_{1 \le r \le \ell} \int_{V_r \setminus \{k_r\}} \|A_r(k)^{-1}\|_{\mathcal{L}(\mathbb{C}^{m_r})} \, \mathrm{d}k < \infty,$$

where $\mathcal{L}(\mathbb{C}^{m_r})$ is the algebra of linear operators on \mathbb{C}^{m_r} .

Remark 5.4.3.

- (i) Thanks to Proposition 5.7.2 in Section 5.7, Assumption (A1) is satisfied if A is either self-adjoint or a real operator of even order.¹¹
- (ii) When the rank d of G is greater than 2, Assumption (A2) holds at a generic spectral edge (see Section 5.6).

To formulate the results, we need to define appropriate spaces of solutions of polynomial growth satisfying the conditions that are associated with a rigged divisor μ .

Definition 5.4.4. Given any $p \in [1, \infty]$ and $N \in \mathbb{R}$, we define

$$L_p(\mu, A, N) := L(\mu, A_N^p),$$

¹¹Here A is real means that Au is real whenever u is real.

where the operator A_N^p stands for A with the domain

$$\operatorname{Dom} A_N^p = \{ u \in V_N^p(X) \mid Au \in C_c^\infty(X) \}.$$

In other words, $L_p(\mu, A, N)$ is the space

$$\{u \in \text{Dom}_{D^+} A^p_N \mid \exists \tilde{u} \in \mathcal{D}'(X) : \tilde{u} = u \text{ on } X \setminus D^+, A\tilde{u} \in L^+, (u, L^-) = 0\}.$$

We thus restrict the growth of a function at infinity, impose the divisor conditions, and require that it satisfies the homogeneous equation outside of a compact.

Remark 5.4.5. Consider $u \in L_p(\mu, A, N)$. Let K be a compact domain in X such that $\bigcup_{g \in G} gK = X$. Define $G_{K,D^+} := \{g \in G \mid \text{dist}(gK, D^+) \ge 1\}$, where we use the notation $\text{dist}(\cdot, \cdot)$ for the distance between subsets arising from the Riemannian distance on X. Since Au = 0 on $X \setminus D^+$, the condition " $u \in \text{Dom}_{D^+} A_N^p$ " can be written equivalently as follows:

$$\{ \|u\|_{L^2(gK)} \cdot \langle g \rangle^{-N} \}_{g \in G_{K,D^+}} \in \ell^p(G_{K,D^+}).$$

By Schauder estimates (see Proposition 5.7.7), this condition can be rephrased as follows:

$$\sup_{\substack{x: \operatorname{dist}(x, D^+) \ge 1}} \frac{|u(x)|}{\operatorname{dist}(x, D^+)^N} < \infty, \quad \text{when} \quad p = \infty,$$

$$\int_{\substack{x: \operatorname{dist}(x, D^+) \ge 1}} \frac{|u(x)|^p}{\operatorname{dist}(x, D^+)^{pN}} \, \mathrm{d}\mu_X(x) < \infty, \quad \text{when} \quad 1 \le p < \infty.$$
(5.32)

So, depending on the sign of N, this condition controls how u grows or decays at infinity. Our first main result is the next theorem.

Theorem 5.4.6. Assume that either $p = \infty$ and $N \ge 0$ or $p \in [1, \infty)$ and N > d/p. Let p' be the Hölder conjugate of p. Then under Assumption A, the following LiouvilleRiemann-Roch inequality holds:

$$\dim L_p(\mu, A, N) \ge \dim V_N^p(A) + \deg_A(\mu) + \dim L_{p'}(\mu^{-1}, A^*, -N),$$
(5.33)

where dim $V_N^p(A)$ can be computed via Theorems 5.2.10 and 5.2.14.

Remark 5.4.7.

This is an extension to include Liouville conditions of the Riemann-Roch inequality (5.23). One might wonder why in comparison with the "μ-index"

$$\dim L_p(\mu, A, N) - \dim L_p(\mu^{-1}, A^*, -N),$$

the above inequality only involves the dimension of the kernel $V_N^p(A)$, rather than a full index. The reason is that in this case, the dimension is indeed the full index (the corresponding range co-dimension being zero).

Also, Assumption (A1) forces the Fredholm index of A(0) on M to vanish (see [47, Theorem 4.1.4]). Therefore, ind_M A(k) (k ∈ C^d) and the L²-index of A (by Atiyah's theorem [7]) are zero as well.

The next consequence of Theorem 5.4.6 is useful.

Theorem 5.4.8. If dim $V_N^p(A) + \deg_A(\mu) > 0$, then there exists a nonzero element in the space $L_p(\mu, A, N)$. In other words, there exists a nontrivial solution of the growth described in (5.32) that also satisfies the conditions on "zeros" and "singularities" imposed by the rigged divisor μ .

To state the remaining results, we will need the following definition related to various types of divisors. Namely, one can consider divisors containing only "zeros" or "singularities" from a given divisor.

Definition 5.4.9.

- The divisor $(\emptyset, 0; \emptyset, 0)$ is called **the trivial divisor**.
- Let μ = (D⁺, L⁺; D⁻, L⁻) be a rigged divisor on X. Then the positive part μ⁺ and the negative part μ⁻ of μ are defined as the tuples (D⁺, L⁺; Ø, 0) and (Ø, 0; D⁻, L⁻), respectively. Hence, for a rigged divisor μ, μ⁺ (resp. μ⁻) is trivial whenever D⁺ = Ø and L⁺ is zero (resp. D⁻ = Ø and L⁻ is zero).

Our next result shows that if μ^- is trivial, the Liouville-Riemann-Roch inequality becomes an equality:

Theorem 5.4.10.

Consider a rigged divisor μ and its positive part μ^+ . Under the assumption of Theorem 5.4.6, the space $L_{p'}((\mu^+)^{-1}, A^*, -N)$ is trivial and

$$\dim L_p(\mu^+, A, N) = \dim V_N^p(A) + \deg_A(\mu^+).$$
(5.34)

In particular,

$$\dim L_p(\mu, A, N) \le \dim V_N^p(A) + \deg_A(\mu^+).$$
(5.35)

In other words, the inequality (5.35) gives an upper bound for the dimension of the space $L_p(\mu, A, N)$ (see (5.33) for its lower bound) in terms of the degree of the positive part of the divisor μ and the dimension of the space $V_N^p(A) = L_p(\mu_0, A, N)$, where μ_0 is the trivial divisor. By (5.34), one can see that this upper bound is sharp for divisors containing only "singularities".

When μ^- is nontrivial, determining the triviality of the space $L_{p'}(\mu^{-1}, A^*, -N)$ is more complicated. In the next proposition, we consider two situations: if the degree of
μ^+ is sufficiently large, then the space $L_{p'}(\mu^{-1}, A^*, -N)$ degenerates to zero, while the spaces $L_{p'}(\mu^{-1}, A^*, -N)$ can have arbitrarily large dimensions if μ^+ is trivial.

Here we recall that a differential operator has the strong unique continuation property if any local smooth solution which has a vanishing order of infinity at some point vanishes everywhere.

Proposition 5.4.11.

(a) For any $N \ge 0$, $p \in [1, \infty]$, and $d \ge 3$,

 $\sup \dim L_{p'}(\mu^{-1}, -\Delta_{\mathbb{R}^d}, -N) = \infty,$

where the supremum is taken over all divisors μ with trivial positive parts.

(b) Let the assumption of Theorem 5.4.6 be satisfied and A* have the strong unique continuation property. Let also the covering X be connected and a point x₀ of X be given. Let us consider p ∈ [1,∞], N ∈ ℝ, a compact nowhere dense subset D⁻ such that x₀ ∉ D⁻, and a finite dimensional subspace L⁻ of E'_D(X). Then there exists M > 0 depending on the data (A, p, N, x₀, D⁻, L⁻) such that dim L_{p'}(μ⁻¹, A*, -N) = 0 for any rigged divisor μ = (D⁺, L⁺; D⁻, L⁻) satisfying x₀ ∈ D⁺ ⊆ X \ D⁻ and L⁺_M := span_C{∂^α_xδ(· - x₀)}_{0≤|α|≤M} ⊆ L⁺ ⊂ E'_{D+}(X).

The second part of the above proposition is a reformulation of [32, Proposition 4.3]. Theorem 5.4.6 can be improved if an additional assumption is imposed.

Theorem 5.4.12. Besides Assumption \mathcal{A} , we suppose further that for each $1 \leq r \leq \ell$, the function $k \in V_r \mapsto ||A_r(k)^{-1}||^2_{\mathcal{L}(\mathbb{C}^{m_r})}$ is integrable ¹². If one of the following two conditions

¹²e.g., this occurs at generic edges when $d \ge 5$.

- $p \ge 2$ and $N \ge 0$
- $p \in [1, 2)$ and 2pN > (2 p)d

is satisfied, then the inequality (5.33) holds. In particular, for any rigged divisor μ , we have

$$\dim L_2(\mu^+, A, 0) = \deg_A(\mu^+)$$
(5.36)

and

$$\dim L_2(\mu, A, 0) = \deg_A(\mu) + \dim L_2(\mu^{-1}, A^*, 0).$$
(5.37)

Remark 5.4.13. When μ is trivial, the equality (5.36) means the absence of non-zero L^2 solutions, and thus generically, spectral edges are not eigenvalues. Here the condition on
integrability of $||A_r(k)^{-1}||^2_{\mathcal{L}(\mathbb{C}^{m_r})}$ is not required.

One can show that the following L^2 -solvability result (or Fredholm alternative) follows from (5.37).

Proposition 5.4.14. Let D be a non-empty compact nowhere dense subset of X and a finite dimensional subspace $L \subset \mathcal{E}'_D(X)$ be given. Define the finite dimensional subspace $\tilde{L} = \{v \in \mathcal{E}'_D(X) \mid A^*v \in L\}$. Consider any $f \in C_c^{\infty}(X)$ satisfying $(f, \tilde{L}) = 0$. Under the assumptions of Theorem 5.4.12 for the periodic operator A of order m, the following two statements are equivalent:

- (i) f is orthogonal to each element in the space L₂(μ, A*, 0), where μ is the rigged divisor (D, L; 0, Ø).
- (ii) The inhomogeneous equation Au = f has a (unique) solution u in $H^m(X)$ such that (u, L) = 0.

We present now examples which show that when μ^+ is trivial, the Liouville-Riemann-Roch equality may hold in some cases, while failing miserably in others. The proofs of the propositions below can be found in Section 5.5.

Proposition 5.4.15. Consider the Laplacian $-\Delta$ on \mathbb{R}^d , $d \ge 3$. For any $N \ge 0$ and positive integer ℓ , there exists a rigged divisor μ such that μ^+ is trivial and

$$\dim L_{\infty}(\mu, -\Delta, N) \ge \ell + \dim V_N^{\infty}(-\Delta) + \deg_{-\Delta}(\mu) + \dim L_1(\mu^{-1}, -\Delta, -N).$$
 (5.38)

Now let A = A(x, D) be an elliptic constant-coefficient homogeneous differential operator of order m on \mathbb{R}^d that satisfies Assumption \mathcal{A} . Consider two non-negative integers $M_0 \ge M_1$. We fix a point x_0 in X and we define $D^- := \{x_0\}$ as well as the following finite dimensional vector subspace of $\mathcal{E}'_{D^-}(\mathbb{R}^d)$:

$$L^{-} := \operatorname{span}_{\mathbb{C}} \{ \partial^{\alpha} \delta(\cdot - x_0) \}_{M_1 \le |\alpha| \le M_0}$$

Let μ be a rigged divisor on \mathbb{R}^d of the form $(D^+, L^+; D^-, L^-)$, where D^+ is a nowhere dense compact subset of \mathbb{R}^d that does not contain the point x_0 and L^+ is a finite dimensional subspace of $\mathcal{E}'_{D^+}(\mathbb{R}^d)$.¹³

Proposition 5.4.16. Assume that one of the following two conditions holds:

- $1 \le p < \infty$, $N > d/p + M_0$.
- $p = \infty, N \ge M_0$.

Then dim $L_{p'}(\mu^{-1}, A^*, -N) = 0$ and the Liouville-Riemann-Roch equality holds:

$$\dim L_p(\mu, A, N) = \dim V_N^p(A) + \deg_A(\mu).$$

¹³A particular case is when $M_1 = 0$ and μ^+ is trivial, i.e., μ becomes the point divisor $x_0^{-(M_0+1)}$ (see Definition 5.3.4).

5.4.2 Empty Fermi surface

In this case, the Liouville theorem becomes trivial, because the emptiness of the Fermi surface implies that there is no non-zero, polynomially growing solution (see [50, Theorem 4.3]), which is an analog of the Schnol theorem (see e.g., [19, 29, 78]). One can obtain a Liouville-Riemann-Roch type result by combining the Riemann-Roch and the Schnol theorems. It is shown in [70] that a more general and stronger statement than the Schnol theorem holds for any C^{∞} -bounded uniformly elliptic operator on a manifold of bounded geometry and subexponential growth ¹⁴. In this setting, a corresponding Liouville-Riemann-Roch theorem can also be proven. We consider only the case of C^{∞} -bounded uniformly elliptic operators ¹⁵ on a co-compact Riemannian covering.

We begin with some definitions from [70] first. Let \mathcal{X} be a co-compact connected Riemannian covering and \mathcal{M} be its base. The deck group G is a countable and finitely generated discrete group (not necessarily abelian). Let $d_{\mathcal{X}}(\cdot, \cdot)$ be the G-invariant Riemannian distance on \mathcal{X} . Due to the compactness of \mathcal{M} , there exists $r_{inj} > 0$ (injectivity radius) such that for every $r \in (0, r_{inj})$ and every $x \in \mathcal{X}$, the exponential geodesic mapping $\exp_x : T\mathcal{X}_x \to \mathcal{X}$ is a diffeomorphism of the Euclidean ball B(0, r) centered at 0 with radius r in the tangent space $T\mathcal{X}_x$ onto the geodesic ball $B_{\mathcal{X}}(x, r)$ centered at xwith the same radius r in \mathcal{X} . Taking $r_0 \in (0, r_{inj})$, the geodesic balls $B_{\mathcal{X}}(x, r)$, where $0 < r \leq r_0$, are called **canonical charts** with x-coordinate in the charts.

Definition 5.4.17. [70,71]

(i) A differential operator P of order m on X is C[∞]-bounded if in every canonical chart, P can be represent as ∑_{|α|≤m} a_α(x)∂^α_x. Here the coefficient a_α(x) is smooth such that for any multi-index β, |∂^β_xa_α(x)| ≤ C_{αβ} where the constants C_{αβ} are inde-

¹⁴One could find details about analysis on manifolds of bounded geometry in e.g., [21, 70, 71].

¹⁵Such operators are not necessarily periodic.

pendent of the chosen canonical coordinate.

(ii) A differential operator P of order m on \mathcal{X} is **uniformly elliptic** if

$$|P_0^{-1}(x,\xi)| \le C|\xi|^{-m}, \quad (x,\xi) \in T^*\mathcal{X}, \xi \ne 0.$$

Here $T^*\mathcal{X}$ is the cotangent bundle of \mathcal{X} , $P_0(x,\xi)$ is the principal symbol of the operator P, and $|\xi|$ is the length of the covector (x,ξ) with respect to the metric on $T^*\mathcal{X}$ induced by the given Riemannian metric on \mathcal{X} .

(iii) \mathcal{X} is of **subexponential growth** if the volumes of balls of radius r grow subexponetially as $r \to \infty$, i.e., for any $\epsilon > 0$ and r > 0,

$$\sup_{x \in \mathcal{X}} \operatorname{vol} B(x, r) = O(\exp(\epsilon r)).$$

Here $vol(\cdot)$ is the Riemannian volume on \mathcal{X} .

(iv) Let x_0 be a fixed point in \mathcal{X} . A differential operator P on \mathcal{X} satisfies **Strong Schnol Property** (SSP) if the following statement is true: If there exists a non-zero solution u of the equation $Pu = \lambda u$ such that for any $\epsilon > 0$

$$u(x) = O(\exp\left(\epsilon d_{\mathcal{X}}(x, x_0)\right))$$

then λ is in the spectrum of *P*.

Clearly, any G-periodic elliptic differential operator with smooth coefficients on \mathcal{X} is C^{∞} -bounded uniformly elliptic.

We now turn to a brief discussion of growth of groups (see e.g., [62]). Let us pick a finite, symmetric generating set S of G. The **word metric** associated to S is denoted by

 $d_S: G \times G \to \mathbb{R}$, i.e., for every pair (g_1, g_2) of two group elements in G, $d_S(g_1, g_2)$ is the length of the shortest representation in S of $g_1^{-1}g_2$ as a product of generators from S. Let e be the identity element of G. The volume function of G associated to S is the function $\operatorname{vol}_{G,S} : \mathbb{N} \to \mathbb{N}$ defined by assigning to every $n \in \mathbb{N}$ the cardinality of the open ball $B_{G,S}(e, n)$ centered at e with radius n in the metric space (G, d_S) . Although the values of this volume function depend on the choice of the generating set S, its asymptotic growth type is independent from it. The group G is said to be of **subexponential growth** if

$$\lim_{n \to \infty} \frac{\ln \operatorname{vol}_{G,S}(n)}{n} = 0.$$

It is known that the deck group G is of subexponential growth if and only if the covering \mathcal{X} is so (see e.g., [68, Proposition 2.1]). Virtually nilpotent groups clearly have polynomial growth ¹⁶. Thus, any virtually nilpotent co-compact Riemannian covering \mathcal{X} is of subexponential growth. Groups with intermediate growth, which were constructed by Grigorchuk [30], provide other non-trivial examples of Riemannian coverings with subexponential growth.

Theorem 5.4.18. [70, Theorem 4.2] If \mathcal{X} is of subexponential growth, then any C^{∞} -bounded uniformly elliptic operator on \mathcal{X} satisfies (SSP).

Remark 5.4.19. A School type theorem can be established without the subexponential growth condition, if the growth of a generalized eigenfunction is controlled in an integral (over an expanding ball), rather than point-wise sense, see [12, Theorem 3.2.2 for the quantum graph case and more general remarks and references in Section 3.8].

Henceforth, the word length of g in G is defined as $|g| := d_S(e, g)$. Similarly, we also ¹⁶The celebrated Gromov's theorem [34] shows that virtually nilpotent groups are the only ones with polynomial growth. say that a positive function $\varphi: G \to \mathbb{R}^+$ is of **subexponential growth** if

$$\lim_{|g|\to\infty}\frac{\ln\varphi(g)}{|g|}=0.$$

Again, this concept does not depend on the choice of the finite generating set S (see [62, Theorem 1.3.12]).

Definition 5.4.20. Let φ be a positive function defined on the deck group G such that both φ and its inverse φ^{-1} are of subexponential growth. Let us denote by S(G) the set of all such φ on G. Then for any $p \in [1, \infty]$ and $\varphi \in S(G)$, we define:

$$\mathcal{V}^p_{\varphi}(\mathcal{X}) = \{ u \in C^{\infty}(\mathcal{X}) \mid \{ \|u\|_{L^2(g\mathcal{F})} \varphi^{-1}(g) \}_{g \in G} \in \ell^p(G) \},$$

where \mathcal{F} is a fundamental domain for \mathcal{M} in \mathcal{X} . Also, let P_{φ}^{p} be the operator P with the domain $\{u \in \mathcal{V}_{\varphi}^{p}(\mathcal{X}) \mid Pu \in C_{c}^{\infty}(\mathcal{X})\}$. Then we denote by $L_{p}(\mu, P, \varphi)$ the space $L(\mu, P_{\varphi}^{p})$, where μ is a rigged divisor on \mathcal{X} . In a similar manner, we also define the space $L_{p}(\mu, P^{*}, \varphi)$, where P^{*} is the transpose of P. In particular, if $G = \mathbb{Z}^{d}$ and $\varphi(g) = \langle g \rangle^{N}$, $\mathcal{V}_{\varphi}^{p}(\mathcal{X})$ is the space $V_{N}^{p}(\mathcal{X})$ introduced in Definition 5.2.6, while $L_{p}(\mu, P, \varphi)$ coincides with the space $L_{p}(\mu, P, N)$ appeared in Definition 5.4.4.

We now state our result.

Theorem 5.4.21. Consider any Riemannian co-compact covering \mathcal{X} of subexponential growth with a discrete deck group G. Let P be a C^{∞} -bounded uniformly elliptic differential operator P of order m on \mathcal{X} such that $0 \notin \sigma(P)$. Let us denote by φ_0 the constant function 1 defined on G. Then the following statements are true:

(a) For each rigged divisor μ on \mathcal{X} , $L_p(\mu, P, \varphi) = L_{\infty}(\mu, P, \varphi_0)$, where $p \in [1, \infty]$ and $\varphi \in \mathcal{S}(G)$. Thus, all the spaces $L_p(\mu, P, \varphi)$ are the same.

- (b) dim $L_{\infty}(\mu, P, \varphi_0) = \deg_P(\mu) + \dim L_{\infty}(\mu^{-1}, P^*, \varphi_0).$
- (c) If $\mu = (D^+, L^+; \emptyset, 0)$, dim $L_{\infty}(\mu, P, \varphi_0) = \deg_P(\mu)$.

Now let $D \subset \mathcal{X}$ be a compact nowhere dense subset, and L be a finite dimensional subspace of $\mathcal{E}'_D(\mathcal{X})$. Denote $\mu := (D, L; 0, \emptyset)$. We also define the space $\tilde{L} := \{u \in \mathcal{E}'_D(\mathcal{X}) \mid P^*u \in L\}$.

As in Corollary 5.4.14, we obtain:

Corollary 5.4.22. Let the assumptions of Theorem 5.4.21 hold and a function $f \in C_c^{\infty}(X)$ be given such that $(f, \tilde{L}) = 0$. The following statements are equivalent:

- (i) f is orthogonal to the vector space $L_{\infty}(\mu, P^*, \varphi_0)$.
- (ii) There exists a unique solution u of the inhomogeneous equation Pu = f such that (u, L) = 0 and $u \in \mathcal{V}^p_{\varphi}(\mathcal{X})$ for some $p \in [1, \infty]$ and $\varphi \in \mathcal{S}(G)$.
- (iii) The equation Pu = f admits a unique solution u which has subexponential decay and satisfies (u, L) = 0.
- (iv) The equation Pu = f admits a unique solution u which has exponential decay and satisfies (u, L) = 0.

Remark 5.4.23.

- (i) Comparing to the Riemann-Roch formula (5.24), the Fredholm index of P does not appear in the formula in Theorem 5.4.21 (b) since P is invertible in this case.
- (ii) When μ is trivial, Theorem 5.4.21 (c) becomes Theorem 5.4.18 and our Corollary 5.4.22 is an analog of [47, Theorem 4.2.1] for the co-compact Riemannian coverings of subexponential growth.

5.5 **Proofs of the main results**

First, we introduce some notions for the proofs.

Definition 5.5.1. For each $s, N \in \mathbb{R}$ and $p \in [1, \infty]$, we denote by $V_{s,N}^p(X)$ the vector space consisting of all function $u \in C^{\infty}(X)$ such that for some compact subset K of X satisfying (5.9), the sequence $\{ \|u\|_{H^s(gK)} \langle g \rangle^{-N} \}_{g \in G}$ belongs to $\ell^p(G)$. We put

$$V_{s,N}^p(A) := V_{s,N}^p(X) \cap \operatorname{Ker} A.$$

Moreover, let $A_{s,N}^p$ be the elliptic operator A with the following domain

Dom
$$A_{s,N}^p = \{ u \in V_{s,N}^p(X) \mid Au \in C_c^\infty(X) \}.$$

When s = 0, we get back the notions of $V_N^p(X)$ and $V_N^p(A)$ in Definition 5.2.6, and the definitions of the operator A_N^p and its domain Dom A_N^p appeared in Definition 5.4.4.

Proof of Theorem 5.4.6

Proof. Fixing a pair (p, N) satisfying the assumption of the statement. We recall that A is an elliptic differential operator of order m on X. Let \mathcal{F} be the closure of a fundamental domain for G-action on X. We also pick a compact neighborhood $\hat{\mathcal{F}}$ in X so that $\mathcal{F} \subseteq \hat{\mathcal{F}}$ and $\hat{\mathcal{F}}$ satisfies the conclusion of Proposition 5.7.7. Our proof consists of several steps.

Step 1. We claim that given $p \in [1,\infty], N \in \mathbb{R}$, and any rigged divisor $\mu = (D^+, L^+; D^-, L^-)$, one should have

$$L(\mu, A_{m,N}^p) = L(\mu, A_N^p) = L_p(\mu, A, N).$$
(5.39)

Indeed, it suffices to show that $L(\mu, A_N^p) \subseteq L(\mu, A_{m,N}^p)$.

Consider $u \in L(\mu, A_N^p)$. Due to Remark 5.4.5, this implies that

$$\{\|u\|_{L^2(g\hat{\mathcal{F}})} \cdot \langle g \rangle^{-N}\}_{G_{\hat{\mathcal{F}},D^+}} \in \ell^p(G_{\hat{\mathcal{F}},D^+}),$$
(5.40)

where $G_{\hat{\mathcal{F}},D^+} = \{g \in G \mid \text{dist} (g\hat{\mathcal{F}},D^+) \geq 1\}$. Let $\mathcal{O} := X \setminus D^+$, then Au = 0 on \mathcal{O} and moreover, the set $G^{\mathcal{O}} = \{g \in G \mid g\hat{\mathcal{F}} \cap D^+ = \emptyset\}$ contains $G_{\hat{\mathcal{F}},D^+}$. By applying Proposition 5.7.7, for any $g \in G_{\hat{\mathcal{F}},D^+}$, one derives

$$||u||_{H^m(g\mathcal{F})} \lesssim ||u||_{L^2(g\hat{\mathcal{F}})}.$$
 (5.41)

By (5.40) and (5.41), $\{\|u\|_{H^m(g\mathcal{F})} \cdot \langle g \rangle^{-N}\}_{G_{\mathcal{F},D^+}} \in \ell^p(G_{\mathcal{F},D^+})$. By using Remark 5.4.5 again, this shows that $u \in L(\mu, A^p_{m,N})$. This proves(5.39).

This claim shows that instead of dealing with A_N^p , it suffices to work with $A_{m,N}^p$. We now introduce its 'adjoint' $(A_{m,N}^p)^*$ (in the sense of Subsection 5.3.1).

Definition 5.5.2. We denote by $(A_{m,N}^p)^*$ the elliptic operator A^* with the domain

Dom
$$(A^p_{m,N})^* = \{ v \in V^{p'}_{m,-N}(X) \mid A^* v \in C^{\infty}_c(X) \},\$$

where 1/p + 1/p' = 1. In another word, $(A_{m,N}^p)^* = (A^*)_{m,-N}^{p'}$.

We also define

$$\operatorname{Dom}' A^p_{m,N} = \operatorname{Dom}' (A^p_{m,N})^* := C^\infty_c(X).$$

Clearly, $C_c^{\infty}(X) \subseteq \text{Dom } A^p_{m,N}, \text{Dom } (A^p_{m,N})^* \subseteq C^{\infty}(X)$ (see (5.15)).

In the next steps, we will apply Theorem 5.3.10 to the operators $A_{m,N}^p$ and $(A_{m,N}^p)^*$.

Step 2. In order to apply Theorem 5.3.10, first, we need to check properties $(\mathcal{P}1) - (\mathcal{P}4)$. Obviously, the first three properties $(\mathcal{P}1) - (\mathcal{P}3)$ hold by definition. To show $(\mathcal{P}4)$, let $u \in \text{Dom } A^p_{m,N}$ and $v \in \text{Dom } (A^p_{m,N})^*$.

Note that since the operator A is G-periodic, one has

$$||Au||_{L^2(g\mathcal{F})} \lesssim ||u||_{H^m(g\mathcal{F})}$$
 and $||A^*v||_{L^2(g\mathcal{F})} \lesssim ||v||_{H^m(g\mathcal{F})}$, for any $g \in G$.

Now by Hölder's inequality, we have

$$\left| \sum_{g \in G} \langle Au, v \rangle_{L^{2}(g\mathcal{F})} \right| \leq \sum_{g \in G} \left| \langle Au, v \rangle_{L^{2}(g\mathcal{F})} \right| \leq \sum_{g \in G} \|Au\|_{L^{2}(g\mathcal{F})} \cdot \|v\|_{L^{2}(g\mathcal{F})}$$
$$\leq \|\{\|Au\|_{L^{2}(g\mathcal{F})} \langle g \rangle^{-N}\}_{g \in G} \|_{\ell^{p}(G)} \cdot \|\{\|v\|_{L^{2}(g\mathcal{F})} \langle g \rangle^{N}\}_{g \in G} \|_{\ell^{p'}(G)}$$
$$\lesssim \|\{\|u\|_{H^{m}(g\mathcal{F})} \langle g \rangle^{-N}\}_{g \in G} \|_{\ell^{p}(G)} \cdot \|\{\|v\|_{H^{m}(g\mathcal{F})} \langle g \rangle^{N}\}_{g \in G} \|_{\ell^{p'}(G)}$$
$$<\infty.$$
(5.42)

Similarly,

$$\left|\sum_{g\in G} \langle u, A^*v \rangle_{L^2(g\mathcal{F})}\right| \leq \|\{\|u\|_{H^m(g\mathcal{F})} \langle g \rangle^{-N}\}_{g\in G}\|_{\ell^p(G)} \cdot \|\{\|v\|_{H^m(g\mathcal{F})} \langle g \rangle^N\}_{g\in G}\|_{\ell^{p'}(G)} < \infty.$$

Hence, both $\langle A^p_{m,N}u,v\rangle$ and $\langle u,(A^p_{m,N})^*v\rangle$ are well-defined.

Our goal is to show that these two quantities are equal. To do this, for each $r \in \mathbb{N}$, we define

$$G_r := \{g \in G \mid |g| \ge r\},\$$

and \mathcal{F}_r as the union of all shifts of \mathcal{F} by deck group elements whose word length do not exceed r, i.e.,

$$\mathcal{F}_r := \bigcup_{g \notin G_{r+1}} g\mathcal{F} = \bigcup_{|g| \le r} g\mathcal{F}.$$

Obviously, $\mathcal{F}_r \Subset \mathcal{F}_{r+1}$ for any $r \in \mathbb{N}$ and the union of these subsets \mathcal{F}_r is the whole covering X. Let $\phi_r \in C_c^{\infty}(X)$ be a cut-off function such that $\phi_r = 1$ on \mathcal{F}_r and $\operatorname{supp} \phi_r \Subset$ \mathcal{F}_{r+1} . Furthermore, all derivatives of ϕ_r are uniformly bounded with respect to r. In particular, the following estimates hold for any smooth function w on X and any $g \in G$:

$$\|\phi_r w\|_{H^m(g\mathcal{F})} \lesssim \|w\|_{H^m(g\mathcal{F})}, \quad \|(1-\phi_r)w\|_{H^m(g\mathcal{F})} \lesssim \|w\|_{H^m(g\mathcal{F})}.$$
 (5.43)

Put $u_r := \phi_r u$ and $v_r := \phi_r v$. Since u_r and v_r are compactly supported smooth functions on X, $\langle Au_r, v_r \rangle = \langle u_r, A^*v_r \rangle$. Therefore, it is enough to demonstrate

$$\langle Au_r, v_r \rangle \to \langle Au, v \rangle \quad \text{and} \quad \langle u_r, A^* v_r \rangle \to \langle u, A^* v \rangle,$$
(5.44)

as $r \to \infty$. By symmetry, we only need to show the first convergence of (5.44). We use the triangle inequality to reduce (5.44) to checking

$$\lim_{r \to \infty} \langle A(u - u_r), v \rangle = \lim_{r \to \infty} \langle Au_r, (v - v_r) \rangle = 0.$$
(5.45)

We repeat the argument of (5.42) for the pairs of functions $((1 - \phi_r)u, v)$ and $(\phi_r u, (1 - \phi_r)v)$, and then use (5.43) to derive

$$\begin{aligned} |\langle Au_{r}, (v - v_{r})\rangle| + |\langle A(u - u_{r}), v\rangle| \\ &\leq \sum_{g \in G} |\langle A(\phi_{r}u), (1 - \phi_{r})v\rangle_{L^{2}(g\mathcal{F})}| + |\langle A((1 - \phi_{r})u), v\rangle_{L^{2}(g\mathcal{F})}| \\ &= \sum_{|g| \geq r+1} |\langle A(\phi_{r}u), (1 - \phi_{r})v\rangle_{L^{2}(g\mathcal{F})}| + |\langle A((1 - \phi_{r})u), v\rangle_{L^{2}(g\mathcal{F})}| \\ &\lesssim ||\{\|\phi_{r}u\|_{H^{m}(g\mathcal{F})}\langle g\rangle^{-N}\}_{g \in G_{r+1}}\|_{\ell^{p}(G_{r+1})} \cdot ||\{\|(1 - \phi_{r})v\|_{H^{m}(g\mathcal{F})}\langle g\rangle^{N}\}_{g \in G_{r+1}}\|_{\ell^{p'}(G_{r+1})} \\ &+ ||\{\|(1 - \phi_{r})u\|_{H^{m}(g\mathcal{F})}\langle g\rangle^{-N}\}_{g \in G_{r+1}}\|_{\ell^{p}(G_{r+1})} \cdot ||\{\|v\|_{H^{m}(g\mathcal{F})}\langle g\rangle^{N}\}_{g \in G_{r+1}}\|_{\ell^{p'}(G_{r+1})} \\ &\lesssim ||\{\|u\|_{H^{m}(g\mathcal{F})}\langle g\rangle^{-N}\}_{g \in G_{r+1}}\|_{\ell^{p}(G_{r+1})} \cdot ||\{\|v\|_{H^{m}(g\mathcal{F})}\langle g\rangle^{N}\}_{g \in G_{r+1}}\|_{\ell^{p'}(G_{r+1})}. \end{aligned}$$

$$(5.46)$$

Since $u \in V_{m,N}^p(X)$ and $v \in V_{m,-N}^{p'}(X)$, it follows that as $r \to \infty$, either

$$\|\{\|u\|_{H^m(g\mathcal{F})}\langle g\rangle^{-N}\}_{g\in G_{r+1}}\|_{\ell^p(G_{r+1})}$$

or

$$\|\{\|v\|_{H^m(g\mathcal{F})}\langle g\rangle^N\}_{g\in G_{r+1}}\|_{\ell^{p'}(G_{r+1})}$$

converges to zero (depending on either p or p' is finite), while the remaining term is bounded. Thus, we always have

$$\lim_{r \to \infty} \|\{\|u\|_{H^m(g\mathcal{F})} \langle g \rangle^{-N}\}_{g \in G_{r+1}} \|_{\ell^p(G_{r+1})} \cdot \|\{\|v\|_{H^m(g\mathcal{F})} \langle g \rangle^N\}_{g \in G_{r+1}} \|_{\ell^{p'}(G_{r+1})} = 0.$$

This fact and (5.46) imply (5.45). Hence, the property $(\mathcal{P}4)$ holds for $A_{m,N}^p$ and $(A_{m,N}^p)^*$.

Step 3. Clearly, $\operatorname{Ker} A_{m,N}^p = \{u \in \operatorname{Dom} A_{m,N}^p \mid Au = 0\} = V_{m,N}^p(A) = V_N^p(A)$. The latter equality is due to Schauder estimates (see (5.41) in Step 1). Also, $\operatorname{Ker}(A_{m,N}^p)^* = V_{m,-N}^{p'}(A^*) = V_{-N}^{p'}(A^*) = 0$ by Theorem 5.2.16. Hence, the kernels of $A_{m,N}^p$ and $(A_{m,N}^p)^*$ are of finite dimensional.

To prove that $A_{m,N}^p$ is Fredholm on X, we only need to show that $\operatorname{Im} A_{m,N}^p = C_c^{\infty}(X) = (\operatorname{Ker} (A_{m,N}^p)^*)^{\circ}$. Given any $f \in C_c^{\infty}(X)$, we want to find a solution u of the equation Au = f such that $u \in V_N^p(X)$. If such a solution u is found then automatically, u is in $V_{m,N}^p(X)$ by the same argument in Step 1 and the fact that Au = 0 on the complement of the compact support of f. Thus, f must belong to the range of $A_{m,N}^p$ and the proof is then finished. So our remaining task is to find such a solution u. This can be done as the following way. First, we pick a cut-off function η_r such that $\eta = 1$ around k_r and $\operatorname{supp} \eta_r \Subset V_r$, where V_r is the neighborhood of k_r appearing in Assumption \mathcal{A} . Define

$$\eta = \sum_{r=1}^{\ell} \eta_r.$$

Note that the Floquet transform $\mathbf{F}f$ is smooth in (k, x) since $f \in C_c^{\infty}(X)$. We decompose $\mathbf{F}f = \eta \mathbf{F}f + (1 - \eta)\mathbf{F}f$. Since the operator A(k) is invertible when $k \notin F_{A,\mathbb{R}}$, the operator function

$$\widehat{u}_0(k) := A(k)^{-1}((1 - \eta(k))\mathbf{F}f(k))$$

is well-defined and smooth in (k, x). By Theorem 5.7.5, the function $u_0 := \mathbf{F}^{-1} \widehat{u_0}$ has rapid decay. We recall when $k \in V_r$, the Riesz projection $\Pi_r(k)$ is defined in *Assumption* \mathcal{A} . Clearly, $0 \notin \sigma(A(k)_{|R(1-\Pi_r(k))})$, where we use the notation R(T) to denote the range of an operator T. Now the operator function

$$\widehat{v}_r(k) := \eta_r(k) (A(k)_{|R(1-\Pi_r(k))})^{-1} (1-\Pi_r(k)) \mathbf{F} f(k)$$

is also smooth and thus, the function $v_r := \mathbf{F}^{-1} \hat{v_r}$ has rapid decay by Theorem 5.7.5 again. In particular, $u_0, v_r (1 \le r \le \ell)$ are in the space $V_0^{\infty}(X)$.

Let us fix $1 \le r \le \ell$. For any $k \in V_r \setminus \{k_r\}$, due to ($\mathcal{A}2$), we can define the operator function $\widehat{w_r}(k) := \eta_r(k)(\mathcal{A}(k)_{|\mathcal{R}(\Pi_r(k))})^{-1}\Pi_r(k)\mathbf{F}f(k)$, which is in the range of $\Pi_r(k)$. By expanding $\Pi_r(k)\mathbf{F}f(k)$ in terms of the basis $(f_j(k))_{1\le j\le m_r}$, it is easy to see that

$$\|\widehat{w_r}(k)\|_{L^2_k(X)} \lesssim \max_{1 \le j \le m_r} \|A_r(k)^{-1} f_j(k)\|_{L^2_k(X)} \cdot \|\mathbf{F}f(k)\|_{L^2_k(X)}.$$

From this and the integrability condition in (A2), we obtain

$$\int_{\mathbb{T}^d} \|\widehat{w_r}(k)\|_{L^2_k(X)} \, \mathrm{d}k \lesssim \int_{V_r \setminus \{k_r\}} \|\mathbf{F}f(k)\|_{L^2_k(X)} \cdot \|(A_r(k)^{-1}f_j(k))_{1 \le j \le m_r}\|_{\ell^{\infty}(\mathbb{C}^{m_r})} \, \mathrm{d}k \\
\lesssim \sup_{k \in V_r} \|\mathbf{F}f(k)\|_{L^2_k(X)} \cdot \int_{V_r \setminus \{k_r\}} \|A_r(k)^{-1}\|_{\mathcal{L}(\mathbb{C}^{m_r})} \, \mathrm{d}k < \infty.$$

Hence, $\widehat{w_r} \in L^1(\mathbb{T}^d, \mathcal{E}^0)$.

Summing up, the function $\widehat{u} := \widehat{u}_0 + \sum_{1 \leq r \leq \ell} (\widehat{v}_r + \widehat{w}_r)$ belongs to $L^1(\mathbb{T}^d, \mathcal{E}^0)$, and moreover, it satisfies the equation

$$\begin{aligned} A(k)\widehat{u}(k) &= A(k)\widehat{u_0}(k) + \sum_{1 \le r \le \ell} A(k)(\widehat{v_r}(k) + \widehat{w_r}(k)) \\ &= (1 - \eta(k))\mathbf{F}f(k) + \sum_{1 \le r \le \ell} \eta_r(k)\mathbf{F}f(k) = \mathbf{F}f(k) \end{aligned}$$

From the above equation, $\hat{u}(k, x)$ is smooth in x for each quasimomentum k. We define $u := \mathbf{F}^{-1}\hat{u}$ by using the formula (5.58). According to Lemma 5.7.6, $u \in L^2_{loc}(X)$. For any $\phi \in C^{\infty}_c(X)$, we can use Fubini's theorem to get

$$\begin{split} \langle u, A^* \phi \rangle_{L^2(X)} &= \int_X \mathbf{F}^{-1} \widehat{u}(k, x) \cdot A^* \phi(x) \, \mathrm{d}\mu_X(x) \\ &= \frac{1}{(2\pi)^d} \int_{\mathbb{T}^d} \int_X \widehat{u}(k, x) \cdot A^* \phi(x) \, \mathrm{d}\mu_X(x) \, \mathrm{d}k = \frac{1}{(2\pi)^d} \int_{\mathbb{T}^d} \int_X A \widehat{u}(k, x) \cdot \phi(x) \, \mathrm{d}\mu_X(x) \, \mathrm{d}k \\ &= \frac{1}{(2\pi)^d} \int_{\mathbb{T}^d} \int_X A(k) \widehat{u}(k, x) \cdot \phi(x) \, \mathrm{d}\mu_X(x) \, \mathrm{d}k \\ &= \frac{1}{(2\pi)^d} \int_{\mathbb{T}^d} \int_X \mathbf{F} f(k, x) \cdot \phi(x) \, \mathrm{d}\mu_X(x) \, \mathrm{d}k = \langle f, \phi \rangle_{L^2(X)}. \end{split}$$

Hence, u is a weak solution of the inhomogeneous equation Au = f on X. Elliptic regularity then implies that u is a classical solution and therefore, $u \in V_0^{\infty}(X)$ due to Lemma 5.7.6 again. If either $N \ge 0$ when $p = \infty$ or N > d/p when $p \in [1, \infty)$, we always have $V_0^{\infty}(X) \subseteq V_N^p(X)$. Thus, this shows that $A_{m,N}^p$ is a Fredholm operator on X.

Step 4. From Step 2 and Step 3, the operator $A_{m,N}^p$ satisfies the assumption of Theorem 5.3.10. Then the Liouville-Riemann-Roch inequality (5.33) follows immediately from (5.39) and Theorem 5.3.10.

Remark 5.5.3. Assumption (\mathcal{A}_2) is needed to guarantee the validity of the Liouville-Riemann-Roch inequality (5.4.6) (at least when $p = \infty$ and N = 0). Indeed, consider $-\Delta$ in \mathbb{R}^2 . Let μ be the point divisor ({0}, $L, \emptyset, 0$), where $L = \mathbb{C}\delta_0$. It is not difficult to see that the space $L_{\infty}(\mu, -\Delta, 0)$ contains only constant functions since the standard fundamental solution $u_0(x) = -\frac{1}{2\pi} \ln |x|$ is not bounded at infinity. Clearly, $L_1(\mu^{-1}, -\Delta, 0)$ is trivial. Hence, we have:

$$\dim L_{\infty}(\mu, -\Delta, 0) = 1 < 2 = \deg_{-\Delta}(\mu) + \dim V_0^{\infty}(-\Delta) + \dim L_1(\mu^{-1}, -\Delta, 0).$$

Proof of Theorem 5.4.10

Proof. From Step 3 in the proof of Theorem 5.4.6, the operator $A_{m,N}^p$ is Fredholm on X and Im $A_{m,N}^p = C_c^{\infty}(X) = \text{Dom}'(A_{m,N}^p)^*$. Now we can apply Corollary 5.3.13 to finish the proof of the equality (5.34). The upper bound estimate (5.35) follows from (5.34) and the trivial inclusion $L_p(\mu, A, N) \subseteq L_p(\mu^+, A, N)$.

Proof of Proposition 5.4.11

Proof. (a) We define $\mu_0 := (\emptyset, 0; D^-, L^-)$. Now suppose for contradiction that for any M > 0, the space $L_{p'}(\mu_M^{-1}, A^*, -N)$ is non-trivial for some rigged divisor $\mu_M = (D^+, L_M^+; D^-, L^-)$ such that $\mathcal{L}_M^+ \subseteq L_M^+$. Note that $L_{p'}(\mu_M^{-1}, A^*, -N)$ is a subspace of $L_{p'}(\mu_0^{-1}, A^*, -N)$. It follows from Proposition 5.4.10 that $L_{p'}(\mu_0^{-1}, A^*, -N)$ is a finite dimensional vector space and thus, we equip it with any norm $\|\cdot\|$. Thus, there is a sequence $\{u_M\}_{M\in\mathbb{N}}$ in $L_{p'}(\mu_0^{-1}, A^*, -N)$ such that $\|u_M\| = 1$ and $(u_M, L_M^+) = 0$. In particular, $(u_M, \mathcal{L}_M^+) = 0$ and therefore, $\partial^{\alpha} u_M(x_0) = 0$ for any $0 \le |\alpha| \le M$. By passing to a subsequence if necessary, there exists $v \in L_{p'}(\mu_0^{-1}, A^*, -N)$ for which $\lim_{M\to\infty} \|u_M - v\| = 0$. It is clear that $\|w\|_{C^M(K)} \le \|w\|$ for any w in $L_{p'}(\mu_0^{-1}, A^*, -N)$, $M \ge 0$, and compact subset $K \Subset X \setminus D^-$. Hence, for any multi-index $\alpha, \partial^{\alpha} v(x_0) = \lim_{M\to\infty} \partial^{\alpha} u_M(x_0) = 0$. As a local smooth solution of A^* , v must vanish on $X \setminus D^-$ due to the strong unique continuation property of A^* . Consequently, v = 0 as an element in $L_{p'}(\mu_0^{-1}, A^*, -N)$ and this gives us a contradiction to the fact that $\|v\| = 1$.

(b) It suffices to prove the statement for the case p = ∞. If r ≥ 0, we define the corresponding point divisor μ_r := (Ø, 0; {0}, L_r⁻), where

$$L_r^- := \left\{ \sum_{|\alpha| \le r} c_\alpha \partial^\alpha \delta(\cdot - 0) \mid c_\alpha \in \mathbb{C} \right\}.$$

Let us consider the function $v_{\alpha}(x) := \partial^{\alpha}(|x|^{2-d})$ for each multi-index α such that $|\alpha| > N + 2$. It is clear that $|v_{\alpha}(x)| \leq |x|^{-|\alpha|-d+2}$ for $x \neq 0$. Therefore, we obtain

$$\sum_{g \in \mathbb{Z}^d} \|v_{\alpha}\|_{L^2([0,1)^d + g)} \cdot \langle g \rangle^N \lesssim \sum_{g \in \mathbb{Z}^d} \langle g \rangle^{-|\alpha| - d + 2 + N} < \infty.$$

Since $|x|^{2-d}$ is a fundamental solution of $-\Delta$ on \mathbb{R}^d (up to some multiplicative constant), v_{α} belongs to the space $L_1(\mu_r^{-1}, -\Delta_{|\mathbb{R}^d}, -N)$ provided that $N + 2 < |\alpha| \le r$. Now let us pick some multi-indices $\alpha_1, \ldots, \alpha_{r-N-2}$ such that $|\alpha_j| = N + 2 + j$ for any $1 \le j \le r - N - 2$. By homogeneity, the functions v_{α_j} are linearly independent as smooth functions on $\mathbb{R}^d \setminus \{0\}$. By letting $r \to \infty$, this completes our proof.

Remark 5.5.4.

- (i) There are large classes of elliptic operators that have the strong unique continuation property, e.g., elliptic operators of second-order with smooth coefficients or elliptic operators with real analytic coefficients.
- (ii) Note that the finiteness of the real Fermi surface F_{A,ℝ} would imply the weak unique continuation property of A*, i.e., A* does not have any non-zero compactly supported solution (see e.g., [47]). We do not know whether the first statement of Proposition 5.4.11 holds if one drops the strong unique continuation property of A*.

Proof of Theorem 5.4.12

Proof. The proof is similar to the proof of Theorem 5.4.6, except for Step 3 where we have a minor modification. Let us keep the same argument and notions in Step 3. The goal here is to prove the solvability of the equation Au = f, where $f \in C_c^{\infty}(X)$ and $u \in V_N^p(X)$. Under the L^2 -assumption, the functions \widehat{w}_r $(1 \leq r \leq \ell)$, which are defined in Step 3, belong to $L^2(\mathbb{T}^d, \mathcal{E}^0)$. Thus, $\widehat{u} \in L^2(\mathbb{T}^d, \mathcal{E}^0)$ and then by Theorem 5.7.5, u is in $L^2(X)$ as well as Au = f. This means that $u \in V_0^2(X)$. If $p \geq 2, N \geq 0$, the inclusion $V_0^2(X) \subseteq V_N^2(X)$ is obvious while if $p \in [1, 2), N > (1/p - 1/2)d$, one can use Hölder's inequality to obtain the inclusion $V_0^2(X) \subset V_N^p(X)$. This completes the proof of the first statement. In particular, when p = 2, N = 0, both of the operators A_0^2 and $(A_0^2)^* = (A^*)_0^2$ are Fredholm. Therefore, we obtain the equality (5.37) since it is known that dim $V_0^2(A) = \dim V_0^2(A^*) = 0$ (see Theorem 5.2.16 (a)).

Remark 5.5.5.

- (a) The integrability of ||A_r(k)⁻¹||²_{L(C^{mr})} is needed in Theorem 5.4.12. For example, let us consider A = -Δ on ℝ^d (d < 5) and the point divisor μ representing a simple pole at 0. Then the fundamental solution c_d|x|^{2-d} does not belong to L₂(μ, -Δ, 0) and thus, dim L₂(μ, -Δ, 0) = 0 < 1 = deg_{-Δ}(μ). Therefore, the equalities (5.36) and (5.37) do not hold in this case.
- (b) Under the assumption of Theorem 5.4.12, the Liouville-Riemann-Roch inequality
 (5.33) holds for any N ≥ 0 if and only if p ≥ 2. Indeed, suppose that d ≥ 5 and p < 2, then (2 d)p ≥ -d and therefore,

$$\int_{|x|\ge 1} |x|^{(2-d)p} \,\mathrm{d}x = \infty.$$

This implies that $L_p(\mu, -\Delta, 0) = \{0\}$, where μ is also the point divisor 0^1 . So (5.33) fails since dim $L_p(\mu, -\Delta, 0) = 0 < 1 = \deg_{-\Delta}(\mu)$.

Proof of Proposition 5.4.14

Proof. We evoke the operators $A_{m,0}^2$ and $(A_{m,0}^2)^*$ and their corresponding domains from the proof of Theorem 5.4.6. Now we recall from our discussion in Subsection 5.3 the notations of the operators

$$\widetilde{A_{m,0}^2}: \Gamma(X,\mu^{-1},A_{m,0}^2) \to \widetilde{\Gamma}_{\mu^{-1}}(X,A_{m,0}^2)$$

and

$$(\widetilde{A_{m,0}^2})^* : \Gamma(X, \mu, (A_{m,0}^2)^*) \to \widetilde{\Gamma}_{\mu}(X, (A_{m,0}^2)^*)$$

are extensions of the two operators $A_{m,0}^2$ and $(A_{m,0}^2)^*$ with respect to the divisors μ^{-1} and μ , respectively. From (5.37) and Remark 5.3.12, we obtain the duality $L_2(\mu, A^*, 0)^\circ = (\operatorname{Ker}(\widetilde{A_{m,0}^2)^*})^\circ = \operatorname{Im}\widetilde{A_{m,0}^2}$. Put in another word, f is orthogonal to \widetilde{L} and $L_2(\mu, A^*, 0)$ if and only if f = Au for some u in the space $\Gamma(X, \mu^{-1}, A_{m,0}^2) = \{v \in H^m(X) \mid Av \in C_c^\infty(X), \langle v, L \rangle = 0\}$. This proves the equivalence of (i) and (ii).

Proof of Proposition 5.4.15

Proof. If $\ell \in \mathbb{N}$, we choose ℓ distinct points z_1, \ldots, z_ℓ in \mathbb{R}^d and define $\mu_\ell := (\emptyset, 0; D^-, L^-)$, where $D^- = \{z_1, \ldots, z_\ell\}$ and

$$L^{-} = \left\{ \sum_{1 \le j \le \ell} \sum_{1 \le \alpha \le d} c_{j\alpha} \frac{\partial}{\partial x_{\alpha}} \delta(x - z_{j}) \mid c_{j\alpha} \in \mathbb{C} \right\}.$$

In terms of the notations in Example 5.3.14, $k = 0, l = \ell$.

Let us recall now the spaces $L(\mu, -\Delta)$ and $L(\mu^{-1}, -\Delta)$ from Example 5.3.14. By their definition, it is clear that $L_{\infty}(\mu, -\Delta, 0) = \mathbb{C}$. Hence,

$$\dim L_{\infty}(\mu, -\Delta, 0) = 1 = \dim L(\mu, -\Delta) + \dim V_0^{\infty}(-\Delta).$$
(5.47)

On the other hand, if $v \in L_1(\mu^{-1}, -\Delta, 0)$ then

$$\lim_{R \to \infty} \sum_{|g| \ge R} \|v\|_{L^2([0,1)^d + g)} = 0.$$

It follows that as $|g| \to \infty$, $||v||_{L^2([0,1)^d+g)} \to 0$. Using elliptic regularity, this is equivalent to the condition $\lim_{|x|\to\infty} v(x) = 0$. Thus, $L_1(\mu^{-1}, -\Delta, 0)$ is a subspace of $L(\mu^{-1}, -\Delta)$. Define

$$v_{j\alpha}(x) = \frac{\partial}{\partial x_{\alpha}} |x - z_j|^{2-d}.$$

Then $v_{j\alpha} \in L(\mu^{-1}, -\Delta)$ (see Example 5.3.14). We now claim that $v_{j\alpha} \notin L_1(\mu^{-1}, -\Delta, 0)$ for any $1 \le j \le \ell$ and $1 \le \alpha \le d$. Suppose for contradiction that

$$v_{j\alpha}(x) = (2-d) \frac{(x_{\alpha} - (z_j)_{\alpha})}{|x - z_j|^d} \in L_1(\mu^{-1}, -\Delta, 0).$$

This implies that for some R > 0, we have

$$V_{\alpha,R} := \sum_{g \in \mathbb{Z}^d, |g| \ge R} \left(\int_{[0,1)^d + g} \frac{|x_\alpha - (z_j)_\alpha|^2}{|x - z_j|^{2d}} \, \mathrm{d}x \right)^{1/2} < \infty.$$

But this gives us a contradiction, since

$$V_{\alpha,R} \gtrsim \sum_{g \in \mathbb{Z}^d, |g| \ge R} \frac{|g_\alpha|}{|g|^d} \ge \sum_{g \in \mathbb{Z}^d, g_\alpha \neq 0, |g| \ge R} \frac{1}{|g|^d} = \infty.$$

From the above claim and the linearly independence of the functions $v_{j\alpha}$ as smooth functions on $\mathbb{R}^d \setminus D^-$, it follows that

$$\dim L_1(\mu^{-1}, -\Delta, 0) \le \dim L(\mu^{-1}, -\Delta) - d\ell.$$
(5.48)

From (5.31), (5.47) and (5.48), we conclude

$$d\ell + \dim L_1(\mu^{-1}, -\Delta, 0) + \deg_{-\Delta}(\mu) + \dim V_0^{\infty}(-\Delta)$$

$$\leq \dim L(\mu^{-1}, -\Delta) + \deg_{-\Delta}(\mu) + \dim V_0^{\infty}(-\Delta) = \dim L_{\infty}(\mu, -\Delta, 0).$$
(5.49)

This yields the inequality (5.38).

Remark 5.5.6.

(i) Note that the examples we constructed in Proposition 5.4.15 also show that the Liouville-Riemann-Roch inequality can be strict for other cases. Let us briefly mention some of them here.

Case 1. $p = \infty, N \ge 0$.

If $(d-1)\ell + 1 \ge \dim V_N^{\infty}(-\Delta)$, one obtains from (5.49) that

$$\dim L_1(\mu^{-1}, -\Delta, -N) + \deg_{-\Delta}(\mu) + \dim V_N^{\infty}(-\Delta) + \ell$$
$$\leq \dim L_1(\mu^{-1}, -\Delta, 0) + \deg_{-\Delta}(\mu) + \dim V_0^{\infty}(-\Delta) + d\ell \leq \dim L_{\infty}(\mu, -\Delta, N).$$

Case 2. $1 \le p < \infty, N > d/p$.

Note that dim $L_p(\mu, -\Delta, N) \ge 1$ since it contains constant solutions. Again, each function $v_{j\alpha}$ does not belong to the space $L_{p'}(\mu^{-1}, -\Delta, -N)$. In fact, for $R > 2|z_j|$ large enough and p > 1, we get

$$\sum_{g \in \mathbb{Z}^d, |g| > R} \|v_{j\alpha}\|_{L^2([0,1)^d + g)}^{p'} \cdot \langle g \rangle^{p'N} \gtrsim \sum_{\substack{\min_{1 \le l \le d} g_l > R}} \langle g \rangle^{p'(N-d)} = \infty.$$

The case p = 1, N > d - 1 is proved similarly. Now as in the proof of (5.49), we

obtain the inequality

$$\dim L_{p'}(\mu^{-1}, -\Delta, -N) + \deg_{-\Delta}(\mu) + \dim V_N^p(-\Delta) + \ell$$
$$\leq \dim L(\mu^{-1}, -\Delta) - d\ell + \deg_{-\Delta}(\mu) + \dim V_N^p(-\Delta) + \ell \leq \dim L_p(\mu, -\Delta, N),$$

provided that $(d-1)\ell + 1 \ge \dim V_N^p(-\Delta)$.

In physics, the functions $v_{j\alpha}$ in $L(\mu^{-1}, -\Delta)$ are the potentials of dipoles located at the equilibrium positions z_j .

(ii) We can also modify our example in Proposition 5.4.15 to obtain examples for (5.38) in the case of point divisors. For instance, we could take the point divisor μ = (Ø, 0; D⁻, L⁻), where D⁻ = {z₁,..., z_ℓ} and L⁻ = span_C {∂^αδ(x-z_j)}_{1≤j≤ℓ,0≤|α|≤1}. Similarly, L_∞(μ, -Δ, 0) = L₁(μ⁻¹, -Δ, 0) = {0} and moreover, deg_{-Δ}(μ) = ℓ(d+1). Hence, dim L_∞(μ, -Δ, 0) = (ℓ(d+1) - 1) + deg_{-Δ}(μ) + dim V₀[∞](-Δ) + dim L₁(μ⁻¹, -Δ, 0).

Additionally, our method can also be adapted without much difficulty to provide more examples of the inequality (5.38) when the positive parts μ^+ and negative parts μ^- of the rigged divisors μ are required to be both non-trivial.

Proof of Proposition 5.4.16

Proof. By translation invariance of A, we can suppose without loss of generality that $x_0 = 0$. Now we fix a pair (p, N) as in the assumption of the statement. We recall from Definition 5.5.1 the notations of the operator $A_{m,N}^p$ and its corresponding domain Dom $A_{m,N}^p$. We will apply Corollary 5.3.13 to the operator $P := A_{m,N}^p$ in order to prove the statement. From our assumption on the operator A and the pair (p, N) and from the conclusion of Step 3 of the proof of Theorem 5.4.6, we only need to show the following claim: If u is a smooth function on \mathbb{R}^d such that $|u(x)| \leq \langle x \rangle^N$ and $\langle Au, \tilde{L}^- \rangle = 0$, then

there is a polynomial v of degree M_0 satisfying Av = 0 and $\langle u - v, L^- \rangle = 0$. Indeed, if this claim holds true, v will belong to the space $\text{Dom } A^p_{m,N}$ due to our condition on p and N. This will fulfill all the necessary assumptions of Corollary 5.3.13 in order to apply it.

To prove the claim, we first introduce the following polynomial:

$$v(x) := \sum_{M_1 \le |\alpha| \le M_0} \frac{\partial^{\alpha} u(0)}{\alpha!} x^{\alpha}.$$

Then, $\langle v - u, g \rangle = 0$ if $g = \partial^{\alpha} \delta(\cdot - 0)$ and $M_1 \leq |\alpha| \leq M_0$. Let $a(\xi)$ be the symbol of the constant-coefficient differential operator A(x, D), i.e.,

$$A = A(x, D) = \sum_{|\alpha|=m} \frac{1}{\alpha!} \partial_{\xi}^{\alpha} a(0) D^{\alpha}.$$

Define $\tilde{M}_j := \max\{0, M_j - m\}$ for $j \in \{0, 1\}$. A straightforward calculation gives us:

$$\begin{split} A(x,D)v(x) &= i^m \sum_{|\alpha|=m} \sum_{|\beta|=\tilde{M}_1}^{\tilde{M}_0} \frac{1}{\alpha!} \frac{1}{\beta!} \partial_{\xi}^{\alpha} a(0) \partial_x^{\alpha+\beta} u(0) \cdot x^{\beta} \\ &= \sum_{|\beta|=\tilde{M}_1}^{\tilde{M}_0} \frac{1}{\beta!} \left(\sum_{|\alpha|=m} \frac{1}{\alpha!} \partial_{\xi}^{\alpha} a(0) \cdot D^{\alpha} (\partial_x^{\beta} u)(0) \right) \cdot x^{\beta} \\ &= \sum_{|\beta|=\tilde{M}_1}^{\tilde{M}_0} \frac{1}{\beta!} A \partial^{\beta} u(0) \cdot x^{\beta}. \end{split}$$

Because $\partial^{\beta} \delta(\cdot - 0) \in \tilde{L}^{-}$ when $\tilde{M}_{1} \leq |\beta| \leq \tilde{M}_{0}$, we obtain $A \partial^{\beta} u(0) = \partial^{\beta} A u(0) = 0$ for such multi-indices β . Now we conclude that Av = 0, which proves our claim.

Remark 5.5.7.

(i) In Proposition 5.4.16, if d > m then any elliptic real-constant-coefficient homogeneous differential operator A of order m on \mathbb{R}^d satisfies Assumption \mathcal{A} . Notice that

m must be even. Since $F_{A,\mathbb{R}} = \{0\}$, it is not hard to see from Theorem 5.2.10 that $\dim V_N^p(A) = \dim V_N^p((-\Delta)^{m/2})$. In particular, if μ is the point divisor $x_0^{-(M_0+1)}$, the Liouville-Riemann-Roch formula becomes

$$\dim L_p(\mu, A, N) = \begin{cases} h_{d, [N]}^{(m)} - h_{d, M_0}^{(m)}, & \text{if } p = \infty. \\ h_{d, \lfloor N - d/p \rfloor}^{(m)} - h_{d, M_0}^{(m)}, & \text{otherwise}, \end{cases}$$

though this can be proven elementarily. Here for $A, B, C \in \mathbb{N}$, we denote by $h_{A,B}^{(C)}$ the quantity $\binom{A+B}{A} - \binom{A+B-C}{A}$, where we adopt the agreement in Definition 5.2.11.

(ii) As a special case of Proposition 5.4.16, the Liouville-Riemann-Roch equality for the Laplacian operator could occur when μ⁻ is not trivial (compare with Theorem 5.4.10). As we have seen so far, the corresponding spaces L_{p'}(μ⁻¹, -Δ, -N) in this situation are trivial too. It is worth mentioning that it is possible to obtain the Liouville-Riemann-Roch equality in certain cases for which the dimensions of the spaces L_{p'}(μ⁻¹, -Δ, -N) are as large as possible. For instance, let p = ∞ and r ≥ N+3, we define μ := (Ø, 0; D⁻, L⁻) with D⁻ = {0} and L⁻ = span_C{∂^αδ(· - 0)}_{|α|=r}. Then clearly, L_∞(μ, -Δ, N) = V_N[∞](-Δ). From the proof of the second part of Proposition 5.4.11,

$$L_1(\mu^{-1}, -\Delta, -N) = \operatorname{span}_{\mathbb{C}} \{\partial^{\alpha}(|x|^{2-d})\}_{|\alpha|=r}$$

= $\{u \in C^{\infty}(\mathbb{R}^d \setminus \{0\}) \mid -\Delta u \in L^-, \lim_{|x| \to \infty} |u(x)| = 0\}.$

By Theorem 5.3.10, it is easy to see that the dimension of this space is equal to the degree of the divisor μ^{-1} (see also Example 5.3.14). Thus, dim $L_{\infty}(\mu, -\Delta, N) =$

 $\dim L_1(\mu^{-1}, -\Delta, -N) + \deg_{-\Delta}(\mu) + \dim V_N^{\infty}(-\Delta) \text{ and as } r \to \infty,$

$$\dim L_1(\mu^{-1}, -\Delta, -N) = \binom{d+r-1}{d-1} - \binom{d+r-3}{d-1} \to \infty.$$

Proof of Theorem 5.4.21

Proof. The key lemma of our proof is the following statement:

Lemma 5.5.8. Let us consider $p_1, p_2 \in [1, \infty]$ and two positive functions φ_1 and φ_2 in S(G) such that one of the following two conditions holds:

- $p_1^{-1} + p_2^{-1} \ge 1$ and $\varphi_1 \varphi_2$ is bounded on G.
- $p_1^{-1} + p_2^{-1} \le 1$ and $\varphi_1^{-1}\varphi_2^{-1}$ is bounded on G.

Then the following Riemann-Roch formula holds:

$$\dim L_{p_1}(\mu, P, \varphi_1) = \deg_P(\mu) + \dim L_{p_2}(\mu^{-1}, P^*, \varphi_2),$$

where μ is any rigged divisor on \mathcal{X} .

Proof of Lemma 5.5.8.

Proof. We follow the strategy of the proof of Theorem 5.4.6. As in Definition 5.5.1, for each $s \in \mathbb{R}$, $\varphi \in \mathcal{S}(G)$ and $p \in [1, \infty]$, let us introduce the following space

$$\mathcal{V}_{s,\varphi}^{p}(\mathcal{X}) := \{ u \in C^{\infty}(\mathcal{X}) \mid \{ \|u\|_{H^{s}(g\mathcal{F})} \cdot \varphi(g) \}_{g \in G} \in \ell^{p}(G) \}.$$

and we denote by $P^p_{m,\varphi}$ the operator P with the domain

Dom
$$P^p_{m,\varphi} := \{ u \in \mathcal{V}^p_{m,\varphi}(\mathcal{X}) \mid Pu \in C^{\infty}_c(\mathcal{X}) \}$$

For the elliptic differential operator P^* , we also use the corresponding notations $(P^*)_{m,\varphi}^p$ and $\text{Dom}(P^*)_{m,\varphi}^p$.

Now let us fix a pair of two real numbers (p_1, p_2) and a pair of two functions (φ_1, φ_2) satisfying the conditions of Lemma 5.5.8. From now on, we will consider the operator $P_{m,\varphi_1}^{p_1}$ and its "adjoint" $(P_{m,\varphi_1}^{p_1})^* := (P^*)_{m,\varphi_2}^{p_2}$. As before, we define $\text{Dom}'(P_{m,\varphi_1}^{p_1}) = \text{Dom}'(P^*)_{m,\varphi_2}^{p_2} = C_c^{\infty}(\mathcal{X})$.

Our goal is to verify the assumptions of Theorem 5.3.10 for the operator $P_{m,\varphi_1}^{p_1}$ and its adjoint $(P^*)_{m,\varphi_2}^{p_2}$. The proof also goes through four steps as in the proof of Theorem 5.4.6. We consider two cases.

Case 1. $p_1^{-1} + p_2^{-1} \ge 1, \varphi_1 \varphi_2 \lesssim 1$. The proof of Step 1 stays exactly the same as before (see Remark 5.7.8). For Step 2, the first three properties $(\mathcal{P}1) - (\mathcal{P}3)$ are also obvious. For the property $(\mathcal{P}4)$, we want to show that whenever $u \in \text{Dom } P_{m,\varphi_1}^{p_1}$ and $v \in \text{Dom}(P^*)_{m,\varphi_2}^{p_2}$,

$$\langle Pu, v \rangle = \langle u, P^*v \rangle. \tag{5.50}$$

Because P and P^* are C^{∞} -bounded, we can repeat the approximation procedure and obtain similar estimates from the proof from Theorem 5.4.6 for showing the identity (5.50) whenever $u \in \text{Dom } P_{m,\varphi_1}^{p_1}$ and $v \in \text{Dom}(P^*)_{m,\varphi_1^{-1}}^{p'_1}$. On the other hand, $\mathcal{V}_{m,\varphi_2}^{p_2}(\mathcal{X}) \subseteq$ $\mathcal{V}_{m,\varphi_1^{-1}}^{p'_1}(\mathcal{X})$ and hence, $\text{Dom}(P^*)_{m,\varphi_2}^{p_2} \subseteq \text{Dom}(P^*)_{m,\varphi_1^{-1}}^{p'_1}$. With this inclusion, it is enough to conclude the property ($\mathcal{P}4$) in this case and this finishes Step 2. For Step 3, first, it is clear that the kernels of $P_{m,\varphi_1}^{p_1}$ and $(P^*)_{m,\varphi_2}^{p_2}$ are both trivial since P and P^* satisfy (SSP) (see Theorem 5.4.18). So the rest is to verify the Fredholm property for both operators $P_{m,\varphi_1}^{p_1}$ and $(P^*)_{m,\varphi_2}^{p_2}$, i.e., to prove that

$$\operatorname{Im} P^{p_1}_{m,\varphi_1} = \operatorname{Im}(P^*)^{p_2}_{m,\varphi_2} = C^{\infty}_c(\mathcal{X}).$$
(5.51)

Let us prove (5.51) for the operator $P_{m,\varphi_1}^{p_1}$ since the other identity is proved similarly. We denote by $G_P(x, y)$ the Green's function of P at the level $\lambda = 0$, i.e., $G_p(x, y)$ is the Schwartz kernel of the resolvent operator P^{-1} . It is known that $G_P(x, y) \in C^{\infty}(\mathcal{X} \times \mathcal{X} \setminus \Delta)$, where $\Delta = \{(x, x) \mid x \in \mathcal{X}\}$. Moreover, all of its derivatives have exponential decay off the diagonal (see [71, Theorem 2.2]). However, it is more convenient for us to use its L^2 -norm version, i.e., [71, Theorem 2.3]: there exists $\varepsilon > 0$ such that for every $\delta > 0$ and every multi-indices α, β , one can find a constant $C_{\alpha\beta\delta} > 0$ such that

$$\int_{x:d_{\mathcal{X}}(x,y)\geq\delta} |\partial_x^{\alpha}\partial_y^{\beta}G_P(x,y)|^2 \exp\left(\varepsilon d_{\mathcal{X}}(x,y)\right) d\mu_{\mathcal{X}}(x) \leq C_{\alpha\beta\delta}.$$
(5.52)

Here the derivatives $\partial_x^{\alpha}, \partial_y^{\beta}$ are taken with respect to canonical coordinates and the constants $C_{\alpha\beta\delta}$ do not depend on the choice of such canonical coordinates. Note that these decay estimates (5.52) still work in the case of exponential growth of the volume of the balls on \mathcal{X} . Let $f \in C_c^{\infty}(\mathcal{X})$ and K be its compact support in \mathcal{X} . We introduce

$$u(x) := P^{-1}f(x) = \int_{\mathcal{X}} G_P(x, y) f(y) \,\mathrm{d}\mu_{\mathcal{X}}(y),$$

where $\mu_{\mathcal{X}}$ is the Riemannian measure on \mathcal{X} . Thus, $u \in L^2(\mathcal{X})$ since P^{-1} is a bounded operator on $L^2(\mathcal{X})$. It is clear that u is a weak solution of the equation Pu = f, and hence, by regularity, u is a smooth solution. We only need to prove that $u \in \mathcal{V}_{m,\varphi_1}^1(\mathcal{X}) \subseteq$ $\mathcal{V}_{m,\varphi_1}^{p_1}(\mathcal{X})$. Let us consider any g in $G_{\overline{F},K} := \{g \in G \mid \text{dist}(g\overline{F},K) > 1\}$. Since \mathcal{X} is quasi-isometric to the metric space (G, d_S) via the orbit action by the Švarc-Milnor lemma, it is not hard to see that there are constants $C_1, C_2 > 0$ such that for every $g \in G_{\overline{F},K}$ and every $(x, y) \in g\mathcal{F} \times K$, one has

$$2C_1|g| - C_2 \le d_{\mathcal{X}}(x, y) \le (2C_1)^{-1}|g| + C_2.$$
(5.53)

Taking $\delta = 1$, we can find $\varepsilon > 0$ so that the decay estimates (5.52) satisfy. Now using Hölder's inequality, (5.52) and (5.53), we derive

$$\begin{aligned} \|u\|_{H^{m}(g\mathcal{F})} &\lesssim \sup_{g \in G_{\bar{\mathcal{F}},K}} \max_{|\alpha| \leq m} \left(\int_{g\mathcal{F}} \left| \int_{K} |\partial_{x}^{\alpha} G_{P}(x,y)| \cdot |f(y)| \, \mathrm{d}\mu_{\mathcal{X}}(y) \right|^{2} \mathrm{d}\mu_{\mathcal{X}}(x) \right)^{1/2} \\ &\lesssim \|f\|_{L^{2}(\mathcal{X})} \cdot \max_{|\alpha| \leq m} \left(\int_{g\mathcal{F}} \int_{K} |\partial_{x}^{\alpha} G_{P}(x,y)|^{2} \, \mathrm{d}\mu_{\mathcal{X}}(y) \, \mathrm{d}\mu_{\mathcal{X}}(x) \right)^{1/2} \\ &\lesssim \|f\|_{L^{2}(\mathcal{X})} \cdot \exp\left(-2C_{1}\varepsilon|g|\right) \max_{|\alpha| \leq m} \sup_{y \in K} \left(\int_{g\mathcal{F}} |\partial_{x}^{\alpha} G_{P}(x,y)|^{2} \exp\left(\varepsilon d_{\mathcal{X}}(x,y)\right) \, \mathrm{d}\mu_{\mathcal{X}}(x) \right)^{1/2} \\ &\lesssim \|f\|_{L^{2}(\mathcal{X})} \cdot \exp\left(-2C_{1}\varepsilon|g|\right) \lesssim \|f\|_{L^{2}(\mathcal{X})} \cdot \varphi_{1}(g) \cdot \exp\left(-C_{1}\varepsilon|g|\right). \end{aligned}$$

Note that the above estimates hold up to multiplicative constants which are uniform with respect to $g \in G_{\overline{F},K}$. Therefore, u belongs to $\mathcal{V}_{m,\varphi_1}^{p_1}(\mathcal{X})$, and this proves (5.51). In particular, the Fredholm indices of the operators $P_{m,\varphi_1}^{p_1}$ and $(P^*)_{m,\varphi_2}^{p_2}$ vanish. Now we are able to apply Theorem 5.3.10 to finish the proof of Lemma 5.5.8 in this case.

Case 2. $p_1^{-1} + p_2^{-1} \leq 1, \varphi_1^{-1}\varphi_2^{-1} \leq 1$. Consider a rigged divisor μ on \mathcal{X} . By assumptions, $L_{p'_2}(\mu, P, \varphi_2^{-1}) \subseteq L_{p_1}(\mu, P, \varphi_1)$ and $L_{p'_1}(\mu^{-1}, P^*, \varphi_1^{-1}) \subseteq L_{p_2}(\mu^{-1}, P^*, \varphi_2)$. From these inclusions and Case 1, we obtain

$$\dim L_{p_1}(\mu, P, \varphi_1) = \deg_P(\mu) + \dim L_{p'_1}(\mu^{-1}, P^*, \varphi_1^{-1}) \le \deg_P(\mu) + L_{p_2}(\mu^{-1}, P^*, \varphi_2)$$
$$= \dim L_{p'_2}(\mu, P, \varphi_2^{-1}) \le \dim L_{p_1}(\mu, P, \varphi_1).$$

Since all of the above inequalities must become equalities, this yields the corresponding Riemann-Roch formula in this case.

We use Lemma 5.5.8 to prove all of the statements now. First, one can get the identity in the second statement of Theorem 5.4.21 by taking $p_1 = p_2 = \infty$ and $\varphi_1 = \varphi_2 = \varphi_0$ in Lemma 5.5.8. Also, due to Theorem 5.4.18, there is no non-zero solution of P^* with subexponential growth. This implies that if $\mu^{-1} = (\emptyset, 0; D^+, L^+)$, the space $L_{\infty}(\mu^{-1}, P^*, \varphi_0)$ is trivial. Thus, the third statement follows immediately from the second statement. For the first statement, let us consider $p \in [1, \infty]$ and a function $\varphi \in S(G)$. Now from Lemma 5.5.8 and the second statement, one gets:

• If φ is bounded,

$$\dim L_1(\mu, P, \varphi) = \deg_P(\mu) + \dim L_{\infty}(\mu^{-1}, P^*, \varphi_0) = \dim L_{\infty}(\mu, P, \varphi_0).$$
(5.54)

• If φ^{-1} is bounded and $1 \le p \le \infty$,

$$\dim L_p(\mu, P, \varphi) = \deg_P(\mu) + \dim L_{\infty}(\mu^{-1}, P^*, \varphi_0) = \dim L_{\infty}(\mu, P, \varphi_0).$$
(5.55)

We consider three cases.

- Case 1. If φ is bounded, the two spaces $L_1(\mu, P, \varphi)$ and $L_{\infty}(\mu, P, \varphi_0)$ are the same since their dimensions are equal to each other by (5.54). Moreover, $L_1(\mu, P, \varphi) \subseteq$ $L_p(\mu, P, \varphi) \subseteq L_p(\mu, P, \varphi_0) \subseteq L_{\infty}(\mu, P, \varphi_0)$. This means all these spaces are the same.
- Case 2. If φ^{-1} is bounded, $L_{\infty}(\mu, P, \varphi_0) \subseteq L_{\infty}(\mu, P, \varphi)$. Using (5.55) with $p = \infty$, we have $L_{\infty}(\mu, P, \varphi_0) = L_{\infty}(\mu, P, \varphi)$. Moreover, (5.55) also yields that all the spaces $L_p(\mu, P, \varphi)$, where $1 \leq p \leq \infty$, must have the same dimension and therefore, they are the same space, which coincide $L_{\infty}(\mu, P, \varphi_0)$.
- Case 3. If neither φ nor φ^{-1} is bounded, we can consider the function $\phi := \varphi + \varphi^{-1}$. Clearly, ϕ is in S(G) and $\phi \ge 2$. Then according to Case 1 and Case 2,

$$L_p(\mu, P, \phi) = L_{\infty}(\mu, P, \varphi_0) = L_p(\mu, P, \phi^{-1}).$$

Also, $L_p(\mu, P, \phi^{-1}) \subseteq L_p(\mu, P, \varphi) \subseteq L_p(\mu, P, \phi)$ since $\phi^{-1} \leq \varphi \leq \phi$. This means that $L_p(\mu, P, \varphi) = L_{\infty}(\mu, P, \varphi_0)$.

Proof of Corollary 5.4.22

Proof. As in the proof of Corollary 5.4.14, the equivalence of the first three statements is an easy consequence of Theorem 5.4.21 and Remark 5.3.12. It is obvious that (iv) implies (iii). To see the converse, one can repeat the argument in the proof of Lemma 5.5.8 to show that the solution $u = P^{-1}f$ has exponential decay due to (5.52). By the unique solvability of the equation Pu = f in $L^2(\mathcal{X})$, (iii) implies (iv).

5.6 Applications of Liouville-Riemann-Roch theorems to specific operators

In this part, we will try to look at some examples where one can apply the results in Section 3.2.5.

Let us begin with self-adjoint operators. Suppose that A is a bounded from below and self-adjoint periodic elliptic operator of order m on an abelian co-compact covering X. We know that the symmetric operator A with the domain $C_c^{\infty}(X)$ is essentially self-adjoint in $L^2(X)$. Unless confusion can arise, we always use the same notation A for the unique self-adjoint extension ¹⁷ of A in $L^2(X)$.

To apply the results from Section 5.4, we will reformulate Assumption A in Section 5.4.

Assumption \mathcal{A}'

Suppose that $F_{A,\mathbb{R}} = \{k_1, \dots, k_\ell\}$ (modulo G^* -shifts). Let $\{\lambda_{r,j}\}_{j=\overline{1,m_r}}$ be the set of dispersion branches that equals to 0 at the quasimomentum k_r $(1 \le r \le \ell)$. There exists

¹⁷The domain of this extension is the Sobolev space $H^m(X)$.

a family of pairwise disjoint neighborhoods V_r of k_r such that the function $k \in V_r \mapsto \max_{1 \le j \le m_r} |\lambda_{r,j}(k)|^{-1}$ is L^1 -integrable.¹⁸

Clearly, Assumption A and Assumption A' are equivalent if A is self-adjoint.

Notation 5.6.1. Let N be a natural number.

• We denote by $h_{d,N}$ the dimension of harmonic polynomials of order at most N in *d*-variables, i.e.,

$$h_{d,N} := \dim V_N^{\infty}(-\Delta_{\mathbb{R}^d}) = \binom{d+N}{d} - \binom{d+N-2}{d}.$$

• We also denote by $c_{d,N}$ the dimension of the space of all homogeneous polynomials of degree N in d variables, i.e.,

$$c_{d,N} := \binom{d+N}{d} - \binom{d+N-1}{d} = \binom{d+N-1}{N}.$$

5.6.1 Periodic operators with nondegenerate spectral edges

Let λ be an energy level that coincides with one of the spectral edges of a gap in the spectrum of A. By shifting, we can suppose that $\lambda = 0$.

One can expect the Fermi surfaces at the spectral edges to be "normally" finite and hence, Liouville type results are applicable in these situations. We make the following assumption.

Assumption \mathcal{B} . There exists a band function $\lambda_j(k)$ such that for each quasimomentum k_r in the real Fermi surface $F_{A,\mathbb{R}}$, one has

(B1) 0 is a simple eigenvalue of the operator $A(k_r)$.

¹⁸Note that for each $k \in V_r \setminus \{k_r\}, \lambda_{r,j}(k) \neq 0$ since $V_r \cap F_{A,\mathbb{R}} = \{k_r\}$.

(B2) The Hessian matrix $\operatorname{Hess} \lambda_j(k_0)$ is non-degenerate.

For self-adjoint second-order periodic elliptic operators, it is commonly believed in mathematics and physics literature (see e.g, [48]) that generically (with respect to the coefficients and other free parameters of the operator), extrema of band functions are isolated, attained by a single band and have non-degenerate Hessians.

Suppose now that the free abelian rank d of the deck group G is greater than 2. The non-degeneracy assumption ($\mathcal{B}2$) implies the integrability of function $|\lambda_j(k)|^{-1}$ over a small neighborhood of $F_{A,\mathbb{R}}$. Hence, Assumption \mathcal{A} follows from Assumption \mathcal{B} .

Due to Theorem 5.2.10, the dimension of the space $V_N^{\infty}(A)$ is equal to $\ell h_{d,[N]}$ (see Notation 5.6.1). Applying the results in Section 5.4, we obtain the following results for a 'generic' self-adjoint periodic elliptic operator A of second-order:

Theorem 5.6.2. Suppose $d \ge 3$ and $N \in \mathbb{R}$. Let $\mu = (D^+, L^+; D^-, L^-)$ be a rigged divisor on X. Recall that $\mu^+ = (D^+, L^+; \emptyset, 0)$.

(a) If $N \ge 0$ then

$$\ell h_{d,[N]} + \deg_A(\mu) + \dim L_1(\mu^{-1}, A^*, -N) \le \dim L_\infty(\mu, A, N)$$

 $\le \ell h_{d,[N]} + \deg_A(\mu^+),$

and

$$\dim L_{\infty}(\mu^+, A, N) = \ell h_{d,[N]} + \deg_A(\mu^+).$$

(b) If $p \in [1, \infty)$ and N > d/p then

$$\ell h_{d,\lfloor N-d/p\rfloor} + \deg_A(\mu) + \dim L_{p'}(\mu^{-1}, A^*, -N) \le \dim L_p(\mu, A, N)$$
$$\le \ell h_{d,\lfloor N-d/p\rfloor} + \deg_A(\mu^+),$$

$$\dim L_p(\mu^+, A, N) = \ell h_{d, \lfloor N - d/p \rfloor} + \deg_A(\mu^+)$$

(c) For $d \ge 5$, the inequality

$$\deg_{A}(\mu) + \dim L_{p'}(\mu^{-1}, A^{*}, -N) \le \dim L_{p}(\mu, A, N) \le \deg_{A}(\mu^{+})$$

holds if we assume one of the following conditions:

- $p \in [1, 2), N \in (d(2-p)/(2p), d/p].$
- $p \in [2, \infty), N \in [0, d/p].$
- **Example 5.6.3.** 1. Let $A = A^* = -\Delta + V$ be a periodic Schrödinger operator on a cocompact abelian cover X with real-valued electric potential V. The domain of A is the Sobolev space $H^2(X)$, and thus, A is self-adjoint. Suppose that 0 is the bottom of its spectrum. It is well-known [43] that Assumption B holds in this situation. Hence, all the conclusions of Theorem 5.6.2 hold with $\ell = 1$ since $F_{A,\mathbb{R}} = \{0\}$ (modulo $2\pi\mathbb{Z}^d$ -shifts).
 - In [13, 14], a deep analysis of the dispersion curves at the bottom of the spectrum was developed for a wide class of periodic elliptic operators of second-order on R^d. Namely, let Γ be a lattice in R^d and Γ* be its dual lattice, then these operators admit the following *regular factorization*:

$$A = \overline{f(x)b(D)^*g(x)b(D)f(x)},$$

where $b(D) = \sum_{j=1}^{d} -i\partial_{x_j}b_j : L^2(\mathbb{R}^d, \mathbb{C}^n) \to L^2(\mathbb{R}^d, \mathbb{C}^m)$ is a linear homogeneous differential operator whose coefficients b_j are constant $m \times n$ -matrices of rank n

(here $m \ge n$), f is a Γ -periodic and invertible $n \times n$ matrix function such that fand f^{-1} are in $L^{\infty}(\mathbb{R}^d)$, and g is a Γ -periodic and positive definite $m \times m$ -matrix function such that for some constants $0 < c_0 \le c_1, c_0 \mathbf{1}_{m \times m} \le g(x) \le c_1 \mathbf{1}_{m \times m}, x \in$ \mathbb{R}^d . The existence of this factorization implies that the first band function attains a simple and nondegenerate minimum with value 0 at *only* the quasimomentum k = 0 (modulo Γ^* -shifts). This covers the previous example since it was noted from [13] that for a periodic metric g(x) and a periodic potential V(x), the periodic Schrödinger operator D(g(x)D) + V(x) admits a regular factorization.

3. Consider a self-adjoint periodic magnetic Schrödinger operator in \mathbb{R}^n (n > 2)

$$H = (-i\nabla + A(x))^2 + V(x),$$

where A(x) and V(x) are real-valued periodic magnetic and electric potentials, respectively. Using a gauge transformation, we can always assume the following normalization condition on A (without effecting our results):

$$\int_{\mathbb{T}^n} A(x) \, \mathrm{d}x = 0.$$

Note that the transpose of H is the magnetic Schrödinger operator

$$H^* = (-i\nabla - A(x))^2 + V(x).$$

From the discussion of [50] (see also [47, Theorem 3.1.7]), the lowest band function of H has a unique nondegenerate extremum at a single quasimomentum k_0 (modulo G^* -shifts) if the magnetic potential A is small enough, e.g., $||A||_{L^r(\mathbb{T}^n)} \ll 1$ or some r > n. Thus, we obtain the same conclusion as the case without magnetic potential. It is crucial that one has to assume the smallness of the magnetic potential since there are examples [69] showing that the bottom of the spectrum are degenerate if the magnetic potential is large enough.

We end this part by proving the following upper-semicontinuity property for families of periodic elliptic operators with non-degenerate spectral edges.

Corollary 5.6.4. Let A_0 be a periodic elliptic operator on a co-compact abelian covering X with the deck group $G = \mathbb{Z}^d$, where $d \ge 5$, such that its real Fermi surface $F_{A_0,\mathbb{R}}$ consists of finitely many simple and non-degenerate minima of the j^{th} -band function of A_0 $(j \ge 1)$. Let B be a symmetric and periodic differential operator on X such that B is A_0 -bounded. We consider the perturbation $A_z = A_0 + zB$, $z \in \mathbb{R}$. By standard perturbation theory, there exists a continuous function $\lambda(z, k)$ defined for small z and all quasimomenta k such that $k \mapsto \lambda(z, k)$ is the j^{th} band function of A_z and $\lambda_j(z, k)$ is analytic in z. Let λ_z be the minimum value of the band function $\lambda(z, k)$ of the perturbation A_z . Then for any rigged divisor μ , there exists $\varepsilon > 0$ such that

$$\dim L_2(\mu, A_z - \lambda_z, 0) \le \dim L_2(\mu, A_0, 0),$$

for any z satisfying $|z| < \varepsilon$.

The corresponding statement for non-degenerate maxima also holds.

Proof. In a similar manner to the proof of Proposition 5.4.14, for the rigged divisor μ , the corresponding extension operator $(A_z)_{m,0}^2$: $\Gamma(X, \mu, (A_z)_{m,0}^2) \rightarrow \tilde{\Gamma}_{\mu}(X, (A_z)_{m,0}^2)$ is Fredholm. As in the proof of [79, Theorem 2], we can deduce the upper-semicontinuity of dim Ker $(A_z)_{m,0}^2$ by using [79, Theorem 1] and [79, Theorem 3]. Since Ker $(A_z)_{m,0}^2 = L_2(\mu, A_z, 0)$, this finishes our proof.

5.6.2 Periodic operators with Dirac points

An important situation in solid state physics and material sciences is when two branches of the dispersion relation touch to form a conical junction point, which is called a Dirac point. Two-dimensional massless Dirac operators or 2D-Schrödinger operators with honeycomb symmetric potentials are typical models of periodic operators with conical structures [11, 23, 35, 51].

Let us consider a self-adjoint periodic elliptic operator A such that there are two branches λ_+ and λ_- of the dispersion relation of A that meet at 0 and form a Dirac cone. Equivalently, we can assume that locally around each quasimomentum k_r in the Fermi surface $F_{A,\mathbb{R}}$, for some $c_r \neq 0$, one has

$$\lambda_{+}(k) = c_{r}|k - k_{r}| \cdot (1 + O(|k - k_{r}|),$$
$$\lambda_{-}(k) = -c_{r}|k - k_{r}| \cdot (1 + O(|k - k_{r}|).$$

Then, the functions $|\lambda_+|^{-1}$ and $|\lambda_-|^{-1}$ are integrable over a small neighborhood of $F_{A,\mathbb{R}}$ provided that d > 1. Hence, we conclude:

Theorem 5.6.5. Suppose $d \ge 2$. Assume that the Fermi surface $F_{A,\mathbb{R}}$ consists of ℓ Dirac conical points. Let us recall the notation $c_{d,N}$ in Notation 5.6.1. Then as in Theorem 5.6.5, we have

(a) If $N \ge 0$ then

$$2\ell c_{d,[N]} + \deg_A(\mu) + \dim L_1(\mu^{-1}, A^*, -N) \le \dim L_\infty(\mu, A, N) \le 2\ell c_{d,[N]} + \deg_A(\mu^+),$$

and

$$\dim L_{\infty}(\mu^{+}, A, N) = 2\ell c_{d,[N]} + \deg_{A}(\mu^{+}).$$
(b) If $p \in [1, \infty)$ and N > d/p then

$$2\ell c_{d,\lfloor N-d/p\rfloor} + \deg_A(\mu) + \dim L_{p'}(\mu^{-1}, A^*, -N) \le \dim L_p(\mu, A, N)$$
$$\le 2\ell c_{d,\lfloor N-d/p\rfloor} + \deg_A(\mu^+),$$

and

$$\dim L_p(\mu^+, A, N) = 2\ell c_{d,\lfloor N-d/p \rfloor} + \deg_A(\mu^+).$$

(c) For $d \ge 3$ and a pair (p, N) satisfying the condition in Theorem 5.6.5 (c), the same conclusion of Theorem 5.6.5 (c) also holds.

Example 5.6.6. In this example, we consider Schrödinger operators with honeycomb lattice potentials in \mathbb{R}^2 . Let us recall briefly some notions from [23]. The triangular lattice $\Lambda_h = \mathbb{Z}\mathbf{v}_1 \oplus \mathbb{Z}\mathbf{v}_2$ is spanned by the basis vectors:

$$\mathbf{v}_1 = a\left(\frac{\sqrt{3}}{2}, \frac{1}{2}\right)^t, \quad \mathbf{v}_2 = a\left(\frac{\sqrt{3}}{2}, -\frac{1}{2}\right)^t (a > 0).$$

The dual lattice is $\Lambda_h^* = \mathbb{Z}\mathbf{k}_1 \oplus \mathbb{Z}\mathbf{k}_2$, where

$$\mathbf{k}_{1} = \frac{4\pi}{a\sqrt{3}} \left(\frac{1}{2}, \frac{\sqrt{3}}{2}\right)^{t}, \quad \mathbf{k}_{2} = \frac{4\pi}{a\sqrt{3}} \left(\frac{1}{2}, -\frac{\sqrt{3}}{2}\right)^{t}.$$

We define

$$\mathbf{K} = \frac{1}{3}(\mathbf{k}_1 - \mathbf{k}_2), \quad \mathbf{K}' = -\mathbf{K}.$$

The Brillouin zone \mathcal{B}_h , a fundamental domain of the quotient \mathbb{R}^2/Λ_h^* , can be chosen as a hexagon in \mathbb{R}^2 such that all six vertices of this hexagon fall into two groups:

- 1. K type-vertices: \mathbf{K} , $\mathbf{K} + \mathbf{k}_2$, $\mathbf{K} \mathbf{k}_1$.
- 2. **K'** type-vertices: **K'**, $\mathbf{K'} \mathbf{k}_2$, $\mathbf{K'} + \mathbf{k}_1$.

Note that these groups of vertices are invariant under the clockwise rotation \mathcal{R} by $2\pi/3$. Let potential $V \in C^{\infty}(\mathbb{R}^2)$ be real, Λ_h -periodic, and such that there exists a point $x_0 \in \mathbb{R}^2$ such that V is inversion symmetric (i.e., even) and \mathcal{R} -invariant with respect to x_0 (see [23, Remark 2.4] for some construction of such potentials). Now consider the Schrödinger operator $H^{\varepsilon} = -\Delta + \varepsilon V$ ($\varepsilon \in \mathbb{R}$). One of the main results of [23] is that except possibly for ε in a countable and closed set \tilde{C} , the dispersion relation of H^{ε} has conical singularities at each vertex of \mathcal{B}_h . Assume that λ_j^{ε} , $j \in \mathbb{N}$, are the band functions of the operator H^{ε} for each $\varepsilon \in \mathbb{R}$. Then according to [23, Theorem 5.1], when $\varepsilon \notin \tilde{C}$, there exists some $j \in \mathbb{N}$ such that the Fermi surface $F_{H^{\varepsilon},\lambda_j^{\varepsilon}(\mathbf{K})}$ of the operator H^{ε} at the level $\lambda_j^{\varepsilon}(\mathbf{K})$ contains (at least) two Dirac points located at the quasimomenta \mathbf{K} and \mathbf{K}' (modulo shifts by vectors in the dual lattice Λ_h^*). Now our next corollary is a direct consequence of [23, Theorem 5.1] and our previous discussion:

Corollary 5.6.7. Let μ be a rigged divisor on \mathbb{R}^2 and V be a honeycomb lattice potential such that the following Fourier coefficient $V_{1,1}$ of the potential V

$$V_{1,1} := \int_{\mathbb{R}^2/\Lambda_h} e^{-i(\mathbf{k}_1 + \mathbf{k}_2) \cdot x} V(x) \, \mathrm{d}x$$

is nonzero. Then for $\varepsilon \notin \tilde{C}$, there exists $j \in \mathbb{N}$ such that the following inequalities hold:

1.
$$p = \infty, N \ge 0$$
:

$$\dim L_{\infty}(\mu, H^{\varepsilon} - \lambda_{j}^{\varepsilon}(\mathbf{K}), N) \geq 4([N] + 1) + \deg_{H^{\varepsilon} - \lambda_{j}^{\varepsilon}(\mathbf{K})}(\mu) + \dim L_{1}(\mu^{-1}, H^{\varepsilon} - \lambda_{j}^{\varepsilon}(\mathbf{K}), -N),$$

$$\dim L_{\infty}(\mu^{+}, H^{\varepsilon} - \lambda_{j}^{\varepsilon}(\mathbf{K}), N) \geq 4([N] + 1) + \deg_{H^{\varepsilon} - \lambda_{j}^{\varepsilon}(\mathbf{K})}(\mu^{+}).$$

2.
$$1 \le p < \infty, N > 2/p$$
:

$$\dim L_p(\mu, H^{\varepsilon} - \lambda_j^{\varepsilon}(\mathbf{K}), N) \ge 4(\lfloor N - 2/p \rfloor + 1) + \deg_{H^{\varepsilon} - \lambda_j^{\varepsilon}(\mathbf{K})}(\mu) + \dim L_{p'}(\mu^{-1}, H^{\varepsilon} - \lambda_j^{\varepsilon}(\mathbf{K}), -N),$$

$$\dim L_p(\mu^+, H^{\varepsilon} - \lambda_j^{\varepsilon}(\mathbf{K}), N) \ge 4(\lfloor N - 2/p \rfloor + 1) + \deg_{H^{\varepsilon} - \lambda_j^{\varepsilon}(\mathbf{K})}(\mu^+).$$

Moreover, there is some $\varepsilon_0 > 0$ *so that for all* $\varepsilon \in (-\varepsilon_0, \varepsilon_0) \setminus \{0\}$ *, we have*

- $\varepsilon V_{1,1} > 0$: the above j can be chosen as j = 1.
- $\varepsilon V_{1,1} < 0$: the above j can be chosen as j = 2.

Remark 5.6.8. Other important results on the existence of Dirac points in the dispersion relations of Schrödinger operators with periodic potentials on honeycomb lattices are established for instance in [51] for quantum graph models of graphene and carbon nanotube materials and in [11] for many interesting models including both discrete, quantum graph, and continuous ones.

5.6.3 Non-self-adjoint second order elliptic operators

We now consider a class of possibly non-self-adjoint second-order elliptic operators arising from probability theory. Let A be a G-periodic linear elliptic operator of secondorder acting on functions u in $C^{\infty}(X)$ such that in any coordinate system $(U; x_1, \ldots, x_n)$, the operator A can be represented in the form

$$A = \sum_{1 \le i,j \le n} a_{ij}(x)\partial_{x_i}\partial_{x_j} + \sum_{1 \le j \le n} b_j(x)\partial_{x_j} + c(x),$$
(5.56)

where the coefficients a_{ij} , b_j , c are *real*, smooth, and periodic. We notice that the coefficient c(x) of zeroth-order of A is globally defined on X since it is the image of the constant function 1 via A.

Definition 5.6.9. [3, 56, 64]

(a) A function u on X is called G-multiplicative with exponent $\xi \in \mathbb{R}^d$ if it satisfies

$$u(g \cdot x) = e^{\xi \cdot g} u(x), \quad \forall x \in X, g \in G.$$

In other words, u is a Bloch function with quasimomentum $i\xi$.

(b) The generalized principal eigenvalue of A is defined by

$$\Lambda_0 = \sup\{\lambda \in \mathbb{R} \mid (A - \lambda)u = 0 \text{ for some positive solution } u\}.$$

The principal eigenvalue is a generalized version of the bottom of the spectrum in the self-adjoint case (see e.g., [2]).

For an operator A of the type (5.56) and any $\xi \in \mathbb{R}^d$, it is known from [3, 47, 56, 64] that there exists a unique real number $\Lambda(\xi)$ such that the equation $(A - \Lambda(\xi))u = 0$ has a *positive* G-multiplicative solution u. We list some known properties of this important function $\Lambda(\xi)$. The reader can find proofs in [3, 56, 64] (see also [50, Lemma 5.7]).

Proposition 5.6.10. (a) $\Lambda_0 = \max_{\xi \in \mathbb{R}^d} \Lambda(\xi).$

- (b) The function $\Lambda(\xi)$ is strictly concave, real analytic, bounded from above, and its gradient vanishes at only its maximum point. The Hessian of $\Lambda(\xi)$ is nondegenerate at all points.
- (c) $\Lambda(\xi)$ is the principal eigenvalue with multiplicity one of the operator $A(i\xi)$.
- (d) $\Lambda_0 \ge 0$ if and only if A admits a positive periodic (super-) solution, which is also equivalent to the existence of a positive G-multiplicative solution of the equation Au = 0.

(e) $\Lambda_0 = 0$ if and only if there is exactly one normalized positive solution u of the equation Au = 0.

We are interested in studying Liouville-Riemann-Roch type results for such operators A satisfying $\Lambda(0) \ge 0$, which implies that A has a positive solution.

- **Example 5.6.11.** 1. Divergence type operators with no zeroth-order coefficient satisfy $\Lambda_0 = \Lambda(0) = 0.$
 - 2. If the zeroth-order coefficient c(x) is nonnegative, Λ(0) is also nonnegative.¹⁹ Indeed, let u be a positive and periodic solution of the equation Au = Λ(0)u. If Λ(0) < 0, it follows from the equation that u is a positive and periodic subsolution of A on X. By the strong maximum principle, u must be constant. This means that 0 ≤ cu = Au < 0, which is a contradiction!</p>

Before stating the main result of this subsection, let us provide a key lemma.

Lemma 5.6.12. (a) If $\Lambda(0) > 0$ then $F_{A,\mathbb{R}} = \emptyset$.

- (b) If $\Lambda(0) = 0$ then $F_{A,\mathbb{R}} = \{0\}$ (modulo G^* -shifts). In this case, there exists an open strip V in \mathbb{C}^d containing the imaginary axis $i\mathbb{R}^d$ such that for any $k \in V$, there is exactly one (isolated and nondegenerate eigenvalue) point $\lambda(k)$ in $\sigma(A(k))$ that is close to 0. The dispersion function $\lambda(k)$ is analytic in V and $\lambda(ik) = \Lambda(k)$ if $k \in \mathbb{R}^d$. Moreover,
 - When Λ₀ > 0, k = 0 is a noncritical point of the dispersion λ(k) in V ∩ ℝ^d as well as of the function Λ in ℝ^d.
 - When $\Lambda_0 = 0$, k = 0 is a nondegenerate extremum of the dispersion $\lambda(k)$ in $V \cap \mathbb{R}^d$ as well as of the function Λ in \mathbb{R}^d .

¹⁹In general, the converse is not true: the zeroth-order coefficient of the transpose A^* is not necessarily nonnegative while the principal eigenvalue of the operator $A^*(0)$ is the same as $\Lambda(0)$.

Proof. The statements of this lemma are direct consequences of [50, Lemma 5.8], Kato-Rellich theorem (see e.g., [66, Theorem XII.8]), and Proposition 5.6.10.

Theorem 5.6.13. Let A be a periodic elliptic operator of second-order with real and smooth coefficients on X such that $\Lambda(0) \ge 0$. Let μ be a rigged divisor on X and μ^+ be the positive part of μ .

- (a) If $\Lambda(0) > 0$, dim $L_{\infty}(\mu^+, A, \varphi) = \deg_A(\mu^+)$ and dim $L_{\infty}(\mu, A, \varphi) = \deg_A(\mu) + \dim L_{\infty}(\mu^{-1}, A^*, \varphi^{-1})$ for any function $\varphi \in \mathcal{S}(G)$ (see Definition 5.4.20).
- (b) If $\Lambda_0 > \Lambda(0) = 0$ and $d \ge 2$, then we have:
 - For any $N \ge 0$, dim $L_{\infty}(\mu^+, A, N) = c_{d,[N]} + \deg_A(\mu^+)$ and

$$c_{d,[N]} + \deg_A(\mu) + \dim L_1(\mu^{-1}, A^*, -N) \le \dim L_\infty(\mu, A, N)$$

 $\le c_{d,[N]} + \deg_A(\mu^+)$

• For any $p \in [1, \infty)$, N > d/p, dim $L_p(\mu^+, A, N) = c_{d,\lfloor N - d/p \rfloor} + \deg_A(\mu^+)$,

$$c_{d,\lfloor N-d/p\rfloor} + \deg_A(\mu) + \dim L_{p'}(\mu^{-1}, A^*, -N) \le \dim L_p(\mu, A, N)$$
$$\le c_{d,\lfloor N-d/p\rfloor} + \deg_A(\mu^+).$$

• For $d \ge 3$ and a pair (p, N) satisfying the condition in Theorem 5.6.5 (c),

$$\deg_{A}(\mu) + \dim L_{p'}(\mu^{-1}, A^{*}, -N) \le \dim L_{p}(\mu, A, N) \le \deg_{A}(\mu^{+}).$$

(c) If $\Lambda_0 = \Lambda(0) = 0$ and $d \ge 3$, then we have:

• For any $N \ge 0$, dim $L_{\infty}(\mu^+, A, N) = h_{d,[N]} + \deg_A(\mu^+)$,

$$h_{d,[N]} + \deg_A(\mu) + \dim L_1(\mu^{-1}, A^*, -N) \le \dim L_\infty(\mu, A, N)$$

 $\le h_{d,[N]} + \deg_A(\mu^+)$

• For any $p \in [1, \infty)$, N > d/p, $\dim L_p(\mu^+, A, N) = h_{d, \lfloor N - d/p \rfloor} + \deg_A(\mu^+)$,

$$h_{d,\lfloor N-d/p\rfloor} + \deg_A(\mu) + \dim L_{p'}(\mu^{-1}, A^*, -N) \le \dim L_p(\mu, A, N)$$
$$\le h_{d,\lfloor N-d/p\rfloor} + \deg_A(\mu^+).$$

• For $d \ge 5$ and a pair (p, N) satisfying the condition in Theorem 5.6.5 (c),

$$\deg_A(\mu) + \dim L_{p'}(\mu^{-1}, A^*, -N) \le \dim L_p(\mu, A, N) \le \deg_A(\mu^+).$$

Proof. To compute the dimensions of the spaces $V_N^p(A)$, we use Lemma 5.6.12 to apply Theorem 5.2.10 and Theorem 5.2.14. Then Theorem 5.6.13 would follow immediately from Theorem 5.4.21, Lemma 5.6.12, Theorem 5.4.6, Proposition 5.4.10, and Theorem 5.4.12.

5.7 Some auxiliary statements and proofs of technical lemmas

5.7.1 Some properties of Floquet functions on abelian coverings

We recall briefly another construction of Floquet functions on X. As we discussed in Subsection 5.2.2, it suffices to define the Bloch function $e_k(x)$ with quasimomentum k and the powers $[x]^j$ on X, where $j \in \mathbb{Z}_+^d$. To do this, we first recall the notion of additive functions on abelian coverings (Definition 4.2.7). We fix an additive function h. Write $h = (h_1, \ldots, h_d)$. Let $j = (j_1, \ldots, j_d) \in \mathbb{Z}_+^d$ be a multi-index. We define

$$[x]^{j} := h(x)^{j} = \prod_{m=1}^{d} h_{m}(x)^{j_{m}},$$

and

$$e_k(x) := \exp\left(ik \cdot h(x)\right).$$

Clearly, $e_k(g \cdot x) = e^{ik \cdot g} e_k(x)$. Then a Floquet function u of order N with quasimomentum k is of the form

$$u(x) = e_k(x) \sum_{|j| \le N} p_j(x) [x]^j,$$

where p_j is smooth and periodic. Observe that the notion of Floquet functions is independent of the choice of h. Namely, u is also a Floquet function with the same order and quasimomentum with respect to another additive function \tilde{h} . Indeed, the difference $w := h - \tilde{h}$ between two additive functions h and \tilde{h} is periodic. Hence, one can rewrite

$$u(x) = e^{ik \cdot \tilde{h}(x)} \sum_{|j| \le N} e^{ik \cdot w(x)} p_j(x) \prod_{m=1}^d (\tilde{h}_m(x) + w_m(x))^{j_m}$$

$$= e^{ik \cdot \tilde{h}(x)} \sum_{|j| \le N} e^{ik \cdot w(x)} p_j(x) \sum_{j' \le j} {j \choose j'} w(x)^{j-j'} \tilde{h}(x)^{j'}$$

$$= e^{ik \cdot \tilde{h}(x)} \sum_{|j| \le N} \tilde{p}_j(x) \tilde{h}(x)^j,$$

(5.57)

where $\tilde{p}_j(x) := \sum_{j \le j'} {j' \choose j} e^{ik \cdot w(x)} p_{j'}(x) w(x)^{j'-j}$ is periodic. The following simple lemma is needed later.

Lemma 5.7.1. Let K be a compact neighborhood in X. Then for any multi-index $j \in \mathbb{Z}_+^d$,

there exists a constant C such that for any $x \in K$ and $g \in \mathbb{Z}^d$, one has

$$\left| [g \cdot x]^j - g^j \right| \le C \langle g \rangle^{|j|-1}$$

Proof.

$$\left| [g \cdot x]^{j} - g^{j} \right| = \left| \prod_{m=1}^{d} (h_{m}(x) + g_{m})^{j_{m}} - \prod_{m=1}^{d} g_{m}^{j_{m}} \right| \le C \langle g \rangle^{|j|-1},$$

for some C > 0 depending on $||h||_{L^{\infty}(K)}$.

5.7.2 Some basic facts about the family $\{A(k)\}_{k \in \mathbb{C}^d}$

In this part, let us always identify elements in $L^2(M)$ with their corresponding periodic functions in $L^2_{loc}(X)$ and vice versa. We discuss briefly how the two definitions of the fiber operators A(k) in this chapter and Chapter 4 (see Definition 4.2.14) are indeed equivalent. Note that the corresponding definition in Chapter 4 requires the introduction of additive functions on abelian coverings. However, each of these definitions has its own advantage.

Consider an additive function h on the abelian covering X (see Definition 4.2.7). Let \mathcal{U}_k be the mapping that multiplies a function f(x) in $L_k^2(X)$ by $e^{-ik \cdot h(x)}$. Thus, \mathcal{U}_k is an invertible bounded linear operator in $\mathcal{L}(L_k^2(X), L^2(M))$ and its inverse is given by the multiplication operator by $e^{ik \cdot h(x)}$. Note that the operator norms of \mathcal{U}_k and \mathcal{U}_k^{-1} are bounded by $e^{|\Im k| \cdot ||h||_{L^{\infty}(\overline{\mathcal{F}})}}$, where \mathcal{F} is the fundamental domain appearing in the definition of the inner product of the Hilbert space $L_k^2(X)$ in Definition 5.2.3.

Define the following elliptic operator $\widehat{A}(k) := \mathcal{U}_k A(k) \mathcal{U}_k^{-1}$. The operator $\widehat{A}(k)$, with the Sobolev space $H^m(M)$ as the domain, is a closed and unbounded operator in $L^2(M)$. For each complex quasimomentum k, the two linear operators $\widehat{A}(k)$ and A(k) are similar and thus, their spectra are identical. In terms of spectral information, it is no need to distinguish A(k) and its equivalent model $\widehat{A}(k)$. One of the benefits of working with the later model is that $\widehat{A}(k)$ acts on the k-independent domain of periodic functions on X. The following proposition gives a simple sufficient condition on the principal symbol of the operator A so that the spectra of A(k) are discrete. More general criteria on the discreteness of spectra can be found in the book [4].

Proposition 5.7.2. If A has real principal symbol, then for each $k \in \mathbb{C}^d$, A(k), as an unbounded operator on $L^2(E_k)$, has discrete spectrum, i.e., its spectrum consists of isolated eigenvalues with finite (algebraic) multiplicities.

Proof. Let B be the real part of the operator A. Since A has real principal symbol, the principal symbols of A and B are the same. By pushing down to operators on M, the differential operator $\widehat{A}(0) - \widehat{B}(0)$ is of lower order. Also, the principal symbols of the operators $\widehat{A}(k)$ and $\widehat{A}(0)$ are identical. Thus, we see that $\widehat{A}(k)$ is a perturbation of the self-adjoint elliptic operator $\widehat{B}(0)$ by a lower order differential operator on the compact manifold M. It follows from [4] that the spectrum of $\widehat{A}(k)$ must be discrete. This finishes the proof.

Now if one assumes that $\sigma(A(k))$ is discrete, then the family of operators $\{\widehat{A}(k)\}_{k\in\mathbb{C}^d}$, with compact resolvents, is analytic of type (A) in the sense of Kato [39]. Therefore, this family satisfies the upper-semicontinuity of the spectrum (see e.g., [39, 66]). We provide this statement here without a proof.

Proposition 5.7.3. Consider $k_0 \in \mathbb{C}^d$. If Γ is a compact subset of the complex plane such that $\Gamma \cap \sigma(A(k_0)) = \emptyset$, then there exists $\delta > 0$ depending on Γ and k_0 such that $\Gamma \cap \sigma(A(k)) = \emptyset$, for any k in the ball $B_{k_0}(\delta)$ centered at k_0 with radius δ .

Remark 5.7.4. (i) The Hilbert bundle \mathcal{E}^m becomes the trivial bundle $\mathbb{C}^d \times H^m(M)$ via the holomorphic bundle isomorphism defined from the linear maps \mathcal{U}_k , where $k \in \mathbb{C}^d$.

(ii) In general, one can use the analytic Fredholm theorem to see that the essential spectrum²⁰ of A(k) is empty for any k ∈ C^d, but this is not enough to conclude that these spectra are discrete in the non-self-adjoint case. For example, if we consider the Z-periodic elliptic operator A = e^{2iπx}D_x on R, a simple argument shows that

$$\sigma(A(k)) = \begin{cases} \mathbb{C}, & \text{if } k \in 2\pi\mathbb{Z} \\ \emptyset, & \text{otherwise.} \end{cases}$$

A similar example for the higher-dimensional case \mathbb{R}^d , d > 1 can be cooked up easily from the above example.

5.7.3 Some properties of Floquet transforms on abelian coverings

We now state some useful properties of the Floquet transform \mathbf{F} on abelian coverings. First, due to (5.4), one can see that the Floquet transform $\mathbf{F}f(k, x)$ of a nice function f, e.g., $f \in C_c^{\infty}(X)$, is periodic in the quasimomentum variable k and moreover, it is a quasiperiodic function in the x-variable, i.e.,

$$\mathbf{F}f(k, g \cdot x) = \gamma_k(g) \cdot \mathbf{F}f(k, x) = e^{ik \cdot g} \cdot \mathcal{F}f(k, x), \text{ for any } (g, x) \in G \times X.$$

It follows that $\mathbf{F}f(k, \cdot)$ belongs to $H_k^s(X)$, for any k and s. Therefore, it is enough to regard the Floquet transform of f as a smooth section of the Hilbert bundle \mathcal{E}^s over $(\mathbb{T}^*)^d$ (identified with the Brillouin zone B).

Let $K \Subset X$ be a domain such that $\bigcup_{g \in G} gK = X$. Then given any real number s, we denote by $\mathcal{C}^s(X)$ the Frechet space consists of all functions $\phi \in H^s_{loc}(X)$ such that for any $N \ge 0$, one has

$$\sup_{g \in G} \|\phi\|_{H^s(gK)} \cdot \langle g \rangle^N < \infty.$$

²⁰Here we use the definition of the essential spectrum of an operator T as the set of all $\lambda \in \mathbb{C}$ such that $T - \lambda$ is not Fredholm.

In terms of Definition 5.2.6, $\bigcap_{N\geq 0} V_N^{\infty}(X) = C^0(X) \cap C^{\infty}(X)$. The following theorem contains a Paley-Wiener type result for the Floquet transform. The details can be found in [47–50].

- **Theorem 5.7.5.** (a) The Floquet transform \mathbf{F} is an isometric isomorphism between the Sobolev space $H^s(X)$ and the space $L^2((\mathbb{T}^*)^d, \mathcal{E}^s)$ of L^2 -integrable sections of the vector bundle \mathcal{E}^s .
- (b) The Floquet transform \mathbf{F} expands the periodic elliptic operator A of order m in $L^2(X)$ into a direct integral of the fiber operators A(k) over $(\mathbb{T}^*)^d$.

$$\boldsymbol{F} A \boldsymbol{F}^{-1} = \int_{(\mathbb{T}^*)^d}^{\oplus} A(k) \, \mathrm{d}k$$

Equivalently, $\mathbf{F}(Af)(k) = A(k)\mathbf{F}f(k)$ for any $f \in H^m(X)$.

(c) The Floquet transform

$$\mathbf{F}: \mathcal{C}^s(X) \to C^\infty((\mathbb{T}^*)^d, \mathcal{E}^s)$$

is a topological isomorphism, where $C^{\infty}((\mathbb{T}^*)^d, \mathcal{E}^s)$ is the space of smooth sections of the vector bundle \mathcal{E}^s . Furthermore, under the Floquet transform \mathbf{F} , the operator $A : \mathcal{C}^m(X) \to \mathcal{C}^0(X)$ becomes the corresponding morphism of sheaves of smooth sections arising from the holomorphic Fredholm morphism A(k) between the two holomorphic Hilbert bundles \mathcal{E}^m and \mathcal{E}^0 over the torus $(\mathbb{T}^*)^d$, i.e., it is an operator from $C^{\infty}((\mathbb{T}^*)^d, \mathcal{E}^m)$ to $C^{\infty}((\mathbb{T}^*)^d, \mathcal{E}^0)$ such that it acts on the fiber of \mathcal{E}^m at k as the fiber operator $A(k) : H^m_k(X) \to L^2_k(X)$. (d) The inversion \mathbf{F}^{-1} of the Floquet transform is given by the formula

$$f(x) = \frac{1}{(2\pi)^d} \int_{(\mathbb{T}^*)^d} \mathbf{F} f(k, x) \,\mathrm{d}k,$$
(5.58)

provided that one can make sense both sides of (5.58) (as functions or distributions).

We prove a simple analog of the Riemann-Lebesgue lemma for the Floquet transform.

Lemma 5.7.6. (a) Let $\widehat{f}(k, x)$ be a function in $L^1((\mathbb{T}^*)^d, \mathcal{E}^0)$. Then the inverse Floquet transform $f := \mathbf{F}^{-1}\widehat{f}$ belongs to $L^2_{loc}(X)$ and

$$\sup_{g\in G} \|f\|_{L^2(g\mathcal{F})} < \infty.$$

Here \mathcal{F} is a fixed fundamental domain. Moreover, one also has

$$\lim_{|g| \to \infty} \|f\|_{L^2(g\mathcal{F})} = 0.$$

(b) If $f \in V_0^1(X)$ then $\mathbf{F}f(k, x) \in C((\mathbb{T}^*)^d, \mathcal{E}^0)$.

Proof. We recall that \mathcal{E}^0 can be considered as $L^2(\mathcal{F})$. To prove the first statement, we use the identity (5.58) and the Minkowski's inequality to obtain

$$\begin{split} \|f\|_{L^{2}(g\mathcal{F})} &= \frac{1}{(2\pi)^{d}} \left\| \int_{(\mathbb{T}^{*})^{d}} \mathbf{F}f(k,\cdot) \,\mathrm{d}k \right\|_{L^{2}(g\mathcal{F})} = \frac{1}{(2\pi)^{d}} \left\| \int_{(\mathbb{T}^{*})^{d}} e^{ik \cdot g} \mathbf{F}f(k,\cdot) \,\mathrm{d}k \right\|_{L^{2}(\mathcal{F})} \\ &\leq \frac{1}{(2\pi)^{d}} \int_{(\mathbb{T}^{*})^{d}} \|\mathbf{F}f(k,\cdot)\|_{L^{2}(\mathcal{F})} \,\mathrm{d}k = \frac{1}{(2\pi)^{d}} \|\mathbf{F}f(k,x)\|_{L^{1}((\mathbb{T}^{*})^{d},\mathcal{E}^{0})} < \infty. \end{split}$$

For the fact $\lim_{|g|\to\infty} ||f||_{L^2(g\mathcal{F})} = 0$, one can modify easily the usual proof of the Riemann-Lebesgue lemma, i.e., by using Theorem 5.7.5 (a) and then approximating $\mathbf{F}f$ by a sequence of functions in $L^2((\mathbb{T}^*)^d, \mathcal{E}^0)$.

The second statement follows directly from (5.4) and triangle inequalities.

5.7.4 A Schauder type estimate

For convenience, we state a well-known Schauder type estimate for solutions of a periodic elliptic equation Au = 0, which we needed to refer to several times in this text. We also sketch its proof for the sake of self-containedness.

Proposition 5.7.7. Let K be a compact neighborhood of a non-empty domain in the covering X. Then there exists a compact neighborhood \hat{K} in X such that $K \Subset \hat{K}$ and the following statement holds: Suppose that \mathcal{O} is an open subset of X such that $\hat{K} \subset \mathcal{O}$. Define $G^{\mathcal{O}} := \{g \in G \mid g\hat{K} \subset \mathcal{O}\}$. Then for any $\alpha \in \mathbb{R}^+$, there exists C > 0 depending on α, K, \hat{K} such that

$$\|u\|_{H^{\alpha}(gK)} \le C \cdot \|u\|_{L^{2}(q\hat{K})},\tag{5.59}$$

for any $g \in G^{\mathcal{O}}$ and any $u \in C^{\infty}(\mathcal{O})$ satisfying the equation Au = 0 on \mathcal{O} .

Proof. Let *B* be an almost local ²¹ pseudodifferential parametrix of *A* such that *B* commutes with actions of the deck group *G* (see e.g., [47, Lemma 2.1.1] or [71, Proposition 3.4]). Hence, BA = 1 + T for some almost-local and periodic smoothing operator *T* on *X*. This implies that for some compact neighborhood \hat{K} (depending on the support of the Schwartz kernel of *T* and the subset *K*) and for any $\alpha \ge 0$, one can find some C > 0 so that for any smooth function v on a neighborhood of \hat{K} , one gets

$$||Tv||_{H^{\alpha}(K)} \le C \cdot ||v||_{L^{2}(\hat{K})}$$

In particular, for any $g \in G^{\mathcal{O}}$ and $u \in C^{\infty}(\mathcal{O})$,

$$||Tu^g||_{H^{\alpha}(K)} \le C \cdot ||u^g||_{L^2(\hat{K})},$$
(5.60)

²¹I.e., for some $\varepsilon > 0$, the support of the Schwartz kernel of *B* is contained in an ε -neighborhood of the diagonal of $X \times X$.

where u^g is the g-shift of the function u on \mathcal{O} . Since T is G-periodic, from (5.60), we obtain

$$||Tu||_{H^{\alpha}(gK)} \le C \cdot ||u||_{L^{2}(g\hat{K})}.$$
(5.61)

The important point here is the uniformity of the constant C with respect to $g \in G^{\mathcal{O}}$.

Assuming that Au = 0 on \mathcal{O} . Thus, u = BAu - Tu = -Tu on \mathcal{O} . This identity and (5.61) imply the following estimate

$$||u||_{H^{\alpha}(gK)} = ||Tu||_{H^{\alpha}(gK)} \le C \cdot ||u||_{L^{2}(g\hat{K})}, \quad \forall g \in G^{\mathcal{O}}.$$

_	 _
	٦

Remark 5.7.8. We would like to emphasize that Proposition 5.7.7 holds in a more general context. Namely, it is true for any C^{∞} -bounded uniformly elliptic operator P on a co-compact Riemannian covering \mathcal{X} with a discrete deck group G. In this setting, P is invertible modulo a uniform smoothing operator T on \mathcal{X} (see [71, Definition 3.1] and [71, Proposition 3.4]). Now the estimate (5.61) follows easily from the uniformly boundedness of the derivatives of any order of the Schwartz kernel of T on canonical coordinate charts and a routine argument of partition of unity. Another approach is to invoke uniform local apriori estimates [71, Lemma 1.4].

5.7.5 A variant of Dedekind's lemma

It is a well-known theorem by Dedekind (see e.g., [58, Lemma 2.2]) that distinct unitary characters of an abelian group G are linearly independent as functions from G to a field \mathbb{F} . The next lemma is a refinement of Dedekind's lemma when $\mathbb{F} = \mathbb{C}$. We notice that a proof by induction method can be found in [65, Lemma 4.4]. For the sake of completeness, we will provide our analytic proof using Stone-Weierstrass's theorem. **Lemma 5.7.9.** Consider a finite number of distinct unitary characters $\gamma_1, \ldots, \gamma_\ell$ of the abelian group \mathbb{Z}^d . Then there are vectors g_1, \ldots, g_ℓ in \mathbb{Z}^d and C > 0 such that for any $v = (v_1, \ldots, v_\ell) \in \mathbb{C}^\ell$, we have

$$\max_{1 \le s \le \ell} \left| \sum_{r=1}^{\ell} v_r \cdot \gamma_r(g_s) \right| \ge C \cdot \max_{1 \le r \le \ell} |v_r|.$$

Proof. By abuse of notation, we can regard $\gamma_1, \ldots, \gamma_\ell$ as distinct points of the torus $(\mathbb{T}^*)^d$.

For each tuple (g_1, \ldots, g_ℓ) in $(\mathbb{Z}^d)^\ell$, let $W(g_1, \ldots, g_\ell)$ be a $\ell \times \ell$ -matrix whose (s, r)entry $W(g_1, \ldots, g_\ell)_{s,r}$ is $\gamma_s^{g_r}$, for any $1 \leq r, s \leq \ell$. We equip \mathbb{C}^ℓ with the maximum norm. Then the conclusion of the lemma is equivalent to the invertibility of some operator $W(g_1, \ldots, g_\ell)$ acting from \mathbb{C}^ℓ to \mathbb{C}^ℓ .

Suppose for contradiction that the determinant function det $W(g_1, \ldots, g_\ell)$ is zero on $(\mathbb{Z}^d)^\ell$, i.e., for any $g_1, \ldots, g_\ell \in \mathbb{Z}^d$, one has

$$0 = \det W(g_1, \dots, g_\ell) = \sum_{\sigma \in S_\ell} \operatorname{sign}\left(\sigma\right) \cdot \left(\gamma_{\sigma(1)}^{g_1} \dots \gamma_{\sigma(\ell)}^{g_\ell}\right),$$

where S_{ℓ} is the permutation group on $\{1, \ldots, \ell\}$. Thus, the above relation also holds for any trigonometric polynomial $P(\gamma_1, \ldots, \gamma_{\ell})$ on $(\mathbb{T}^*)^{d\ell}$, i.e.,

$$\sum_{\sigma \in S_{\ell}} \operatorname{sign} \left(\sigma \right) \cdot P(\gamma_{\sigma(1)}, \dots, \gamma_{\sigma(\ell)}) = 0.$$

By using the fact that the trigonometric polynomials are dense in $C((\mathbb{T}^*)^{d\ell})$ in the uniform topology (Stone-Weierstrass theorem), we conclude that

$$\sum_{\sigma \in S_{\ell}} \operatorname{sign}\left(\sigma\right) \cdot f(\gamma_{\sigma(1)}, \dots, \gamma_{\sigma(\ell)}) = 0,$$
(5.62)

for any continuous function f on $(\mathbb{T}^*)^{d\ell}$.

Now for each $1 \leq r \leq \ell$, let us select some smooth cutoff functions ω_r supported on a neighborhood of the point γ_r such that $\omega_r(\gamma_s) = 0$ whenever $s \neq r$. We define $f \in C((\mathbb{T}^*)^{d\ell})$ as follows

$$f(x_1,\ldots,x_\ell) := \prod_{r=1}^\ell \omega_r(x_r), \quad x_1,\ldots,x_\ell \in (\mathbb{T}^*)^d.$$

Hence, $f(\gamma_{\sigma(1)}, \ldots, \gamma_{\sigma(\ell)})$ is non-zero if and only if σ is the trivial permutation. By substituting f into (5.62), we get a contradiction, which proves our lemma.

5.7.6 **Proofs of technical statements**

In this part, we will use the notation \mathcal{F} for the closure of a fundamental domain for G-action on the covering X.

Proof of Theorem 5.2.14.

Proof. If $u \in V_{\lfloor N-d/p \rfloor}^{\infty}(A)$ then $u \in V_{N_0}^{\infty}(A)$ for some nonnegative integer N_0 such that $N_0 < N - d/p$. Thus,

$$\sum_{g \in G} \|u\|_{L^2(g\mathcal{F})}^p \langle g \rangle^{-pN} \lesssim \sum_{g \in G} \langle g \rangle^{p(N_0 - N)} < \infty.$$

Thus, $V^{\infty}_{\lfloor N-d/p \rfloor}(A) \subseteq V^p_N(A)$.

Now suppose that $F_{A,\mathbb{R}} = \{k_1, \ldots, k_\ell\}$ (modulo G^* -shifts), where $\ell \in \mathbb{N}$. It suffices for us to show that

$$V_N^p(A) \subseteq V_{|N-d/p|}^\infty(A).$$
(5.63)

A key ingredient of the proof of (5.63) is the following statement.

Lemma 5.7.10. Suppose that $\mathcal{N} > d/p$.

- (i) If $u \in V^p_{\mathcal{N}}(A) \cap V^{\infty}_{\mathcal{M}}(A)$ for some $0 \leq \mathcal{M} < \mathcal{N} + 1 d/p$ then $u \in V^{\infty}_{\mathcal{M}'}(A)$ for some $\mathcal{M}' < \mathcal{N} d/p$. In particular, $u \in V^{\infty}_{\mathcal{N} d/p}(A)$.
- (ii) $V^p_{\mathcal{N}}(A) \cap V^{\infty}_{\mathcal{N}+1-d/p}(A) \subseteq V^{\infty}_{\mathcal{N}-d/p}(A).$

Instead of proving Lemma 5.7.10 immediately, first, let us assume its validity. We now prove (5.63). Consider $u \in V_N^p(A)$.

Case 1. p > 1.

We prove by induction that if $0 \le s \le d - 1$, then

$$u \in V_{N+d/p-(s+1)/p}^{p}(A) \cap V_{N-s/p}^{\infty}(A).$$
(5.64)

The statement holds for s = 0 since clearly, $V_N^p(A) \subseteq V_N^{\infty}(A)$ and $N + d/p - 1/p \ge N$. Now suppose that (5.64) holds for s such that $s + 1 \le d - 1$. Since 1 - 1/p > 0, we can apply Lemma 5.7.10 (i) to u and the pair $(\mathcal{N}, \mathcal{M}) = (N + d/p - (s + 1)/p, N - s/p)$ to deduce that $u \in V_{N-(s+1)/p}^{\infty}(A)$. Therefore, (5.64) also holds for s + 1. In the end, we have $u \in V_N^p(A) \cap V_{N-(d-1)/p}^{\infty}(A)$. Again, by Lemma 5.7.10 (i), u belongs to $V_{\mathcal{M}'}^{\infty}(A)$ for some $\mathcal{M}' < N - d/p$. In other words, u is in $V_{\lfloor N - d/p \rfloor}^{\infty}(A)$.

As in Case 1, we apply Lemma 5.7.10 (ii) and induction method to prove that $u \in V_{N+d-1-s}^1(A) \cap V_{N-s}^{\infty}(A)$ for any $0 \le s \le d-1$. Hence, $u \in V_N^1(A) \cap V_{N+1-d}^{\infty}(A)$. Due to Lemma 5.7.10 (ii) again, one has $u \in V_N^1(A) \cap V_{N-d}^{\infty}(A)$. Applying Lemma 5.7.10 (i) to u and the pair $(\mathcal{N}, \mathcal{M}) = (N, N-d)$, we conclude that $u \in V_{\lfloor N-d \rfloor}^{\infty}(A) = V_{N-(d+1)}^{\infty}(A)$.

Therefore, Theorem 5.2.14 follows from Lemma 5.7.10. Our proof of Lemma 5.7.10 consists of several steps.

(i) Step 1. The lemma is trivial if u = 0. So we assume that u is non-zero. Since $V_{\mathcal{N}}^{p}(A) \subseteq V_{\mathcal{N}}^{\infty}(A)$, we can apply Theorem 5.2.8 (ii) to represent $u \in V_{\mathcal{N}}^{p}(A)$ as a

Case 2. p = 1.

finite sum of Floquet solutions of A, i.e.,

$$u = \sum_{r=1}^{\ell} u_r.$$

Here u_r is a Floquet function of order $M_r \leq \mathcal{N}$ with quasimomentum k_r . Let N_0 be the highest order among all the orders of the Floquet functions u_r appearing in the above representation. Without loss of generality, we can suppose that there exists $r_0 \in [1, \ell]$ such that for any $r \leq r_0$, the orders M_r of u_r is maximal among all the orders of these Floquet functions. Thus, $M_r = N_0 \leq \mathcal{M}$ when $r \in [1, r_0]$. To prove our lemma, it suffices to show that $N_0 < \mathcal{N} - d/p$.

Step 2. According to Proposition 5.7.7, we can pick a compact neighborhood $\hat{\mathcal{F}}$ of \mathcal{F} such that for any $\alpha \geq 0$,

$$\|u\|_{H^{\alpha}(g\mathcal{F})} \le C \cdot \|u\|_{L^{2}(g\hat{\mathcal{F}})}$$

for some C > 0 which is independent of $g \in G$. Let $\alpha > n/2$, the Sobolev embedding theorem yields the estimate

$$\|u\|_{C^0(g\mathcal{F})} \lesssim \|u\|_{L^2(g\hat{\mathcal{F}})}, \quad \forall g \in G.$$

$$(5.65)$$

From (5.65) and the fact that $u \in V^p_{\mathcal{N}}(A)$, we obtain

$$\sup_{x \in \mathcal{F}} \left(\sum_{g \in G} |u(g \cdot x)|^p \langle g \rangle^{-p\mathcal{N}} \right) \lesssim \sum_{g \in G} ||u||_{L^2(g\hat{\mathcal{F}})}^p \langle g \rangle^{-p\mathcal{N}} < \infty.$$
(5.66)

Step 3. By definition (see Subsection 5.2.2), one can write

$$u(x) = \sum_{r=1}^{\ell} u_r(x) = \sum_{r=1}^{r_0} e_{k_r}(x) \sum_{|j|=N_0} a_{j,r}(x) [x]^j + O(|x|^{N_0-1}).$$

Here each function $a_{j,r}$ is *G*-periodic and the remainder term $O(|x|^{N_0-1})$ is an exponential polynomial with periodic coefficients of order at most $N_0 - 1$. Hence, for any $(g, x) \in G \times \mathcal{F}$, we get

$$u(g \cdot x) = \sum_{r=1}^{r_0} e^{ik_r \cdot g} \sum_{|j|=N_0} e_{k_r}(x) a_{j,r}(x) [g \cdot x]^j + O(\langle g \rangle^{N_0 - 1}).$$

Since $N_0 - 1 \leq \mathcal{M} - 1 < \mathcal{N} - d/p$, the series

$$\sum_{g \in \mathbb{Z}^d} O(\langle g \rangle^{p(N_0 - 1)}) \langle g \rangle^{-p\mathcal{N}}$$

converges. From this and (5.66), we deduce

$$\sup_{x\in\mathcal{F}}\sum_{g\in G}\left|\sum_{r=1}^{r_0}e^{ik_r\cdot g}\sum_{|j|=N_0}e_{k_r}(x)a_{j,r}(x)[g\cdot x]^j\right|^p\langle g\rangle^{-p\mathcal{N}}<\infty.$$

By Lemma 5.7.1, $|[g \cdot x]^j - g^j| = O(\langle g \rangle^{N_0-1})$ for any multi-index j such that $|j| = N_0$. This implies that

$$\sup_{x\in\mathcal{F}}\sum_{g\in G}\left|\sum_{|j|=N_0}\sum_{r=1}^{r_0}e_{k_r}(x)a_{j,r}(x)e^{ik_r\cdot g}g^j\right|^p\langle g\rangle^{-p\mathcal{N}}<\infty.$$
(5.67)

Step 4. We will use Lemma 5.7.9 to reduce the condition (5.67) to the one without exponential terms $e^{ik_r \cdot g}$, i.e., we can assume that $F_{A,\mathbb{R}} = \{0\}$ (modulo G^* -shifts).

Indeed, let $\gamma_1, \ldots, \gamma_{r_0}$ be distinct unitary characters of G, which are defined via the

identities $\gamma_r(g) = e^{ik_r \cdot g}$, where $r \in \{1, \ldots, r_0\}$ and $g \in G$. Now due to Lemma 5.7.9, there are $g_1, \ldots, g_{r_0} \in G$ and a constant C > 0 such that for any vector $(v_1, \ldots, v_{r_0}) \in \mathbb{C}^{r_0}$, we have the following inequality

$$C \cdot \max_{1 \le s \le r_0} \left| \sum_{r=1}^{r_0} v_r \cdot e^{ik_r \cdot g_s} \right| \ge \max_{1 \le r \le r_0} |v_r|.$$
(5.68)

Hence, for any $(g, x) \in G \times \mathcal{F}$ and $1 \leq s \leq r_0$, we apply (5.68) to the vector

$$(v_1,\ldots,v_{r_0}) := \left(\sum_{|j|=N_0} e_{k_r}(x)a_{j,r}(x)(g+g_s)^j \langle g+g_s \rangle^{-\mathcal{N}} e^{ik_r \cdot g}\right)_{1 \le r \le r_0}$$

to deduce that

$$\max_{1 \le r \le r_0} \left| \sum_{|j|=N_0} e_{k_r}(x) a_{j,r}(x) (g+g_s)^j \right|^p \langle g+g_s \rangle^{-pN} \\
= \max_{1 \le r \le r_0} \left| \sum_{|j|=N_0} e_{k_r}(x) a_{j,r}(x) (g+g_s)^j \langle g+g_s \rangle^{-N} e^{ik_r \cdot g} \right|^p \\
\lesssim \max_{1 \le s \le r_0} \left| \sum_{r=1}^{r_0} \left(\sum_{|j|=N_0} e_{k_r}(x) a_{j,r}(x) (g+g_s)^j \langle g+g_s \rangle^{-N} e^{ik_r \cdot g} \right) e^{ik_r \cdot g_s} \right|^p \quad (5.69) \\
\lesssim \sum_{s=1}^{r_0} \left| \sum_{r=1}^{r_0} \sum_{|j|=N_0} e_{k_r}(x) a_{j,r}(x) (g+g_s)^j \cdot e^{ik_r \cdot (g+g_s)} \right|^p \cdot \langle g+g_s \rangle^{-pN}$$

Summing the estimate (5.69) over $g \in G$, we derive

$$\max_{1 \leq r \leq r_0} \sup_{x \in \mathcal{F}} \sum_{g \in G} \left| \sum_{|j|=N_0} e_{k_r}(x) a_{j,r}(x) g^j \right|^p \langle g \rangle^{-p\mathcal{N}}$$

$$= \max_{1 \leq r,s \leq r_0} \sup_{x \in \mathcal{F}} \sum_{g \in G} \left| \sum_{|j|=N_0} e_{k_r}(x) a_{j,r}(x) (g+g_s)^j \right|^p \langle g+g_s \rangle^{-p\mathcal{N}}$$

$$\lesssim \sup_{x \in \mathcal{F}} \sum_{s=1}^{r_0} \sum_{g \in G} \left| \sum_{r=1}^{r_0} \sum_{|j|=N_0} e_{k_r}(x) a_{j,r}(x) (g+g_s)^j \cdot e^{ik_r \cdot (g+g_s)} \right|^p \cdot \langle g+g_s \rangle^{-p\mathcal{N}}$$

$$\lesssim \sup_{x \in \mathcal{F}} \sum_{g \in G} \left| \sum_{r=1}^{r_0} \sum_{|j|=N_0} e_{k_r}(x) a_{j,r}(x) g^j \cdot e^{ik_r \cdot g} \right|^p \cdot \langle g \rangle^{-p\mathcal{N}} < \infty.$$
(5.70)

From (5.67) and (5.70), we get

$$\sum_{g \in G} \left| \sum_{|j|=N_0} e_{k_r}(x) a_{j,r}(x) g^j \right|^p \cdot \langle g \rangle^{-p\mathcal{N}} < \infty,$$
(5.71)

for any $1 \leq r \leq r_0$ and $x \in \mathcal{F}$.

Step 5. We prove the following claim: If P is a non-zero homogeneous polynomial of degree N_0 in d-variables such that $N_0 < N + 1 - d/p$ and

$$\sum_{g \in \mathbb{Z}^d} |P(g)|^p \cdot \langle g \rangle^{-p\mathcal{N}} < \infty,$$
(5.72)

then $N_0 < \mathcal{N} - d/p$.

Our idea is to approximate the series in (5.72) by the integral

$$\mathcal{I} := \int_{\mathbb{R}^d} |P(z)|^p \cdot \langle z \rangle^{-p\mathcal{N}} \, \mathrm{d}z.$$

In fact, for any $z \in [0,1)^d + g$, one can use the triangle inequality and the assumption

that the order of P is N_0 to achieve the following estimate

$$|P(z)|^{p} \leq 2^{p-1} \left(|P(g)|^{p} + |P(z) - P(g)|^{p} \right) \lesssim |P(g)|^{p} + O(\langle g \rangle^{p(N_{0}-1)}).$$

Integrating the above estimate over the cube $[0,1)^d + g$ and then summing over all $g \in \mathbb{Z}^d$, we deduce

$$\mathcal{I} = \sum_{g \in \mathbb{Z}^d} \int_{[0,1)^d + g} |P(z)|^p \cdot \langle z \rangle^{-p\mathcal{N}} \, \mathrm{d}z \lesssim \sum_{g \in \mathbb{Z}^d} |P(g)|^p \cdot \langle g \rangle^{-p\mathcal{N}} + \sum_{g \in \mathbb{Z}^d} \langle g \rangle^{p(N_0 - 1 - \mathcal{N})} < \infty,$$

where we have used (5.72) and the condition $(N_0 - 1 - N)p < -d$.

We now rewrite our integral \mathcal{I} in polar coordinates as follows

$$\mathcal{I} = \int_{0}^{\infty} \int_{\mathbb{S}^{d-1}} |P(r\omega)|^p \langle r \rangle^{-p\mathcal{N}} r^{d-1} \,\mathrm{d}\omega \,\mathrm{d}r = \int_{0}^{\infty} \langle r \rangle^{-p\mathcal{N}} r^{d-1+pN_0} \,\mathrm{d}r \cdot \int_{\mathbb{S}^{d-1}} |P(\omega)|^p \,\mathrm{d}\omega.$$

Suppose for contradiction that $(N_0 - \mathcal{N})p \ge -d$. Then it follows that

$$\int_{0}^{\infty} \langle r \rangle^{-p\mathcal{N}} r^{d-1+pN_0} \, \mathrm{d}r = \infty.$$

Thus, the finiteness of \mathcal{I} implies that

$$\int_{\mathbb{S}^{d-1}} |P(\omega)|^p \,\mathrm{d}\omega = 0.$$

Hence, $P(\omega) = 0$ for any $\omega \in \mathbb{S}^{d-1}$. By homogeneity, P must be zero (contradiction). This shows our claim.

Step 6. Since u is non-zero, there are some $r \in \{1, ..., r_0\}$ and $x \in \mathcal{F}$ such that the

following homogeneous polynomial of degree N_0 in \mathbb{R}^d

$$P(z) := \sum_{|j|=N_0} e_{k_r}(x) a_{j,r}(x) z^j$$

is non-zero. Thanks to (5.71) and the condition $N_0 \leq \mathcal{M} < \mathcal{N} + 1 - d/p$ (see Step 1), the inequality $N_0 < \mathcal{N} - d/p$ must be satisfied according to Step 5. This finishes the proof of the first part of the lemma.

(ii) Consider $u \in V_{\mathcal{N}}^{p}(A) \cap V_{\mathcal{N}+1-d/p}^{\infty}(A)$. In particular, for any $\varepsilon > 0$, $u \in V_{\mathcal{N}+\varepsilon}^{p}(A) \cap V_{\mathcal{N}+1-d/p}^{\infty}(A)$. Using Lemma 5.7.10 (i), u is in the space $V_{\mathcal{M}_{\varepsilon}}^{\infty}(A)$ for some $\mathcal{M}_{\varepsilon} < (\mathcal{N}+\varepsilon) - d/p$. By letting $\varepsilon \to 0^{+}$, we conclude that $u \in V_{\mathcal{N}-d/p}^{\infty}(A)$. This yields the second part of the lemma.

Proof of Theorem 5.2.16.

Proof. (a) Consider $u \in V_N^p(A)$. Due to Theorem 5.2.14 and the condition that $N \leq d/p$,

$$V_N^p(A) \subseteq V_{d/p+1/2}^p(A) = V_0^\infty(A).$$

Using Theorem 5.2.8 (ii),

$$u(x) = \sum_{r=1}^{\ell} e_{k_r}(x) a_r(x),$$
(5.73)

for some periodic functions $a_r(x)$.

Using (5.65) and the assumption that $u \in V_N^p(A)$, we derive

$$\sup_{x \in \mathcal{F}} \sum_{g \in G} \left| \sum_{r=1}^{\ell} e_{k_r}(x) a_r(x) e^{ik_r \cdot g} \right|^p \cdot \langle g \rangle^{-pN} < \infty.$$
(5.74)

Then one can modify the argument in Step 4 of the proof of Lemma 5.7.10 (from the estimate (5.67) to (5.71)) to get

$$\max_{1 \le r \le \ell} \sup_{x \in \mathcal{F}} |e_{k_r}(x)a_r(x)|^p \cdot \sum_{g \in \mathbb{Z}^d} \langle g \rangle^{-pN} < \infty.$$

Hence, the assumption $-pN \ge -d$ forces that $\max_{1 \le r \le \ell} \sup_{x \in \mathcal{F}} |e_{k_r}(x)a_r(x)| = 0$. Thus, u must be zero.

(b) Let u be an arbitrary element in V[∞]_N(A). Since N < 0, we can assume that u has the form (5.73). To prove that u = 0, it is enough to show that e_{k_r}(x)a_r(x) = 0 for any x ∈ F and 1 ≤ r ≤ ℓ. One can repeat the same argument of the previous part to prove this claim. However, we will provide a different proof using Fourier analysis on the torus (T^{*})^d.

For each $x \in \mathcal{F}$, we introduce the following distribution on $(\mathbb{T}^*)^d$

$$f(k) := \sum_{r=1}^{\ell} e_{k_r}(x) a_r(x) \delta(k - k_r),$$

where $\delta(\cdot - k_r)$ is the Dirac delta distribution on the torus $(\mathbb{T}^*)^d$ at the quasimomentum k_r . Taking Fourier series, we obtain

$$\hat{f}(g) = \sum_{r=1}^{\ell} e_{k_r}(x) a_r(x) e^{-ik_r \cdot g}.$$

As in (5.74), the assumption $u \in V_N^{\infty}(A)$ is equivalent to

$$\sup_{g\in\mathbb{Z}^d} \left| \hat{f}(g) \right| \cdot \langle g \rangle^{-N} < \infty.$$

Let ϕ be a smooth function on $(\mathbb{T}^*)^d$. Using Parseval's identity and Hölder's inequality,

we have

$$\left|\sum_{r=1}^{\ell} e_{k_r}(x) a_r(x) \phi(k_r)\right| = \left|\langle f, \phi \rangle\right| = \left|\sum_{g \in \mathbb{Z}^d} \hat{f}(g) \hat{\phi}(-g)\right| \lesssim \sum_{g \in \mathbb{Z}^d} |\hat{\phi}(g)| \cdot \langle g \rangle^N.$$
(5.75)

We pick $\delta > 0$ small enough such that $k_s \notin B(k_r, 2\delta)$ if $s \neq r$. Then we choose a cut-off function ϕ_r such that $\operatorname{supp} \phi_r \subseteq B(k_r, 2\delta)$ and $\phi_r = 1$ on $B(k_r, \delta)$. For $1 \leq r \leq \ell$, we define functions in $C^{\infty}((\mathbb{T}^*)^d)$ as follows:

$$\phi_r^{\varepsilon}(k) := \phi_r(\varepsilon^{-1}k), \quad (0 < \varepsilon < 1).$$

To bound the Fourier coefficients of ϕ_r^{ε} in terms of ε , we use integration by parts. Indeed, for any nonnegative real number *s*,

$$(2\pi)^{d} |\widehat{\phi}_{r}^{\varepsilon}(g)| = \left| \int_{(\mathbb{T}^{*})^{d}} \phi_{r}^{\varepsilon}(k) e^{-ik \cdot g} dk \right| = \varepsilon^{d} \cdot \left| \int_{B(k_{r}, 2\delta)} \phi_{r}(k) e^{-i\varepsilon k \cdot g} dk \right|$$
$$= \varepsilon^{d} \langle \varepsilon g \rangle^{-2[s]-2} \cdot \left| \int_{B(k_{r}, 2\delta)} (1 - \Delta)^{[s]+1} \phi_{r}(k) \cdot e^{-ik \cdot \varepsilon g} dk \right|$$
$$\leq \varepsilon^{d} \langle \varepsilon g \rangle^{-2[s]-2} \cdot \sup_{k} \left| (1 - \Delta)^{[s]+1} \phi_{r}(k) \right| \lesssim \varepsilon^{d-s} \langle g \rangle^{-s}.$$

In the last inequality, we make use of the fact that $\langle \varepsilon g \rangle^{-2[s]-2} \leq \langle \varepsilon g \rangle^{-s} \leq \varepsilon^{-s} \langle g \rangle^{-s}$ whenever $\varepsilon \in (0, 1)$. In particular, by choosing any $s \in (\max(0, N + d), d)$, one has

$$|\widehat{\phi}_r^{\varepsilon}(g)| \cdot \langle g \rangle^N \lesssim \varepsilon^{d-s} \cdot \langle g \rangle^{N-s}.$$
(5.76)

Now we substitute $\phi := \phi_r^{\varepsilon}$ in (5.75), use (5.76), and then take $\varepsilon \to 0^+$ to derive

$$|e_{k_r}(x)a_r(x)| \lesssim \lim_{\varepsilon \to 0} \varepsilon^{d-s} \sum_{g \in \mathbb{Z}^d} \langle g \rangle^{N-s} = 0,$$

which completes the proof.

Proof of Corollary 5.3.13.

Proof. It suffices to prove that $\operatorname{Im} \tilde{P} = \tilde{\Gamma}_{\mu}(\mathcal{X}, P)$ since from (5.29), dim Ker $\tilde{P}^* = \operatorname{codim} \operatorname{Im} \tilde{P} = 0$ and then the conclusion of Corollary 5.3.13 holds true as we remarked in Section 5.3. Now given any $f \in \tilde{\Gamma}_{\mu}(\mathcal{X}, P)$, one has $\langle f, \tilde{L}^- \rangle = 0$ and f = Pu for some $u \in \operatorname{Dom} P$ since $\operatorname{Im} P = \operatorname{Dom}' P^*$. According to the assumption, we can find a solution w = u - v in $\operatorname{Dom} P$ of the equation Pw = Pu - Pv = f such that $\langle w, L^- \rangle = 0$. Let w_0 be the restriction of w on $\mathcal{X} \setminus D^+$. Clearly, w_0 belongs to the space $\Gamma(\mathcal{X}, \mu, P)$. Since Pw is smooth on $\mathcal{X}, \tilde{P}w_0 = Pw = f$ by the definition of the extension operator \tilde{P} . This shows that $f \in \operatorname{Im} \tilde{P}$, which finishes the proof. \Box

Remark 5.7.11. In the special case $D^- = \emptyset$, $L^- = \{0\}$, one can prove the Riemann-Roch equality (5.24) directly, i.e., without referring to the extension operators \tilde{P} and $\tilde{P^*}$. For reader's convenience, let us present this short proof following from [32]. We define the space $\Gamma(\mu, P) := \{u \in \mathcal{D}'(\mathcal{X}) \mid u \in \text{Dom } P \setminus D^+, Pu \in L^+\}$. Then it is easy to check that the following sequences are exact:

$$0 \to \tilde{L}^+ \xrightarrow{i} \Gamma(\mu, P) \xrightarrow{r} L(\mu, P) \to 0$$

$$0 \to \operatorname{Ker} P \xrightarrow{i} \Gamma(\mu, P) \xrightarrow{P} L^+ \to 0,$$

where *i* and *r* are natural inclusion and restriction maps. Here the surjectivity of *P* from $\Gamma(\mu, P)$ to L^+ is a consequence of the existence of a properly supported pseudodifferential parametrix of *P* (modulo a properly supported smoothing operator) and $C_c^{\infty}(\mathcal{X}) \subseteq$ Dom' $P^* = \text{Im } P$. Note that Ker $P^* = \{0\}$. Hence, it follows that dim $L(\mu, P) =$ dim $\Gamma(\mu, P) - \dim \tilde{L}^+ = \dim \text{Ker } P + \dim L^+ - \dim \tilde{L}^+ = \operatorname{ind} P + \deg_P(\mu)$.

5.8 Concluding remarks

- Let {A_z}_z and {μ_z}_z be given families of periodic elliptic operators A_z satisfying the assumptions of Theorem 5.4.12 and of rigged divisors μ_z. Under appropriate conditions on these families depending "nicely" on the parameter z, the uppersemicontinuity in z of the two functions dim L₂(μ_z, A_z, 0) and dim L₂(μ_z⁻¹, A_z*, 0) can be deduced from the Liouville-Riemann-Roch equality (5.37).
- From Remark 5.5.3, we know that Assumption (A2) cannot be dropped in Theorem 5.4.6. Besides the example in Remark 5.5.3, we give a heuristic explanation here. It is known that if {A_t} is a family of Fredholm operators that is continuous with respect to the parameter t, the kernel dimension dim Ker A_t is upper semicontinuous. The idea in combining Riemann-Roch and Liouville theorems by considering dimensions of spaces of solutions with polynomial growth as some Fredholm indices would imply that in this direction, the upper-semicontinuity property should hold for these dimensions. On the other hand, in [69], there exists a continuous family {M_t} of periodic operators on ℝ² such that for each N ≥ 0, dim V_N[∞](M_t) = 2 dim V_N[∞](M_{2√3}) if 2√3 < t < 2√3 + ε for some ε > 0 and thus, dim V_N(M_t) is not upper-semicontinuous at t = 2√3 (see [50]). In this example, the minimum 0 of the lowest band λ₁(k) of the operator M_{2√3} is degenerate [69]. This explains why our approach requires the "non-degeneracy" type condition (A2) for avoiding some intractable cases like the previous example. Notice that in general, Assumption A and Liouville type results are not stable under small perturbations.
- The results in this chapter extend to linear elliptic systems and operators between vector bundles.

6. SUMMARY AND CONCLUSIONS

In Chapter 3, we obtained the Green's function asymptotics of periodic elliptic operators of second-order on Euclidean spaces. Namely, we computed explicitly the leading term of the Green's function $G_{\lambda}(x, y)$ of a "generic" periodic elliptic operator of secondorder when λ is close enough to a gap edge located at a high symmetry point of the Brillouin zone. The case when λ is below the bottom of the spectrum was known before [9,60]. However, the techniques used there could not apply directly to our situation here. We took an approach based on [52] and the calculation in the discrete version [81].

In Chapter 4, we generalized the results of Chapter 3 and [52] to the setting of abelian coverings. One of the main tools here is the use of additive functions on abelian coverings (Definition 4.2.7) for describing such asymptotics. These asymptotics capture a well-known idea (due to Gromov): a co-compact regular cover and its deck transformation group look similar at large scales.

In Chapter 5, we studied some possible combinations between the Riemann-Roch and the Liouville type results for periodic elliptic operators on noncompact abelian coverings with compact bases. Our idea is to employ a very general version of the Riemann-Roch formula for elliptic operators on noncompact manifolds developed in [32, 33]. Riemann-Roch type theorems allow the solutions to have prescribed zeros or poles at a given divisor. On the other hand, Liouville type results address the finite-dimensionality and structure of the space of solutions with a prescribed polynomial growth on noncompact manifolds. It is thus natural to try to combine the Riemann-Roch formula and the Liouville type results. In this direction, we were able to achieve two different groups of results:

 Liouville-Riemann-Roch formulas for uniformly bounded elliptic operators on cocompact Riemannian coverings outside of the spectra. (2) Liouville-Riemann-Roch inequalities and their applications for periodic elliptic operators on co-compact abelian Riemannian coverings at the edges of the spectra.

We have also obtained "an L^p -analog" of the Liouville type results of [49, 50].

REFERENCES

- S. Agmon, On kernels, eigenvalues, and eigenfunctions of operators related to elliptic problems, Comm. Pure Appl. Math. 18 (1965), 627–663.
- [2] _____, On positivity and decay of solutions of second order elliptic equations on Riemannian manifolds, Liguori, Naples, 1983.
- [3] _____, On positive solutions of elliptic equations with periodic coefficients in Rⁿ, spectral results and extensions to elliptic operators on Riemannian manifolds, Differential equations (Birmingham, Ala., 1983) 92 (1984), 7–17.
- [4] _____, Lectures on elliptic boundary value problems, AMS Chelsea Publishing, Providence, RI, 2010.
- [5] A. Ancona, Some results and examples about the behavior of harmonic functions and Green's functions with respect to second order elliptic operators, Nagoya Math. J. 165 (2002), 123–158.
- [6] N. W. Ashcroft and N. D. Mermin, *Solid state physics*, Holt, Rinehart and Winston, New York-London, 1976.
- [7] M. F. Atiyah, *Elliptic operators, discrete groups and von Neumann algebras*, Colloque "Analyse et Topologie" en l'Honneur de Henri Cartan (Orsay, 1974) (1976), 43–72. Astérisque, No. 32–33.
- [8] M. Avellaneda and F.-H. Lin, Un théorème de Liouville pour des équations elliptiques à coefficients périodiques, C. R. Acad. Sci. Paris Sér. I Math. 309 (1989), no. 5, 245–250.

- [9] M. Babillot, *Théorie du renouvellement pour des chaînes semi-markoviennes tran*sientes, Ann. Inst. H. Poincaré Probab. Statist. 24 (1988), no. 4, 507–569.
- [10] J. M. Barbaroux, J. M. Combes, and P. D. Hislop, *Localization near band edges for random schrödinger operators*, Helv. Phys. Acta **70** (1997), no. 1-2, 16–43. Papers honouring the 60th birthday of Klaus Hepp and of Walter Hunziker, Part II (Zürich, 1995).
- [11] G. Berkolaiko and A. Comech, Symmetry and dirac points in graphene spectrum, To appear in Journal of Spectral Theory, available at https://arxiv.org/abs/ 1412.8096.
- [12] G. Berkolaiko and P. Kuchment, *Introduction to quantum graphs*, AMS, Providence, RI, 2013.
- [13] M. Birman and T. Suslina, *Threshold effects near the lower edge of the spectrum for periodic differential operators of mathematical physics*, Systems, approximation, singular integral operators, and related topics (Bordeaux, 2000), Oper. Theory Adv. Appl. **129** (2001), 71–107.
- [14] _____, Periodic second-order differential operators. Threshold properties and averaging, Algebra i Analiz 15 (2003), no. 5, 1–108.
- [15] J. Brüning and T. Sunada, On the spectrum of gauge-periodic elliptic operators, Astérisque 210 (1992), 6, 65–74. Méthodes semi-classiques, Vol. 2 (Nantes, 1991).
- [16] _____, On the spectrum of periodic elliptic operators, Nagoya Math. J. **126** (1992), 159–171.

- [17] I. Chavel, *Riemannian geometry*, Second Edition, Cambridge Studies in Advanced Mathematics, vol. 98, Cambridge University Press, Cambridge, 2006. A modern introduction.
- [18] J. M. Combes and L. Thomas, Asymptotic behaviour of eigenfunctions for multiparticle schrödinger operators, Comm. Math. Phys. 34 (1973), 251–270.
- [19] H. L. Cycon, R. G. Froese, and B. Simon, *Schrödinger operators with application to quantum mechanics and global geometry*, Study, Texts and Monographs in Physics, Springer-Verlag, Berlin, 1987.
- [20] M. S. P. Eastham, *The spectral theory of periodic differential equations*, Texts in Mathematics (Edinburgh), Scottish Academic Press, Edinburgh; Hafner Press, New York, 1973.
- [21] J. Eichhorn, *Global analysis on open manifolds*, Nova Science Publishers, Inc., New York, 2007.
- [22] L. C. Evans, *Partial differential equations*, Graduate Studies in Mathematics, vol. 19, American Mathematical Society, Providence, RI, 1998.
- [23] C. L. Fefferman and M. I. Weinstein, *Honeycomb lattice potentials and Dirac points*, J. Amer. Math. Soc. 25 (2012), no. 4, 1169–1220.
- [24] J. Feldman, H. Knörrer, and E. Trubowitz, Asymmetric Fermi surfaces for magnetic Schrödinger operators, Comm. Partial Differential Equations 25 (2000), no. 1-2, 319–336.
- [25] N. Filonov and I. Kachkovskiy, On the structure of band edges of 2d periodic elliptic operators. arXiv:1510.04367, preprint.

- [26] I. M. Gel'fand, Expansion in characteristic functions of an equation with periodic coefficients, Doklady Akad. Nauk SSSR (N.S.) 73 (1950), 1117–1120.
- [27] M. Ghomi, Gauss map, topology, and convexity of hypersurfaces with nonvanishing curvature, Topology 41 (2002), no. 1, 107–117.
- [28] D. Gilbarg and N. S. Trudinger, *Elliptic partial differential equations of second order*, Second, Grundlehren der Mathematischen Wissenschaften, vol. 224, Springer-Verlag, Berlin, 1983.
- [29] I. M. Glazman, Direct methods of qualitative spectral analysis of singular differential operators, Translated from the Russian by the IPST staff, Israel Program for Scientific Translations, Jerusalem, 1965; Daniel Davey & Co., Inc., New York, 1966. MR0190800
- [30] R. I. Grigorchuk, Degrees of growth of finitely generated groups and the theory of invariant means, Izv. Akad. Nauk SSSR Ser. Mat. 48 (1984), no. 5, 939–985.
- [31] M. Gromov, *Hyperbolic groups*, Math. Sci. Res. Inst. Publ., vol. 8, Springer, New York, 1987.
- [32] M. Gromov and M. A. Shubin, *The Riemann-Roch theorem for elliptic operators* 16 (1993), 211–241.
- [33] _____, The Riemann-Roch theorem for elliptic operators and solvability of elliptic equations with additional conditions on compact subsets, Invent. Math. 117 (1994), no. 1, 165–180.
- [34] M. Gromov, *Groups of polynomial growth and expanding maps*, Inst. Hautes Études Sci. Publ. Math. **53** (1981), 53–73.

- [35] V. V. Grushin, Application of the multiparameter theory of perturbations of Fredholm operators to Bloch functions, Mat. Zametki 86 (2009), no. 6, 819–828.
- [36] V. Guillemin, Notes on elliptic operators, Lecture Notes, MIT (Spring 2005). http: //math.mit.edu/~vwg/classnotes-spring05.pdf.
- [37] J. M. Harrison, P. Kuchment, A. Sobolev, and B. Winn, On occurrence of spectral edges for periodic operators inside the Brillouin zone, J. Phys. A 40 (2007), no. 27, 7597–7618.
- [38] L. Hörmander, *The analysis of linear partial differential operators*. *I*, Grundlehren der Mathematischen Wissenschaften, vol. 256, Springer-Verlag, Berlin, 1983. Distribution theory and Fourier analysis.
- [39] T. Kato, *Perturbation theory for linear operators*, Second, Springer-Verlag, Berlin-New York, 1976. Grundlehren der Mathematischen Wissenschaften, Band 132.
- [40] M. Kha, P. Kuchment, and A. Raich, Green's function asymptotics near the internal edges of spectra of periodic elliptic operators. Spectral gap interior, To appear in Journal of Spectral Theory, available at https://arxiv.org/abs/1508. 06703.
- [41] M. Kha, Green's function asymptotics of periodic elliptic operators on abelian coverings of compact manifolds, available at https://arxiv.org/abs/1511.
 00276.
- [42] _____, A short note on additive functions on co-compact riemannian normal coverings., available at https://arxiv.org/abs/1511.00185.
- [43] W. Kirsch and B. Simon, *Comparison theorems for the gap of Schrödinger operators*,J. Funct. Anal. **75** (1987), no. 2, 396–410.

- [44] F. Klopp and J. Ralston, *Endpoints of the spectrum of periodic operators are generically simple*, Methods Appl. Anal. 7 (2000), no. 3, 459–463.
- [45] T. Kobayashi, K. Ono, and T. Sunada, *Periodic Schrödinger operators on a manifold*, Forum Math. 1 (1989), no. 1, 69–79.
- [46] M. Kotani and T. Sunada, Albanese maps and off diagonal long time asymptotics for the heat kernel, Comm. Math. Phys. 209 (2000), no. 3, 633–670.
- [47] P. Kuchment, *Floquet theory for partial differential equations*, Operator Theory: Advances and Applications, vol. 60, Birkhäuser Verlag, Basel, 1993.
- [48] _____, *An overview of periodic elliptic operators*, Bulletin (New Series) of the American Mathematical Society **53** (July 2016), no. 3, 343–414.
- [49] P. Kuchment and Y. Pinchover, Integral representations and Liouville theorems for solutions of periodic elliptic equations, J. Funct. Anal. 181 (2001), no. 2, 402–446.
- [50] _____, *Liouville theorems and spectral edge behavior on abelian coverings of compact manifolds*, Trans. Amer. Math. Soc. **359** (2007), no. 12, 5777–5815.
- [51] P. Kuchment and O. Post, *On the spectra of carbon nano-structures*, Comm. Math. Phys. 275 (2007), no. 3, 805–826.
- [52] P. Kuchment and A. Raich, Green's function asymptotics near the internal edges of spectra of periodic elliptic operators. Spectral edge case, Math. Nachr. 285 (2012), no. 14-15, 1880–1894.
- [53] O. A. Ladyzhenskaya and N. N. Ural'tseva, *Linear and quasilinear elliptic equations*, Translated from the Russian by Scripta Technica, Inc. Translation editor: Leon Ehrenpreis, Academic Press, New York-London, 1968.
- [54] H. B. Lawson Jr. and M.-L. Michelsohn, *Spin geometry*, Princeton Mathematical Series, vol. 38, Princeton University Press, Princeton, NJ, 1989.
- [55] J. M. Lee, Introduction to smooth manifolds, Second Edition, Graduate Texts in Mathematics, vol. 218, Springer, New York, 2013.
- [56] V. Ya. Lin and Y. Pinchover, *Manifolds with group actions and elliptic operators*, Mem. Amer. Math. Soc. **112** (1994), no. 540, vi+78.
- [57] W. Lück, Survey on geometric group theory, Münster J. Math. 1 (2008), 73–108.
- [58] P. Morandi, *Field and Galois theory*, Graduate Texts in Mathematics, vol. 167, Springer-Verlag, New York, 1996.
- [59] J. Moser and M. Struwe, On a Liouville-type theorem for linear and nonlinear elliptic differential equations on a torus, Bol. Soc. Brasil. Mat. (N.S.) 23 (1992), no. 1-2, 1–20.
- [60] M. Murata and T. Tsuchida, Asymptotics of Green functions and Martin boundaries for elliptic operators with periodic coefficients, J. Differential Equations 195 (2003), no. 1, 82–118.
- [61] _____, Asymptotics of Green functions and the limiting absorption principle for elliptic operators with periodic coefficients, J. Math. Kyoto Univ. 46 (2006), no. 4, 713–754.
- [62] P. W. Nowak and G. Yu, *Large scale geometry*, EMS Textbooks in Mathematics, European Mathematical Society (EMS), Zürich, 2012.
- [63] Y. Pinchover, On nonexistence of any λ₀-invariant positive harmonic function, a counterexample to Stroock's conjecture, Comm. Partial Differential Equations 20 (1995), no. 9-10, 1831–1846.

- [64] R. G. Pinsky, Second order elliptic operators with periodic coefficients: criticality theory, perturbations, and positive harmonic functions, J. Funct. Anal. 129 (1995), no. 1, 80–107.
- [65] E. Randles and L. Saloff-Coste, Convolution powers of complex functions on Z^d,
 To appear in Revista MatemÄatica Iberoamericana (2016), available at http://arxiv.org/abs/1507.03501.
- [66] M. Reed and B. Simon, Methods of modern mathematical physics. IV. Analysis of operators, Academic Press, New York-London, 1978.
- [67] M. Ruzhansky and V. Turunen, *Pseudo-differential operators and symmetries*, Pseudo-Differential Operators. Theory and Applications, vol. 2, Birkhäuser Verlag, Basel, 2010. Background analysis and advanced topics.
- [68] L. Saloff-Coste, Analysis on Riemannian co-compact covers, Surveys in differential geometry. Vol. IX (2004), 351–384.
- [69] R. G. Shterenberg, An example of a periodic magnetic Schrödinger operator with a degenerate lower edge of the spectrum, Algebra i Analiz **16** (2004), no. 2, 177–185.
- [70] M. A. Shubin, Weak bloch property and weight estimates for elliptic operators, SÃI'minaire ÃL'quations aux dÃI'rivÃI'es partielles (Polytechnique) (Unknown Month 1989), 1–20 (eng).
- [71] _____, Spectral theory of elliptic operators on noncompact manifolds, Astérisque 207 (1992), 5, 35–108. Méthodes semi-classiques, Vol. 1 (Nantes, 1991).
- [72] _____, *Pseudodifferential operators and spectral theory*, Second, Springer-Verlag, Berlin, 2001. Translated from the 1978 Russian original by Stig I. Andersson.

- [73] A. Sobolev, Periodic operators: the method of gauge transform, Lectures at the I. Newton Institute, https://www.newton.ac.uk/event/pepw01/timetable (2015).
- [74] T. Sunada, A periodic Schrödinger operator on an abelian cover, J. Fac. Sci. Univ. Tokyo Sect. IA Math. 37 (1990), no. 3, 575–583.
- [75] _____, Group C*-algebras and the spectrum of a periodic Schrödinger operator on a manifold, Canad. J. Math. 44 (1992), no. 1, 180–193.
- [76] M. E. Taylor, *Partial differential equations I. Basic theory*, Second Edition, Applied Mathematical Sciences, vol. 115, Springer, New York, 2011.
- [77] J. A. Thorpe, *Elementary topics in differential geometry*, Undergraduate Texts in Mathematics, Springer-Verlag, New York, 1994. Corrected reprint of the 1979 original.
- [78] È. È. Šnol, On the behavior of the eigenfunctions of Schrödinger's equation, Mat. Sb. (N.S.) 42 (84) (1957), 273-286; erratum 46 (88) (1957), 259.
- [79] L.-C. Wang, *Stability in Gromov-Shubin index theorem*, Proc. Amer. Math. Soc. 125 (1997), no. 5, 1399–1405.
- [80] C. H. Wilcox, *Theory of Bloch waves*, J. Analyse Math. **33** (1978), 146–167.
- [81] W. Woess, *Random walks on infinite graphs and groups*, Cambridge Tracts in Mathematics, vol. 138, Cambridge University Press, Cambridge, 2000.
- [82] M. G. Zaĭdenberg, S. G. Kreĭn, P. A. Kučment, and A. A. Pankov, *Banach bundles and linear operators*, Uspehi Mat. Nauk **30** (1975), no. 5(185), 101–157.

APPENDIX A

A TOPOLOGICAL APPROACH TO DEFINING ADDITIVE FUNCTIONS ON CO-COMPACT RIEMANNIAN NORMAL COVERINGS

A.1 Introduction

Let X be a connected smooth Riemannian manifold of dimension n equipped with an isometric, properly discontinuous, free, and co-compact action of a discrete group G. Notice that G is finitely generated due to the Švarc-Milnor lemma and hence, $Hom(G, \mathbb{R})$ is finite dimensional. Furthermore, the orbit space M := X/G is a compact Riemannian manifold when equipped with the metric pushed down from X. The action of an element $g \in G$ on $x \in X$ is denoted by $g \cdot x$. Let π be the covering map from X onto M. Thus, $\pi(g \cdot x) = \pi(x)$ for any $(g, x) \in G \times X$.

Following [56], we define the class of additive functions on the covering X as follows:

Definition A.1.1.

A real smooth function u on X is said to be additive if there is a homomorphism
 α : G → ℝ such that

$$u(g\cdot x)=u(x)+\alpha(g),\quad\text{for all}\quad(g,x)\in G\times X.$$

- We denote by $\mathcal{A}(X)$ the space of all additive functions on X.
- A map h from X to ℝ^m (m ∈ ℕ) is called a vector-valued additive function on X if every component of h belongs to A(X).

We remark that additive functions on co-compact coverings appeared in various results such as studying the structure of positive *G*-multiplicative type solutions [3, 56], describ-

ing the off-diagonal long time asymptotics of the heat kernel [46] and the Green's function asymptotics of periodic elliptic operators [41] on a noncompact abelian covering of a compact Riemannian manifold.

A direct construction of additive functions on X can be found in either [50, Section 3] or [56, Remark 2.6]. However, this construction depends on the choice of a fundamental domain for the base M in X. A more invariant approach to defining additive functions on coverings was mentioned briefly in [3, 50]. Our goal in this note is to present the full details of this approach for any co-compact covering.

A.2 Additive functions on co-compact normal coverings

We begin with the following notion (see [3, 50]):

Definition A.2.1. Let $H_{DR}^1(M)$, $H_{DR}^1(X)$ be De Rham cohomologies of M and X, respectively. We denote by $\Omega^1(M; G)$ the image in $H_{DR}^1(M)$ of the set of all closed differential 1-forms ω on M (modulo the exact ones) such that their lifts ω to X are exact. In other words, $\Omega^1(M; G)$ is the kernel of the homomorphism

$$\pi^*: H^1_{DR}(M) \to H^1_{DR}(X),$$

where π^* is the induced homomorphism of the covering map $\pi: X \to M$.

By De Rham's theorem, $\Omega^1(M;G)$ is a finite dimensional vector space. Indeed, more is true:

Lemma A.2.2. $\Omega^1(M, G) \cong \operatorname{Hom}(G, \mathbb{R}).$

Proof. By Hurewicz's theorem (see e.g., [55]), the homologies $H_1(M)$ and $H_1(X)$ are isomorphic to the abelianizations of fundamental groups $\pi_1(M)$ and $\pi_1(X)$, respectively. We can identify De Rham cohomologies $H_{DR}^1(M)$ and $H_{DR}^1(X)$ with $Hom(\pi_1(M), \mathbb{R})$ and $\operatorname{Hom}(\pi_1(X), \mathbb{R})$, correspondingly. Since X is a normal covering of M, $\pi_1(X)$ is a normal subgroup of $\pi_1(M)$ and moreover, the following sequence

$$0 \to \pi_1(X) \to \pi_1(M) \to G \to 0$$

is exact. Because $Hom(\cdot, \mathbb{R})$ is a contravariant exact functor, we deduce the exactness of the following sequence of vector spaces:

$$0 \to \operatorname{Hom}(G, \mathbb{R}) \to H^1_{DR}(M) \to H^1_{DR}(X) \to 0.$$

Hence, $\Omega^1(M, G)$ is isomorphic to $Hom(G, \mathbb{R})$.

Fixing a base point $x_0 \in X$. For any closed 1-form ω on M such that its lift to X is exact, there exists a unique function $f_{\omega} \in C^{\infty}(X, \mathbb{R})$ such that $\pi^* \omega = df_{\omega}$ and $f_{\omega}(x_0) = 0$. Equivalently,

$$f_{\omega}(x) = \int_{x_0}^x \pi^* \omega, \quad \forall x \in X.$$

Lemma A.2.3. For such 1-form ω , we have

- *i)* Fix any $g \in G$, then $f_{\omega}(g \cdot x) f_{\omega}(x)$ is independent of $x \in X$.
- ii) If $\pi^* \omega = 0$ then $\omega = 0$.

Proof.

i) For each g ∈ G, let L_g be the diffeomorphism of X that maps any element x to the element g · x. Since π ∘ L_g = π, we get df_ω = π^{*}ω = L_g^{*}π^{*}ω = L_g^{*}df_ω = dL_g^{*}f_ω = d(f_ω ∘ L_g). Thus, d(f_ω - f_ω ∘ L_g) = 0 and so, f_ω ∘ L_g - f_ω is constant since X is connected.

ii) Fix any point p ∈ M. We pick an evenly covered open subset U of M such that it contains p. Then there is a smooth local section σ : U → X, i.e., π ∘ σ = id_{|U} (see e.g., [55, Proposition 4.36]). Hence, ω(p) = σ*π*ω(p) = 0.

On $\mathcal{A}(X)$, we introduce an equivalent relation \sim as follows: $f_1 \sim f_2$ in $\mathcal{A}(X)$ if and only if $f_1 - f_2 = f \circ \pi$ for some function $f \in C^{\infty}(M, \mathbb{R})$.

By Lemma A.2.3 (i), the map $\omega \mapsto f_\omega$ induces the following linear map

$$\Lambda : \Omega^{1}(M, G) \to \mathcal{A}(X) / \sim$$

$$[\omega] \mapsto [f_{\omega}], \qquad (A.1)$$

where the notation $[\omega]([f_{\omega}])$ stands for the equivalent class of $\omega(f_{\omega})$ in $\Omega^{1}(M, G)(\mathcal{A}(X)/\sim)$, correspondingly. We now claim that $[\omega] = 0$ if and only if $[f_{\omega}] = 0$, and hence Λ is an injective linear map. Indeed, due to Lemma A.2.3 (ii), the condition that ω is exact is equivalent to $\pi^{*}\omega = d(f \circ \pi)$ for some $f \in C^{\infty}(M, \mathbb{R})$. But this is the same as $df_{\omega} = d(f \circ \pi)$, or $[f_{\omega}] = 0$.

Consider an additive function f on X. According to the definition, there exists a unique group homomorphism $\ell_f : G \to \mathbb{R}$ such that $f(g \cdot x) = f(x) + \ell_f(g)$ for any $g \in G, x \in X$. Then the map $f \mapsto \ell_f$ induces the linear map

$$\Upsilon : \mathcal{A}(X) / \sim \to \operatorname{Hom} (G, \mathbb{R})$$

$$[f] \mapsto \ell_f,$$
(A.2)

which is injective.

Then the composition $\Upsilon \circ \Lambda$ is also injective. By Lemma A.2.2, $\dim_{\mathbb{R}} \Omega^1(M, G) = \dim_{\mathbb{R}} \operatorname{Hom}(G, \mathbb{R}) < \infty$. These facts together imply that the linear maps Υ and Λ are

isomorphism. We conclude:

Theorem A.2.4. The three vector spaces $\Omega^1(M, G)$, $\mathcal{A}(X)/\sim$ and $\operatorname{Hom}(G, \mathbb{R})$ are isomorphic.

In particular, we obtain:

Corollary A.2.5. Assume that $G = \mathbb{Z}^d$. Then there is a smooth \mathbb{R}^d -valued function h on X such that for any $(g, x) \in \mathbb{Z}^d \times X$,

$$h(g \cdot x) = h(x) + g. \tag{A.3}$$

The following proposition says that given any additive function u on X, one can pick a harmonic additive function f such that f - u is G-periodic.

Proposition A.2.6. For any ℓ in $\text{Hom}(G, \mathbb{R})$, there exists a unique (modulo a real constant) *harmonic* function f on X such that for any $(g, x) \in G \times X$, we have

$$f(g \cdot x) = f(x) + \ell(g). \tag{A.4}$$

Proof. First, we show the existence part. Due to Theorem A.2.4, let \tilde{f} be a function on X satisfying $\tilde{f}(g \cdot x) = \tilde{f}(x) + \ell(g)$ for any $(g, x) \in G \times X$. We recall the isomorphism Λ defined in (A.1). We put $\alpha := \Lambda^{-1}([\tilde{f}]) \in \Omega^1(M, G)$. By the Hodge theorem, there exists a unique harmonic 1-form ω on M such that $[\omega] = \alpha$ in $H^1_{DR}(M)$. Let f be a smooth function such that $f \in \mathcal{A}(X)$ and $\pi^*\omega = df$. Then f satisfies (A.4) since $[f] = [\tilde{f}]$ in $\mathcal{A}(X)/\sim$. Thus, it is sufficient to show that f is harmonic on X. We denote by δ_X , Δ_X and δ_M , Δ_M the codifferential and Laplace-Beltrami operators on X and M, respectively. Since the covering map π is a local isometry between X and M, its pullback π^* intertwines the codifferential operators, i.e.,

$$\delta_X \pi^* = \pi^* \delta_M$$

Since $\Delta_M \omega = 0$, it follows that $\delta_M \omega = 0$. Thus,

$$\Delta_X f = \delta_X df = \delta_X \pi^* \omega = \pi^* \delta_M \omega = 0.$$

For the uniqueness part, let f_1 and f_2 be any two harmonic functions on X such that (A.4) holds for each of these functions. Since $f_1 - f_2$ is G-periodic, it can be pushed down to a real function f on M. Moreover, $\pi^* \Delta_M f = \Delta_X \pi^* f = \Delta_X (f_1 - f_2) = 0$. Therefore, f must be constant since it is a harmonic function on a compact, connected Riemannian manifold M. Thus, $f_1 - f_2$ is constant.

Corollary A.2.7. Fixing a base point x_0 in X. Then to each $\alpha \in \text{Hom}(G, \mathbb{R})$, there exists a unique harmonic function f_{α} defined on X such that $f_{\alpha}(x_0) = 0$ and $\Upsilon([f_{\alpha}]) = \alpha$, where Υ is introduced in (A.2). Consequently,

$$\mathcal{A}(X) = \bigsqcup_{\alpha \in \operatorname{Hom}(G,\mathbb{R})} \{ f_{\alpha} + \varphi \mid \varphi \text{ is periodic } \}.$$

Remark A.2.8. When $G = \mathbb{Z}^d$, the Albanese pseudo-metric d_G introduced in [46, Section 2] is just the pseudo-distance arising from any harmonic vector-valued additive function h satisfying (A.3), i.e., $d_G(x, y) = |h(x) - h(y)|$ for any $x, y \in X$.